



PhD Thesis

Author:

Brunnel Eduardo Guzmán Ortiz

PhD Programme in Industrial Engineering and Production

Supervisors:

Dr. Raúl Poler Escoto Dr. Beatriz Andrés Navarro

July 2022

Acknowledgement

Here I wish to thank all those who have helped me while conducting my doctoral thesis. First of all, I wish to thank God for everything he has given me, his love and mercy is invaluable, and my life would not be the same without knowing that I can trust Him. Thank you for your guidance and orientation, and thank you for allowing me to know your existence.

I would like to thank my supervisors Dr. Raul Poler and Dr. Beatriz Andres, who offered me their trust, support and motivation. I thank Dr. Raul Poler because his classes motivated me to do my doctorate. The opportunity to work with and learn from Raul has been very valuable to me. I thank him for his patience, optimism, guidance, willingness to teach me and, above all, his friendship. I thank Dr. Beatriz for all her teachings, for always motivating and inspiring me to overcome the difficulties that arose, and for always caring about me. I thank Raul and Beatriz, who beyond their theoretical and practical teachings, have taught me by their example, their honesty, respect and work.

I must also thank my colleagues at the CIGIP, especially Raquel for her advice, for all the encouragement she has always given me and her concern about me.

I thank my family, Melissa, Stefania, Chesco, Amelia, Puko, Puka and Ramona, for all their affection, love, unconditional support and patience. I thank them for their laughter and company, for taking care of me at difficult times, and for always being by my side. Without them, none of this would have been possible. I especially thank my mother Martha for her infinite love, and for all her teachings, advice and dedication. Thank you for trusting and believing in me, thank you for being the main promoter of my dreams, and thank you for all those bad nights when I was studying.

I also thank all my family, my cousins and friends who were by my side to support me in everything, who always took the time to talk to me and ask me how I was, and always encouraged me.

Abstract

Optimisation in manufacturing companies is especially important, due to the large investments they make, as sometimes these investments do not obtain the expected return because the profit margins of products are very tight. Therefore, companies seek to maximise the use of productive and financial resources by minimising lost time and, at the same time, improving process flows while meeting market needs.

The planning process is a critical activity for companies. This task involves great challenges due to market changes, alterations in production processes within the company and in the supply chain, and changes in legislation, among others.

Planning of replenishment, production and distribution plays a critical role in the performance of manufacturing companies because ineffective planning of suppliers, production processes and distribution systems contributes to higher product costs, longer lead times and less profits. Effective planning is a complex process that encompasses a wide range of activities to ensure that equipment, materials and human resources are available in the right time and the right place.

Motivated by the complexity of planning in manufacturing companies, this thesis studies and develops quantitative tools to help planners in the replenishment, production and delivery planning processes. From this perspective, realistic models and efficient methods are proposed to support decision making in industrial companies, mainly in small- and medium-sized enterprises (SMEs).

The contributions of this thesis represent a scientific breakthrough based on a comprehensive literature review about replenishment, production and distribution planning that helps to understand the main models and algorithms used to solve these plans, and highlights trends and future research directions. It also provides a holistic framework to characterise models and algorithms by focusing on production planning, scheduling and sequencing. This thesis also proposes a decision support tool for selecting an algorithm or solution method to solve concrete replenishment, production and distribution planning problems according to their complexity, which allows planners to not duplicate efforts modelling or programming solution techniques. Finally, new state-of-the-art mathematical models and solution approaches are developed, such as matheuristic algorithms, which combine mathematical programming and metaheuristic techniques. The new models and algorithms comprise improvements in computational performance terms, and include realistic features of real-world problems faced by manufacturing companies. The mathematical models have been validated with a case of an important company in the automotive sector in Spain, which allowed to evaluate the practical relevance of these novel models using large instances, similarly to those existing in the company under study. In addition, the matheuristic algorithms have been tested using free and open-source tools. This also helps to contribute to the practice of operations research, and provides insight into how to deploy these solution methods and the computational time and gap performance that can be obtained by using free or open-source software.

Resumen

La optimización en las empresas manufactureras es especialmente importante, debido a las grandes inversiones que realizan, ya que a veces estas inversiones no obtienen el rendimiento esperado porque los márgenes de beneficio de los productos son muy ajustados. Por ello, las empresas tratan de maximizar el uso de los recursos productivos y financieros minimizando el tiempo perdido y, al mismo tiempo, mejorando los flujos de los procesos y satisfaciendo las necesidades del mercado.

El proceso de planificación es una actividad crítica para las empresas. Esta tarea implica grandes retos debido a los cambios del mercado, las alteraciones en los procesos de producción dentro de la empresa y en la cadena de suministro, y los cambios en la legislación, entre otros.

La planificación del aprovisionamiento, la producción y la distribución desempeña un papel fundamental en el rendimiento de las empresas manufactureras, ya que una planificación ineficaz de los proveedores, los procesos de producción y los sistemas de distribución contribuye a aumentar los costes de los productos, a alargar los plazos de entrega y a reducir los beneficios. La planificación eficaz es un proceso complejo que abarca una amplia gama de actividades para garantizar que los equipos, los materiales y los recursos humanos estén disponibles en el momento y el lugar adecuados.

Motivados por la complejidad de la planificación en las empresas manufactureras, esta tesis estudia y desarrolla herramientas cuantitativas para ayudar a los planificadores en los procesos de la planificación del aprovisionamiento, producción y distribución. Desde esta perspectiva, se proponen modelos realistas y métodos eficientes para apoyar la toma de decisiones en las empresas industriales, principalmente en las pequeñas y medianas empresas (PYMES).

Las aportaciones de esta tesis suponen un avance científico basado en una exhaustiva revisión bibliográfica sobre la planificación del aprovisionamiento, la producción y la distribución que ayuda a comprender los principales modelos y algoritmos utilizados para resolver estos planes, y pone en relieve las tendencias y las futuras direcciones de investigación. También proporciona un marco holístico para caracterizar los modelos y algoritmos centrándose en la planificación de la producción, la programación y la secuenciación. Esta tesis también propone una herramienta de apoyo a la decisión para seleccionar un algoritmo o método de solución para resolver problemas concretos de la planificación del aprovisionamiento, producción y distribución en función de su complejidad, lo que permite a los planificadores no duplicar esfuerzos de modelización o programación de técnicas de solución. Por último, se desarrollan nuevos modelos matemáticos y enfoques de solución de última generación, como los algoritmos matheurísticos, que combinan la programación matemática y las técnicas metaheurísticas.

Los nuevos modelos y algoritmos comprenden mejoras en términos de rendimiento computacional, e incluyen características realistas de los problemas del mundo real a los que se enfrentan las empresas de fabricación. Los modelos matemáticos han sido validados con un caso de una importante empresa del sector de la automoción en España, lo que ha permitido evaluar la relevancia práctica de estos novedosos modelos utilizando instancias de gran tamaño, similares a las existentes en la empresa objeto de estudio. Además, los algoritmos matheurísticos han sido probados utilizando herramientas libres y de código abierto. Esto también contribuye a la práctica de la investigación operativa, y proporciona una visión de cómo desplegar estos métodos de solución y el tiempo de cálculo y rendimiento de la brecha que se puede obtener mediante el uso de software libre o de código abierto.

Resum

L'optimització a les empreses manufactureres és especialment important, a causa de les grans inversions que realitzen, ja que de vegades aquestes inversions no obtenen el rendiment esperat perquè els marges de benefici dels productes són molt ajustats. Per això, les empreses intenten maximitzar l'ús dels recursos productius i financers minimitzant el temps perdut i, alhora, millorant els fluxos dels processos i satisfent les necessitats del mercat.

El procés de planificació és una activitat crítica per a les empreses. Aquesta tasca implica grans reptes a causa dels canvis del mercat, les alteracions en els processos de producció dins de l'empresa i la cadena de subministrament, i els canvis en la legislació, entre altres.

La planificació de l'aprovisionament, la producció i la distribució té un paper fonamental en el rendiment de les empreses manufactureres, ja que una planificació ineficaç dels proveïdors, els processos de producció i els sistemes de distribució contribueix a augmentar els costos dels productes, allargar els terminis de lliurament i reduir els beneficis. La planificació eficaç és un procés complex que abasta una àmplia gamma d'activitats per garantir que els equips, els materials i els recursos humans estiguen disponibles al moment i al lloc adequats.

Motivats per la complexitat de la planificació a les empreses manufactureres, aquesta tesi estudia i desenvolupa eines quantitatives per ajudar als planificadors en els processos de la planificació de l'aprovisionament, producció i distribució. Des d'aquesta perspectiva, es proposen models realistes i mètodes eficients per donar suport a la presa de decisions a les empreses industrials, principalment a les petites i mitjanes empreses (PIMES).

Les aportacions d'aquesta tesi suposen un avenç científic basat en una exhaustiva revisió bibliogràfica sobre la planificació de l'aprovisionament, la producció i la distribució que ajuda a comprendre els principals models i algorismes utilitzats per resoldre aquests plans, i posa de relleu les tendències i les futures direccions de recerca. També proporciona un marc holístic per caracteritzar els models i algorismes centrant-se en la planificació de la producció, la programació i la seqüenciació. Aquesta tesi també proposa una eina de suport a la decisió per seleccionar un algorisme o mètode de solució per resoldre problemes concrets de la planificació de l'aprovisionament, producció i distribució en funció de la seua complexitat, cosa que permet als planificadors no duplicar esforços de modelització o programació de tècniques de solució. Finalment, es desenvolupen nous models matemàtics i enfocaments de solució d'última generació, com ara els algoritmes matheurístics, que combinen la programació matemàtica i les tècniques metaheurístiques.

Els nous models i algoritmes comprenen millores en termes de rendiment computacional, i inclouen característiques realistes dels problemes del món real a què s'enfronten les empreses de fabricació. Els models matemàtics han estat validats amb un cas d'una important empresa del sector de l'automoció a Espanya, cosa que ha permés avaluar la rellevància pràctica d'aquests nous models utilitzant instàncies grans, similars a les existents a l'empresa objecte d'estudi. A més, els algorismes matheurístics han estat provats utilitzant eines lliures i de codi obert. Això també contribueix a la pràctica de la investigació operativa, i proporciona una visió de com desplegar aquests mètodes de solució i el temps de càlcul i rendiment de la bretxa que es pot obtindre mitjançant l'ús de programari lliure o de codi obert.

Contents

Acknowledgement	
Abstract	
Resumen	V
Resum	VII
Contents	IX
List of Figures	XV
List of Tables	XVII
Chapter 1	19
1. Introduction	19
1.1 Planning in Supply Chain Management	20
1.2 Problem statement and motivation	22
1.2.1 Planning optimization	24
1.3 Research Questions and Objectives	26
1.4 Research Methodology	28
1.4.1 Observation	29
1.4.2 Theoretical construction	29
1.4.3 Experimentation	30
1.4.4 System Development	31
1.5 Research Outline	31
1.6 References	35
Chapter 2	
2. An analysis of reviews of models and algorithms for the optimisation of chain replenishment, production and distribution plans.	supply 38
2.1 Introduction	
2.2 Perspectives from reviews	40
2.3 Methodology	41

2.3.1 Collection	2
2.3.2 Descriptive analysis4	5
2.3.3 Category selection4	6
2.3.4 Validation	8
2.4 Results	8
2.4.1 Perspectives for the analysis48	8
2.4.2 Supply chain processes5	5
2.4.3 Modelling approaches5	5
2.4.4 Solution methods5	7
2.5 Discussion and future research opportunities58	8
2.5.1 Supply chain under uncertainty6	1
2.5.2 Closed loop supply chain62	2
2.5.3 Green supply chain62	3
2.5.4 Sustainable supply chain64	4
2.5.5 Supply chain risk6	5
2.5.6 Resilience in the supply chain6	7
2.5.7 Supply chain management6	7
2.6 Conclusions	0
2.7 References	1
Chapter 3	5
3. Models and algorithms for production planning, scheduling and sequencing problems: a holistic framework and a systematic review	9 5
3.1 Introduction	6
3.2 Literature review methodology78	8
3.2.1 Collecting material	9
3.2.2 Descriptive analysis80	0
3.2.3 Category selection	1
3.2.4 Material evaluation	4

3.3 Results analysis	85
3.3.1 Decision level	85
3.3.2 Plan aggregation	86
3.3.3 Modelling approach and solution techniques	90
3.3.4 Mathematical model objectives	97
3.3.5 Applications area and enterprise integration level	103
3.3.6 Solution quality and problem scale	107
3.4 Discussion and perspectives	
3.5 Conclusions and future research	
3.6 References	115
Chapter 4	125
4. A decision-making tool for algorithm selection based on a fuzzy approach to solve replenishment, production and distribution planning p	TOPSIS roblems. 125
4.1 Introduction	126
4.2 Algorithm selection problem literature review	
4.3 Solution methodology	
4.3.1 Fuzzy Set Theory and fuzzy numbers	
4.3.2 The Fuzzy TOPSIS Method	
4.4 The methodological approach for algorithm selection problem .	
4.4.1 Stage 1 - Define criteria and alternatives	
4.4.2 Stage 2 - Problem statement	
4.4.3 Stage 3 - Application of the Fuzzy TOPSIS Method	
4.5 Sensitivity analysis	
4.6 Conclusions	
4.7 References	
Chapter 5	
5. An MILP model for the lot-sizing/scheduling of automotive plastic com with raw materials and packaging availability.	ponents 171

5.1 Introduction	172
5.2 Literature review	173
5.3 Problem description and formulation	175
5.4 Computational experiments	181
5.4.1 Generating datasets	181
5.4.2 Computational results	181
5.5 Conclusion	185
5.6 References	186
Chapter 6	188
6. A MILP model for reusable containers management in Automo components supply chain	tive plastic 188
6.1 Introduction	189
6.2 Literature review	190
6.3 Problem definition	190
6.3.1 A MILP model for Reusable Containers Management	193
6.3.2 Numerical Experiment	196
6.4 Collaboration scheme	197
6.5 Conclusions	198
6.6 References	198
Chapter 7	200
7. Matheuristic Algorithm for Job-Shop Scheduling Problem Using a Mathematical Model	Disjunctive 200
7.1 Introduction	201
7.2 Literature Review: Matheuristic Resolution Approaches	202
7.3 Job-Shop Scheduling Problem: Disjunctive Mathematical F	ormulation 208
7.4 Materials and Methods	210
7.4.1 Proposed Matheuristic Approach	210
7.4.2 Initial Population	212

7.4.3 Feasibility Tester 21	3
7.4.4 Fitness Function21	5
7.4.5 Selection	6
7.4.6 Crossover Operator21	6
7.4.7 Mutation Operator21	7
7.5 Computational Experiments21	7
7.6 Conclusions and Further Work22	20
7.7 References	21
Chapter 8	25
8. A matheuristic approach to production and distribution planning	25
8.1 Introduction	26
8.2 Related works22	29
8.3 Problem definition23	1
8.3.1 Notation23	2
8.4 Matheuristic solution method23	4
8.4.1 Initial population23	6
8.4.2 Evaluation function23	6
8.4.3 Selection	7
8.4.4 Crossover	7
8.4.5 Mutation23	7
8.4.6 The matheuristic approach procedure	8
8.5 Numerical experiments23	8
8.5.1 Experimental results23	9
8.6 Conclusion	4
8.7 References	5
Chapter 9	9
9. Conclusions	9
9.1 Main contributions25	0

9.1.1 The replenishment, production and distribution planning holist conceptual framework25	tic 50
9.1.2 Methodology for selecting algorithms to solve planning problem	ns 51
9.1.3 Models and algorithms for replenishment, production an distribution planning25	nd 52
9.2 Future research lines25	56
9.3 Reference25	57
Funding Acknowledgement25	58

List of Figures

Figure 1.1. Classification of plans (based on: the SCOR model [2] and Andres	et al.
[3])	20
Figure 1.2. Research Methodology proposed by Nunamaker et al. [25] and	
analogously with the research objectives	29
Figure 1.3. Research questions, objectives and structure of this thesis	34
Figure 2.1. Search criteria and selecting articles	43
Figure 2.2. Classification of articles by publication year.	45
Figure 2.3 Number of published articles per year cited in the analysed review	ed
articles	49
Figure 3.1. Structured literature review process.	79
Figure 3.2. Distribution of the reviewed papers according to year	80
Figure 3.3. Distribution of articles for publication year and journal	81
Figure 3.4. Distribution of the scale of problems and solution quality	. 109
Figure 4.1. Fuzzy triangular number	.136
Figure 4.2. A methodological approach for algorithm selection problem	.139
Figure 4.3. Replenishment, production and distribution planning problem types	pes.
	.143
Figure 4.4. Hierarchical structure for algorithm selection.	.144
Figure 5.1. Outline of the Source and Make Plans	.175
Figure 5.2. Representation of the realistic lot-sizing/scheduling model with r	aw
materials and packaging availability	. 183
Figure 6.1. Closed loop of reusable containers scheme	. 192
Figure 6.2. Flow chart of negotiation process	. 197
Figure 7.1. Flow chart of the matheuristic algorithm.	. 211
Figure 7.2. Individual's structure.	. 212
Figure 7.3. Feasible solution representation and its Gantt chart.	. 214
Figure 7.4. Infeasible solution representation and its Gantt chart.	. 214
Figure 7.5. Repaired solution representation and its Gantt chart	. 214
Figure 7.6. Position of genes on integer chromosomes.	. 215
Figure 7.7. An example of a swap mutation operator.	. 217
Figure 7.8. Experimental results	. 219
Figure 7.9. Box plot of all the methods.	. 220
Figure 8.1. Illustration of the integrated production and distribution planning	g
problem	. 232
Figure 8.2. The flow chart of the proposed matheuristic approach	. 235

Figure 8.3. Single-Point Crossover	237
Figure 8.4. The Swap mutation operator	237
Figure 8.5. Time spent by matheuristic to find a feasible solution	241
Figure 8.6. Comparison of matheuristic and MILP-CBC performance	242
Figure 8.7. Comparison of matheuristic and Gurobi performance	244

List of Tables

Table 2.1. Perspectives on supply chain analyses.	40
Table 2.2. Database search criteria.	42
Table 2.3. The selected review articles	43
Table 2.4. Distribution of the articles based on journals	45
Table 2.5. Types and subtypes of plans according to the SCOR model	46
Table 2.6. Classification of categories according to the taxonomies proposed b	у
[14, 21, 22]	47
Table 2.7. Analysis of the focus of the research works.	50
Table 2.8. Classification according to SCOR models	55
Table 2.9. Types of models found in the reviews	56
Table 2.10. Types of resolution methods	57
Table 2.11. Summary of future research lines from different supply chain	
perspectives	58
Table 2.12. Summary of future research lines from a supply chain managemer	nt
perspective considering several sectors	60
Table 3.1. Framework proposed to represent production planning, scheduling	and
sequencing problems	82
Table 3.2. The decision-making levels of the reviewed works	86
Table 3.3. Plan type and plan aggregation of the reviewed works	88
Table 3.4. The modelling approach and solution techniques of the reviewed	
works	94
Table 3.5. Production planning, scheduling and sequencing objectives	98
Table 3.6. The main objectives of the proposed models	101
Table 3.7. Industry sectors and application type	106
Table 3.8. Problem scales and solutions quality.	110
Table 4.1. Research studies addressing the algorithm selection problem	131
Table 4.2. Criteria for algorithm selection.	140
Table 4.3. Linguistic scales to assess the criteria and alternatives (Chen [97]).	146
Table 4.4. Decision makers' linguistic assessment of the criteria.	146
Table 4.5. Decision matrix with the aggregated scores	148
Table 4.6. Aggregate fuzzy weights for each criterion.	149
Table 4.7. Distances between alternatives	149
Table 4.8. Closeness quotient and algorithms ranking	150
Table 4.9. Quantitative results of the sensitivity analysis.	152
Table 5.1.Notation	177

 Table 5.3. MILP model results. Table 5.4. Numerical results of instance S1: backorders, inventories, stockout, lot-sizing. 184 Table 5.5. Numerical results of instance S1: scheduling. 184 Table 5.6. Numerical results of instance S1: use and inventory of materials. 185 Table 6.1. Nomenclature for Reusable Containers Management model
 Table 5.4. Numerical results of instance S1: backorders, inventories, stockout, lot-sizing. 184 Table 5.5. Numerical results of instance S1: scheduling. 184 Table 5.6. Numerical results of instance S1: use and inventory of materials. 185 Table 6.1. Nomenclature for Reusable Containers Management model
lot-sizing
Table 5.5. Numerical results of instance S1: scheduling.184Table 5.6. Numerical results of instance S1: use and inventory of materials.185Table 6.1. Nomenclature for Reusable Containers Management model193Table 6.2 Results of MILP model for Reusable Containers Management.196Table 7.1 Literature review of matheuristic to solve the job-shop problem.207Table 7.2. The mathematical notations used in the JSP formulation.209Table 7.3. The job-shop scheduling problem data.211Table 7.4. Sequence of jobs on machines.213Table 7.5. Precedence of jobs with binary array z_{ii} 215
Table 5.6. Numerical results of instance S1: use and inventory of materials185Table 6.1. Nomenclature for Reusable Containers Management model
Table 6.1. Nomenclature for Reusable Containers Management model
Table 6.2 Results of MILP model for Reusable Containers Management.196Table 7.1 Literature review of matheuristic to solve the job-shop problem.207Table 7.2. The mathematical notations used in the JSP formulation.209Table 7.3. The job-shop scheduling problem data.211Table 7.4. Sequence of jobs on machines.213Table 7.5. Precedence of jobs with binary array z_{ii} 215
Table 7.1 Literature review of matheuristic to solve the job-shop problem207Table 7.2. The mathematical notations used in the JSP formulation209Table 7.3. The job-shop scheduling problem data
Table 7.2. The mathematical notations used in the JSP formulation.209Table 7.3. The job-shop scheduling problem data.211Table 7.4. Sequence of jobs on machines.213Table 7.5. Precedence of jobs with binary array z_{ii} 215
Table 7.3. The job-shop scheduling problem data.211Table 7.4. Sequence of jobs on machines.213Table 7.5. Precedence of jobs with binary array z_{ii} 215
Table 7.4. Sequence of jobs on machines
Table 7.5. Precedence of jobs with binary array z_{ii}
······································
Table 7.6. GA-LP parameters
Table 7.7. Comparison of how the proposed approaches perform218
Table 8.1. Classification of matheuristic approaches
Table 8.2. Hierarchical matheuristic classification [12]228
Table 8.3. The MILP model nomenclature232
Table 8.4. Performance comparison between CBC and Matheuristic240

Abstract:

This chapter presents an overview of the research conducted in this thesis, starting with the description of the different types of plans, including replenishment, production and distribution, followed by a description of the problem statement and motivation. Subsequently, research questions are raised and the objectives pursued by this research are defined. The research methodology is shown, which begins with theoretical construction to the development of systems. Finally, the structure in which this thesis is organised is presented and corresponds to the research papers presented or published in different scientific journals.

1.1 Planning in Supply Chain Management

Supply chain planning deals with the organisation and coordination of processes, and seeks to respond to demand, replenishment, production and distribution at the required time and place and in the right quantity. Planning provides support for decision making because companies make a large number of decisions that need to be continuously addressed. These decisions range from replenishment, resource utilisation for production and distribution of finished products to sales planning [1].

The Supply Chain Operations Reference (SCOR) model [2] describes the different planning schemes, which can be organised according to distinct modules [3], as shown in Figure 1.1.



Figure 1.1. Classification of plans (based on: the SCOR model [2] and Andres et al. [3]).

According to SCOR [2], supply chain planning consists of three main schemes, which are replenishment, production and distribution, all of which are governed by demand, or its forecast, in a planning horizon. Within the replenishment scheme, we find the following modules:

- Procurement planning: it focuses on the process of identifying, compiling purchasing needs, procurement schedules, as well as identifying and selecting suppliers for the procurement of a variety of goods [4].
- Inventory Planning: it aims to minimise the total storage cost and to reduce the costs of keeping products in stock. It is responsible for the design of policies and procedures for inventory storage and includes raw materials, components and final products. In addition, it is responsible for inventory replenishment by the acquisition of raw materials or finished products. This plan helps to sufficient quantity of items to produce or sell to exist [5, 6].
- Material requirements planning: it focuses on determining in detail the right quantity at the right time of the materials, components and subassemblies needed to produce a final product, and plans their production or purchase based on a planning horizon [7].

The modules within the production scheme are detailed below:

- Production planning: it is responsible for determining the required level of production, inventory and manpower according to a planning horizon from actual or planned demand [3].
- Production scheduling: it establishes the resource utilisation schedule, i.e. it is a matter of assigning specific start and end times for resource allocation with capacity constraints, precedence and compliance dates being relevant to this plan [3, 8].
- Production sequencing: it specifies the order in which resources are used in a shared workstation [8, 9].

With the distribution scheme, there are the following modules:

 Distribution Planning: it determines the flow of goods between the customers or centres of a distribution network. In this plan, the best alternatives for distribution are evaluated, an analysis of the logistic flows (demographic characteristics and travel patterns), and even the design of a network, are carried out [10].

 Transport planning: it is in charge of establishing truck loads by respecting the design of trucks, in addition to the routing of vehicles, and seeking to respect delivery times, legal restrictions for drivers and ecological legislation [1].

The different above-described planning modules can share information and provide feedback to one another. To do so, there are several architectures that have the main advantage of sharing data within an integrated module and not only independent databases that can cause redundancies. Integrated and collaborative modules are detailed below:

- Replenishment and production planning: it takes care of acquiring items or services to turn them into finished products based on demand [4].
- Production planning and distribution planning: it determines the quantity of production and distribution of each item in a planning horizon. This type of plan generally depends on capacity and delivery times [10].
- Inventory planning and distribution planning: it involves the routing of vehicles and inventory management where inventory allocation and shortterm transport are determined, i.e., it deals with the distribution of a product from a facility to customers. The aim of this plan is no customer shortages. So it determines how much to deliver, the customer to deliver to and routes, by seeking to minimise distribution costs [11, 12].
- Replenishment, production and distribution planning. This approach type is characterised by the collaboration between plans, ranging from procurement of materials by calculating requirements to the transformation of these materials into finished products, to meet expected or actual demand and, finally, to be delivered to individual customers or a distribution network by calculating vehicle capacity and possible delivery routes [3].

1.2 Problem statement and motivation

Over the years, the business world has faced different types of crises of economic, social and health kinds. One of the biggest exponents of the economic crisis that the business world faced was the Great Economic Recession that hit European countries hard between 2008 and 2013 [13]. Now the outbreak of COVID-19 has once again tested companies' resilience. According to BBC World's online publication "Coronavirus and the economy: three key differences in the Great

Recession of 2008" [14], the current crisis caused by the pandemic and the consequences of the virus could be comparable for some countries to the 2008 global financial crisis. The common features between the Great Recession and the pandemic were the great uncertainty that it generated, uncertainty that was marked by economic recovery and employment.

Some of the problems faced by production companies in different crises were caused by uncertainty, and by the speed with which they had to adapt to new working conditions and respond to new market demands. For example, with the COVID-19 crisis, textile companies had to change their production and start processing textiles for protective masks, clothing for medical staff, among others. In response to this new demand, textile companies faced the problem of not only planning deliveries and shipments, but also changes in production processes and operations. All this made such tasks challenges because they had to generate optimal production plans that minimised downtime, and they had to plan how much and when to produce.

Other companies like beverage and non-essential food supply companies, e.g. carbonated soft drinks, faced limited space in their warehouses for finished products, raw materials and components, as well as changes in their shipments because borders were closed by some countries [15]. In some cases delivering raw materials and components on time became a very difficult task. This meant that companies had to react quickly and plan their sales, which resulted in downtime due to not receiving components to be able to produce and not knowing how to respond to change.

Meat product companies also had to overcome new challenges because they had to introduce new operations into the production cycle related to personnel hygiene, such as workers' additional hand washing, antiseptic treatment and taking body temperature before they started a working day. In addition, they had to reduce the number of people on a shift and change from a 2-shift mode to a 3shift mode. They also had to introduce an additional cleaning operation of production and processing plants with disinfectant. All this meant that many companies could not plan in such a way to not reduce production volumes.

According to Myro [16], the manufacturing industry in Spain was affected by the pandemic as it had a knock-on effect on other productive activities, especially the most directly affected ones: hotels, leisure and restaurants, as well as transport, commerce and construction. The most affected goods were consumer durables and capital goods. The same author estimates that the automotive industry in April 2020 suffered a year-on-year decline of 98.7%, components, parts

and accessories (88%), and bodywork and trailers (62.5%). Furniture, jewellery and games and toys recorded falls of more than 70%, as did footwear and clothing, followed closely by textile finishing. Metal industries contracted nearly 50%, slightly more than construction materials, while machinery industries saw their output fall by nearly 40%. In chemical and food industries, the drop in production was not as high, except for artificial fibres and plastics for the former, or bakery products, pasta and beverages for the latter, which fell by more than 20%. However, Spain and other European Union (EU) countries could obtain funding from the Recovery Plan proposed by the European Commission. According to Torres and Férnandez [17], forecasts are still subject to a very high degree of uncertainty and, therefore, to a much higher than usual margin of error; not only because of the possibility of new restrictions on certain economic activities, but also because of the great uncertainty regarding the behaviour of certain macroeconomic variables in such an unusual situation. For example, one can only make more or less well-founded, assumptions about the level of precautionary savings, which will be a key determinant of the extent of recovery in consumption.

Many manufacturing companies in both the pandemic and the Great Recession did not have the arrangements to adapt quickly to new working conditions and change their sourcing, production and distribution plans while optimising their costs. Adaptability, flexibility and rapid response to change are the basis for success in today's global economy. Today the success of most companies, especially small- and medium-sized enterprises (SMEs), is linked to the efficiency of their production processes. The importance of optimising company processes has led researchers and governments to look for methods to help companies to improve and optimise their production processes.

1.2.1 Planning optimization

The task of optimally utilising all available resources and efficiently planning replenishment, production and distribution has become a key element for many manufacturing companies. In this context, situations like changes in the production process, the emergence of new labour resource constraints, delays from supplier components replenishment, and changes in deliveries and shipments, lead many companies to make mistakes in their decision-making processes due to lack of efficient mathematical tools to support planning process optimisation. Nowadays combinatorial optimisation problems constantly arise and in many forms in not only the manufacturing sector, but also in different economic and social sectors. This is because there are lots of services driven by optimisation algorithms; for example, public transport, delivery services, shift

scheduling in hospitals, amongst others. These algorithms allow to better use available resources and guarantee a service level [18].

In a more general context, decision support systems continue to be very popular in industrial engineering thanks to their ability to help decision makers manage their resources despite various imposed replenishment, manufacturing and delivery constraints. However, manufacturing systems are complex because they connect many technical and human resources. These systems often seek greater efficiency and, as a result, new strategies are developed. Presently, such systems must be able to adapt quickly to market needs.

In summary, one of the most effective methods for solving materials replenishment, production and delivery planning problems are automated systems based on mathematical optimisation methods (finding the minimum and maximum of the objective function) [19]. There are many reported applications and events that document the success and importance of this research area. Optimisation is about minimising cost, time, distance or risk, and about maximising quality, satisfaction, profit, among others. Finding the best solution for certain optimisation problems of scientific or industrial importance often results in computational representations that are of intractable complexity.

This is why particular interest is shown in heuristic, metaheuristic and matheuristic algorithms in the optimisation field, which serve as intelligent strategies to design or improve general optimising procedures for solving complex problems with excellent performance.

According to this reality, more and more researchers are striving to study and develop systems and techniques given the fact that most combinatorial optimisation problems are NP-Hard, and there is no deterministic polynomial-time algorithm for such intractable problem. This scenario means that some algorithms cannot finish in practical computational times once the problem size has become too large. In practice, finding optimal solutions for these problems remains a research topic, and searching for solutions that are superior in quality to previously known best solutions is extremely valuable.

The use of approximation algorithms is the main alternative to solve this kind of problems because it allows a solution close to the optimal one in reasonable computational times to be obtained. The application of algorithms to optimisation problems has been very important in recent decades. Its main advantage is flexibility in the problems to be solved and the robustness of the provided solutions because this allows algorithms to be applied to a wide range of problems.

This causes combined or hybrid techniques to start to become successful in view of the fact that choosing an appropriate combination of algorithms can be key to achieve better performance in a solution [20]. Hence a combination of algorithms can provide sufficiently good solutions in practical computational times [21].

In fields like computer science, artificial intelligence and operations research, many of these problems have been addressed for some years now without them interacting much. However in recent years, due to the fourth industrial revolution (Industry 4.0)[22], these groups have joined forces and started to look at the progress that all these areas can make by seeking to develop optimisation techniques that are faster, robuster and easier to maintain.

In this context, the central focus of this thesis lies in the need to develop tools to improve the use of available resources, avoid waste and improve the sustainability of services in companies. Therefore, this thesis is framed in the secondary or industrial sector of the economy. Hence, this thesis is framed in the secondary or industrial sector of the economy. This sector is the cornerstone for a country's economic growth, for which various optimisation methods are researched in depth by placing emphasis on resolution methods, and on developing strategies for certain classes of NP-Hard combinatorial optimisation problems. Given the aim to design and implement new models and algorithms from the base created by European project C2NET (Cloud Collaborative Manufacturing Networks) [23], the intention is to examine the combination of various methods, such as metaheuristic or heuristic methods, to make the best of the characteristics of each technique type to obtain a flexible and easy-to-use system for companies.

1.3 Research Questions and Objectives

The current environment in which companies operate is highly competitive, and generally changes rapidly. So, companies must design, manufacture and distribute their products at the same speed, while being efficient and reducing costs at the same time. This leads companies to seek to continuously improve their operations.

Companies' economic performance is linked with technological research and innovation. Both these aspects are of decisive importance today because we are facing a new technological revolution of unusual magnitude that is driven by advances in digitalisation and the need for a firm and large-scale fight for environmental sustainability. Computer systems, mathematical models and algorithms for decision-making support are tools now available to companies that

aim to improve their performance, which turns the use of this technology into a competitive advantage.

Mathematical models make a representation of a system or a real problem by means of mathematical equations, these equations try to represent the components of the system and the solution of these equations serve to predict the changes in the system or the changes that can undergo in time [24]. There are different types of methods to solve different types of mathematical models, such as optimiser, heuristic, metaheuristic and matheuristic algorithms.

By considering the importance of providing advanced models and algorithms that allow SMEs to achieve the integral optimisation of their production assets, and respond more quickly and efficiently to the market, the following research questions are posed to support the design of models and algorithms oriented to solve real-world replenishment, production and distribution plans efficiently; that is, in adequate computational times with high-quality solutions. With the eagerness to integrate the objective of this research, the following general research question (GRQ) is posed:

GRQ: What suitable approach could efficiently solve replenishment, production and distribution planning, which are computationally difficult to solve by exact solvers?

The following research questions (RQ) derive from this general research question:

- **RQ1.** How can replenishment, production and distribution planning problems and solution methods be categorised?
- **RQ2.** How can the most suitable algorithm or solution method be selected to solve replenishment, production and distribution planning according to its complexity?
- **RQ3.** What new algorithms should be developed to solve real replenishment, production and distribution planning problems?

In order to answer the research questions, the general research objective (GRO) is defined:

GRO. Design and implement new models and algorithms for the calculation of replenishment, production and distribution plans in industrial companies.

From the GRO, the following specific objectives (O) can be derived:

01. Propose a taxonomy to categorise the optimisation problems for replenishment, production and distribution plans.

02. Propose a framework to evaluate the different types of models and algorithms that can solve the identified replenishment, production and distribution planning problems.

03. Develop a decision support tool to properly select an algorithm or solution method for a replenishment, production and distribution planning problem.

04. Develop and implement new models and algorithms to solve real-world problems

05. Validate the applicability of the developed models and algorithms by applying them to real cases.

1.4 Research Methodology

To answer the research questions, and to fulfil the stated objectives, a multimethod approach that combines quantitative and qualitative research is adopted, which is incorporated into an overall methodological framework based on the multimethod approach proposed by Nunamaker et al. [25]. The selected research methodology consists of four stages: (i) observation; (ii) theory building; (iii) experimentation; (iv) systems development (see Figure 1.2).

This methodology was used to achieve the proposed objectives and, additionally, this approach was used in the development of each of the chapters of this doctoral thesis. The main characteristic of this methodology is that it has a scheme that can be used sequentially, and each stage can be feedback to the other stages. Figure 1.2 shows how each stage is used for each of the objectives of the doctoral thesis. The following subsections describe each of the stages and how they have been used to achieve the objectives.



Figure 1.2. Research Methodology proposed by Nunamaker et al. [25] and analogously with the research objectives.

1.4.1 Observation

According to the methodological approach proposed by Nunamaker et al. [25], *Observation* is used to inquire about the topic under study. By exploring the study field, this approach helps to create generalisations or hypotheses that must be demonstrated by the research. Observation in this research is used to meet all the set objectives. The aim of this stage is to study the mathematical models proposed in the literature to represent replenishment, production and distribution planning problems and their characteristics, and to investigate the used solution methods and to study the relation between models and algorithms. The observation stage provides the thesis with an overview of the entire spectrum of methods used in planning problems.

1.4.2 Theoretical construction

This research methodology stage develops concepts, conceptual frameworks and mathematical models [25].

This stage tackles objectives *O1*, *O2*, *O3* and *O4*. Based on the results obtained from the analysis of the planning problems, and after gaining an overview of the whole spectrum of models and algorithms used in planning problems, a taxonomy for classifying optimization problems and a holistic framework for evaluating the different types of models and algorithms are proposed. The taxonomy and the holistic framework allowed defining the most relevant classification dimensions of planning problems with which this thesis intends to provide new insights into the characteristics of problems.

After defining the different dimensions, and knowing that the literature details many resolution methods for optimisation problems, a decision support tool is generated to put these resolution methods to best use. Here the aim is to seek to develop a methodology to select an algorithm or resolution method given a portfolio of algorithms. In this way, the users who use the decision support tool will obtain a ranking of algorithms or solution methods that can be used to solve replenishment, production and delivery planning problems from different complexities. This will help to avoid having to duplicate efforts in generating or programming different algorithms and then evaluating their solutions.

Following this stage, different mathematical formulations are generated with which to seek to capture companies' needs. For this purpose, solutions are given to problems with a combined replenishment and production planning approach and with a combined replenishment and distribution planning approach. In this way, we respond to the existing research gaps in the literature. To validate our formulations, a real case of a company in the automotive sector is used with which attempts are made to perceive how the company under study can achieve cost reduction, which is an important test to validate models. Finally, knowing the computational difficulty of the planning problems that are generally presented as NP-hard or Np-complete, and considering that the most advanced solvers have difficulties in generating good quality solutions for different mathematical models, the aim is to develop matheuristics algorithms for different planning problems, such as the sequencing problem and the combined production and distribution approach, and to use free and open-source solvers because they can be useful for companies that do not wish to incur costs associated with licences.

1.4.3 Experimentation

In this stage, experiments or computational simulations, laboratory or field tests can be carried out. Experimentation is used to validate theories. The experimental stage is guided by the theoretical basis and facilitated by the development of systems, and this stage can help to refine theories, models or simulations [25].

This approach is used in objectives *O3*, *O4* and *O5*, which validate the mathematical models and matheuristics algorithms using large datasets similar to those used by companies. In doing so, we aim to validate our formulations while seeking to obtain high-quality solutions in acceptable resolution times. The proposed models are evaluated in a real-world problem, and realistic instances are also created to assess the industrial relevance of the proposed matheuristics algorithms.

1.4.4 System Development

In this stage, systems are designed, the architecture of systems is built, and prototypes are created. Here theoretical knowledge is adapted with technological advances, and the viability of concepts is demonstrated by creating prototypes. The difficulties in this stage mean that concepts or theories are raised again or modified. The success of this stage represents the development of a product and technological transfer to companies; that is to say, the success of the theories, concepts and models raised in the previous stages [25]. This approach is used for the *O3, O4 and O5* in which a set of data generators is developed (see under [26–28]), along with programmes containing the algorithms and methodology for selecting algorithms.

1.5 Research Outline

The chapters of this thesis comprise a set of selected scientific articles that answer the research questions posed in previous sections and are aligned with the previously defined objectives. An overview of the main aspects addressed in the articles is detailed below.

Chapter 1 provides an overview of the research work conducted in this thesis. This chapter outlines the motivation and the problem statement, in addition to formulating the research questions, objectives and the research methodology employed.

Chapter 2 presents an analysis of the different models and algorithms for optimising replenishment, production and distribution plans in the supply chain based on the latest literature reviews in this field. It provides an overview of models and algorithms used to solve different types of problems encountered in the supply chain. This review presents a starting point for the selection of models and algorithms to solve these types of supply chain problems and discusses several lines of future research. This chapter aims to answer *RQ1* by generating a taxonomy to categorize and analyse the different types of plans to meet *O1*

Chapter 3 provides a holistic conceptual framework for production scheme. It includes production planning, scheduling and sequencing. The proposed conceptual framework answers *RQ1* and fulfils *O2* by studying the different levels of aggregation. and disaggregation of plans, modelling approaches to represent different types of plans and their characteristics, solution approaches with adopted algorithms, application areas, levels of intra- and interenterprise integration, dataset sizes used to validate models and algorithms, development tools and the quality of solutions obtained in relation to the problem data size.

Chapter 4 provides a decision support tool for algorithm selection by a fuzzy TOPSIS approach to help decision makers to choose the best algorithm to solve replenishment, production and delivery planning problems in manufacturing companies. This tool contributes to solving *RQ2*, since the selection of an algorithm in an attempt to solve an optimisation problem is a difficult task. In replenishment, production and distribution planning problems, different methods and techniques with many characteristics are applied. Hence the difficult problem of knowing which algorithm is suitable for a planning problem arises. The main contribution of this chapter to fulfil Objective 3 is to provide a tool to select a solving method, which can range from a solver (commercial or free) to different types of heuristic, metaheuristic and matheuristic algorithms.

Chapter 5 provides a mixed integer linear programming (MILP) model for the lot-sizing/scheduling problem for the manufacture of automotive plastic components. A second-tier supplier in the automotive supply chain is studied. The second-tier supplier produces plastics for automobiles. These plastics are produced in moulds that are assembled on flexible injection moulding machines in parallel. To cover *O4* on the development of new models for real problems, the novelty of this model lies in the fact that it considers the arrival of materials as raw material for the injection of parts into the moulds, and the utilisation of raw materials and the availability of containers for the packaging of finished products. In addition, moulds can only be assembled during specific time periods according to the amount of available labour during each period. The usefulness of this model is demonstrated with randomly generated instances fulfilling *O5*, thus answering *RQ3*.

Chapter 6 responds to *RQ3* and addresses Objectives 4 and 5 by proposing a mixed integer linear programming model to address the production lot-sizing and delivery decisions of a containers fleet in a production system. The model is approached in the automotive supply chain from the second-tier supplier perspective. The model determines the number of cardboard containers that the second-tier supplier should use when production exceeds the number of available

reusable containers required to deliver components from the second- to the firsttier supplier. As the second-tier supplier has a storage space constraint, it has to limit production to the warehouse size, but always in line with the first-tier supplier's demand plan. In addition, the second-tier supplier is limited in terms of the number of empty reusable containers that the first-tier supplier delivers to the second-tier supplier, which are used to package and deliver the injected plastic components. The filled reusable containers are delivered to the first-tier supplier in accordance with the demand plan for plastic components.

Chapter 7 focuses on developing a new efficient matheuristic to solve the machine scheduling and sequencing problem. This chapter fulfils Objectives 4 and 5 and solves *RQ3*. From a mathematical point of view, this problem is classified as NP-hard. The matheuristic algorithm is proposed for the job-shop problem by combining a genetic algorithm with a disjunctive mathematical model, and the open-source solver Coin-OR Branch & Cut is used. The matheuristic algorithm, in which an optimal model and a metaheuristic algorithm interoperate, provides efficient solutions to be found and reduces the computation time by using an open-source optimisation solver combined with a genetic algorithm.

Chapter 8 presents a matheuristic approach to solve the combined production and distribution planning problem. Given the complexity of this problem, the combination of a genetic algorithm and a mixed integer linear programming model is proposed. The matheuristic algorithm was tested using the open-source solver Coin-OR Branch & Cut. The computational results revealed that the presented matheuristic algorithm can be used to solve real-size problems. Objectives 4 and 5 are met in this chapter and RQ3 is answered.

Finally, **Chapter 9** summarises the contributions of this thesis and the most important results obtained and offers directions for future research.

An outline summarizing the relationship between the research questions, the objectives and the chapters of the thesis is shown in Figure 1.3



Figure 1.3. Research questions, objectives and structure of this thesis.

34 |

1.6 References

- H. Stadtler, "Supply chain management and advanced planning Basics, overview and challenges," *Eur. J. Oper. Res.*, vol. 163, no. 3, pp. 575–588, 2005, doi: 10.1016/j.ejor.2004.03.001.
- [2] Supply Chain Council, *Supply Chain Operations Reference (SCOR) Model*. 2012.
- [3] B. Andres, R. Poler, L. Saari, J. Arana, J. V. Benaches, and J. Salazar, "Optimization Models to Support Decision-Making in Collaborative Networks: A Review," in *Closing the Gap Between Practice and Research in Industrial Engineering, Lecture Notes in Management and Industrial Engineering,* 2018, pp. 249–258.
- [4] A. Orbegozo, B. Andres, J. Mula, M. Lauras, C. Monteiro, and M. Malheiro, "An Overview of Optimization Models for Integrated Replenishment and Production Planning Decisions," in *Closing the Gap Between Practice and Research in Industrial Engineering*, 2018, pp. 239–247.
- [5] F. Campuzano and J. Mula, *Supply Chain Simulation: A System Dynamics Approach for Improving Performance*. 2011.
- [6] Y. Akçay, Y. Li, and H. P. Natarajan, "Category Inventory Planning With Service Level Requirements and Dynamic Substitutions," *Prod. Oper. Manag.*, vol. 29, no. 11, pp. 2553–2578, 2020, doi: 10.1111/poms.13240.
- [7] B. Ram, M. R. Naghshineh-Pour, and X. Yu, "Material requirements planning with flexible bills-of-material," *Int. J. Prod. Res.*, vol. 44, no. 2, pp. 399–415, 2006, doi: 10.1080/00207540500251505.
- [8] E. Guzman, B. Andres, and R. Poler, "Models and algorithms for production planning, scheduling and sequencing problems: a holistic framework and a systematic review," *J. Ind. Inf. Integr.*, p. 100287, 2021, doi: 10.1016/j.jii.2021.100287.
- D. R. Kiran, "Sequencing and line balancing," *Prod. Plan. Control*, pp. 345–356, 2019, doi: 10.1016/b978-0-12-818364-9.00024-x.
- [10] B. Andres, R. Sanchis, J. Lamothe, L. Saari, and F. Hauser, "Integrated production-distribution planning optimization models: A review in collaborative networks context," *Int. J. Prod. Manag. Eng.*, vol. 5, no. 1, pp. 31– 38, 2017, doi: 10.4995/ijpme.2017.6807.
- [11] S. Hartmut, C. Kilger, and M. Herbert, Supply Chain Management and Advanced Planning: Concepts, Models, Software and Case Studies. Springer, Berlin, Heidelberg, 2015.
- [12] A. Campbell, L. Clarke, A. Kleywegt, and M. Savelsbergh, "The Inventory Routing Problem," in *Crainic T.G., Laporte G. (eds) Fleet Management and Logistics*,

Springer, Boston, MA, 1998.

- [13] J. M. Ampuero-Nuño and J. Martín-Fernández, "Impacto de la crisis económica sobre la percepción de la salud en la población española," *Rev. Clínica Med. Fam.*, vol. 14, no. 2, pp. 57–63, 2021.
- [14] BBC News Mundo, "Coronavirus y la economía: 3 diferencias clave entre la Gran Recesión de 2008 y la actual crisis causada por la pandemia - BBC News Mundo." https://www.bbc.com/mundo/noticias-52987816 (accessed Dec. 16, 2020).
- [15] S. K. Paul and P. Chowdhury, "A production recovery plan in manufacturing supply chains for a high-demand item during COVID-19," *Int. J. Phys. Distrib. Logist. Manag.*, 2020, doi: 10.1108/IJPDLM-04-2020-0127.
- [16] R. Myro, "La industria española ante el COVID-19," *Cuad. Inf. Económica*, vol. 277, pp. 41–53, 2020, [Online]. Available: https://dialnet.unirioja.es/servlet/articulo?codigo=7501196.
- [17] R. Torres and M. J. Fernández, "Se inicia la recuperación, pero persisten las incertidumbres," *Cuad. Inf. Económica*, vol. 277, pp. 1–8, 2020.
- [18] T. Urli, "Hybrid meta-heuristics for combinatorial optimization," *Constraints*, vol. 20, no. 4, p. 473, 2015, doi: 10.1007/s10601-015-9209-7.
- [19] J. F. Shapiro, "Mathematical programming models and methods for production planning and scheduling," *Handbooks Oper. Res. Manag. Sci.*, vol. 4, no. C, pp. 371–443, 1993, doi: 10.1016/S0927-0507(05)80188-4.
- [20] G. R. Raidl, J. Puchinger, and C. Blum, "Metaheuristic Hybrids," in *Handbook of Metaheuristics*, M. Gendreau and J.-Y. Potvin, Eds. Cham: Springer International Publishing, 2019, pp. 385–417.
- [21] G. Cabrera-Guerrero, C. Lagos, C. Castañeda, F. Johnson, F. Paredes, and E. Cabrera, "Parameter tuning for local-search-based matheuristic methods," *Complexity*, vol. 2017, 2017, doi: 10.1155/2017/1702506.
- H. Kagermann, W.-D. Lukas, and W. Wahlster, "Industrie 4.0: Mit dem Internet der Dinge auf dem Weg zur 4. industriellen Revolution," VDI Nachrichten, no. 13, pp. 3–4, 2011, [Online]. Available: http://www.vdinachrichten.com/Technik-Gesellschaft/Industrie-40-Mit-Internet-Dinge-Weg-4-industriellen-Revolution.
- [23] H2020 Project C2NET, "'Cloud Collaborative Manufacturing Networks' (C2NET)." 2017, [Online]. Available: http://cordis.europa.eu/project/rcn/193440_en.html.
- [24] C. L. Dym, *Principles of mathematical modeling*, 2nd ed. Amsterdam ; Elsevier Academic Press, 2004.
- [25] J. F. Nunamaker, M. Chen, and T. D. M. Purdin, "Systems development in information systems research," *J. Manag. Inf. Syst.*, vol. 7, no. 3, pp. 89–106, 1990, doi: 10.1080/07421222.1990.11517898.
- [26] B. Andrés Navarro, B. E. Guzmán Ortiz, and R. Poler Escoto, "Synthetic input data generator for a MILP model for lot-sizing and scheduling on parallel flexible injection machines with setup common operators." Universitat Politècnica de València, 2021, [Online]. Available: http://hdl.handle.net/10251/161635.
- [27] B. Andrés Navarro, B. E. Guzmán Ortiz, and R. Poler Escoto, "Synthetic input data generator for a MILP model for lot-sizing and scheduling on parallel flexible injection machines." Universitat Politècnica de València, 2021, [Online]. Available: http://hdl.handle.net/10251/161636.
- [28] B. E. Guzmán Ortiz, B. Andrés Navarro, and R. Poler Escoto, "Synthetic input data generator for a MILP model for lot-sizing and scheduling of automotive plastic components with availability of raw materials and packaging." Universitat Politècnica de València, 2021, [Online]. Available: http://hdl.handle.net/10251/172395.
- [29] E. Guzmán, R. Poler, and B. Andres, "Un análisis de revisiones de modelos y algoritmos para la optimización de planes de aprovisionamiento, producción y distribución de la cadena de suministro," *Dir. y Organ.*, vol. 70, pp. 28–52, 2020, doi: https://doi.org/10.37610/dyo.v0i70.567.
- [30] E. Guzman, B. Andres, and R. Poler, "A Decision-Making Tool for Algorithm Selection Based on a Fuzzy TOPSIS Approach to Solve Replenishment, Production and Distribution Planning Problems," *Mathematics*, vol. 10, no. 9, 2022, doi: 10.3390/math10091544.
- [31] E. Guzman, B. Andres, and R. Poler, "A MILP Model for Reusable Containers Management in Automotive Plastic Components Supply Chain," in 22nd IFIP WG 5.5 Working Conference on VIRTUAL ENTERPRISES, PRO-VE 2021, 2021, p. 8p, [Online]. Available: https://hal-emse.ccsd.cnrs.fr/emse-03338406.
- [32] E. Guzman, B. Andres, and R. Poler, "Matheuristic Algorithm for Job-Shop Scheduling Problem Using a Disjunctive Mathematical Model," *Computers*, vol. 11, no. 1, pp. 1–16, 2022, doi: 10.3390/computers11010001.

Chapter 2

An analysis of reviews of models and algorithms for the optimisation of supply chain replenishment, production and distribution plans.

Guzmán, E, R Poler, and B Andres. 2020. "Un Análisis de Revisiones de Modelos y Algoritmos Para La Optimización de Planes de Aprovisionamiento, Producción y Distribución de La Cadena de Suministro." Dirección y Organización 70: 28–52. https://doi.org/https://doi.org/10.37610/dyo.v0i70.567.

Abstract:

This section provides an analysis of the different models and algorithms for the optimisation of sourcing, production and delivery plans in the supply chain based on the latest literature reviews in this field. It aims to provide researchers with a starting point to select models and algorithms to solve this type of problems in the supply chain and to present several future research lines.

2.1 Introduction

Since 1982, when Keith Oliver, a consultant at Boaz Allen Hamilton, introduced the term "supply chain management" during an interview for the *Financial Times* [1], different models and methods for supply chain design and management have been published in various studies, which are generally of the primary type (original research), while others of the secondary type (literature reviews) and tertiary studies (analyses of literature reviews) are scarce. In their review article on the closed-loop supply chain, Kazemi et al. [2] mention that, due to the volume of constantly appearing publications and their growth in recent years, it is difficult to maintain a broad overview of the study area, which makes the contribution of secondary or tertiary studies that help to synthesise and structure each research study necessary.

This article provides a tertiary study to conduct a systematic analysis of literature reviews on models and algorithms for the optimisation of procurement, production and distribution plans in the supply chain. It proposes useful guidelines for the research and development of new models and algorithms in these areas. The following research questions were posed for this study:

RQ1: What are the perspectives or approaches proposed in the literature on the supply chain?

RQ2: What is the current status of research into the optimisation of procurement, production and distribution plans?

RQ3: What are the most relevant reviews in the optimising procurement, production and distribution plans context in the supply chain?

RQ4: How can procurement, production and distribution plans be conceptualised?

RQ5: How can the quantitative methods used in the optimisation of procurement, production and distribution plans be conceptualised?

RQ6: What quantitative methods have been used in the literature for procurement, production and distribution plans?

RQ7: What future research lines can be proposed based on existing reviews and what research areas can be targeted for future studies?

To answer these research questions, a systematic analysis of literature reviews that focus on supply chain planning problems was conducted. It evaluated 168 articles and 17 were selected for their in-depth analysis. The selection criteria were based mainly on the reviews that had analysed mathematical models or quantitative methods, and had used a methodological process in the review. The found research works generally suggest review approaches, such as systematic,

narrative and meta-analysis. However, this study focuses on systematic reviews to facilitate comparisons of the quantitative methods used in the different supply chain approaches. In addition, the SCOR model is employed to classify the analyses produced by the selected articles.

This analysis aims to concisely summarise the type of planning addressed in the reviewed articles, and the models and methods followed to solve not only optimisation problems, but also algorithms and approaches. This research also aims to identify gaps and opportunities to propose future research directions in the supply chain operations planning field.

This article is organised as follows: Section 2.2 shows the approaches or perspectives studied in the literature and their definition. Section 2.3 explains the methodology followed in the systematic analysis. Section 2.4 presents the research findings. Section 2.5 shows the observations, perspectives and orientations given by researchers. Finally, Section 2.6 presents the conclusions and directions for future research.

2.2 Perspectives from reviews.

The supply chain concept has been described by several authors, who express similar features in their definitions. In turn these definitions are adapted to the analysis or research field. One definition that aptly summarises the term is that by Ivanov and Sokolov [3]: a supply chain: *"is a network of organizations, flows and processes wherein a number of various enterprises (suppliers, manufacturers, distributors and retailers) collaborate (cooperate and coordinate) along the entire value chain to acquire raw materials, to convert these raw materials into specified final products, and to deliver these final products to customers"*.

It is important to highlight the approaches or perspectives on supply chains and the direction that researchers are currently taking. Table 2.1 describes the supply chain research areas found in the literature. The definitions of these terms are intended to answer RQ1.

Perspective	Definition
Supply Chain Management	Supply chain management is the management of material, information and financial flows through the supply chain. It includes the coordination and collaboration of processes and activities across different functions, such as marketing, sales, production, product design, procurement, logistics, finance and information technology in the supply chain [4].

Table 2.1. Perspectives on supply chain analyses.

Table 2.1. Continued. Perspectives on supply chain analyses.					
Perspective Definition					
Supply Chain under Uncertainty	Supply chains that address uncertainty express how uncertainty propagates up and down and influences their performance. Uncertainty affects demand volumes and supply capacities that are subject to economic instability and market fluctuations in addition to other endogenous and exogenous factors [5].				
Closed-loop Supply Chain Management	A closed-loop supply chain can be seen as the traditional supply chain supplemented with reverse operations for recovered products that are reprocessed and eventually re-enter the supply chain. Closed-loop supply chain management describes the design, control and operation of a system to maximise value creation throughout the product's entire life cycle with a dynamic recovery of value from different types and volumes of returns over time [6].				
Sustainable Supply Chain Management	Sustainable supply chain management is an extension of the traditional supply chain management concept, and it adds economic, environmental and social/ethical aspects by taking into account the requirements of customers and other stakeholders [7, 8].				
Supply Chain Risk Management	Supply chain risk management is a multidisciplinary area where research and practice are based on at least three domains: supply chain management, enterprise risk management (identification, assessment, mitigation, response), and crisis management [4, 9].				
Green Supply Chain Management	It is the integration of environmental thinking into supply chain management, including product design, material selection and sourcing, manufacturing processes, delivery of the final product to consumers, and end-of-life management [10].				
Supply Chain Resilience	it addresses a supply chain's adaptive capacity to reduce the likelihood of facing sudden disruptions, resisting the propagation of disruptions while maintaining control over structures and functions, and recovering and responding through immediate and effective reactive plans to transcend the disruption and to restore the supply chain to a robust state of operations [11].				

Table 2.1. Continued.	Perspectives on	supply chain	analyses.
	i ciopeccites on	Supply chuin	unuiysesi

2.3 Methodology

The literature review forms part of the basic research structure. The objective of this study is to systematically analyse the reviews carried out on supply chains by taking into account the relation with the applied approach by identifying its key elements and the types of methods studied to date. This section provides an answer to RQ2.

In order to fulfil this objective, we follow the methodology proposed by Seuring and Müller [12] and Seuring and Gold [13], in which four steps are established to carry out a systematic review: 1) collection of material; 2) descriptive analysis; 3) selection of categories; 4) evaluation of the material.

2.3.1 Collection

The scope of the study starts with the selection of databases, and SCOPUS and Web of Science were selected. The employed search criteria are shown in Table 2.2. These search criteria were adapted to each specific database to ensure the robustness of the search.

Identification	Description
Databases	SCOPUS, Web of Science
Language	English
Scientific Areas	all
Magazines	all
Types of articles	literature review
Search field	title, abstract, keywords
Search date	February 2019
Years searched	2009-2019

Table 2.2. Database search criteria.

In order to ensure the adequate selection of articles, the search was limited to English-language journals, and excluded conference reviews, books, etc. Furthermore, the search was narrowed down by applying keywords based on the research questions (see Figure 2.1).

The search period was adjusted to the last 10 years as the research purpose was to find out what future research lines (for the next decade) were proposed by the authors. Some of the studied authors [14–17] analysed literature reviews and primary works. In line with the aforementioned researchers, Brandenburg et al. [14] analysed reviews from 1999 to 2012, Ho et al. [17] did so from 2006 to 2012, Malviya and Kant [15] from 2007 to 2013 and Barbosa-Póvoa et al. [18] from 2007 to 2016.







Figure 2.1 summarises the selection process of the literature reviews herein considered. From the first search in the selected databases, 168 references were obtained, and the elimination of duplicates left 123 articles. Reading the abstracts of the 123 articles allowed us to assess whether the articles matched our research questions. By applying this criterion, the number of articles dropped to 62, which were considered for further reviews. The full text of these 62 articles was reviewed in-depth by analysing their response to the following questions: does the study present models or methods to solve supply chain optimisation problems?; does the study describe a relation with procurement, production and distribution plans?; does the study describe a perspective and a sector of application?; does the study propose future research directions?. After evaluating this research, we identified three articles that were cited, but not captured, by our search criteria. These three articles were not grouped as literature reviews, but as articles. After reading these articles, we identified their relevance and assessed whether they matched our research questions. The collection yielded 65 articles that were analysed in-depth. Subsequently, 48 articles were excluded because they did not adequately match our research questions. Finally, we selected 17 to be studied (see Table 2.3).

Table 2.3. The selected review articles

Year	Authors	Title
2009	David Peidro, Josefa Mula, Raúl	Quantitative models for supply chain planning
	Poler, Francisco-Cruz Lario	under uncertainty: a review.
2010	Josefa Mula, David Peidro, Manuel	Mathematical programming models for supply
	Díaz-Madroñero, Eduardo Vicens	chain production and transport planning.

	Table 2.3. Continued. The selected review articles.						
Year	Authors	Title					
2014	Dennis Stindt, Ramin Sahamie	Review of research on closed loop supply chain management in the process industry.					
2014	Marcus Brandenburg, Kannan Govindan, Joseph Sarkis, Stefan Seuring	Quantitative models for sustainable supply chain management: Developments and directions.					
2015	Rakesh Kumar Malviya, Ravi Kant	Green supply chain management (GSCM): a structured literature review and research implications.					
2015	Çağrı Sel, Bilge Bilgen	Quantitative models for supply chain management within dairy industry: a review and discusion.					
2015	Kannan Govindan, Hamed Soleimani, Devika Kannan	Reverse logistics and closed-loop supply chain: A comprehensive review to explore the future.					
2015	William Ho, Tian Zheng, Hakan Yildiz, Srinivas Talluric	Supply chain risk management: a literature review.					
2016	Muhammad Salman Habib, Young Hae Lee, Muhammad Saad Memon	Mathematical Models in Humanitarian Supply Chain Management: A Systematic Literature Review.					
2016	Wladimir E. Soto-Silva, Esteve Nadal-Roig, Marcela C. González, Lluis M. Pla-Aragones	Operational research models applied to the fresh fruit supply chain.					
2017	Carlos Franco, Edgar Alfonso- Lizarazo	A Structured Review of Quantitative Models of the Pharmaceutical Supply Chain.					
2018	Nasim Zandi Atashbar, Nacima Labadie, Christian Prins	Modelling and optimisation of biomass supply chains: a review.					
2018	Faiza Hamdi, Ahmed Ghorbel, Faouzi Masmoudi, Lionel Dupont	Optimization of a supply portfolio in the context of supply chain risk management: literature review.					
2018	Cigdem Gonul Kochan, David R. Nowicki	Supply chain resilience: a systematic literature review and typological framework.					
2018	Ana Paula Barbosa-Póvoa, Cátia da Silva, Ana Carvalho	Opportunities and challenges in sustainable supply chain: An operations research perspective.					
2019	Isabel Mundi, M. M. E. Alemany, Raúl Poler & Vicente S. Fuertes- Miquel	Review of mathematical models for production planning under uncertainty due to lack of homogeneity: proposal of a conceptual model.					
2019	J.B. Oliveiraa, M. Jin, R.S. Lima, J.E. Kobza, J.A.B. Montevechia	The role of simulation and optimization methods in supply chain risk management: Performance and review standpoints.					

2.3.2 Descriptive analysis

In this phase, the 17 selected articles were analysed to answer RQ3. The evaluated characteristics were: publication date, type of publication and journal. Of the journals to which publications belonged, the *European Journal of Operational Research* and the *International Journal of Production Research* stood out (Table 2.4).

,	
Journal	Articles n=17
European Journal of Operational Research	5
International Journal of Production Research	3
Mathematical Problems in Engineering	1
Benchmarking: An International Journal	1
Complexity	1
European J. Industrial Engineering,	1
Flexible Services and Manufacturing Journal	1
The International Journal of Advanced Manufacturing Technology	1
International Journal of Physical Distribution & Logistics	1
Management	
Journal of Intelligent Manufacturing	1
Simulation Modelling Practice and Theory	1

Table 2.4. Distribution of the articles based on journals.

Analysing the publication dates provides information on the evolution of the research carried out into different supply chain perspectives. The selected articles covered a 10-year period, and it is noteworthy that 27.8% were published in 2018. The year 2015 was equally significant with 22.2%, while the remaining dates represented less than 11.1% (see Figure 2.2).



2.3.3 Category selection

The main analysis themes and structural dimensions included in this selection are described in Tables 2.5 and 2.6. The processes taking place in the supply chain can be classified with the SCOR (*Supply Chain Operations Reference*) model. This approach describes business processes to classify them into three categories, i.e. Procurement (S, *Source*), Production/ Manufacturing/ Manufacturing (M, *Make*), Distribution (D, *Deliver*), to which subtypes and schemes are added [19]. This classification solves RQ4 (Table 2.5).

Supply chain	Type of plan	Subtypes	Schemes				
Supplier	Procure- ment (S)	S/ Material requirements plan S/ Raw material inventory plan- ning S/Procurement planning S/ Procurement planning S/ Inventory planning	Procure- ment & Production (SM)	S/Inventory planning & M/ Produc- tion planning S/ Material requirements plan & M/ Produc- tion planning	Procure-	S/ Procure- ment plan- ning & M/ Production planning & D/ Distribu- tion planning	
Produc- tion/ Manufac- turing	Produc- tion (M)	M/Production planning M/ Production scheduling M/Production Sequencing		M/ Produc- tion Planning &	ment & Production & Distribu- tion (SMD)	S/Inventory planning & M/ Produc- tion planning	
Customer	Distribu- tion (D)	D/Demand planning D/Distribution planning D/Order promis- ing D/Transport planning	Production & Distribu- tion (MD)	D/ Distribu- tion Planning M/ Produc- tion Planning & D/ Transport Planning		& D/ Distri- bution plan- ning	

Table 2.5. Types and subtypes of plans according to the SCOR model.

Andres et al. [20] propose a classification of models and algorithms used to solve procurement, production and distribution plans as part of European Project "*Cloud Collaborative Manufacturing Networks* (*C2NET*)", presented in the deliverable "*Taxonomy of optimisation and simulation solutions for Manufacturing*

and Logistics Processes" [21]. Table 2.6 answers RQ5. It shows this classification and other parameters considered to assess the reviews.

Types of models	Analytical, Simulation, Mathematic	cal.				
Multicriteriadecisionmaking(MCDM);AnalyticHierProcess(AHP);AnalyticNetworkProcess(ANP);TechnicAnalyticalOrder ofPreference bySimilarity toIdealSolution(TElimination and ChoiceExpressingReality(ELECTRE);MultOptimisationandCompromiseSolution(VIKOR)						
Simulation	Agent-Based Simulation (ABS); Discrete Event Simulation (DES); System Dynamics Simulation (SDS); Monte Carlo Simulation (Monte Carlo Simulation); Other Simulation Methods					
Mathematical	Linear Programming (LP); Integer Linear Programming (ILP); Mixed Integer Linear Programming (MILP); Non-Linear Programming (NLP); Mixed Integer/Integer Non-Linear Programming; Quadratic Programming (QP); Dynamic Programming (DP); Stochastic Programming (SP); Robust Programming (RP); Constraint Programming (CP); Multi-Objective Linear Programming (MOLP) / Multi-Objective Non-Linear Problem (MONLP) / Multi-Objective Mixed-Integer Non-Linear Programming (MOMINLP); Hybrid Approach (HA): Euzzy Programming (EP): Goal Programming (GP).					
Resolution	Optimising Algorithms (OA), Heur	istics (HA), Metaheuristics (MA)				
methods						
Optimisers (OA)	Heuristics (HA)	Metaheuristics (MA)				
OA/ Decomposition OA/ Lomnicki	HA/ Multi-Objective Maste	er MA/Colony Optimisation				

Table 2.6. Classification of categories according to the taxonomies proposed by [14,21, 22].

2.3.4 Validation

In this phase, the analysed articles were validated according to the selected characteristics, approaches and dimensions by the hierarchical agglomerative grouping technique with a deductive and inductive process.

The assessment ensured that reviews were appropriately and sufficiently wellinformed to, thus, condition articles to match the parameters considered for classification purposes. This was one of the reasons why several publications were discarded. The analysis helped to examine the organisation, categorisation and structure, and the main findings of the systematic review, to motivate future research lines.

2.4 Results

This section describes the results obtained with the categorisation and evaluation of the selected reviewed articles by analysing elements like types of used schemes, models and resolution methods.

2.4.1 Perspectives for the analysis

The selected reviewed articles present different analysis perspectives: supply chain management, supply chain under uncertainty, closed-loop supply chain management, sustainable supply chain management, supply chain risk management, green supply chain management and supply chain resilience. This research analyses 2,309 articles over an approximate search period from 1983 to 2019. Table 2.7 shows which sectors these studies focus on. Several research works focus on a specific sector by studying all the publications that refer to the sector in relation to the analysed perspective. Other reviews analyse all the sectors from the analysis perspective. It is important to highlight that most articles analyse the reviews that concentrate between 2007 and 2013, which represent 60.64% of the articles during this period (Figure 2.3), where 2012 is the year with the most articles: 241 (10.75%).

Another aspect analysed by reviews is the decision-making levels during the planning process, where research analyses strategic, tactical and operational levels. Table 2.7 indicates which studies consider this aspect.



Chapter 2. An analysis of reviews of models and algorithms for the optimisation of supply chain replenishment, production and distribution plans.

Figure 2.3 Number of published articles per year cited in the analysed reviewed articles.

Author	Journal	Re- viewed articles	Analysis period	Analysed perspective	Study of decision levels	Analysed sector	Study focus
Peidro et al. [23]	The Interna- tional Journal of Advanced Manufacturing Technology	103	1988- 2007	Supply chain under uncer- tainty	x	Multisectoral	It focuses on supply chain planning under uncertainty by adopting quantitative ap- proaches.
Mula et al. [24]	European Jour- nal of Opera- tional Research	44	1989- 2009	Supply chain	x	Multisectoral	It studies mathematical pro- gramming models for supply chain production and transport planning, based on the analysis of eight aspects: structure, decision level, mod- elling approach, purpose, in- formation sharing, model limitations, novelty provided, practical application.
Bran- den- burg et al. [14]	European Jour- nal of Opera- tional Research	134	1994- 2012	Sustainable management of your sup- ply chain	-	Multisectoral	It studies quantitative models for sustainable supply chain management.

Tab	le 2.7. Ana	lysis of the	e focus of the	research	n works.
-----	-------------	--------------	----------------	----------	----------

	Table 2.7. Continued. Analysis of the focus of the research works.									
Author	Journal	Re- viewed articles	Analysis period	Analysed perspective	Study of decision levels	Analysed sector	Study focus			
Stindt and Sa- hamie [6]	Flexible Ser- vices and Manufacturing Journal	167	1994- 2012	Closed loop supply chain management	-	Chemical Industry, Metal/Steel Industry, Construction Indus- try, Paper/Pulp and Paper Industry, Pharmaceutical Industry, Plas- tics/Polymers In- dustry, Textile Industry	Describes quantitative ap- proaches to closed-loop supply chain planning in the process industry			
Malviya and Kant [15]	Benchmarking: An Interna- tional Journal	177	1998- 2013	Green supply chain man- agement	-	Multisectoral	It identifies and analyses research methods, data analysis techniques, multicriteria decision- making methods and the main industries actively involved in green supply chain management.			

Chapter 2. An analysis of reviews of models and algorithms for the optimisation of supply chain replenishment, production and distribution plans.

Author	Journal	Re- viewed articles	Analysis period	Analysed perspective	Study of decision levels	Analysed sector	Study focus
Sel and Bilgen [25]	European J. Industrial En- gineering	78	1983- 2013	Supply chain management	x	Dairy	It explores quantitative models for production planning and schedul- ing, distribution planning and vehi- cle routing problems (VRP) in dairy supply chain management.
Govinda n et al. [26]	European Journal of Operational Research	382	2007- 2013	Closed loop supply chain	x	Multisectoral	It presents a literature review of the optimisation methods used in re- verse logistics and closed loop sup- ply chains.
Ho et al. [17]	International Journal of Production Research	159	2003- 2013	Supply chain risk manage- ment	-	Multisectoral	It examines quantitative and quali- tative methods of supply chain risk management according to four main processes: identification, as- sessment, mitigation and monitor- ing of risks.
Habib et al. [27]	Mathemati- cal Problems in Engineer- ing	140	2005- 2015	Supply chain management	-	Humanitar- ian	It explores the mathematical opti- misation techniques and algorithms developed in humanitarian supply chain operations.

Table 2.7. Continued. Analysis of the focus of the research work

Chapter 2. An analysis	of reviews of	of models and	l algorithms	for the	optimisation	of supply	chain	replenishment,	production	and
distribution plans.										

	Table 2.7. Continued. Analysis of the focus of the research works.									
Author	Journal	Re- viewed articles	Analysis period	Analysed perspective	Study of decision levels	Analysed sector	Study focus			
Soto- Silva et al. [28]	European Journal of Operational Research	28	1976- 2015	Supply chain	x	Fresh fruit	It focuses on operational research models to solve decision problems related to the fresh fruit supply chain.			
Franco and Al- fonso- Lizarazo [29]	Complexity	46	1984- 2016	Supply chain	x	Pharmacist	It analyses network design, in- ventory models and optimisation of a pharmaceutical supply chain.			
Zandi Atashba r et al. [30]	International Journal of Production Research	110	1996- 2017	Supply chain	x	Biomass	It describes models, methods and solution approaches to optimise biomass supply chains.			
Hamdi et al. [31]	Journal of In- telligent Man- ufacturing	124	2003- 2014	Supply chain risk manage- ment	-	Multisectoral	It describes optimisation tech- niques and the quantitative, qualitative and hybrid approach to supply chain risk management.			

Table 2.7. Continued. Analysis of the focus of the research works

Tuble Eff Continueur Analysis of the focus of the rescuren works									
Author	Journal	Re- viewed articles	Analysis period	Analysed perspec- tive	Study of decision levels	Analysed sector	Study focus		
Kochan and Nowicki [22]	International Journal of Physical Dis- tribution & Logistics Management	228	2003- 2017	Supply chain re- silience	-	Multisectoral	It analyses quantitative and simu- lation methods; reviews defini- tions related to supply chain resilience and identifies measures and assessment techniques.		
Barbosa -Póvoa et al. [18]	European Journal of Operational Research	220	2007- 2016	Sustaina- ble sup- ply chain	x	Multisectoral	It refers to operational research methods to support sustainable supply chain activities.		
Mundi et al. [32]	International Journal of Production Research	87	1995- 2018	Supply chain un- der un- certainty	x	Petroleum/Agri- food/Remanufac- turing/ Timber sec- tor/Mining sec- tor/Ceramic sector	It analyses mathematical models for production planning under un- certainty due to non-homogene- ity.		
Oliveira et al. [33]	Simulation Modelling Practice and Theory	52	2000- 2017	Supply chain risk manage- ment	-	Multisectoral	It investigates simulation and op- timisation methods for the supply chain risk management approach.		

Table 2.7. Continued. Analysis of the focus of the research worl	ks.
------------------------------------------------------------------	-----

2.4.2 Supply chain processes

The selected papers focus their analysis on addressing procurement, production and distribution issues. Many of these articles conceptualise, categorise and describe the plans and models involved in supply chain planning processes. However, each publication includes its own methodology and classification criteria. To present a comparative table, the analysed planning processes are classified into the SCOR model categories, as shown in Table 2.8.

Supply chain process								
Author	Procurement	Production	Distribution					
	(Source)	(Make)	(Deliver)					
Peidro et al. [23]	х	x	х					
Mula et al. [24]	Х	х	х					
Stindt and Sahamie [6]	-	х	-					
Brandenburg et al. [14]	х	х	х					
Malviya and Kant [15]	х	х	х					
Sel and Bilgen [25]	х	x	х					
Govindan, Soleimani, and Kannan [26]	х	x	х					
Ho et al. [17]	х	х	х					
Habib, Lee, and Memon [27]	х	-	х					
Soto-Silva et al. [28]	х	х	х					
Franco and Alfonso-Lizarazo [29]	х	x	х					
Zandi Atashbar, Labadie, and Prins [30]	х	х	х					
Hamdi et al. [31]	х	x	х					
Kochan and Nowicki [22]	-	-	-					
Barbosa-Póvoa, da Silva, y Carvalho [16]	х	x	x					
Mundi et al. [32]	х	x	х					
Oliveira et al. [33]	х	x	х					

Table 2.8. Classification according to SCOR models.

2.4.3 Modelling approaches

Similarly, the models used to solve optimisation problems are classified to answer RQ6. This provides the analytical models (AHP, ANP, TOPSIS, ELECTRE, VIKOR) analysed by the majority of researchers. The same can be stated of the simulation methods detailed in Table 2.6. The most widely used mathematical modelling approaches are linear programming methods, mixed integer linear programming,

non-linear programming, stochastic programming, and multi-objective programming with its variants and fuzzy programming. Table 2.9 describes the types of models analysed in the reviews.

Generally speaking, researchers use different solvers, such as CPLEX or Gurobi, to solve models. However, only a few studies consider this aspect in their reviews [6, 28, 31, 34].

Author							Тур	es of	fmo	dels					
Author	Α	S	1	2	3	4	5	6	7	8	9	10	11	12	13
Peidro et al. [23]	х	х	x	х	x	х		х	х			х	х	x	
Mula et al. [24]		х	х	х	х	х			х			х	х	х	
Stindt and Sahamie [6]	х		x		х				х				x		
Brandenburg et al. [14]	x	х	x		x	x		x	х				х		х
Malviya and Kant [15]	х	х	х												
Sel and Bilgen [25]	х	х	x	х	х				х		х	х	х	х	
Govindan, Soleimani, and Kannan [26]	х	x	x		х	х		х						х	х
Ho et al. [17]	х	х	x	x	х	x	х	х	х		х	х	х	х	х
Habib, Lee, and Memon [27]		x	x	х	х	х								х	
Soto-Silva et al. [28]		х	x	х	х	х		х	х			х	х		
Franco and Alfonso- Lizarazo [29]		x			х			х	х					x	
Zandi Atashbar, Labadie, and Prins [30]	х	х	x		х	х	x		x	х		х		х	х
Hamdi et al. [31]	х	х	x		х	х			x			x	х	х	x
Kochan and Nowicki [22]	x	x		х	х				х			x	x	х	x
Barbosa-Póvoa, da Silva, and Carvalho [16]		x													
Mundi et al. [32]			x			х			х			х	х	х	
Oliveira et al. [33]	х	х	x	х	х	х	х		х	х		х		х	

Table 2.9. Types of models found in the reviews.

- A. Analytical Models
- S. Simulation models
- 1. Linear Programming (LP)
- 2. Integer Linear Programming (ILP),
- 3. Mixed Integer Linear Programming (MILP),
- 4. Non-Linear Programming (NLP)/Mixed integer/Integer nonlinear programming
- 5. Quadratic Programming (QP),
- 6. Dynamic Programming (DP),
- 7. Stochastic Programming (SP),
- 8. Robust Programming (RP),
- 9. Constraint Programming (CP),
- Multi-Objective Linear Programming/Multi-Objective Non-Linear Programming/Multi-Objective Mixed-Integer Non-Linear Programming (MOLP)/Multi-Objective Non-Linear Problem (MONLP)/Multi-Objective Mixed-Integer Non-Linear Programming (MOMINLP),
- 11. Hybrid Approach (HA)
- 12. Fuzzy programming (FP)
- 13. Goal Programming (GP)

2.4.4 Solution methods

Nowadays, analysing the calculation methods followed to solve optimisation problems is extremely relevant because it is a very broad research field and many industry sectors emphasise its development. Table 2.10 describes the algorithms used in publications by bearing in mind that the classification is done in relation to the resolution methods proposed in Table 2.6, which specifies the categorisation of each algorithm type. Researchers generally propose using heuristic and metaheuristic algorithms for solving large problems (real problems), which leaves a large area to be explored: the so-called matheuristic algorithms. It is noteworthy that only two research works, namely Habib et al. (2016) and Zandi Atashbar et al. (2018), compare which methods or solution techniques solve optimisation models.

Author	Optimiser	Heuristic	Metaheuristic	Matheuristic
Peidro et al. [23]	х	х	х	
Mula et al. [24]		х	х	
Stindt and Sahamie [6]	х	х	х	
Brandenburg et al.[14]		х	х	
Malviya and Kant [15]		х		
Sel y Bilgen [25]		х	х	
Govindan, Soleimani, and Kannan [26]	Х	Х	x	

Table 2.10. Types of resolution methods.

Models and algorithms for the optimisation of replenishme	ent, production and distribution
plans in industrial enterprises.	

Table 2.10. Continued. Types of resolution methods.								
Author	Optimiser	Heuristic	Metaheuristic	Matheuristic				
Ho et al.[17]			х					
Habib, Lee, and Memon [27]	х	х	х					
Soto-Silva et al.[28]		х	х					
Franco and Alfonso-Lizarazo [29]	х	х	x					
Zandi Atashbar, Labadie, and Prins [30]	х	х	x	x				
Hamdi et al. [31]	х	х	х	х				
Kochan and Nowicki [22]								
Barbosa-Póvoa, da Silva, and Carvalho[16]			x					
Mundi et al. [32]								
Oliveira et al. [33]	х	х	х					

2.5 Discussion and future research opportunities

This section discusses the results of the review and research lines proposed by the analysed reviews. Moreover, the studies in this analysis are compared to the perspective or approach to which they are addressed. Table 2.11 briefly summarises the main research lines proposed by the reviews. This section answers RQ7.

Table 2.11. Summary of future research lines from different supply chain perspectives.

Perspective	Future research lines	References
Supply chain under uncertainty	 Accurate identification of sources of uncertainty Development of optimisation models and solution techniques to address demand uncertainty for products with non-homogeneities Aggregate production planning study for reconfigurable manufacturing systems Study of aggregate production planning in the process industry in uncertain or deterministic scenarios by considering a rolling horizon Application to real cases 	[24, 33, 35]

	perspectives.	
Perspective	Future research lines	References
Closed loop supply chain	 Development of models that look specifically at a particular industry rather than a general application Use of approaches such as Interval Approaches and Chaos Theory Application of two-stage stochastic approaches Use of non-deterministic approaches Use of robust optimisation 	[2, 6, 27]
Green supply chain	 Cross-sectorial industry benchmarking studies Cross-continental comparative industry studies Application to real cases 	[35]
Sustainable supply chain	 Analysis of social aspects, and their quantification and integration with economic and environmental aspects Environmental risk management Study of hybrid approaches by combining metaheuristics, matheuristics or other types of more efficient methods Application to sectors like pharmaceuticals, clothing, energy and transport Sustainability analysis in areas like transport and warehousing 	[14, 16]
Supply chain risk	 Infrastructure, financial and information risk analysis Investigation of production and process risk Measurement of the relation between risk factors and the type of risk by relating it to the probability of occurrence Validation of risk management methods to assess and select the best risk mitigation strategies by examining between individual and integrated strategies Analysis of the supply chains of services such as banking, insurance, medicine, etc. Use of simulation methods to reproduce risk dynamics and risk impacts 	[32, 34, 37]

Table 2.11. Continued. Summary of future research lines from different supply chainperspectives.

perspectives.		
Perspective	Future research lines	References
Supply chain risk	 Development of risk management models, such as the Ant Colony model or the Normal Boundary Intersection method Use of stochastic or fuzzy programming techniques Hybridisation of mathematical programming approaches with artificial intelligence techniques for automated decision support based on prediction and learning 	[32, 34, 37]
Resilience in the supply chain	 Analysis of the relation linking green supply chain, resilience and resilience analyses in Industry 4.0 Use of analytical methods like AHP, ANP, TOPSIS, ELECTRE, VIKOR Explore <i>Multicriteria Decision-Making</i> (MCDM) methods. Development of two-stage stochastic programming techniques with multiple objectives and use of <i>Constraint Programming</i> 	[23, 38]

 Table 2.11. Continued. Summary of future research lines from different supply chain perspectives.

Table 2.12 specifies some sectors that propose interesting lines of research in supply chain management.

Supply chain management				
Sector	Future research lines	References		
Humanitarian	 Realistic modelling that takes into account real- life aspects, such as policies, practices and procedures for disaster management and debris processing Study of humanitarian supply chain management by adopting green supply chain policies 	[27]		
Dairy	Study of models that consider issues like traceability and food safety aspects, and other aspects like waste minimisation	[25]		

 Table 2.12. Summary of future research lines from a supply chain management perspective considering several sectors.

management perspective considering several sectors			
Sector	Future research lines	References	
Fresh fruit	Design of holistic approaches to design and manage agricultural and perishable supply chains Study organic fruit production Study integrated simulation and optimisation techniques by considering the evolution from a single-criteria to a multicriteria approach	[28]	
Pharmaceuticals	 Study of quantitative methods for the healthcare area or for hospital logistics studies Use of combined or hybrid techniques to approximate the functioning of the pharmaceutical supply chain to real-life cases Use of techniques like fuzzy programming or robust programming 	[29]	
Biomass	Use of decomposition or relaxation techniques, metaheuristic techniques or evolutionary algorithms, and simulation-based optimisation methods	[30]	

Table 2.12. Continued. Summary of future research lines from a supply chain

Below a detailed analysis of future research lines from different perspectives in the supply chain and supply chain management from a sectoral perspective is found.

2.5.1 Supply chain under uncertainty

Research into supply chain uncertainty converges insofar as the full set of potential sources of uncertainty is not yet identified regardless of the study field. Similarly, the analysed reviews indicate that studies applied to real cases are lacking.

In light of this, two reviews study the supply chain from the uncertainty perspective [24, 33]. Although these two studies were conducted 10 years apart, they both agree that future work should develop fuzzy programming models to handle imprecise and/or unavailable data. It is worth noting that Mundi et al. [32] use the work of [23] as a basis for their research, but add the non-homogeneity feature to outputs. Peidro et al. [23] suggest other research lines, such as developing new hybrid models by combining models based on both fuzzy set theory and simulation, which can consider predictive control approaches. They also highlight that modelling approaches are not applied to real cases. These

authors emphasise using models based on artificial intelligence and analysing uncertainty management mechanisms from a quantitative perspective. They also suggest developing optimisation models, while Mundi et al. [32] indicate developing optimisation models and solution techniques to address uncertainty in the demand for products with non-homogeneity.

The supply chain perspective under uncertainty is a widely studied topic. One example of it is the review by Jamalnia et al. [34], which analyses aggregate production planning under uncertainty. It concludes that fuzzy programming models are the most widely studied, followed by stochastic programming. So we conclude that the future research lines identified by Peidro et al. [23] have been addressed by many authors and remain valid. However, this research area still contains gaps to be bridged, such as those mentioned in the study by Mundi et al. [32] and those presented by Jamalnia et al. [34]. The latter mentions that there are very few studies on aggregate production planning for reconfigurable manufacturing systems, and also in the process industry (oil refineries, beverage manufacturing and chemical processes) in uncertain or deterministic scenarios that consider a rolling horizon.

2.5.2 Closed loop supply chain

The closed loop supply chain presents an area with many research opportunities. Recent publications, like that of Kazemi et al. [2], state that most studies into the closed-loop supply chain focus only on direct economic consequences, such as operating costs and profits from sales. However, the closed-loop supply chain seeks to mitigate environmental damage and risks, and to improve social life. So modelling environmental, social and economic parameters together can be an important research area.

All these observations also relate to previous studies, like that of Govindan et al. [26], which also suggests studying non-deterministic approaches, such as Interval Approaches and Chaos Theory, to deal with uncertainty, and considering that different non-deterministic approaches can be integrated. They also contemplate employing two-stage stochastic approaches and robust optimisation to deal with the uncertainty that characterises real situations. They propose utilising non-linear programming and convex optimisation to deal with real problems. Govindan et al. [26] propose investigating the application of approximation algorithms or hybrid algorithms because they can present an acceptable solution for solving complex problems in shorter computational times.

Finally, they propose analysing uncertainty forecasting methods as their findings are mostly conceptual.

In the same vein, Stindt and Sahamie [6] propose developing sector-specific quantitative models and solution methods for each sector of the process or manufacturing industry, which take into account each sector's circumstances individually because general parameters of this industry have been analysed in models.

We highlight that research opportunities in this field are still to be exploited, which leads to the conclusion that the closed-loop supply chain is making very rapid progress in relation to the literature. One example of this is the study by Kazemi et al. [2], which analyses 94 studies between 2000 and 2017, and updates the study by Govindan et al. [26]. Both reviews agree with Stindt and Sahamie [6] about models developed specifically for particular industry types are lacking. The authors complement their findings by emphasising that the models that may work in one specific sector might not work in another. In turn, they consider using methods, such as Chaos Theory or Rough Set Theory, as well as robust optimisation, to deal with uncertainty.

2.5.3 Green supply chain

Industries are becoming increasingly committed to incorporate the "green" concept into supply chains. This means incorporating the environmental concept into all product stages, from design to end-of-life. In the final stage, products can be recycled, reused or reclaimed [38].

One of the main findings in the green supply chain context is that most authors propose simulation methods to model problems, while mathematical models are used to a limited extent [15]. Conversely to this finding, recent publications like that by Tseng et al. [39] reveal that more articles propose mathematical models for optimising decision making in green supply chain management.

All in all, green supply chain management is an area that, according to Tseng et al. [39], is being shown a lot of interest given the pressure placed by governments on companies to improve environmental performance. Although the future research lines set out in 2015 to 2019 have already been covered, according to Tseng et al. [39] some areas still need to be explored. The line of research suggesting cross-sectoral comparative industry studies and cross-continental comparative studies is worth mentioning, as such research is limited in the

literature. Tseng et al. [39] also emphasise that studies presenting real-life data are lacking.

2.5.4 Sustainable supply chain

Sustainable supply chain management is an approach that pushes companies to improve from a sustainability perspective, i.e. companies taking care to integrate economic, social and environmental objectives into supply chain processes [40].

In relation to this topic, Brandenburg et al. [14] have studied the quantitative methods used in sustainable supply chain management. They describe how the models that have dealt with this issue are related mainly to manufacturing industries, such as the automotive and textile industries. Very few studies are framed within the chemical and petroleum sectors. These authors also identify that no attention is paid to environmental risk management and the incorporation of social aspects into the proposed models. Finally, Brandenburg et al. [14] also propose analysing sustainability in areas like transport and storage by highlighting that evolutionary algorithms and techniques, such as dynamic programming and local search methods, have not been studied in-depth.

A more up-to-date review of this topic is presented by Barbosa-Povoa et al. [18]. They identify how mathematical programming methods are the most widely used, followed by simulation techniques. As for combining techniques, the most widely used combination is mathematical programming and simulation techniques, followed by analytical methods and metaheuristics. Barbosa-Povoa et al. [18] agrees with Brandenburg et al. [14] that only a few models address social aspects.

Barbosa-Povoa et al. [18] highlight that optimisation techniques peaked in publications in 2015 and real industrial applications using optimisation techniques in their publications markedly grew. This work agrees with Brandenburg et al. [14] that the manufacturing sector (automotive, textile, etc.) has been well-studied. However in the 4 years difference between both reviews, publications proposing optimisation models applied to the process industry have increased and occupy first place, as Barbosa-Povoa et al. [18] mention. This also agrees with Brandenburg et al. [14] insofar as social aspects should be studied and quantified, and they propose studying hybrid approaches by combining metaheuristics, matheuristics, or other types of more efficient methods. For industrial approaches, sectors like pharmaceuticals, clothing, energy and transport require further study.

In short, sustainable supply chain management is an area that presents important research lines, such as those mentioned by Barbosa-Povoa et al. [18] who, in many of their research works, coincide with and extend the research by Brandenburg et al. [14]. These research lines are generally related to the fact that some industry sectors need to be studied and their models must combine economic, social and environmental aspects by employing robust techniques. It is also worth highlighting what Barbosa-Povoa et al. [18] mention, who highlight the existence of a research gap in topics, such as closed-loop supply chains and reverse supply chain management, that address economic, social and environmental issues in supply chain processes.

We conclude that reverse supply chain management is a growing topic with significant research areas. However, the search conducted in this article found no studies that analyse this topic in detail and can answer our research questions.

2.5.5 Supply chain risk

According to Baryannis et al. [36], researchers' interest in supply chain risk management has increased significantly in the last two decades for three reasons: (i) adoption of philosophies such as "*lean manufacturing*" or "*just-in-time*" in manufacturing processes, which has left less scope for error and change; (ii) exposure to more risks because companies are becoming more global and less vertical; (iii) likelihood of disasters or events occurring and disrupting the global supply chain, such as natural disasters, shortages, e.g.: raw materials, etc.

In this context, studies like those by Ho et al. [17] recommend analysing infrastructural risks, such as transport, financial and information risk, and examining manufacturing or production risks and process risks. They also propose measuring the relation between risk factors and the type of risk by relating this relation to probability of occurrence. Ho et al. [17] emphasise the validation of conceptual frameworks and risk management methods to evaluate and select the best risk mitigation strategies by examining between individual and integrated strategies to, thus, measure their efficiency and effectiveness. These authors also propose analysing supply chains of services, such as banking, insurance, medicine, etc.

Hamdi et al. [31] studied supplier selection from the supply chain risk management approach. They note that quantitative research is predominant in this field, especially in mathematical formulation terms. AHP approaches to supplier selection, and stochastic optimisation approaches used to model fluctuations and incoming changes in supply chains, are also prevalent. These authors propose investigating combining techniques to minimise the supplier

risk, i.e. by analysing the case of simultaneous disruption of local, global and semiglobal suppliers.

Oliveira et al. [33] address the issue of risk mitigation strategies by concluding that there is no alignment between risk response and risk solution approaches because risk response strategies must be consistent with risk mitigation solutions. Their study analyses the application of simulation tools, optimisation models and performance measurement systems for supply chain risk management. On simulation and optimisation, these authors propose using simulation methods to reproduce risk dynamics and risk impacts, as well as risk management models, such as Ant Colony and the Normal Boundary Intersection method. As these methods do not appear in the reviewed articles, Oliveira et al. [33] suggest not only improving the simulation-based optimisation perspective, but also developing Six Sigma-based risk measures to assess the critical impacts of risk on supply chain performance.

As a result of this research, we conclude that the supply chain risk management area presents several interesting research lines, such as those put forward by Baryannis et al. [36], who reviewed 276 articles on supply chain risk management and its relation with artificial intelligence techniques. Of the 276 articles analysed by Baryannis et al. [36], 84% focus on risk response in models that avoid or mitigate risk and uncertainty effects, 4% combine the risk response with some form of risk assessment, 4% combine risk assessment with risk identification, 3% address risk identification individually, 2% assess risks and 3% use holistic approaches to risk identification, assessment and response. These authors also describe that most followed techniques are mathematical programming, which falls in line with the conclusions drawn by the study of Hamdi et al. [31]. In this sense, both reviews conclude that researchers have been more inclined to develop mathematical programming techniques. Artificial Intelligence techniques such as agent- and network-based models, automatic reasoning, machine learning and Big Data analysis are less applied. Both papers identify that techniques like stochastic programming or fuzzy programming are scarcely used. Finally, they mention that the artificial intelligence field for proactive and predictive risk management in supply chains is a marvellous research opportunity and propose investigating hybridisation between mathematical programming approaches together with an artificial intelligence technique to help automated decision making based on prediction and learning.

2.5.6 Resilience in the supply chain

One issue related to supply chain risk management is supply chain resilience. In this context, the study by Kochan and Nowicki [22] mentions that no unanimous supply chain resilience definition exists. Yet according to Burnard et al. [41], resilience offers companies a dynamic capacity to anticipate, respond and adapt to risks and threats.

On resilience in the supply chain, one of the most recent studies is that by Hosseini et al. [37], in which interesting research areas like using analytical methods, including AHP, ANP, TOPSIS, ELECTRE, VIKOR are raised, but have not been analysed in the supply chain resilience context. It also recommends future research lines to focus on exploring multicriteria decision-making (MCDM) methods, and conclude that such methods are applicable for assessing supply chain networks, and in relation to resilience, ecological and organisational criteria.

Another aspect proposed by Hosseini et al. [37] is to focus on two-stage stochastic programming with multiple objectives because supply chain resilience is intrinsically related to stochastic aspects likes capacity loss, capacity recovery, resource restoration and the expected recovery time. They also propose developing robust optimisation models which, according to their research, is yet to be addressed. These authors also propose employing Constraint Programming in the supply chain resilience context. They generally propose studying the relation among green supply chains, resilience and resilience analyses in Industry 4.0. Ultimately, the future research areas raised by these recent research works offer researchers several study possibilities.

2.5.7 Supply chain management.

Supply chain management is a very large area that is applicable to different sectors and involves distinct approaches. This section provides details of some sectors that propose interesting research lines.

2.5.7.1 Sector: Humanitarian

A new area that is beginning to emerge is the humanitarian supply chain. According to Habib et al. [27], this sector has a large unexplored research area as most of the articles on these topics emerged after the 2004 Indian Ocean earthquake, and they confirm that the commercial scope of supply chain management with a focus on sustainability has been considerably investigated, but humanitarian supply chain management has not. The review by Habib et al. [27] concludes that the formulation of humanitarian supply chain models is

unrealistic and difficult to apply. These authors propose that humanitarian supply chain management should adopt green supply chain policies and, when generating new models, several real-life aspects like disaster management and debris processing policies, practices and procedures should be taken into account.

2.5.7.2 Sector: dairy

Sel and Bilgen [25] review and discuss quantitative models for supply chain management in the dairy industry. Most of the articles analysed by these authors have used mixed integer programming models, which is why they propose that these models should take into account key aspects for the sector, such as traceability and food safety. Other future research lines are defined in line with waste minimisation issues and Constraint Programming in production and distribution planning, but have not yet been addressed. In the same vein, Sel and Bilgen [25] mention that very few publications use the heuristics, metaheuristics or hybridisations of these methods. They also propose applying heuristics based on relaxed mixed integer programming models and rolling horizon approaches. These authors state that there are very few combinations of mathematical programming models with approximate solution methods in the dairy supply chain context. Other proposed further research suggestions include the development of stochastic or multiparametric programming models.

2.5.7.3 Sector: fresh fruit

Soto-Silva et al. [28] review the operations research models applied to the fresh fruit supply chain. These authors state that holistic approaches for the design and management of agricultural supply chains are lacking, and the same is true for perishable products. Topics like optimal crop scheduling and discrete event logistics simulation have been addressed in this field. Sustainability and environmental issues, as well as food security issues, are given a high profile. Soto-Silva et al. [28] conclude that the problems encountered in their review are related to transport, planning and allocation in production and distribution stages, with linear programming, integer linear programming or mixed integer linear programming models being the most widely used. In this area, and as mentioned in the review, the analysed articles are real cases. Indeed the authors propose analysing organic fruit production and, in optimisation issues, using integrated simulation and optimisation techniques and moving from a single-criteria approach to a multicriteria one.

2.5.7.4 Sector: pharmaceuticals

Franco and Alfonso-Lizarazo [29] analyse the pharmaceutical supply chain by mentioning that it is not an exhaustively studied topic mainly in the quantitative methods proposed in the healthcare area or hospital logistics studies. These authors describe how most of the analysed studies apply a stochastic form of uncertainty in some supply chain stage, and only 20% correspond to supply chain optimisation models, 52% to network design and 28% to inventory problems. Of all the presented models, 92% stochastically use demand data and the rest do so deterministically. They also point out that very few articles employ fuzzy programming, Franco and Alfonso-Lizarazo [29] highlight that a few articles apply fuzzy programming or robust programming, and propose combining techniques to approximate the functioning of pharmaceutical supply chains to real-life cases.

2.5.7.5 Sector: biomass

Zandi Atashbar et al. [30] mention that most research studies into supply chain management in the biomass sector come from agriculture or chemistry. Researchers in these areas are experts in these topics, but very few specialists are found in the operations research area because they are involved in the biomass sector, which is an important research area and for input of knowledge acquired from other industrial sector areas. Zandi Atashbar et al. [30] also identify how decomposition or relaxation techniques and metaheuristic techniques or evolutionary algorithms are barely used, and propose applying simulation-based optimisation methods.

This section analyses review articles that deal with the supply chain management concept by considering different relevant sectors. However, other authors propose reviews in the supply chain management context regardless of the sector. Thus Mula et al. [24] study the supply chain independently of the sector by identifying that proposals to simultaneously optimise production and transport planning are lacking, and propose that the generation of these models should take into account transport characteristics like: environmental restrictions and transport mode considerations. These authors also propose integrating tactical or operational decision levels and a collaborative planning structure to manage information sharing with supply chain partners. Mula et al. [24] also recognise a future research line, that of integrating optimisation tools: simulation, fuzzy optimisation, multi-agent systems and evolutionary algorithms.

In short, the number of studies that analyse models and algorithms for the optimisation of supply chain procurement, production and distribution plans continues to increase. Notwithstanding, a small number of works clearly address

the review of specific areas, with generally a few developments with real practical applications in the research area herein addressed.

2.6 Conclusions

This study reviews quantitative models and operations research methods to solve optimisation problems related to procurement, production and distribution in the supply chain to gain a better understanding of the supply chain areas with currently research trends. After contemplating the reviews in the literature, as far as we know, no review focuses on analysing these reviews, which poses a pressing need as research growth makes it difficult for researchers to be clear about not only the supply chain area that needs to be investigated, but also the tools employed to solve supply chain optimisation problems in supply chains.

Consequently, this study provides a comprehensive analysis of the current research status by considering that these studies focus specifically on a particular area, and present a content analysis that comprises a differentiation of the categories that answer the posed research questions. In particular, supply chain processes are identified, consolidated and synthesised with the SCOR model, and methods and models are put forward in the reviews to solve optimisation problems by exploring key issues, research trends and the developments made in the different supply chain areas.

In line with all this and to answer RQ7, the analysis shows that most reviews agree that very few studies employ real data, and that comparative studies between different industry sectors are lacking because success in one area can be extrapolated to another. So it would be interesting to analyse this issue. Many models and conceptual frameworks appear and act as an important theoretical basis. However, it is important to validate these models and to reach a consensus about conceptualisations to, thus, generate an integral vision.

Another important finding is that sustainability and environmental approaches are increasingly considered in modelling. However, there is still a large area to explore because, despite its growth, reviews agree that the variables to which models are subjected need to incorporate the new constraints demanded by today's governments and markets. In addition, robust optimisation and fuzzy programming methods are still a research opportunity in modelling approaches. Similarly at decision levels, it is still important to analyse their combination, and topics like the rolling horizon have only been used in some articles and very few reviews briefly mention the topic. This makes it a significant area to be explored.

Finally, it is very important to highlight that a growing number of articles present mathematical and analytical models, which is an area where research is still to be done: simulation models and their combination with other methods. Many models employ heuristic or metaheuristic methods for solutions, which means that the hybridisation area of these methods or the use of matheuristic methods remains to be explored. Although several studies agree that the most widely used mathematical models are linear/integer/mixed integer programming models, given the size of real problems it would still be interesting to analyse matheuristic methods with large datasets, which are not currently approachable by commercial solvers. This is, hence, a remarkable research area that will allow the transformation of mathematical programming methods into models to be solved with matheuristic algorithms.

2.7 References

- [1] R. B. Handfield and E. L. Nichols, *Introduction to supply chain management*. Upper Saddle River: Prentice Hall, 1999.
- [2] N. Kazemi, N. M. Modak, and K. Govindan, "A review of reverse logistics and closed loop supply chain management studies published in IJPR: a bibliometric and content analysis," *Int. J. Prod. Res.*, vol. 7543, pp. 1–24, 2018, doi: 10.1080/00207543.2018.1471244.
- [3] D. Ivanov and B. Sokolov, *Adaptive supply chain management [electronic resource]*, 1st ed.. London: London, 2010.
- [4] M. S. Sodhi and C. S. Tang, *Managing supply chain risk*, vol. 172. New York, NY: Springer, 2012.
- [5] M. Fathian, J. Jouzdani, M. Heydari, and A. Makui, "Location and transportation planning in supply chains under uncertainty and congestion by using an improved electromagnetism-like algorithm," *J. Intell. Manuf.*, vol. 29, no. 7, pp. 1447–1464, 2018, doi: 10.1007/s10845-015-1191-9.
- [6] D. Stindt and R. Sahamie, "Review of research on closed loop supply chain management in the process industry," *Flex. Serv. Manuf. J.*, vol. 26, no. 1–2, pp. 268–293, 2014, doi: 10.1007/s10696-012-9137-4.
- [7] A. M. Sacaluga, J. G. Arca, J. C. P. Prado, A. J. F. González, and J. A. C. Benavides, "Modelo para la aplicación de la Responsabilidad Social Corporativa en la Gestión de la Cadena de Suministro," *Dir. y Organ.*, vol. 45, pp. 20–31, 2011.
- [8] A. B. Patel and T. N. Desai, "A systematic review and meta-analysis of recent developments in sustainable supply chain management," *Int. J. Logist. Res. Appl.*, vol. 0, no. 0, pp. 1–22, 2018, doi: 10.1080/13675567.2018.1534946.

- [9] B. Fahimnia, C. S. Tang, H. Davarzani, and J. Sarkis, "Quantitative models for managing supply chain risks: A review," *Eur. J. Oper. Res.*, vol. 247, no. 1, pp. 1– 15, 2015, doi: 10.1016/j.ejor.2015.04.034.
- [10] S. K. Srivastava, "Green supply-chain management: A state-of-the-art literature review," Int. J. Manag. Rev., vol. 9, no. 1, pp. 53–80, 2007, doi: 10.1111/j.1468-2370.2007.00202.x.
- J. Pires Ribeiro and A. Barbosa-Povoa, "Supply Chain Resilience: Definitions and quantitative modelling approaches – A literature review," *Comput. Ind. Eng.*, vol. 115, no. November 2017, pp. 109–122, 2018, doi: 10.1016/j.cie.2017.11.006.
- [12] S. Seuring and M. Müller, "From a literature review to a conceptual framework for sustainable supply chain management," *J. Clean. Prod.*, vol. 16, no. 15, pp. 1699–1710, 2008, doi: 10.1016/j.jclepro.2008.04.020.
- [13] S. Seuring and S. Gold, "Conducting content-analysis based literature reviews in supply chain management," *Supply Chain Manag.*, vol. 17, no. 5, pp. 544– 555, 2012, doi: 10.1108/13598541211258609.
- [14] M. Brandenburg, K. Govindan, J. Sarkis, and S. Seuring, "Quantitative models for sustainable supply chain management: Developments and directions," *Eur. J. Oper. Res.*, vol. 233, no. 2, pp. 299–312, 2014, doi: 10.1016/j.ejor.2013.09.032.
- [15] R. K. Malviya and R. Kant, "Green supply chain management (GSCM): a structured literature review and research implications," *Benchmarking An Int. J.*, vol. 22, no. 7, pp. 1360–1394, Oct. 2015, doi: 10.1108/BIJ-01-2014-0001.
- [16] A. P. Barbosa-Póvoa, C. da Silva, and A. Carvalho, "Opportunities and challenges in sustainable supply chain: An operations research perspective," *Eur. J. Oper. Res.*, vol. 268, no. 2, pp. 399–431, 2018, doi: 10.1016/j.ejor.2017.10.036.
- [17] W. Ho, T. Zheng, H. Yildiz, and S. Talluri, "Supply chain risk management: A literature review," *Int. J. Prod. Res.*, vol. 53, no. 16, pp. 5031–5069, 2015, doi: 10.1080/00207543.2015.1030467.
- [18] A. P. Barbosa-Povoa, C. da Silva, and A. Carvalho, "Opportunities and challenges in sustainable supply chain: An operations research perspective," *Eur. J. Oper. Res.*, vol. 268, no. 2, pp. 399–431, Jul. 2018, doi: 10.1016/j.ejor.2017.10.036.
- [19] B. Andres, R. Sanchis, J. Lamothe, L. Saari, and F. Hauser, "Integrated production-distribution planning optimization models: A review in collaborative networks context," *Int. J. Prod. Manag. Eng.*, vol. 5, no. 1, p. 31, 2017, doi: 10.4995/ijpme.2017.6807.
- [20] B. Andres, R. Sanchis, R. Poler, and L. Saari, "A Proposal of Standardised Data Model for Cloud Manufacturing Collaborative Networks," in *Collaboration in a Data-Rich World*, 2017, pp. 77–85.
Chapter 2. An analysis of reviews of models and algorithms for the optimisation of supply chain replenishment, production and distribution plans.

- [21] C2NET, "Taxonomy of optimisation and simulation solutions for Manufacturing and Logistics Processes." 2016.
- [22] C. G. Kochan and D. R. Nowicki, "Supply chain resilience: a systematic literature review and typological framework," *Int. J. Phys. Distrib. Logist. Manag.*, vol. 48, no. 8, pp. 842–865, 2018, doi: 10.1108/IJPDLM-02-2017-0099.
- [23] D. Peidro, J. Mula, R. Poler, and F. C. Lario, "Quantitative models for supply chain planning under uncertainty," *Int. J. Adv. Manuf. Technol.*, vol. 43, no. 3–4, pp. 400–420, 2009, doi: 10.1007/s00170-008-1715-y.
- [24] J. Mula, D. Peidro, M. Díaz-Madroñero, and E. Vicens, "Mathematical programming models for supply chain production and transport planning," *Eur. J. Oper. Res.*, vol. 204, no. 3, pp. 377–390, 2010, doi: 10.1016/j.ejor.2009.09.008.
- [25] Ç. Sel and B. Bilgen, "Quantitative models for supply chain management within dairy industry: a review and discussion," *Eur. J. Ind. Eng.*, vol. 9, no. 5, p. 561, 2015, doi: 10.1504/ejie.2015.071772.
- [26] K. Govindan, H. Soleimani, and D. Kannan, "Reverse logistics and closed-loop supply chain: A comprehensive review to explore the future," *Eur. J. Oper. Res.*, vol. 240, no. 3, pp. 603–626, 2015, doi: 10.1016/j.ejor.2014.07.012.
- [27] M. S. Habib, Y. H. Lee, and M. S. Memon, "Mathematical Models in Humanitarian Supply Chain Management: A Systematic Literature Review," *Math. Probl. Eng.*, vol. 2016, pp. 1–20, 2016, doi: 10.1155/2016/3212095.
- [28] W. E. Soto-Silva, E. Nadal-Roig, M. C. González-Araya, and L. M. Pla-Aragones, "Operational research models applied to the fresh fruit supply chain," *Eur. J. Oper. Res.*, vol. 251, no. 2, pp. 345–355, 2016, doi: 10.1016/j.ejor.2015.08.046.
- [29] C. Franco and E. Alfonso-Lizarazo, "A Structured Review of Quantitative Models of the Pharmaceutical Supply Chain," *Complexity*, vol. 2017, pp. 1–13, 2017, doi: 10.1155/2017/5297406.
- [30] N. Zandi Atashbar, N. Labadie, and C. Prins, "Modelling and optimisation of biomass supply chains: a review," *Int. J. Prod. Res.*, vol. 56, no. 10, pp. 3482– 3506, 2018, doi: 10.1080/00207543.2017.1343506.
- [31] F. Hamdi, A. Ghorbel, F. Masmoudi, and L. Dupont, "Optimization of a supply portfolio in the context of supply chain risk management: literature review," *J. Intell. Manuf.*, vol. 29, no. 4, pp. 763–788, 2018, doi: 10.1007/s10845-015-1128-3.
- [32] I. Mundi, M. M. E. Alemany, R. Poler, and V. S. Fuertes-Miquel, "Review of mathematical models for production planning under uncertainty due to lack of homogeneity: proposal of a conceptual model," *Int. J. Prod. Res.*, vol. 7543, pp. 1–45, 2019, doi: 10.1080/00207543.2019.1566665.

- [33] J. B. Oliveira, M. Jin, R. S. Lima, J. E. Kobza, and J. A. B. Montevechi, "The role of simulation and optimization methods in supply chain risk management: Performance and review standpoints," *Simul. Model. Pract. Theory*, vol. 92, no. June 2018, pp. 17–44, 2019, doi: 10.1016/j.simpat.2018.11.007.
- [34] A. Jamalnia, J.-B. Yang, A. Feili, D.-L. Xu, and G. Jamali, "Aggregate production planning under uncertainty: a comprehensive literature survey and future research directions," *Int. J. Adv. Manuf. Technol.*, vol. 102, no. 1–4, pp. 159–181, 2019, doi: 10.1007/s00170-018-3151-y.
- [35] M. L. Tseng, M. S. Islam, N. Karia, F. A. Fauzi, and S. Afrin, "A literature review on green supply chain management: Trends and future challenges," *Resour. Conserv. Recycl.*, vol. 141, no. June 2018, pp. 145–162, 2019, doi: 10.1016/j.resconrec.2018.10.009.
- [36] G. Baryannis, S. Validi, S. Dani, and G. Antoniou, "Supply chain risk management and artificial intelligence: state of the art and future research directions," *Int. J. Prod. Res.*, vol. 0, no. 0, pp. 1–24, 2018, doi: 10.1080/00207543.2018.1530476.
- [37] S. Hosseini, D. Ivanov, and A. Dolgui, "Review of quantitative methods for supply chain resilience analysis," *Transp. Res. Part E Logist. Transp. Rev.*, vol. 125, no. December 2018, pp. 285–307, 2019, doi: 10.1016/j.tre.2019.03.001.
- [38] M. Gandhi and H. Vasudevan, "Green Supply Chain Management Practices and Its Impact on Business Performance," in *Proceedings of International Conference on Intelligent Manufacturing and Automation*, 2019, pp. 601–611.
- [39] M. L. Tseng, M. S. Islam, N. Karia, F. A. Fauzi, and S. Afrin, "A literature review on green supply chain management: Trends and future challenges," *Resour. Conserv. Recycl.*, vol. 141, no. September 2018, pp. 145–162, 2019, doi: 10.1016/j.resconrec.2018.10.009.
- [40] E. Koberg and A. Longoni, "A systematic review of sustainable supply chain management in global supply chains," *J. Clean. Prod.*, vol. 207, pp. 1084–1098, 2019, doi: 10.1016/j.jclepro.2018.10.033.
- [41] K. J. Burnard and R. Bhamra, "Challenges for organisational resilience," Contin. Resil. Rev., vol. 1, no. 1, pp. 17–25, 2019, doi: 10.1108/crr-01-2019-0008.

Chapter 3

Models and algorithms for production planning, scheduling and sequencing problems: a holistic framework and a systematic review.

Guzman, Eduardo, Beatriz Andres, and Raul Poler. 2021. "Models and Algorithms for Production Planning, Scheduling and Sequencing Problems: A Holistic Framework and a Systematic Review." Journal of Industrial Information Integration, 100287. https://doi.org/10.1016/j.jii.2021.100287.

Abstract:

Production planning, scheduling and sequencing comprise the core of the manufacturing companies' performance. The new and changing market demands make manufacturing a challenge because companies must produce by using the minimum possible number of resources to provide high-quality products and to respond quickly to market demands. Thus the need for efficient production planning, scheduling and sequencing has become a very important research area for companies and researchers in recent decades. We evaluated the current state of such research with a holistic framework that comprised the plans aggregation and disaggregation levels, the modelling approaches to represent the different types of plans and their characteristics, the solution approaches with the adopted algorithms, the application areas, the intra- and inter-enterprise levels of integration, the sizes of the datasets used to validate the models and algorithms, the development tools, and the quality of the solutions obtained in relation to the problems' data size. The systematic literature review is arranged within the framework and grouped around different types of plans, including production planning, scheduling and sequencing, and their combinations. Finally, some gaps in the related research are identified and future research opportunities are proposed.

3.1 Introduction

In recent decades, researchers and industrial professionals have voiced concern about production planning. Several approaches have been developed to formulate and solve production planning problems. Developing models for real problems is a complex task, and the solution procedure is difficult in most cases. For this reason, a plethora of solution techniques and methods has been developed to provide different types of solutions.

The literature describes different models and approaches to solve production planning, scheduling and sequencing problems. The general aim of research works was to determine the resources needed so that production meets customer demands. The production planning problem has been extensively studied because it allows manufacturers to improve enterprise profits by better using manufacturing resources. In fact, the decision-making process in production planning allows not only the resources needed to carry out future manufacturing operations to be determined, but also all the production activities performed to optimise companies' objectives to be effectively coordinated. This allows resources to be allocated to production as and when required at the lowest cost [1].

The scientific literature based on tactical and operational production planning concepts is a vast fruitful area to which plenty of attention has been paid. The number of publications has rapidly increased, and the variety of proposed methods, trends and structures is very wide. These trends need to be aligned to address production planning, scheduling and sequencing problems and solutions in enterprises. The present review seeks to provide both an understanding of the common and unique characteristics of the proposed models of production planning, scheduling and sequencing problems and an accurate classification of different optimisation criteria to solve them. Accordingly, we pose the following research questions:

RQ1. How can production planning, scheduling and sequencing problems be classified?

RQ2. What types of modelling approaches are used in production planning, scheduling and sequencing problems, and what characteristics do they have?

RQ3. What methods or techniques are proposed to solve production planning, scheduling and sequencing problems?

RQ4. What methods or techniques can solve real large-scale problems, and what is the obtained solution quality?

Before investigating the modelling approaches and solution approaches proposed in the literature to deal with production planning, scheduling and sequencing problems, we analysed the existing review works to justify the research need of this paper.

The production planning literature is currently extensive. Nam and Logendran [2] conducted a review of Aggregate Production Planning (APP) from 1950 to 1990 to summarise the various existing techniques within a framework. By reducing searches to papers published in the recent decade, we found that Cheraghalikhani et al. [3] focused on APP methodologies, characteristics and structures of models and solving approaches, and many papers emphasise APP models under uncertainty; see Jamalnia et al. [4].

Many works in the literature discuss techniques, methods, levels, and solution approaches related to production planning. Mula et al. [5] analysed models for production planning under uncertainty by classifying them into four typologies: conceptual models, analytical models, artificial intelligence models and simulation models. Díaz-Madroñero et al. [6] reviewed optimisation models for tactical production planning. These authors analysed different characteristics, including the problem type, aim, number of products, time period, nature of demand, capacity constraints, extensions, modelling approach, solution approach, development tool, application, limitations and benefits.

Although much research has been conducted in the production planning area in general, the analysed reviews differ in several aspects. Some papers are descriptive, which highlights the importance of a specific field. One work worth highlighting is that by Mundi et al. [7], which reviewed production planning models by considering the uncertainty given by lack of homogeneity on products (LHP). These authors classified the reviewed papers according to the sectors affected by LHP inherent uncertainty, the modelled inherent LHP uncertainty types and approaches for modelling. Lage and Filho [8] reported production planning and control (PPC) in remanufacturing by proposing a classification based on four categories: PPC activities, characteristics, remanufacturing subsystemfocused and research type.

Other reviews have analysed production planning from a combined perspective. One example is that by Mula et al. [9], who reviewed mathematical programming models for production and transport planning. They classified papers according to supply chain structure, decision level, modelling approach, purpose, shared information, limitations, novelty and application. Akçcal and Çetinkaya [10] studied quantitative models for inventory and production planning

in closed-loop supply chains. They classified deterministic and stochastic problems according to modelling of demand, return processes and solution methodologies. On the supply chain, Peidro et al. [11] conducted a literature review that focused on supply chain planning under uncertainty by adopting quantitative approaches, similarly to Stindt and Sahamie [12] and Govindan et al. [13], whose reviews considered closed-loop supply chain planning.

After analysing previous literature reviews and, as far as we are aware, we concluded that our paper significantly differs from extant publications. We identified that most authors did not consider the holistic framework herein proposed. Therefore, this review aims to provide an overview of the key elements of production planning, scheduling, and sequencing problems. We propose a holistic framework to characterise all the aggregation and disaggregation levels. We place particular emphasis on the decision-making level at which they are contextualised. Continuing with the analysis of the most relevant aspects when posing a problem, such as planning horizon, type of modelling approach, the objectives pursued by mathematical models and the techniques applied to solve problems. We also analyse applications and evaluate the quality of the problem's solution according to the size of data.

Finally with this systematic review of articles based on the holistic classification framework, we seek to identify current research trends in production planning, scheduling and sequencing, as well as future research gaps and directions.

In order to answer these four research questions, this paper is organised as follows. Section 3.2 describes the methodology followed to perform the systematic literature review and details the proposed framework to review production planning, scheduling and sequencing problems. In Section 3.3, a detailed analysis of aggregation levels, the modelling approach, the solution techniques, the objectives raised, the development tool, the applications area and the quality of solutions, are applied to production planning, scheduling and sequencing problems. Section 3.4 highlights and discusses the main results. Finally, Section 3.5 draws the main conclusions and directions for future research.

3.2 Literature review methodology

The systematic literature review employed a structured methodology, and followed a scientific and transparent process, to reduce papers' selection bias by a thorough literature search. The synthesis that characterised the systematic

literature review allowed existing findings, research guidelines and gaps to be identified [14]. This paper followed a 4-step methodology in accordance with Seuring and Müller [15] and Seuring and Gold [16]: (i) collecting material (Section 3.2.1); (ii) descriptive analysis (Section 3.2.2); (iii) selecting or identifying categories (Section 3.2.3); (iv) evaluating material (Section 3.2.4).

3.2.1 Collecting material

The references collected for this study covered a 20-year (2000-2020) time frame. We conducted searches in December 2020 in the Elsevier SCOPUS and Web of Science citation databases. The collected works included all the English language articles registered as 'Articles', with no limitations set to scientific journals. Searches included the title, abstract and keyword fields, and three search terms were defined. Each term was a combination of the keyword 'Product* Plan*' OR 'Product* Schedul*' OR 'Product* Sequenc*', with an additional keyword: mathematical programming, linear programming, heuristic, metaheuristic or matheuristic (see Figure 3.1). The selected keywords were chosen to collect the most relevant papers. The modifier asterisk was used in the Boolean search as a source word for all the derivative keywords. Figure 3.1 depicts the strategy adopted to follow the structured literature review process.



Figure 3.1. Structured literature review process.

The keyword search gave 2,380 articles after removing duplicates. The abstracts of these articles were reviewed to assess if they matched our research

questions. Throughout this process (i.e., from 2,380 studies to 82), the exclusion criteria for why papers were unrelated to production planning, scheduling and sequencing modelling approaches were as follows:

- Not including the production processes whose approaches cover production planning, manufacturing operational process scheduling and sequencing processes

- Lack of an optimisation model or heuristic, metaheuristic, or matheuristic algorithms. Simulation and analytic methods were excluded from the review. although these methods may appear, they go beyond the scope of this paper.

After analysing abstracts, 82 papers were retained for full reading. Subsequently to this analysis, we added 12 articles that resulted from the backward search process and, thus, resulted in 94 papers. Additional papers were included as they were cited in the articles that derived from the keyword search and were applicable to the research topic. Of this subset of 94 articles, 34 were not considered relevant to the review because they did not satisfactorily answer our research questions. This left 60 papers for the analysis, evaluation and classification processes.

3.2.2 Descriptive analysis

This study analysed 60 scientific papers published between 2000 and 2020, and Figure 3.2 illustrates their publication trend. In turn, a slightly increasing trend in the last 3 years was identified. Some years provided significantly fewer papers than previous years; for example, more articles were retrieved in 2014 than in 2015.



Figure 3.2. Distribution of the reviewed papers according to year.

Most of the articles selected for the final review appeared in 20 different journals. Figure 3.3 shows the distribution of the articles reviewed from these journals. Of the 20 journals, International Journal of Production Research published the most papers with 26.66% of all the reviewed articles. Of the 20 journals, Computers and Industrial Engineering, Computers and Operations Research, European Journal of Operational Research and International Journal of Production Economics were equally representative, and collectively published 43.33% of all the reviewed articles. Overall, 42 publications appeared in the top five journals.



Figure 3.3. Distribution of articles for publication year and journal.

3.2.3 Category selection

In order to answer RQ1, we proposed a holistic framework that summarised the most important aspects characterising production planning, scheduling and

sequencing problems. This holistic framework resulted from integrating the literature reviews [6, 9, 17–19] and from deducting the analysed papers. This framework details a set of categories, such as: decision level, plan aggregation, planning horizon, modelling approach, mathematical model objectives, solution approach, development tool, proposed solution, application area, actual case application, data set size, solution quality. Table 3.1 presents the resulting framework, which represents a significant contribution of this work and can be general applied to any production planning, scheduling and sequencing problem.

Table 3.1. Framework proposed to represent production planning, scheduling and
sequencing problems.

Categories	Analytical categories				
Decision level	Strategical, Tactical, Operational				
Plan aggregation	Aggregated Plan (AP), Master Plan (MP), Dispatching Plan (DP)				
Planning horizon	Day, Week, Month, Year				
	Binary Programming (BP)				
	Constraint Programming (CP)	Multi-Objective Linear			
	Dynamic Programming (DP)	Programming (MOLP)			
	Fuzzy Programming (FP)	Multi-Objective Mixed-Integer			
	Fuzzy Goal Programming (FGP)	Linear Programming (MOMILP)			
	Fuzzy Linear Programming (FLP)	Multi-Objective Mixed-Integer			
	Fuzzy Multi-Objective Linear Programming	(MOMINLP)			
Modellina	(FMOLP)	Multi-Objective Non-Linear			
approach	Goal Programming (GP)	Programming (MONLP)			
approach	Integer Programming (IP)	Non-Linear Programming			
	Integer Linear Programming (ILP)	(NLP)			
	Integer-Weighted Goal Programming (IWGP)	Quad-Objective Mixed Integer Linear Programming (QOMILP)			
	Linear Programming (LP)	Quadratic Programming (QP)			
	Mixed Integer Linear Programming (MILP)	Robust Programming (RP)			
	Mixed Integer Non-Linear Programming (MINLP)	Stochastic Programming (SP)			
Mathematical					
model	Cost, Time, Product, Resources, Service, Sust	ainability			
objectives					

Categories		Analytical categori	ies
	Optimizer Algorithm (OA)	OA/ Branch and Bound (BB) OA/ Branch and Cut (BC) OA/ Criss-cross (CC) OA/ Decomposition strategy (DS) OA/ Lomnicki (LO)	OA/ Lompen Algorithm (LM) OA/ Simplex (SI) OA/ Solution procedure of model P* (SPP*)
Solution approach	Heuristic Algorithm (HA)	HA/ Benders Decomposition (BD) HA/ Beam Search (BM) HA/ Campbell-Dudeck Algorithm (CD) HA/ Decomposition & Aggregation (DA) HA/ Fix-Price-Optimise (FPO) HA/ Greedy (GR) HA/ Iterative Variable Neighbourhood (IVN) HA/ Lagrangian Relaxation (LGR) HA/ Local Improvement Procedure (LIP) HA/ LP and Fix (LF)	HA/ LP Relaxation (LPR) HA/ Minimum Spanning Tree (MS) HA/ Multi-Objective Master Planning Algorithm (MOMPA) HA/ Nawaz, Enscore and Ham (NEH) HA/ Nearest Neighbour (NN) HA/ Primal-Dual Based Heuristic (PDBH) HA/ Relax and Fix (RF) HA/ Relax-Price-Fix (RPF)
approach	Metaheuristic Algorithm (MA)	MA/ Ant Colony Optimisation (ACO) MA/ Evolutionary Computation (EC) MA/ Genetic Algorithm (GA) MA/ GRASP (GR) MA/ Iterated Local Search (ILS) MA/ Iterated Greedy (IG) MA/ Iterated Greedy (IG) MA/ Memetic Algorithm (MA) MA/ Multi-objective Simulated Annealing (MOHSA) algorithm MA/ Non-dominated Sorting Genetic Algorithm II (NSGA-II) MA/ Particle Swarm Optimisation (PSO) MA/ Scatter Search (SS)	MA/ Simulated Annealing (SA) MA/ Subpopulation Genetic Algorithm (SPGA) MA/ Tabu Search (TS) MA/ Tabu Search Grabowski and Wodecki (TSGW) MA/ Variable Tabu Search (VTS) MA/ Variable Tabu Search (VNS) MA/ Variable Neighbourhood Search (VNS) MA/ Variable Neighbourhood Descent (VND) MA/ Weighted Sum Multi- Objective Genetic Algorithm (WMOGA)

Table 3.1. Continued. Framework proposed to represent production planning, scheduling and sequencing problems.

Categories		Analytical catego	ories			
		MTA/ Ant Colony + Mathematical Model (ACO_MM)	MTA/ Iterated Local Search + Mathematical Model (ILS_MM)			
Solution approach	Matheuristic Algorithm (MTA)	MTA/ Biased Random-Key Genetic Algorithm + Mathematical Model	MTA Simulated annealing + Mathematical Model (SA_MM)			
		(BRKGA_MM) MTA/Fixed Variable List Algorithm and Clustering Sequence Algorithm + Mathematical Model (FVLA_CSA_MM)	MTA/ Tabu Search + Mathematical Model (TS_MM)			
		MTA Genetic Algorithm + Mathematical Model (GA_MM)				
Development tool	Programming Languages, Modelling language, Solver					
Proposed solution	Model + Solutio description (M/ Algorithm desc description+ Al	on (MS), Model + Solver+ Soluti AD) Model + Algorithm descrip ription+ Solver+ Solution (MAI gorithm Pseudocode + Solver +	ion (MSS), Model + Algorithm tion+ Solution (MADS), Model + DSS), Model + Algorithm • Solution (MADPCSS)			
Applications area	Sectorial - Transversal					
Real case application	Yes (Y) / No (N)					
Enterprise integration level	Intra-enterprise level – Inter-enterprise level					
Data set size	Small (S) – Medium (M) – Large (L)					
Quality solution	Optimal (OP), N	Optimal (OP), Near – Optimal (N-OP) – Good (GD)				

Table 3.1. Continued. Framework proposed to represent production planning,scheduling and sequencing problems.

3.2.4 Material evaluation

All the articles were evaluated and coded according to the holistic framework proposed in Section 3.2.2. Validation was carried out by considering the characteristics, approaches and level of aggregation of each article. To do so, we used the grouping technique and applied deductive and inductive methods [20]. The evaluation ensured that articles had sufficient information to be validated.

3.3 Results analysis

3.3.1 Decision level

Production planning, scheduling and sequencing problems can be decomposed and classified according to the extent or effect of the decision in time terms [9]. Several authors, such as [6, 21–24] among others, have classified these problems as strategical, tactical and operational problems.

Strategical or long-term planning models address a time period lasting between 5 and 10 years. This decision level implies a wide range of uncertainty, which normally affects enterprises' design, configuration and location. Moreover, strategical decisions deal with the development of new products, the identification of distribution channels, suppliers' selection and the selection of information technology [25].

Tactical planning models aim to plan mid-term activities. These models address planning horizons that last between 1 month or several months and 2 years. The decisions made at the tactical level are planned to be executed and comply with the decisions made at the strategical level. Tactical decisions include activities like production planning, material handling, distribution and storage planning, production capacity allocation, inventory management and maintenance activities [6, 9, 25].

The operational level is characterised by addressing short-term decisions that are generally made weekly, daily or hourly by focusing essentially on sequencing, scheduling, packaging, lot size calculation, routes allocation and vehicle load. This level seeks to guarantee an optimal flow of products along the production chain [23, 25, 26].

We should also bear in mind that distinctions of decision levels cannot always be made because some problems may involve planning at many levels and are incorporated into different decision levels. By way of example, the works of Rasmi et al. [27] present an Aggregate Production planning (APP) problem that incorporate decisions at the strategical and tactical levels in a multi-objective mixed-integer linear program (MOMILP) model, which evaluate economic, social, environmental and cultural aspects for an appliance manufacturer. Moreover, Omar and Teo [28], Xue et al. [29], Aghezzaf et al. [30], Fumero et al. [31] and Fumero, Corsano and Montagna [32] propose tactical-operational decision making, which is often used for models dealing with mid-term decisions that are taken daily, weekly or monthly, and generally up to 1 year.

In line with this, Omar and Teo [28], Fumero et al. [31], Fumero, Corsano and Montagna [32] propose dealing with production planning and scheduling jointly. Xue et al. [29] address production planning and scheduling by the hierarchical production planning approach. Finally, Aghezzaf et al. [30] propose a robust hierarchical production planning approach for master planning and scheduling. Table 3.2 classifies the reviewed works in relation to their decision-making level. Of all the reviewed papers, 28.33% address production planning at the tactical level, and propose solutions to aggregate and master plans. Over half the reviewed papers (61.67%) make decisions at the operational level by addressing scheduling and sequencing problems, 8.33% of the analysed papers deal with planning problems at several decision-making levels, namely tactical and operational, and only 1.66% present strategical and tactical decisions.

Decision level	Reference
Strategical & Tactical	Rasmi et al. [27]
Tactical	RC. Wang & Fang [33]; Leung & Chan [34]; Baykasoglu & Gocken [35]; Sillekens et al. [36]; Mirzapour Al-E-Hashem et al. [37]; Zhang et al. [38]; Ramezanian et al. [39]; Chakrabortty & Akhtar Hasin [40]; Khalili- Damghani & Shahrokh [41]; Makui et al. [1]; Tavaghof-Gigloo et al. [42]; Gholamian et al. [43]; de Kruijff et al. [44]; Mehdizadeh et al. [45]; Djordjevic et al. [46]; Bensmain et al. [47].
Operational	Grabowski & Wodecki [48]; D. Gupta & Magnusson [26]; Nonås & Olsen [49]; Bellabdaoui & Teghem [50]; Hooker [51]; P Doganis & Sarimveis [52]; Gaglioppa et al. [53]; Moon et al. [54]; Philip Doganis & Sarimveis [55]; Fakhrzad & Khademi Zare [56]; Mohammadi et al. [57]; Guimarães et al. [58]; Cheng et al. [59]; Chen et al. [60]; Motta Toledo et al. [61]; Na & Park [62]; Franz et al.[63]; Mattik et al. [64]; Golle et al. [65]; Baumann & Trautmann [66]; Abdeljaouad et al.[67]; Aroui et al.[68]; Zeppetella et al. [69]; Torkaman et al. [70]; Lopes et al. [71]; Woo & Kim [72]; Verbiest et al. [73]; Mönch & Roob [74]; Ekici et al. [75]; Chansombat et al. [76]; de Armas & Laguna [77]; S. Wang et al. [78]; De Smet et al. [79]; Yang & Xu [80]; Otto & Li [81]
Tactical &	Omar & Teo [28]; Xue et al. [29]; Aghezzaf et al. [30]; Fumero et al.[31];
Operational	rumero et al.[52];rodopiù et al. [62]

Table 3.2. The decision-making levels of the reviewed works.

3.3.2 Plan aggregation

Plan Make, identified in SCOR views [83], forms part of one of the most relevant planning decisions for companies. Plan Make aims to achieve effective planning

and management for all production operations, and in such a way to optimise company objectives. It focuses on determining the optimal number of items to be produced, the inventory, and other key production factors, to meet the variable demand in a planning horizon. Plan Make can be divided by considering three different decision-making production levels that comprise production planning, scheduling and sequencing [84].

In manufacturing environments, production planning supports decision makers in determining the use of resources, which are generally decisions made about the quantity to be produced, the inventory level, the required workforce size or the allocation of the necessary assets and resources to carry out the manufacturing process to meet the real or planned demand on a given horizon [85,86]. Production planning problems can cover mid-term or long-term planning horizons using aggregated or disaggregated information. Hax and Meal [87] distinguished production planning problems according to their horizon time and aggregation, which ranged from a long-term aggregation level (aggregated plan) to a short-term detailed level (dispatching plan).

The production planning category distinguishes two types of plans: aggregate plans and master plans. In aggregated plans, the used unit is product families, which refers to the groups of products belonging to the same type that shares similar configurations [29]. Production plans can be disaggregated into more detailed programmes, which define the product quantities to be produced during shorter time periods than the aggregated plan, which are normally weekly or monthly periods [88].

Production planning problems represented 30% of all the reviewed papers, the majority of which dealt with aggregated plans (see Table 3.3). We found only one article, that of de Kruijff, Hurkens, and de Kok [44], which addressed the master plan and proposed a mid-term production planning model for high-tech and low-volume industries.

The literature contains a vast variety of point views when contextualising scheduling and sequencing plans, and some do not clearly indicate how the functions of each one should be carried out. The present work considers that the scheduling plan deals with efficient resources allocation given a set of due dates, release dates, demand for products and operational restrictions to help to decide the number of products to be produced during each time period. Accordingly, the scheduling plan implies finding a way to assign times (at which each operation in the sequence will start and finish), corresponds to the activity of timetabling operations [89], while sequencing plans involves the sequencing of jobs given a

set of shared resources (jobs, materials, machines) so that they meet certain production constraints; such as capacity, production levels, precedence, start and due dates, machine capabilities, machine availabilities, lot-size restrictions, resource requirements and resource availabilities [85, 86, 89] A sequencing plan specifies the order in which jobs are to be processed at a shared workstation.

Considerably more research interest has been shown in scheduling problems and their combinations (45%) because the practical application of such problems to industry is more frequent. Scheduling problems have been combined with production planning problems [28–32], and also with sequencing problems [58, 59, 70, 77]. Fewer sequencing problems have been studied (see Table 3.3) as 25% of the research works analysed these problems. These problems have often been combinatorial and presented as NP-Hard, so most research works have applied specific algorithms to solve them.

A distinctive feature of scheduling and sequence problems is listed at the lowest production hierarchy level, namely at the operational decision-making level. Once sequence and scheduling plans have been computed, they are reflected in production orders. Implementing these orders to start the production of each item is called dispatching [90].

Year Week Sillekens et al. [36]; de Kruijff et al. [44] Mirzapour Al-E-Hashem et al. [37]: Taycobof Giolog et al. [42]:	Plan type	Plan's Aggregation	Planning Horizon	Planning Period	Reference	
Mirzapour Al-E-Hashem et al.			Vear		Sillekens et al. [36]; de Kruijff et al. [44]	
Bensmain et al. [47];	Dandunting			Month	Mirzapour Al-E-Hashem et al. [37]; Tavaghof-Gigloo et al. [42]; Bensmain et al. [47];	
Quarterly Month Leung & Chan, [34]			Quarterly	Month	Leung & Chan, [34]	
Aggregated Month Week [46] Readuction		Aggregated	Month	Week	Makui et al. [1]; Djordjevic et al. [46]	
Planning Plan Week Day Fang et al. [91]	Planning	Plan	Week	Day	Fang et al. [91]	
Month Chakrabortty & Akhtar Hasin [40]	5			Month	Chakrabortty & Akhtar Hasin [40]	
NS ^(*) RC. Wang & Fang [33] Baykasoglu & Gocken [35]; Zhan et al. [38]; Ramezanian et al. [39] Khalili-Damghani & Shahrokh [41] Gholamian et al. [43]; Mehdizade et al. [45]; Rasmi et al. [27]			NS ^(*)	NS	RC. Wang & Fang [33]; Baykasoglu & Gocken [35]; Zhang et al. [38]; Ramezanian et al. [39]; Khalili-Damghani & Shahrokh [41]; Gholamian et al. [43]; Mehdizadeh et al. [45]; Rasmi et al. [27]	
Master Plan Month Week de Kruijff et al. [44]		Master Plan	Master Plan Month Week de Kruijff et al. [44]			

Table 3.3. Plan type and plan aggregation of the reviewed works.

Tab	Table 3.3. Continued. Plan type and plan aggregation of the reviewed works.						
Plan type	Plan's Aggregation	Planning Horizon	Planning Period	Reference			
		Month	Week	Mattik et al. [64]			
	-	Week	Day	Philip Doganis & Sarimveis [55]			
	-		NS	Motta Toledo et al. [61]; De Smet et al. [79]			
	-	Dav	Day	Nonås & Olsen [49]			
		Day	Hour	Doganis & Sarimveis [52]			
Scheduling	Dispatching Plan	NS	NS	Grabowski & Wodecki [48]; Gupta & Magnusson [26]; Hooker [51]; Gaglioppa et al. [53]; Fakhrzad & Khademi Zare, [56]; Zeppetella et al. [69]; Chansombat et al. [76]; Verbiest et al., [73]; S. Wang et al. [78]; Yang & Xu, [80]; Otto & Li, 2020 [81]; Prata, de Abreu, et al., 2020 [92]; Prata, Rodrigues, et al. [93]; Rodoplu et al. [82]			
		Day	Minute	Woo & Kim [72]			
	-	Week	NS	Baumann & Trautmann [66]			
		Day	NS	Franz et al. [63]			
Sequencing	Dispatching Plan	NS	NS	Mohammadi et al. [57]; Bellabdaoui & Teghem [50]; Moon et al. [54]; Chen et al. [60] ; Golle et al. [65] Na & Park [62]; Abdeljaouad et al. [67]; Aroui et al. [68]; Lopes et al. [71]; Mönch & Roob [74]; Ekici et al. [75]			
	Aggregated	Year	Quarterly	Xue et al. [29]; Fumero et al. [31]			
Production	Plan &		Month	Omar & Teo [28]			
Planning &	Plan	Week	Hour	Fumero et al. [32]			
Production Scheduling	Master Plan & Dispatching Plan	Month	Week	Aghezzaf et al. [30]			
Production		Day	Hour	de Armas & Laguna [77]			
Scheduling & Production Sequencing	Dispatching Plan	NS	NS	Guimarães et al. [58]; Cheng et al. [59]; Torkaman et al. [70]			
141							

^(*) NS: not specified.

The analysis performed on the plans' aggregation features enables to provide a concrete definition of production planning, sequencing, and scheduling problems. The proposed delimitation describes the different planning levels. Tactical or medium-term level uses aggregated data, once the results of this phase are available, you can move on to a detailed short-term scheduling phase. Input data for planning problems at the tactical level is generally measured in months or weeks rather than days or hours, as is done in scheduling and sequencing plans. Planning issues at the tactical level seek to minimize production costs, warehousing costs, inventory costs, and others detailed in section 3.3.4. The results obtained from this process generally describe the monthly or weekly production quantities for all products, requiring a number of resources (machines, operations) and capacities. Scheduling and sequencing activities are done in the short term, although, as mentioned above, some papers present a combined approach in which the results of medium-term planning are the input of scheduling or sequencing plans (short-term). Short-term plans seek to optimize each stage and each installation (machines or resources), in shorter time horizons [94].

3.3.3 Modelling approach and solution techniques

The literature describes a wide variety of models and approaches to solve production planning, scheduling and sequencing problems. The analysed works have generally sought to develop models and to apply them to real planning problems using large-sized input data. As this leads to complexity, the procedure to find a solution in data management and computational efficiency terms is difficult. This is why there are different types of techniques to model and solve production planning, scheduling and sequencing problems. The objective of the present paper was to analogously present the mathematical programming methods followed to raise different model types, the techniques to solve them and the software used to treat these problems. Table 3.4 presents the applied modelling approach and solution techniques in the reviewed works to answer RQ1 and RQ2. The first column in Table 3.4 refers to the modelling approach. The analysis allowed us to conclude that mixed integer linear programming (MILP) models were the most widely used to deal with production planning, scheduling and sequencing problems. Indeed 73.33% of the analysed papers adopted this approach, while only two authors resorted to fuzzy linear programming (FLP) [33, 46]. Other modelling approaches indicated during the review included fuzzy goal programming (FGP) [41], multi-objective linear programming (MOLP), mixed integer non-linear programming (MINLP) [47, 79, 40], multi-objective mixed-

integer linear programming (MOMILP) [45], multi-objective mixed-integer nonlinear programming (MOMINLP) [37], quad-objective mixed integer linear programming (QOMILP) [27] and the robust programming (RP) model [1]. Hooker [51] combined two methods: mixed integer linear programming (MILP) and constraint programming (CP). Omar and Teo [28] combined and applied two techniques, firstly MILP to solve the aggregate plan, and then an integer programming (IP) model to disaggregate the plan.

The second column in Table 3.4 refers to the solution algorithms proposed in the reviewed works. Considering the complexity of the models and their applications, different types of techniques appeared to solve distinct production problems. Andres et al. [95] classified these techniques into four groups: (i) optimiser algorithms (OA), which respond to techniques that ensure that the best possible solution is provided, and are commonly integrated into predetermined solutions; (ii) heuristic algorithms (HA), which do not guarantee the optimal solution, but a solution/s that is/are relatively good by coming close to the global optimum [96]; (iii) metaheuristic algorithms (MA), which consist of higher-level heuristics [96] and can provide a sufficiently high-quality solution through an iterative master process that guides and modifies subordinate heuristics (partial search algorithm) operations [6, 97]; (iv) matheuristic algorithms (MTA) represent a hybridisation or combination of heuristic and metaheuristic algorithms and exact methods [98]

Regarding the techniques for solving production planning, scheduling and sequencing problems, 71.66% of the articles described the algorithm used to solve these problems, while 28.33% did not use a specific algorithm, but described the type of commercial solvers employed. Some commercial solvers like Gurobi or Cplex incorporate parametrisation features to efficiently solve optimisation problems. Nevertheless, the reviewed papers did not report any software parameter to provide clues about the algorithms employed in the commercial solver.

The majority of the reviewed articles applied metaheuristic algorithms (37.20%), where genetic algorithms were the predominant metaheuristic procedure. Some authors performed combinations or hybridisations of algorithms, and a summary of the most relevant ones follows. Fakhrzad and Khademi Zare [56] introduced a hybrid genetic algorithm (genetic algorithm + local search) that, jointly with a Lagrangian algorithm, addressed the lot size determination in multistage production scheduling problems. With this hybridisation, the authors obtained near-optimal solutions in a medium dataset. Chen et al. [60] presented a hybrid approach based on two metaheuristic algorithms, the variable

neighbourhood search and particle swarm optimisation (VNPSO), to solve multistage and parallel-machine scheduling problems. This hybrid algorithm was compared to the traditional particle swarm optimisation (PSO) algorithm, and the authors concluded that the obtained solutions and the calculation time were better in the hybrid algorithm than in the traditional PSO for obtaining almost optimal solutions for large instances. Aroui et al. [68] presented two metaheuristic algorithms (genetic algorithms and simulated annealing) and a hybrid algorithm composed of a genetic algorithm and simulated annealing (GASA) to solve a problem to sequence assembly lines of mixed models to minimise workload. The authors tested MILP and algorithms in an industrial case of a truck assembly line. The results obtained from the different algorithms demonstrated that the hybrid algorithm provided better solutions and better calculation times than MILP in large instances. GASA was also better than simulated annealing algorithms but required longer calculation times and SA provides better solutions than the genetic algorithm.

Finally, is worth mentioning the work of Torkaman et al. [70], who proposed a hybrid simulated annealing (HSA) algorithm that used a genetic algorithm to obtain an initial solution. This hybrid algorithm was used to solve multistage, multiperiod and multiproduct lot sizing problems with remanufacturing and sequence-dependent setups and a setup carry-over in a flow shop system. This hybrid algorithm was compared to the four heuristic algorithms and a MILP model. The authors concluded that the MILP model achieved better solutions than the hybrid algorithm when computing small datasets. Nevertheless, the MILP model needed a longer calculation time than HSA. Accordingly, HSA in larger instances was more efficient than the mathematical model and heuristic methods, thus the proposed hybrid algorithm can be used in this type of problem to obtain better calculation times in medium and large datasets.

In terms of heuristic algorithms (30.23%), the most widely used techniques were LP Relaxation (LPR) [32, 38, 53] and Benders decomposition [1, 44, 51]. For optimiser algorithms (16.27%), some frequently used techniques included Branch and Bound (BB) [42, 49, 64]. Finally, MTA (16.27%), the metaheuristic combinations (genetic algorithms) and MILP models were the most frequently used [61,72].

Table 3.4 shows the various combinations or associations of each development tool classified as programming languages, modelling languages and solvers. As regards programming languages, only a few authors (25.00%) indicated the

programming languages that they used to conduct their research, while others simply did not specify (NS) them. The employed languages were C, C#, C++, Julia, Java, Python and Visual Basic, whereas C++ was the most preferred one (46.66%). Modelling languages included AIMMS, GAMS, ILOG, JUMP LINGO, MATLAB and OPL, and 58.33% of the reviewed studies described which modelling language they used, of which LINGO and ILOG were the most frequently reported ones. Of them all, 81.66% informed about the solvers utilised to solve production planning, scheduling and sequencing problems. Solvers were CPLEX, CP Optimiser, LINGO, Xpress, Gurobi and OM Patners, and the most representative ones were CPLEX (50.00%) and LINGO (18.33%).

The proposed solution column summarises the findings of the reviewed articles. For those papers proposing model and solution (MS), readers can find a mathematical programming model and its solution. By way of example, we cite Wang and Fang [33] and Djordjevic et al. [46], who proposed a fuzzy linear programming model, but only indicated the obtained results. Model, solver and solution (MSS) added the solver. In this case, Khalili-Damghani and Shahrokh [41] used a fuzzy goal programming model and solved it by LINGO. Algorithm description (MAD) showed a model and described the algorithm, but these research types were not studied because they went beyond the scope of RQ4.

Model, algorithm description and solution (MADS) showed the model, and described the algorithm and the obtained solution, but not the used solver; e.g., Fang et al. [91] formulated the aggregate production planning problem as an MILP model, and solved it by the Lagrangian relaxation technique (LGR), but did not describe the used solver. Model, algorithm description, solver and solution (MADSS) similar proposed solutions to MADS and included the used solver; e.g., Chen et al. [60] studied a sequencing problem and formulated an MILP model by developing a hybrid approach based on VNS and PSO. The MILP model was formulated with the IBM ILOG CPLEX software package and was solved by BB algorithms, which were implemented in C++. Model, algorithm description, algorithm to MADSS, and there were only seven papers of this type: Mehdizadeh et al. [45] de Kruijff et al. [44]; Gupta and Magnusson [26]; Motta Toledo et al. [61]; Hooker [51]; Franz et al. [63]; Aroui et al. [68].

Table 3.4. The modeling approach and solution techniques of the reviewed works.							
Modelling approach	Algorithm	Programming Languages	Modelling language	Solver	Proposed solution	Reference	
FLP	-	NS	NS	NS	MS	RC. Wang & Fang [33]; Djordjevic et al. [46]	
FGP	-	NS	LINGO	LINGO	MSS	Khalili- Damghani & Shahrokh [41]	
FMOLP	OA/ SI	NS	GAMS	CPLEX	MSS	Gholamian et al. [43]	
FP	MA/ TS	NS	NS	NS	MADS	Baykasoglu & Gocken [35]	
GP	OA/ SI	NS	NS	LINDO	MSS	Leung & Chan [34]	
IP	MTA/ (BRKGA + IP)	C++	NS	LP-Solve	MADS	Mönch & Roob [74]	
ILP	<u>Hybrid</u> MA/ GA + HA/ LGR	Visual Basic	NS	LINGO	MADSS	Fakhrzad & Khademi Zare [56]	
ILP	HA/ IVN	NS	ILOG CPLEZ	CPLEX	MADPCSS	Otto & Li [81]	
	-	NS	NS	OM Partners	MSS	Bellabdaoui & Teghem [50]	
	-	NS	NS	CPLEX	MSS	P Doganis & Sarimveis [52]; Golle et al. [65]	
		NS	OPL	CPLEX	MSS	Aghezzaf et al. [30]	
MILP	-	NS	GAMS	CPLEX	MSS	Philip Doganis & Sarimveis [55] ; Fumero et al. [31]	
	-	С	NS	Gurobi	MSS	Baumann & Trautmann [66]	
	-	NS	ILOG CPLEX	CPLEX	MSS	Zeppetella et al. [69]; Lopes et al. [71]	
	-	NS	NS	Gurobi	MSS	Chansombat et al. [76]	

problems: a holistic framework and a systematic review. Table 3.4. The modelling approach and solution techniques of the reviewed works.

Chapter 3. Models and algorithms for production planning, scheduling and sequencing

works.							
Modelling approach	Algorithm	Programming Languages	Modelling language	Solver	Proposed solution	Reference	
	-	Julia	JUMP	Gurobi	MADS	S. Wang et al. [78]	
	HA/ BD	NS	AIMMS	CPLEX	MADPCSS	de Kruijff et al. [44]	
	HA/ GR	NS	OPL MATLAB	CPLEX	MADPCSS	D. Gupta & Magnusson [26]	
	HA/ LF HA/ RF	NS	ILOG CPLEX	CPLEX	MSS	Sillekens et al. [36]	
	HA/ LGR	C#	NS	NS	MADS	Fang et al. [91]	
	HA/ LPR	NS	NS	CPLEX	MSS	Gaglioppa et al. [53]	
	HA/ LPR	NS	GAMS	CPLEX	MSS	Fumero et al. [32]	
	HA/ LPR HA/ BM	C#	LINGO	LINGO	MADSS	Zhang et al. [38]	
MILD	HA/ NEH	NS	LINGO	LINGO	MADSS	Abdeljaouad et al. [67]	
MILP	MA/ GA	NS	NS	NS	MADS	Moon et al. [54]	
		C++	NS	CPLEX	MADSS	Cheng et al. [59]	
		NS	ILOG CPLEX	CPLEX CP Opti- miser	MADSS	Na & Park [62]	
	MTA/ GA + MILP	C++	NS	CPLEX	MADPCSS	Motta Toledo et al. [61]	
	MA/ MOHSA	NS	MATLAB	NS	MADS	Mohammadi et al. [57]	
	MA/GA MA/ SA <u>Hybrid</u> MA/GA + MA/SA	NS	ILOG CPLEX MATLAB	CPLEX	MADPCSS	Aroui et al. [68]	
	MA/GA MA/ TS	NS	LINGO- MATLAB	LINGO	MADSS	Ramezanian et al. [39]	

plans in industrial enterprises. Table 3.4. Continued. The modelling approach and solution techniques of the reviewed

Models and algorithms for the optimisation of replenishment, production and distribution

works.							
Modelling approach	Algorithm	Programming Languages	Modelling language	Solver	Proposed solution	Reference	
	MA/TSGW	C++	NS	NS	MADS	Grabowski & Wodecki [48]	
	MA/VNS MA/ VTS	NS	ILOG CPLEX	CPLEX	MADPCSS	Franz et al. [63]	
	MA/PSO						
	<u>Hybrid</u>	C++	ILOG CPLEX	CPLEX	MADSS	Chen et al. [60]	
	MA/ VNS + MA/ PSO		CILEX			[00]	
	OA/ BB	NS	NS	CPLEX	MADSS	Nonås & Olsen [49]	
	OA/ BB OA/ SI	NS	NS	FICO Xpress Opti- mizer - CBC	MSS	Tavaghof- Gigloo et al. [42]	
	OA/ SPP*	NS	LINGO	LINGO	MADSS	Xue et al. [29]	
	OA/BB HA/ LPR	NS	OPL	CPLEX	MSS	Mattik et al. [64]	
MILP	MTA/ILS + MILP	NS	NS	Gurobi	MADS	Verbiest et al. [73]	
	MTA/ GA + MILP / MTA/ SA + MILP	NS	ILOG CPLEX	CPLEX	MADS	Woo & Kim [72]	
	MTA/ TS + MILP	C++	NS	CPLEX	MADS	Ekici et al. [75]	
	HA/ RF	Python	NS	CPLEX	MADPCSS	Rodoplu et al. [82]	
	MTA/FVLA_C SA_MM	Julia	JUMP	CPLEX	MADPCSS	Prata et al. [93]	
	MA/ VND - MA/ IG	NS	MATLAB	NS	MADPCSS	Yang & Xu [80]	
	-	Java SE 8	NS	CPLEX	MSS	de Armas & Laguna [77]	
	MTA/ RPF + FPO + MILP	C++	ILOG CPLEX	CPLEX	MADSS	Guimarães et al. [58]	

Chapter 3. Models and algorithms for production planning, scheduling and sequencing problems: a holistic framework and a systematic review.

Table 3.4. Continued. The modelling approach and solution techniques of the reviewed

works.							
Modelling approach	Algorithm	Programming Languages	Modelling language	Solver	Proposed solution	Reference	
MILP	<u>Hybrid</u> MA/ SA + MA/ GA	NS	GAMS MATLAB	CPLEX	MADSS	Torkaman et al. [70]	
MILP – CP	HA/ BD	NS	OPL	CPLEX	MADPCSS	Hooker [51]	
MILP - IWGP	-	NS	LINGO	LINGO	MSS	Omar & Teo [28]	
MINLP	<u>Hybrid</u> MA/GA + HA/ RF based rolling horizon heuristic	NS	LINGO	LINGO	MADPCSS	Bensmain et al. [47];	
	HA/RF	NS	NS	Gurobi	MADPCSS	De Smet et al. [79]	
MOLP	MA/ GA	NS	MATLAB	NS	MADS	Chakrabortty & Akhtar Hasin [40]	
MOMILP	MA/ SPGA MA/ WMOGA MA/ NSGA-II	NS	LINGO MATLAB	LINGO	MADPCSS	Mehdizadeh et al. [45]	
MOMINLP	OA/ BB	NS	LINGO	LINGO	MSS	Mirzapour Al- E-Hashem et al. [37]	
QOMILP	-	NS	NS	NS	MS	Rasmi et al. [27]	
RP	HA/ BD	NS	GAMS	NS	MADS	Makui et al. [1]	

Models and algorithms for the optimisation of replenishment, production and distribution plans in industrial enterprises. Table 3.4. Continued. The modelling approach and solution techniques of the reviewed

3.3.4 Mathematical model objectives

This section reviews mathematical programming models in detail. Mathematical models often describe a problem through the objective function, as well as constraints to define the problem's structure. Therefore, to answer RQ2 and to study the characteristics of problems, we analysed the objective functions of the models proposed in the reviewed papers (see Table 3.6).

Table 3.5 summarises the typical objectives used to support decision making in production planning, scheduling and sequencing problems. The objectives were

classified according to their nature: (i) cost-based objectives (OC), costs or profits representing variables related to monetary units; (ii) time-based objectives (OT) evaluate the time units required to perform certain processes, i.e. jobs, machines, material processing, manufacturing cycles, order processing, etc; (iii) productbased objectives (OP), which intend to improve the efficiency of operations and aim to ensure that manufacturing meets the appropriate quantity and quality to cover customer demands; (iv) resource-based objectives (ORS), which seek to achieve the optimal use of resources, such as people, materials and machinery; (v) service-based objectives (OS), which assess delays, shortage or expiration dates, and the quality of goods and services for final customers; (vi) sustainability-based objectives (OST), which seeks to strike a balance in the utilisation of resources for production at environmental, social and economic dimensions. In order to gain profounder knowledge, the review analysis allowed a group of 64 subtypes of objectives belonging to each defined category (OC, OT, OP, ORS, OS and OST; see Table 3.5) to be identified.

Type of objectives	Subtype	Designation	Subtype	Designation
	Production cost minimisation	OC1	Holding cost minimisation	OC21
	Variable production cost minimisation	OC2	Changeover cost minimisation	0C22
	Remanufacturing cost	OC3	Supply chain cost minimization	OC23
Costs (OC)	Setup cost minimisation	OC4	OC4 Shortage cost minimisation	
	Inventory cost minimisation	OC5	Changing shift model minimisation (cost)	OC25
	Cost to change from production capacity level	OC6	Transportation, inventory and shortage costs minimisation	OC26
	Normal/ Extra time (overtime) production cost minimisation	OC7	Subcontract cost minimisation (outsourcing)	OC27
	Labour minimisation (hiring cost and lay-off cost)	0C8	Fixed cost per unit minimisation	OC28

Table 3.5. Production planning, scheduling and sequencing objectives.

Table 3.5. Continued. Production planning, scheduling and sequencing obj						
Type of objectives	Subtype	Designation	Subtype	Designation		
	Cost of workers' salary minimisation	OC9	Repairs and deterioration machines cost minimisation	OC29		
	Labour training cost	OC10	Machine utilisation cost minimisation	OC30		
	Workforce changing cost (skilled and unskilled workforce)	OC11	Cost's preventive maintenance minimisation	OC31		
	Normal and overtime labour cost minimisation	OC12	Capital cost minimisation	OC32		
	Backorder minimisation (quantity or cost)	OC13	Start-up cost	OC33		
Costs (OC)	Idle time cost minimisation	OC14	Contamination cost	0C34		
	Tardiness penalty costs; earliness penalty costs minimisation	OC15	Cost value of jobs of family	OC35		
	Investment cost minimisation	OC16	Maintenance costs minimisation	OC36		
	Profit maximisation	OC17	Delivery and tardiness costs minimisation	OC37		
	Transport cost minimisation	OC18	Total costs minimisation	OC38		
	Raw Material purchasing cost minimisation	OC19				
	Raw material inventory holding cost	OC20				
Time (OT)	Lead time minimisation	OT1	Mean flow time minimisation of jobs	OT9		
	Production time minimisation	OT2	Time of sequences minimisation	OT10		
	Warehouse time minimisation	OT3	Cycle time minimisation	OT11		

Table 3.5. Continued. Production planning, scheduling and sequencing objectives.						
Type of objectives	Subtype	Designation	Subtype	Designation		
Time (OT)	Preparation times minimisation	OT4	Work overload minimisation	OT12		
	Transition time minimisation	OT5	Makespan minimisation	OT13		
	Setup time minimisation	OT6	Total weighted completion			
	Tardiness minimisation	OT7	time minimisation	OT14		
	Earliness minimisation	OT8				
Products (OP)	Product sold maximisation	OP1	Inventory quantity minimisation	OP4		
	Shortage product minimisation	OP2	Faulty products minimisation	OP5		
	Total production maximisation	OP3	Quality of products maximisation	OP6		
Resources (ORS)	Labour minimisation hiring and lay-off (quantity)	OR1	Machine utilisation maximisation	OR2		
Service (OS)	Customer service level maximisation	OS				
Sustainability (OST)	Environmental issues minimisation	OST1	Social factors minimisation	OST3		
	Cultural elements maximisation	OST2				

The production planning process at the tactical level generally finds objectives related to searches for financial benefits. These financial benefits are represented by reductions in the different cost types. The costs to be minimised at this level are mainly related to production, hiring or firing, inventory and storage, subcontracting, and production in normal time and overtime. In the combined production planning and scheduling approach are less frequent in the literature, since they require greater coordination of operations, so the models and techniques must be incorporated into a single framework [94]. This approach moves from a first phase that uses aggregated data to a second phase that employs more detailed information. Thus, in the aggregated phase, objectives are generally sought to minimise production, inventory, setup, backorder, and normal and overtime costs, while the objectives in the scheduling stage are essentially based on times, such as lead time, setup time, tardiness and earliness.

Production scheduling models focus on optimising the facilities and resources in different areas on shorter planning horizons. The objectives pursued by scheduling models seek to reduce setup, holding and production costs and times, and to minimise makespan and tardiness.

The objectives sought by the combined scheduling and sequencing approaches are primarily to minimise setup and holding costs. Finally, the sequencing problem mainly pursues minimising makespan, tardiness and work overload, but also tends to reduce setup, transition, mean flow, sequence and cycle times.

Ref. / Goals in the objective function	Costs	Time	Product	Resources	Service	Sustain- ability
RC. Wang & Fang	0C1; 0C7; 0C8;					
[33]	OC17					
Grabowski &		OT13				
Wodecki [48]		0115				
D. Gupta &	004.005					
Magnusson [26]	001,000					
Nonås & Olsen [49]		OT7				
Bellabdaoui &		OT10				
Teghem [50]						
Hooker [51]	0C28	017;				
		0113				
0 0 1 [20]	OC1; OC4; OC5;	016;				
Omar & Teo [28]	OC12; OC13	017;				
D Decenic P	001.004	018				
P Doganis &	001;004;					
Sariinveis [52]	000; 0021					
Sarimveis [55]	0021; 0022;					
Carlioppa et al [53]	001.004.					
Gagnoppa et al. [55]	0C5					
Moon et al. [54]		OT7;				
		ОТ9;				
		OT13				
Fakhrzad & Khademi	OC2; OC4;					
Zare [56]	0C17; 0C21					
Leung & Chan [34]	OC17		OP5	OR2		
Baykasoglu &	OC1; OC7;		OP1; OP4	OR1		
Gocken	OC12					
[35]						
Aghezzaf et al. [30]	0C1; 0C5	OT1				
Mirzapour Al-E-	OC1; OC5;		OP2			
Hashem et al. [37]	OC8; OC10;					
	OC18; OC19;					
	0C20; 0C23					

Table 3.6. The main objectives of the proposed models.

Table 3.6. Continued. The main objectives of the proposed models.							
Ref. / Goals in the objective function	Costs	Time	Product	Resources	Service	Sustain- ability	
Sillekens et al. [36]	OC1; OC6;						
	OC8; OC12;						
	OC21; OC28						
Xue et al. [29]	OC1; OC4;						
	OC5; OC8;						
	OC12; OC13						
Mohammadi et al.		OT4;					
[57]		OT5;					
		OT6;					
		OT7					
Ramezanian et al.	OC1; OC4;						
[39]	OC5; OC8;						
	OC13; OC27						
Zhang et al. [38]	OC1; OC5;						
5	OC16						
Chakrabortty &	OC1; OC5;						
Akhtar Hasin [40]	OC7; OC8;						
	OC13; OC27						
Cheng et al. [59]		OT13					
Guimarães et al.	OC4; OC21						
[58]							
Chen et al. [60]		OT13					
Franz et al. [63]		OT12					
Golle et al. [65]		OT12					
Khalili-Damghani &	OC1; OC8;		OP6		OS		
Shahrokh [41]	OC9; OC10;						
	OC13; OC20;						
	OC21						
Mattik et al. [64]		OT7;					
		OT13					
Motta Toledo et al.	OC4; OC5;						
[61]	0C24						
Na & Park [62]		OT7					
Baumann &		OT13					
Trautmann [66]							
Abdeljaouad et al.		OT13					
[67]							
Fumero et al. [31]	OC17						
Gholamian et al.	OC1; OC8;		OP2	OR1			
[43]	OC10; OC19;						
	OC20; OC23;						
	OC26						
Makui et al. [1]	OC1; OC4;						
	OC5; OC8;						
	0C11						

Table 3.6. Continued. The main objectives of the proposed models.

lable 3.6. Continued. The main objectives of the proposed models.						
Ref. / Goals in the objective function	Costs	Time	Product	Resources	Service	Sustain- ability
Tavaghof-Gigloo et	OC12; OC21;					
al. [42]	OC25; OC27					
Aroui et al. [68]		OT12				
Fang et al. [91]	OC1					
Fumero et al. [32]	OC17					
Zeppetella et al. [69]	0C1; 0C17					
de Kruijff et al. [44]	OC13; OC21;					
	OC22					
Lopes et al. [71]		OT11				
Mehdizadeh et al. [45]	OC17; OC29					
Torkaman et al. [70]	OC1; OC3;					
	OC4; OC21					
Chansombat et al.	OC1; OC4;					
[76]	OC14; OC15;					
	OC21; OC31					
Mönch & Roob [74];		OC35				
Verbiest et al. [73];	OC32; OC33;					
	OC34					
Woo & Kim [72]		OT13				
Ekici et al. [75]		OT7;				
		OT8				
de Armas & Laguna [77]			OP3			
Djordjevic et al. [46]		OT1;				
		OT2;				
		OT3;				
		OT4				
Rasmi et al. [27]	OC17					OST1;
						OST2;
						OST3
Bensmain et al. [47]	0C5; 0C36					
S. Wang et al. [78]		OT14				
De Smet et al. [79]	OC4; OC7;					
	OC21					
Yang & Xu [80]	OC37					
Otto & Li [81]	OC38					
Prata [93]	OT13					
Rodoplu et al. [82]	OC1					

Table 3.6. Continued. The main objectives of the proposed models.

3.3.5 Applications area and enterprise integration level

Some important aspects when modelling production planning, scheduling and sequencing problems are the industrial sector, the specific industry and the product type to which the model is proposed. The impact of applying a model

generated for a specific industry to another industry or sector type can be insignificant in some cases, but can be transcendental in others because the required costs and time directly affect the profitability and feasibility of processes. In order to analyse the impact of a model, and by analysing its extrapolation to another industry or sector, the reviewed models were classified into two categories: sectorial and transversal (see Table 3.7). The sectorial category responded to vertical measures and focused on a specific sector or industry. The transversal category referred to all those production operations that have had or could have an impact on multiple manufacturing sectors, and generally responded to horizontal measures [99].

According to Table 3.7, 30% of the articles were classified in the sectorial category and 70% in the transversal category. The problems that arose in the sectorial category generally responded to a specific industry's needs and often used real data: Omar & Teo [28]; Doganis & Sarimveis [55]; Leung & Chan [34]; Mirzapour Al-E-Hashem et al. [37]; Cheng et al. [59]; Chen et al. (2013) [60]; Mattik et al. [64]; Aroui et al. [68]; de Armas & Laguna [77]; Ekici et al. [75]. Other studies did not use real data, but tests with similar data to those of a real case were carried out: Nonås and Olsen [49]; Sillekens et al. [36]; Motta Toledo et al. [61]; Baumann & Trautmann [66], Franz [63]; de Kruijff et al. [44]; Chansombat et al. [76] were also classified as sectorial. The papers classified in the transversal category could be applied or adapted to various industrial sectors. Some studies, such as those by Tavaghof-Gigloo et al. [42] Khalili-Damghani & Shahrokh [41] Makui et al. [1] Djordjevic et al. [46], used real data from one industry (see Table 3.7). Although these studies were validated in a specific industry type, the proposed techniques and approaches could be applied to other sectors. In this category, several works conducted tests with similar data or instances to those of a real case [30, 40, 43, 50, 58, 73].

Enterprise integration is an Industrial Information Integration Engineering (IIIE) category. IIIE is a multidisciplinary research area, according to Chen, 2016 [100], "it is a set of foundational concepts and techniques that facilitate the industrial information integration process". Chen, 2016 [100] and Chen, 2020 [101] in their literature reviews on industrial information integration defined 37 and 27 categories, respectively, one of which was enterprise integration. According to Andres and Poler [102], enterprise integration was classified at two levels: intra- and inter-enterprise levels.

The intra-enterprise level refers to solving production planning, scheduling and sequencing as internal enterprise activities; that is, not sharing information with other supply chain network actors. However, information can be integrated

at different plans aggregation levels computed in the same enterprise. That is, the output of an aggregated plan is integrated into a tactical plan (master plan). In the same way, the master plan solution is integrated and used as the input of operational plans, namely sequencing or scheduling mathematical models. Omar and Teo [28] presented an integrated approach to determine the batches to be processed in a batch processing environment of multiple products and identical parallel machines. This approach was hierarchically divided into three levels. The first level solved the problem in aggregate by focusing on production decisions, inventories and backorders. A second level disaggregated the problem into monthly batches. A third level solved the sequencing of batches on parallel machines. Xue et al. [29] presented a modelling approach that integrated production planning and scheduling for decision support for senior and middle managers. The MILP model described aggregated production planning, family deaggregation and production scheduling with sequence-dependent setup times. Fumero et al. 2016 [31] reported an MILP model that presented a hierarchical approach to integrate different decision-making levels (production planning and scheduling decisions) on multiproduct batch plants. Aghezzaf, et al. [30] provided a model that hierarchically integrated planning decisions from semifinished products at an aggregated level up to finished products; that is, with disaggregated information. Fumero et al. 2017 [32] provided MILP that integrated planning and scheduling for the production planning of multiproduct batch plants in several stages operating in the campaign mode.

The inter-enterprise level is associated with the collaborative planning among different supply chain stakeholders. The majority of papers addressed production planning, scheduling and sequencing from the intra-enterprise perspective, while only one paper, that of Mirzapour Al-E-Hashem et al. [37], considered the collaborative network perspective. Mirzapour Al-E-Hashem et al. [37] contemplated multi-objective aggregate production planning to a multisite, multiperiod, multiproduct aggregate production planning problem. By developing a MOMINLP, this model proposed two objective functions. The first aimed to minimise total supply chain losses and the second to minimise the sum of the maximum amount of shortages between customers' zones during all periods. The computational tests of this model demonstrated its efficiency for supply chain production planning.

Table 3.7. Industry sectors and application type.							
Reference	Sec- torial	Transversal	Real case	Industry application			
RC. Wang & Fang [33]		Х	Ν				
Grabowski & Wodecki [48]		Х	Ν				
Nonås & Olsen [49]	Х		Ν	Maritime and shipyard in- dustry			
D. Gupta & Magnusson [26]		Х	Ν				
Bellabdaoui & Teghem [50]		Х	Y	Steelmaking-continuous casting			
Omar & Teo [28]	Х		Y	Chemical and pharmaceu- tical			
Hooker [51]		Х	Ν				
P Doganis & Sarimveis [52]		Х	Ν	Dairy			
Philip Doganis & Sarimveis [55]	Х		Y	Dairy			
Gaglioppa et al. [53]		Х	Ν	Process Industries			
Moon et al. [54]		Х	Ν				
Leung & Chan [34]	Х		Y	Surface and materials sci- ence			
Fakhrzad & Khademi Zare [56]		Х	Ν				
Baykasoglu & Gocken [35]		Х	Ν				
Mirzapour Al-E-Hashem et al. [37]	Х		Y	Wood and Paper			
Sillekens et al. [36]	Х		Ν	Automotive			
Xue et al. [29]		Х	Ν	Digital Electronic			
Aghezzaf et al. [30]		Х	Y	X-ray film			
Mohammadi et al. [57]		Х	Ν	5			
Ramezanian et al. [39]		Х	Ν				
Zhang et al. [38]		Х	Ν				
Cheng et al. [59]	Х		Y	Solar cell manufacturing			
Chen et al. [60]	Х		Y	Solar cell manufacturing			
Guimarães et al. [58]		Х	Ν	Beverage industry			
Chakrabortty & Akhtar Hasin [40]		Х	Ν	Textile			
Mattik et al. [64]	Х		Y	Steel			
Motta Toledo et al. [61]	X		N	Food (soft drinks)			
Franz et al. [63]	X		N	Automotive			
Na & Park [62]		Х	N				

Table 3.7. Continued. Industry sectors and application type.							
Reference	Sec- torial	Transversal	Real case	Industry application			
Khalili-Damghani &		Х	Y	Automotive colours and			
Shahrokh [41]				resins			
Baumann & Trautmann	Х		Ν	Consumer goods sector,			
[66]							
Golle et al. [65]		Х	Ν				
Abdeljaouad et al. [67]		Х	Ν				
Gholamian et al. [43]		Х	Ν	Wood and Paper			
Tavaghof-Gigloo et al.		Х	Y	Electronics manufacturer			
[42]							
Makui et al. [1]		Х	Y	Paper Industry			
Fumero et al. [31]		Х	Ν				
Aroui et al. [68]	Х		Y	Automotive			
Fang et al. [91]		Х	Ν	Iron and Steel			
Fumero et al. [32]		Х	Ν				
Zeppetella et al. [69]		Х	Ν				
de Kruijff et al. [44]	Х		Ν	High-tech low volume			
Torkaman et al. [70]		Х	Ν	Automotive			
Mehdizadeh et al. [45]		Х	Ν				
Lopes et al. [71]		Х	Ν				
Mönch & Roob [74]		Х	Ν				
Verbiest et al.[73]		Х	Ν	Chemical			
Woo & Kim [72]		Х					
Ekici et al. [75]	Х		Y	Electronics manufacturer			
de Armas & Laguna [77]	Х		Y	Pipe-insulation industry			
Chansombat et al. [76]	Х		Ν	Capital goods			
Djordjevic et al. [46]		Х	Y	Automotive			
Rasmi et al. [27]	Х		Y	Household appliances			
Bensmain et al. [47]		Х	Ν				
S. Wang et al. [78]		Х	Ν	Coating			
De Smet et al. [79]	Х		Ν	Paper			
Yang & Xu [80]		Х	Ν				
Otto & Li [81]		Х	N				
Prata [93]		Х	N				
Rodoplu et al. [82]		Х	N	Textile			

3.3.6 Solution quality and problem scale

Currently, mathematical models seek to capture the most relevant aspects of industry processes in a simplified way. Accordingly, very few or no models reflect all the aspects of a real-world company's processes. The use of mathematical models can be compromised between complexity and reality. Therefore,

employing optimisation algorithms, heuristics, metaheuristics and matheuristics allows the best performance of solutions for real world problems (large problem scales) without committing the efficiency of the required computational resources.

In this context, we classified the quality of solutions into three categories: (i) optimal (OP), characterised by being able to provide exact and optimal solutions; (ii) near-optimal (N-OP), containing solutions that generate an optimisation gap that is generally less than 2%; (iii) good (GD), which encompasses reasonable solutions in time and quality terms without reaching an optimum solution.

Table 3.8 compares the quality of the solutions with the problem scale, which refers to the data size or numerical instances of problems. According to the amount of data used to validate the proposed models, three categories of problem scale were defined: (i) a small dataset, which allows rapid tests to be run; (ii) a medium dataset, which is significantly bigger in size and dimension, and allows optimal or near-optimal solutions to be found in reasonable computation times; (iii) a large dataset, generally corresponding to instances that simulate real data or are in fact real data extracted from company manufacturing processes.

Finally, to answer RQ4, over 55.0% of the papers used small datasets to test the performance of the models and algorithms, most of these documents reached optimal solutions and 20 papers tested the problems with large-scale data. Of these papers, only the work of Chansombat et al. [76] obtained an OP solution. These authors proposed MILP for a problem that integrated production and preventive maintenance scheduling into the capital goods industry. This type of problem is generally solved with metaheuristic methods and they achieved N-OP solutions. However, the mathematical model presented by the authors achieved OP solutions for small, medium and large datasets.

In the area of production planning the use of large-scale dataset was less frequent. in approaches where the production planning and scheduling are jointly modelled and solved, no large-scale dataset studies were tested. However in these two approaches, most studies presented real industrial applications or generated similar data to the real ones of a company or industry, such as those presented by [1,30,34,37,40,42–44,46]. The size of the datasets generated by these studies were useful enough because they represented real cases. Therefore, it was not necessary to create or generate larger datasets. However, half the studies that tested real instances and medium datasets obtained OP results, but the other half obtained N-OP or GD solutions.
Production scheduling and sequencing problems are generally NP-Hard, which makes them difficult to solve when large datasets are considered. In the combined approaches that jointly solved scheduling and sequencing problems, tests were performed primarily with medium datasets and N-OP solutions were obtained (see Figure 3.4).



Figure 3.4. Distribution of the scale of problems and solution quality.

The application of different types of methods for production planning, scheduling and sequence problems, and the amount of data, provided distinct solutions (see Table 3.8); for example, those presented by Grabowski and Wodecki [48]. They proposed a Tabu Search Grabowsk and Wodecki (TSGW) algorithm to address a flow workshop scheduling problem with makespan criteria, and reported N-OP solutions for a problem with a large dataset consisting of 500 jobs and 20 machines. Hooker [51] proposed an algorithm based on decomposition benders for a programming problem. By applying it to a large dataset, they achieved N-OP solutions. Cheng et al. [59] presented a variation of the genetic algorithm, which they have called the hybrid code genetic algorithm. They presented N-OP solutions to problems with large datasets in a multistage and parallel-machine scheduling problem in the solar cell industry. Golle et al. [65] reported two models for sequencing products on a mixed-model assembly line to minimise work overload. These models were tested with different data sizes and provided N-OP solutions with a large dataset using CPLEX. Mattik et al. [64] worked with numerical tests

using parameters and data deriving from a real case in the steel industry and used MILP accompanied by LP relaxation in addition to the Branch and Bound procedures. With these solution approaches, the authors obtained almost OP solutions for medium and large datasets. Baumann and Trautmann [66] presented a hybrid method for the short-term scheduling of make-and-pack production processes and represented the problem through MILP. This model was able to provide N-OP solutions for large-scale and real-life instances for consumer goods companies.

From the review, we conclude that the resolution of production planning, scheduling and sequencing when mathematical model approaches were applied to large dataset instances was not efficient in terms of calculation time and the quality of solutions. Over 71.66% of the reviewed papers used different types of methods and around 28 distinct types of techniques were tested (see Table 3.4). In this regard, matheuristic and metaheuristic algorithms obtained better results in large instances, as well as the hybridisations of metaheuristic algorithms [60,68].

	Problem Scale			Solution Quality		
Authors	Small (S)	Medium (M)	Large (L)	ОР	N-OP	GD
RC. Wang & Fang [33]	х			S		
Grabowski & Wodecki [48]	х	х	х	S-M	L	
Nonås & Olsen [49]	х			S		
D. Gupta & Magnusson [26]	х					S
Bellabdaoui & Teghem [50]	х			S		
Omar & Teo [28]		х		М		
Hooker [51]	х	х	х	S-M	L	
P Doganis & Sarimveis [52]	х			S		
Philip Doganis & Sarimveis [55]	х			S		
Gaglioppa et al. [53]	х	х		S		Μ
Moon et al. [54]	х	х		S		Μ
Leung & Chan [34]	х			S		
Fakhrzad & Khademi Zare [56]	х	х		S	М	
Baykasoglu & Gocken [35]	х				S	
Mirzapour Al-E-Hashem et al. [37]		х		М		
Sillekens et al. [36]	х					S
Xue et al. [29]	х				S	
Aghezzaf et al. [30]	х			S		
Mohammadi et al. [57]		х			М	
Ramezanian et al. [39]	х	х		S		М
Zhang et al. [38]		х				М
Cheng et al. [59]	х	х	х	S	M-L	

Table 3.8. Problem scales and solutions qu	uality
--------------------------------------------	--------

Table 3.8. Continued. Problem scales and solutions quality.
plans in industrial enterprises.
Models and algorithms for the optimisation of replenishment, production and distribution

	Problem Sca		2	, So	Solution Quality	
Authors	Small (S)	Medium (M)	Large (L)	ОР	N-OP	GD
Chen et al. [60]	х	х	х	S	M-L	
Guimarães et al. [58]	х	х	х		S-M	L
Chakrabortty & Akhtar Hasin [40]	х				S	
Mattik et al. [64]		х	х		M-L	
Motta Toledo et al. [61]			х			L
Franz et al. [63]	х	х	х	S	М	L
Na & Park [62]	х	х				S-M
Khalili-Damghani & Shahrokh [41]	х			S		
Baumann & Trautmann [66]		х	х		M-L	
Golle et al. [65]	х	х	х	S	M-L	
Abdeljaouad et al. [67]	х	х	х	S		M-L
Gholamian et al. [43]		х		М		
Tavaghof-Gigloo et al. [42]		х		М		
Makui et al. [1]		х		М		
Fumero et al. [31]	х			S		
Aroui et al. [68]	х		х	S		L
Fang et al. [91]	х	х			S-M	
Fumero et al. [32]	х			S		
Zeppetella et al. [69]	х				S	
de Kruijff et al. [44]		х			М	
Torkaman et al. [70]	х	х			S-M	
Mehdizadeh et al. [45]	х	х		S		М
Mönch & Roob [74]	х		х	S	L	
Verbiest et al. [73]	х	х	х	S		M-L
Woo & Kim [72]	х	х	х	S-M	L	
Lopes et al. [71]		х		М		
Ekici et al. [75]			х			L
de Armas & Laguna [77]		х				М
Chansombat et al. [76]	х	х	х	S-		
				M-L		
Djordjevic et al. [46]	х			S		
Bensmain et al. [47]	х	х	х	S		M-L
S. Wang et al. [78]	х	х				S-M
De Smet et al. [79]	х	х	х	S		M-L
Yang & Xu [80]	х	x	x		S	M-L
Otto & Li [81]			x			L
Prata [93]	х		x		S-L	
Rodoplu et al. [82]	х		x	S	L	

3.4 Discussion and perspectives

The range of experiments carried out in the papers addressing production planning, scheduling and sequencing problems illustrate the vast variety of techniques addressed in the literature to solve such enterprise planning problems. It must be stated that MILP were the most widely used (44 of 60) to represent the different types of production planning, scheduling and sequence Problems. In production planning problems, we found that the majority of models dealt with aggregated plans and applied MILP as the modelling approach, and also applied heuristic and metaheuristic algorithms to solve them. LINGO was identified as the most widely used solver. Production planning problems were mainly classified as transversal models, which allowed their application regardless of industry and sector type. The proposed approaches were validated with small and medium datasets, and collectively achieved optimal solutions. Production scheduling problems were predominantly modelled with MILP models, and heuristic and metaheuristic algorithms were implemented to solve them. The CPLEX commercial software was extensively used to obtain N-OP solutions with medium and large datasets. Although some solutions were almost OP, only a few real cases appeared. In sequencing problems, MILP was still the most widely used modelling approach. The developed models were well tested with metaheuristic algorithms, such as the genetic algorithm and variable neighbourhood search algorithm, for which CPLEX was the most widely used commercial software. Multiple tests were run with a medium dataset, which usually obtained N-OP and GD solutions. Although no predominant sector appeared in our review, we detected that the automotive industry presented real cases.

Of the reviewed papers, only one study applied MOMINLP [37] to a problem of multi-objective aggregate production planning. To solve the proposed MOMINLP, the authors formulated this problem as an MOMINLP model and then transformed it into a linear model. Afterwards, MOMINLP was reformulated as a robust MOLP model, and this robust multi-objective model was then solved as a single-objective problem. Similarly, only two studies employed MINLP approaches [47,79]. MINLP models are generally used to address chemical engineering design problems [103]. Currently, there are different types of solvers to deal with MINLP models, such as AlphaECP, Antigone, AOA, BONMIN, BARON, Couenne, DICOPT, Juniper, LINDO, Minotaur, Muriqui, Pavito, SBB, SCIP and SHOT. These solvers have been tested with different instances by Kronqvist et al. [104]. The use of such a model is an area that involves many researchers who seek to develop solver software. For this reason, it is necessary to further investigate models and algorithms for this problem type [104]. Therefore, modelling MINLP, MOMINLP

and MOMILP for production planning, scheduling and sequencing problems is considered as a novel area to be explored.

Our systematic literature review enabled us to recognise an important research line for solving production planning, scheduling and sequencing problems, which includes the adoption of: (i) hybrid methods, as the combinations or hybridisations of metaheuristic or heuristic algorithms; (ii) the interoperation of mathematical models with metaheuristic or heuristic algorithms, designated in the literature as matheuristic algorithms. The papers applying MTA have been demonstrated to give good results as well as hybrid algorithms. According to Pellerin et al. [105], hybrid metaheuristic algorithms have been extensively studied in the past two decades. These authors [105] also analysed the performance of 36 different hybrid metaheuristic algorithms, applied to a resource-constrained project scheduling problem, and concluded that these techniques gave N-OP solutions quickly and efficiently. Here we found a gap in the literature as the MTA research line has not yet been studied in such depth as the hybrid metaheuristic algorithms area.

Some studies proposing MTA are analysed in Section 3.3.3 and obtained N-OP solutions for planning purposes. Some examples include the work carried out by Woo and Kim [72], in which proposed a combination of an MILP model with a simulated annealing algorithm and a genetic algorithm to deal with a parallel machine scheduling problem with time-dependent deterioration and multiple rate-modifying activities. In this problem, the authors were able to obtain N-OP solutions and suggested researching other matheuristics with other types of combinations to improve the computation time of the algorithms they presented. Verbiest et al. [73] described an MTA made up of an MILP model with an iterative local search algorithm for multiproduct batch plant designs on parallel production lines. With this combination they obtained good results in acceptable times, but proposed furthering their research to extend problems with more restrictions. The work by Ekici et al. [75] presented a combined of an ILP model with a Tabu search algorithm to address the unrelated parallel machine scheduling problem with sequence-dependent setups. This paper used real-world instances to test the proposed matheuristic technique, and this technique provided good solutions for the addressed problem.

Consequently, the design of MTA can be a flexible and useful tool for solving a wide range of planning problems [104, 105]. Thus matheuristic techniques have the advantage of reducing and simplifying problems into smaller problems or subproblems that can be solved using mathematical models and different types of solvers, which also benefits from the synergies among optimisation, heuristic and

metaheuristic techniques [107]. Therefore, future work should aim to validate the efficiency of matheuristics in large instances and in real problems. At present, there is limited evidence for the performance of these techniques. Accordingly, matheuristic techniques offer a wide field to be explored given the different combinations that can be developed.

3.5 Conclusions and future research

Production planning, scheduling and sequencing are usually the most critical activities that a company performs. For companies, the objective of these activities is to use the fewest resources in the shorted possible time to meet demand. In recent years, various methods and solution techniques have appeared in the literature to overcome such problems. We conducted a systematic literature review to offer a comprehensive perspective of production planning, scheduling and sequencing problems published from 2000 to 2020. This review leads to three main contributions. Firstly, from the studied and the analysed articles we present a holistic framework that characterises planning problems. Secondly, we organise and classify the existing papers according to the proposed holistic framework after identifying the aggregation and decision levels, the type of models, the objectives characterising each modelling approach, the followed resolution techniques, the development of tools, the application areas and sectors, the enterprise integration level, the experiments carried out to test real cases, the data size with which the problem was solved, and the quality of the obtained solutions. Finally, our contribution consists in identifying research opportunities.

According to the reviewed topic, future research lines are next determined. This review indicates that a gap still exists in developing mathematical models. Accordingly, novel modelling approaches should be developed to address and associate the parameters related to production and sustainability (for its three pillars: social, economical and environmental), and these should also address uncertain parameters. Another research area is to develop transversal formulations when modelling a planning problem. Transversal formulations could comprise general and modular formulations that can be adapted to the context of the application, and these formulations can be evaluated in different activity sectors.

Additionally, the development of matheuristic algorithms to propose new modelling approaches and solution techniques is needed to avoid large computational efforts, and to obtain GD or N-OP solutions when larger and more complex production planning, sequencing and scheduling problems are posed at

the industrial level. We also recommend studying non-linear mathematical models, using different types of non-linear solvers, and comparing the computational results of these solvers to those solving linear models. Finally, we propose mathematical models being generated from an inter-enterprise perspective as most of the presented papers have focused on intra-enterprise models without considering any type of collaboration between supply chain companies. Considering the importance of collaboration in planning, production scheduling and sequencing terms [102], we suggest that the problems of production planning, scheduling, and sequencing should be treated from a collaborative perspective, in which the different network partners share information. Several authors describe the advantages of inter-enterprise models, such as those presented by Hall and Potts [108], who describe that the implementation of Inter-enterprise architecture can reduce the total cost of the system by 20-25%. In addition, the implementation of inter-enterprise models provides additional benefits, such as harmonisation of processes, alignment of the commercial strategy, reduction of technological costs and risks, improved customer service and better responsiveness [109]. Therefore, the proposed framework will play a major role in guiding future research as it allows the key features of a production planning, scheduling and sequencing problem to be identified.

3.6 References

- [1] A. Makui, M. Heydari, A. Aazami, and E. Dehghani, "Accelerating Benders decomposition approach for robust aggregate production planning of products with a very limited expiration date," *Comput. Ind. Eng.*, vol. 100, pp. 34–51, 2016.
- [2] S. Nam and R. Logendran, "Aggregate production planning A survey of models and methodologies," *Eur. J. Oper. Res.*, vol. 61, pp. 255–272, 1992.
- [3] A. Cheraghalikhani, F. Khoshalhan, and H. Mokhtari, "Aggregate production planning: A literature review and future research directions," *Int. J. Ind. Eng. Comput.*, vol. 10, no. 2, pp. 309–330, 2019.
- [4] A. Jamalnia, J.-B. Yang, A. Feili, D.-L. Xu, and G. Jamali, "Aggregate production planning under uncertainty: a comprehensive literature survey and future research directions," *Int. J. Adv. Manuf. Technol.*, vol. 102, no. 1–4, pp. 159–181, 2019.
- [5] J. Mula, R. Poler, G. S. García-Sabater, and F. C. Lario, "Models for production planning under uncertainty: A review," *Int. J. Prod. Econ.*, vol. 103, no. 1, pp. 271– 285, 2006.

- [6] M. Díaz-Madroñero, J. Mula, and D. Peidro, "A review of discrete-time optimization models for tactical production planning," *Int. J. Prod. Res.*, vol. 52, no. 17, pp. 5171–5205, 2014.
- [7] I. Mundi, M. M. E. Alemany, R. Poler, and V. S. Fuertes-Miquel, "Review of mathematical models for production planning under uncertainty due to lack of homogeneity: proposal of a conceptual model," *Int. J. Prod. Res.*, vol. 7543, pp. 1–45, 2019.
- [8] M. Lage and M. G. Filho, "Production planning and control for remanufacturing: Literature review and analysis," *Prod. Plan. Control*, vol. 23, no. 6, pp. 419–435, 2012.
- [9] J. Mula, D. Peidro, M. Díaz-Madroñero, and E. Vicens, "Mathematical programming models for supply chain production and transport planning," *Eur. J. Oper. Res.*, vol. 204, no. 3, pp. 377–390, 2010.
- [10] E. Akçcal and S. Çetinkaya, "Quantitative models for inventory and production planning in closed-loop supply chains," *Int. J. Prod. Res.*, vol. 49, no. 8, pp. 2373– 2407, 2011.
- [11] D. Peidro, J. Mula, R. Poler, and F. C. Lario, "Quantitative models for supply chain planning under uncertainty," *Int. J. Adv. Manuf. Technol.*, vol. 43, no. 3–4, pp. 400–420, 2009.
- [12] D. Stindt and R. Sahamie, "Review of research on closed loop supply chain management in the process industry," *Flex. Serv. Manuf. J.*, vol. 26, no. 1–2, pp. 268–293, 2014.
- [13] K. Govindan, H. Soleimani, and D. Kannan, "Reverse logistics and closed-loop supply chain: A comprehensive review to explore the future," *Eur. J. Oper. Res.*, vol. 240, no. 3, pp. 603–626, 2015.
- [14] B. Fahimnia, C. S. Tang, H. Davarzani, and J. Sarkis, "Quantitative models for managing supply chain risks: A review," *Eur. J. Oper. Res.*, vol. 247, no. 1, pp. 1– 15, 2015.
- [15] S. Seuring and M. Müller, "From a literature review to a conceptual framework for sustainable supply chain management," *J. Clean. Prod.*, vol. 16, no. 15, pp. 1699–1710, 2008.
- [16] S. Seuring and S. Gold, "Conducting content-analysis based literature reviews in supply chain management," *Supply Chain Manag.*, vol. 17, no. 5, pp. 544–555, 2012.
- [17] M. Brandenburg, K. Govindan, J. Sarkis, and S. Seuring, "Quantitative models for sustainable supply chain management: Developments and directions," *Eur. J. Oper. Res.*, vol. 233, no. 2, pp. 299–312, 2014.
- [18] C. G. Kochan and D. R. Nowicki, "Supply chain resilience: a systematic literature

review and typological framework," *Int. J. Phys. Distrib. Logist. Manag.*, vol. 48, no. 8, pp. 842–865, 2018.

- [19] J. A. Parejo, A. Ruiz-Cortés, S. Lozano, and P. Fernandez, "Metaheuristic optimization frameworks: A survey and benchmarking," *Soft Comput.*, vol. 16, no. 3, pp. 527–561, 2012.
- [20] M. Saunders, P. Lewis, and A. Thornhill, "Research methods for business students." Pearson Education, Harlow, England, 2016.
- [21] A. Gupta and C. D. Maranas, "A hierarchical Lagrangean relaxation procedure for solving midterm planning problems," *Ind. Eng. Chem. Res.*, vol. 38, no. 5, pp. 1937–1947, 1999.
- [22] A. Gupta and C. D. Maranas, "Managing demand uncertainty in supply chain planning," *Comput. Chem. Eng.*, vol. 27, no. 8–9, pp. 1219–1227, 2003.
- [23] H. Min and G. Zhou, "Supply chain modeling: Past, present and future," Comput. Ind. Eng., vol. 43, no. 1–2, pp. 231–249, 2002.
- [24] G. Q. Huang, J. S. K. Lau, and K. L. Mak, "The impacts of sharing production information on supply chain dynamics: A review of the literature," *Int. J. Prod. Res.*, vol. 41, no. 7, pp. 1483–1517, 2003.
- [25] Ş. Y. Balaman, *Basics of Decision-Making in Design and Management of Biomass-Based Production Chains*. 2019.
- [26] D. Gupta and T. Magnusson, "The capacitated lot-sizing and scheduling problem with sequence-dependent setup costs and setup times," *Comput. Oper. Res.*, vol. 32, no. 4, pp. 727–747, 2005.
- [27] S. A. B. Rasmi, C. Kazan, and M. Türkay, "A multi-criteria decision analysis to include environmental, social, and cultural issues in the sustainable aggregate production plans," *Comput. Ind. Eng.*, vol. 132, pp. 348–360, 2019.
- [28] M. K. Omar and S. C. Teo, "Hierarchical production planning and scheduling in a multi-product, batch process environment," *Int. J. Prod. Res.*, vol. 45, no. 5, pp. 1029–1047, 2007.
- [29] G. Xue, O. Felix Offodile, H. Zhou, and M. D. Troutt, "Integrated production planning with sequence-dependent family setup times," *Int. J. Prod. Econ.*, vol. 131, no. 2, pp. 674–681, 2011.
- [30] E. H. Aghezzaf, C. Sitompul, and F. Van Den Broecke, "A robust hierarchical production planning for a capacitated two-stage production system," *Comput. Ind. Eng.*, vol. 60, no. 2, pp. 361–372, 2011.
- [31] Y. Fumero, M. S. Moreno, G. Corsano, and J. M. Montagna, "A multiproduct batch plant design model incorporating production planning and scheduling decisions under a multiperiod scenario," *Appl. Math. Model.*, vol. 40, no. 5–6, pp. 3498–

3515, 2016.

- [32] Y. Fumero, G. Corsano, and J. M. Montagna, "An MILP model for planning of batch plants operating in a campaign-mode," *Ann. Oper. Res.*, vol. 258, no. 2, pp. 415–435, 2017.
- [33] R.-C. Wang and H.-H. Fang, "Aggregate production planning with multiple objectives in a fuzzy environment," *Eur. J. Oper. Res.*, vol. 133, no. 3, pp. 521– 536, 2001.
- [34] S. C. H. Leung and S. S. W. Chan, "A goal programming model for aggregate production planning with resource utilization constraint," *Comput. Ind. Eng.*, vol. 56, no. 3, pp. 1053–1064, 2009.
- [35] A. Baykasoglu and T. Gocken, "Multi-objective aggregate production planning with fuzzy parameters," *Adv. Eng. Softw.*, vol. 41, no. 9, pp. 1124–1131, 2010.
- [36] T. Sillekens, A. Koberstein, and L. Suhl, "Aggregate production planning in the automotive industry with special consideration of workforce flexibility," *Int. J. Prod. Res.*, vol. 49, no. 17, pp. 5055–5078, 2011.
- [37] S. M. J. Mirzapour Al-E-Hashem, H. Malekly, and M. B. Aryanezhad, "A multiobjective robust optimization model for multi-product multi-site aggregate production planning in a supply chain under uncertainty," *Int. J. Prod. Econ.*, vol. 134, no. 1, pp. 28–42, 2011.
- [38] R. Zhang, L. Zhang, Y. Xiao, and I. Kaku, "The activity-based aggregate production planning with capacity expansion in manufacturing systems," *Comput. Ind. Eng.*, vol. 62, no. 2, pp. 491–503, 2012.
- [39] R. Ramezanian, D. Rahmani, and F. Barzinpour, "An aggregate production planning model for two phase production systems: Solving with genetic algorithm and tabu search," *Expert Syst. Appl.*, vol. 39, no. 1, pp. 1256–1263, 2012.
- [40] R. K. Chakrabortty and M. A. Akhtar Hasin, "Solving an aggregate production planning problem by using multi-objective genetic algorithm (MOGA) approach," *Int. J. Ind. Eng. Comput.*, vol. 4, no. 1, pp. 1–12, 2013.
- [41] K. Khalili-Damghani and A. Shahrokh, "Solving a new multi-period multiobjective multi-product aggregate production planning problem using fuzzy goal programming," *Ind. Eng. Manag. Syst.*, vol. 13, no. 4, pp. 369–382, 2014.
- [42] D. Tavaghof-Gigloo, S. Minner, and L. Silbermayr, "Mixed integer linear programming formulation for flexibility instruments in capacity planning problems," *Comput. Ind. Eng.*, vol. 97, pp. 101–110, 2016.
- [43] N. Gholamian, I. Mahdavi, and R. Tavakkoli-Moghaddam, "Multi-objective multiproduct multi-site aggregate production planning in a supply chain under uncertainty: Fuzzy multi-objective optimisation," *Int. J. Comput. Integr. Manuf.*,

vol. 29, no. 2, pp. 149–165, 2016.

- [44] J. T. de Kruijff, C. A. J. Hurkens, and T. G. de Kok, "Integer programming models for mid-term production planning for high-tech low-volume supply chains," *Eur. J. Oper. Res.*, vol. 269, no. 3, pp. 984–997, 2018.
- [45] E. Mehdizadeh, S. T. A. Niaki, and M. Hemati, "A bi-objective aggregate production planning problem with learning effect and machine deterioration: Modeling and solution," *Comput. Oper. Res.*, vol. 91, pp. 21–36, 2018.
- [46] I. Djordjevic, D. Petrovic, and G. Stojic, "A fuzzy linear programming model for aggregated production planning (APP)in the automotive industry," *Comput. Ind.*, vol. 110, pp. 48–63, 2019.
- [47] Y. Bensmain, M. Dahane, M. Bennekrouf, and Z. Sari, "Preventive remanufacturing planning of production equipment under operational and imperfect maintenance constraints: A hybrid genetic algorithm based approach," *Reliab. Eng. Syst. Saf.*, vol. 185, pp. 546–566, 2019.
- [48] J. Grabowski and M. Wodecki, "A very fast tabu search algorithm for the permutation flow shop problem with makespan criterion," *Comput. Oper. Res.*, vol. 31, no. 11, pp. 1891–1909, 2004.
- [49] S. L. Nonås and K. A. Olsen, "Optimal and heuristic solutions for a scheduling problem arising in a foundry," *Comput. Oper. Res.*, vol. 32, no. 9, pp. 2351–2382, 2005.
- [50] A. Bellabdaoui and J. Teghem, "A mixed-integer linear programming model for the continuous casting planning," *Int. J. Prod. Econ.*, vol. 104, no. 2, pp. 260– 270, 2006.
- [51] J. N. Hooker, "Planning and Scheduling by Logic-Based Benders Decomposition," *Oper. Res.*, vol. 55, no. 3, pp. 588–602, 2007.
- [52] P. Doganis and H. Sarimveis, "Optimal scheduling in a yogurt production line based on mixed integer linear programming," *J. Food Eng.*, vol. 80, no. 2, pp. 445– 453, 2007.
- [53] F. Gaglioppa, L. A. Miller, and S. Benjaafar, "Multitask and Multistage Production Planning and Scheduling for Process Industries," *Oper. Res.*, vol. 56, no. 4, pp. 1010–1025, 2008.
- [54] I. Moon, S. Lee, and H. Bae, "Genetic algorithms for job shop scheduling problems with alternative routings," *Int. J. Prod. Res.*, vol. 46, no. 10, pp. 2695– 2705, 2008.
- [55] P. Doganis and H. Sarimveis, "Optimal production scheduling for the dairy industry," Ann. Oper. Res., vol. 159, no. 1, pp. 315–331, 2008.
- [56] M. B. Fakhrzad and H. Khademi Zare, "Combination of genetic algorithm with

Lagrange multipliers for lot-size determination in multi-stage production scheduling problems," *Expert Syst. Appl.*, vol. 36, no. 6, pp. 10180–10187, 2009.

- [57] G. Mohammadi, A. Karampourhaghghi, and F. Samaei, "A multi-objective optimisation model to integrating flexible process planning and scheduling based on hybrid multi-objective simulated annealing," *Int. J. Prod. Res.*, vol. 50, no. 18, pp. 5063–5076, 2012.
- [58] L. Guimarães, D. Klabjan, and B. Almada-Lobo, "Pricing, relaxing and fixing under lot sizing and scheduling," *Eur. J. Oper. Res.*, vol. 230, no. 2, pp. 399–411, 2013.
- [59] C.-Y. Cheng, T.-L. Chen, L.-C. Wang, and Y.-Y. Chen, "A genetic algorithm for the multi-stage and parallel-machine scheduling problem with job splitting-A case study for the solar cell industry," *Int. J. Prod. Res.*, vol. 51, no. 16, pp. 4755–4777, 2013.
- [60] Y.-Y. Chen, C.-Y. Cheng, L.-C. Wang, and T.-L. Chen, "A hybrid approach based on the variable neighborhood search and particle swarm optimization for parallel machine scheduling problems - A case study for solar cell industry," *Int. J. Prod. Econ.*, vol. 141, no. 1, pp. 66–78, 2013.
- [61] C. F. Motta Toledo, L. De Oliveira, R. De Freitas Pereira, P. M. França, and R. Morabito, "A genetic algorithm/mathematical programming approach to solve a two-level soft drink production problem," *Comput. Oper. Res.*, vol. 48, pp. 40–52, 2014.
- [62] H. Na and J. Park, "Multi-level job scheduling in a flexible job shop environment," Int. J. Prod. Res., vol. 52, no. 13, pp. 3877–3887, 2014.
- [63] C. Franz, E. C. Hällgren, and A. Koberstein, "Resequencing orders on mixedmodel assembly lines: Heuristic approaches to minimise the number of overload situations," *Int. J. Prod. Res.*, vol. 52, no. 19, pp. 5823–5840, 2014.
- [64] I. Mattik, P. Amorim, and H. O. Günther, "Hierarchical scheduling of continuous casters and hot strip mills in the steel industry: A block planning application," *Int. J. Prod. Res.*, vol. 52, no. 9, pp. 2576–2591, 2014.
- [65] U. Golle, F. Rothlauf, and N. Boysen, "Car sequencing versus mixed-model sequencing: A computational study," *Eur. J. Oper. Res.*, vol. 237, no. 1, pp. 50– 61, 2014.
- [66] P. Baumann and N. Trautmann, "A hybrid method for large-scale short-term scheduling of make-and-pack production processes," *Eur. J. Oper. Res.*, vol. 236, no. 2, pp. 718–735, 2014.
- [67] M. A. Abdeljaouad, Z. Bahroun, A. Omrane, and J. Fondrevelle, "Job-shop production scheduling with reverse flows," *Eur. J. Oper. Res.*, vol. 244, no. 1, pp. 117–128, 2015.
- [68] K. Aroui, G. Alpan, and Y. Frein, "Minimising work overload in mixed-model

assembly lines with different types of operators: a case study from the truck industry," *Int. J. Prod. Res.*, vol. 55, no. 21, pp. 6305–6326, 2017.

- [69] L. Zeppetella, E. Gebennini, A. Grassi, and B. Rimini, "Optimal production scheduling with customer-driven demand substitution," *Int. J. Prod. Res.*, vol. 55, no. 6, pp. 1692–1706, 2017.
- [70] S. Torkaman, S. M. T. Fatemi Ghomi, and B. Karimi, "Hybrid simulated annealing and genetic approach for solving a multi-stage production planning with sequence-dependent setups in a closed-loop supply chain," *Appl. Soft Comput. J.*, vol. 71, pp. 1085–1104, 2018.
- [71] T. C. Lopes, A. S. Michels, C. G. S. Sikora, R. G. Molina, and L. Magatão, "Balancing and cyclically sequencing synchronous, asynchronous, and hybrid unpaced assembly lines," *Int. J. Prod. Econ.*, vol. 203, no. June, pp. 216–224, 2018.
- [72] Y. Bin Woo and B. S. Kim, "Matheuristic approaches for parallel machine scheduling problem with time-dependent deterioration and multiple ratemodifying activities," *Comput. Oper. Res.*, vol. 95, pp. 97–112, 2018.
- [73] F. Verbiest, T. Cornelissens, and J. Springael, "A matheuristic approach for the design of multiproduct batch plants with parallel production lines," *Eur. J. Oper. Res.*, vol. 273, pp. 933–947, 2018.
- [74] L. Mönch and S. Roob, "A matheuristic framework for batch machine scheduling problems with incompatible job families and regular sum objective," *Appl. Soft Comput. J.*, vol. 68, pp. 835–846, 2018.
- [75] A. Ekici, M. Elyasi, O. Ö. Özener, and M. B. Sarıkaya, "An application of unrelated parallel machine scheduling with sequence-dependent setups at Vestel Electronics," *Comput. Oper. Res.*, vol. 111, pp. 130–140, 2019.
- [76] S. Chansombat, P. Pongcharoen, and C. Hicks, "A mixed-integer linear programming model for integrated production and preventive maintenance scheduling in the capital goods industry," *Int. J. Prod. Res.*, vol. 57, no. 1, pp. 61– 82, 2019.
- [77] J. de Armas and M. Laguna, "Parallel machine, capacitated lot-sizing and scheduling for the pipe-insulation industry," *Int. J. Prod. Res.*, 2019.
- [78] S. Wang, M. Kurz, S. J. Mason, and E. Rashidi, "Two-stage hybrid flow shop batching and lot streaming with variable sublots and sequence-dependent setups," *Int. J. Prod. Res.*, vol. 57, no. 22, pp. 6893–6907, 2019.
- [79] N. De Smet, S. Minner, E. H. Aghezzaf, and B. Desmet, "A linearisation approach to the stochastic dynamic capacitated lotsizing problem with sequencedependent changeovers," *Int. J. Prod. Res.*, vol. 58, no. 16, pp. 4980–5005, 2020.
- [80] S. Yang and Z. Xu, "The distributed assembly permutation flowshop scheduling problem with flexible assembly and batch delivery," *Int. J. Prod. Res.*, vol. 0, no.

0, pp. 1–19, 2020.

- [81] A. Otto and X. Li, "Product sequencing in multiple-piece-flow assembly lines," Omega (United Kingdom), vol. 91, p. 102055, 2020.
- [82] M. Rodoplu, T. Arbaoui, and A. Yalaoui, "A fix-and-relax heuristic for the singleitem lot-sizing problem with a flow-shop system and energy constraints," Int. J. Prod. Res., vol. 58, no. 21, pp. 6532–6552, 2020.
- [83] Supply Chain Council SCC, Supply Chain Operations Reference Model SCOR version 11.0. 2012.
- [84] P. P. M. Stoop and V. C. S. Wiers, "The complexity of scheduling in practice," Int. J. Oper. Prod. Manag., vol. 16, no. 10, pp. 37–53, 1996.
- [85] L. F. Gelders and L. N. Van Wassenhove, "Production planning: a review," Eur. J. Oper. Res., vol. 7, no. 2, pp. 101–110, 1981.
- [86] B. Andres, R. Poler, L. Saari, J. Arana, J.-V. Benaches, and J. Salazar, "Optimization Models to Support Decision-Making in Collaborative Networks: A Review," in *Closing the Gap Between Practice and Research in Industrial Engineering*, 2018, pp. 249–258.
- [87] A. C. Hax and H. C. Meal, "Hierarchical integration of production planning and scheduling," 1973.
- [88] G. E. Vieira and P. C. Ribas, "A new multi-objective optimization method for master production scheduling problems using simulated annealing," *Int. J. Prod. Res.*, vol. 42, no. 21, pp. 4609–4622, 2004.
- [89] F. A. Rodammer, "A recent survey of production scheduling," *IEEE transactions* on systems, man, and cybernetics., vol. 18, no. 6. New York, N.Y. :, p. 841, 1988.
- [90] L. J. Thomas and J. O. McClain, "An overview of production planning," *Handbooks Oper. Res. Manag. Sci.*, vol. 4, no. C, pp. 333–370, 1993.
- [91] C. Fang, X. Liu, P. M. Pardalos, J. Long, J. Pei, and C. Zuo, "A stochastic production planning problem in hybrid manufacturing and remanufacturing systems with resource capacity planning," J. Glob. Optim., vol. 68, no. 4, pp. 851–878, 2017.
- [92] B. de A. Prata, L. R. de Abreu, and J. Y. F. Lima, "Heuristic methods for the singlemachine scheduling problem with periodical resource constraints," *Top*, no. 0123456789, 2020.
- [93] B. de A. Prata, C. D. Rodrigues, and J. M. Framinan, "Customer order scheduling problem to minimize makespan with sequence-dependent setup times," *Comput. Ind. Eng.*, no. September, p. 106962, 2020.
- [94] S. Kreipl and M. Pinedo, "Planning and Scheduling in Supply Chains: An Overview of Issues in Practice," *Prod. Oper. Manag.*, vol. 13, no. 1, pp. 77–92, 2004.
- [95] B. Andres, R. Sanchis, and R. Poler, "A Cloud Platform to support Collaboration

in Supply Networks," Int. J. Prod. Manag. Eng., vol. 4, no. 1, p. 5, 2016.

- [96] M. Gavrilas, "Heuristic and metaheuristic optimization techniques with application to power systems," Int. Conf. Math. Methods Comput. Tech. Electr. Eng. - Proc., pp. 95–103, 2010.
- [97] K. Sörensen, M. Sevaux, and F. Glover, "A History of Metaheuristics," in Handbook of Heuristics, R. Martí, P. Panos, and M. G. C. Resende, Eds. Cham: Springer International Publishing, 2018, pp. 1–18.
- [98] M. A. Boschetti, V. Maniezzo, M. Roffilli, and A. Bolufé Röhler, "Matheuristics: Optimization, simulation and control," *Lect. Notes Comput. Sci. (including Subser. Lect. Notes Artif. Intell. Lect. Notes Bioinformatics)*, vol. 5818 LNCS, pp. 171–177, 2009.
- [99] A. Andreoni and P. Neuerburg, "Manufacturing Competitiveness in South Africa : Matching industrial systems and policies," Int. Conf. Manuf. Led Growth Employ. Equal., no. May, pp. 1–72, 2014.
- [100] Y. Chen, "Industrial information integration—A literature review 2006–2015," J. Ind. Inf. Integr., vol. 2, pp. 30–64, 2016.
- [101] Y. Chen, "A survey on industrial information integration 2016–2019," J. Ind. Integr. Manag., vol. 5, no. 1, pp. 33–163, 2020.
- [102] B. Andres and R. Poler, "A decision support system for the collaborative selection of strategies in enterprise networks," *Decis. Support Syst.*, vol. 91, pp. 113–123, 2016.
- [103] N. V Sahinidis, "Mixed integer nonlinear programming 2018," Optim. Eng., vol. 20, no. 2, pp. 301–306, 2019.
- [104] J. Kronqvist, D. E. Bernal, A. Lundell, and I. E. Grossmann, *A review and comparison of solvers for convex MINLP*, vol. 20, no. 2. Springer US, 2019.
- [105] R. Pellerin, N. Perrier, and F. Berthaut, "A survey of hybrid metaheuristics for the resource-constrained project scheduling problem," *Eur. J. Oper. Res.*, vol. 280, no. 2, pp. 395–416, 2020.
- [106] S. S. G. Perumal, J. Larsen, R. M. Lusby, M. Riis, and K. S. Sørensen, "A matheuristic for the driver scheduling problem with staff cars," *Eur. J. Oper. Res.*, vol. 275, no. 1, pp. 280–294, 2019.
- [107] E. Guzman and R. Poler, "A matheuristic approach for sourcing, production, and delivery plans optimization," in *Organizational Engineering in Industry 4.0. Book of abstracts*, D. A. de la Fuente García, R. Pino Díez, P. Priore Moreno, F. J. Puente García, A. Gómez Gómez, J. Parreño Fernández, M. I. Fernández Quesada, N. García Fernández, R. Rosillo Camblor, and B. Ponte Blanco, Eds. Servicio de Publicaciones de la Universidad de Oviedo, 2019, p. 208.

- [108] N. G. Hall and C. N. Potts, "Supply Chain Scheduling: Batching and Delivery," Oper. Res., vol. 51, no. 4, pp. 566–584, 2003.
- [109] A. Vargas, A. Boza, S. Patel, D. Patel, L. Cuenca, and A. Ortiz, "Inter-enterprise architecture as a tool to empower decision- making in hierarchical collaborative production planning," *Data Knowl. Eng. j*, vol. 105, pp. 5–22, 2016.

Chapter 4

A decision-making tool for algorithm selection based on a fuzzy TOPSIS approach to solve replenishment, production and distribution planning problems.

Guzman, Eduardo, Beatriz Andres, and Raul Poler. 2021. "A decision-making tool for algorithm selection based on a fuzzy TOPSIS approach to solve replenishment, production and distribution planning problems." Mathematics 10 (9). https://doi.org/10.3390/math10091544.

Abstract:

A wide variety of methods and techniques with multiple characteristics are used in solving replenishment, production and distribution planning problems. Selecting a solution method (either a solver or an algorithm) when attempting to solve an optimization problem involves considerable difficulty. Identifying the best solution method among the many available ones is a complex activity that depends partly on human experts or a random trial-and-error procedure. This paper addresses the challenge of recommending a solution method for replenishment, production and distribution planning problems by proposing a decision-making tool for algorithm selection based on the fuzzy TOPSIS approach. This approach considers a collection of the different most commonly used solution methods in the literature, including distinct types of algorithms and solvers. To evaluate a solution method, 13 criteria were defined that all address several important dimensions when solving a planning problem, such as the computational difficulty, scheduling knowledge, mathematical knowledge, algorithm knowledge, mathematical modeling software knowledge and expected computational performance of the solution methods. An illustrative example is provided to demonstrate how planners apply the approach to select a solution method. A sensitivity analysis is also performed to examine the effect of decision-maker biases on criteria ratings and how it may affect the final selection. The outcome of the approach provides planners with an effective.

4.1 Introduction

The supply chain comprises different sequential activities, such as replenishment, production and distribution, which must all be planned and optimized. The main management function of companies is planning [1]. Planning activities aim to effectively coordinate and schedule a company's available resources [2]. Planning is accompanied by a set of decisions to be made by the planning manager; for example, a planner must make decisions about the quantity of materials needed for production by taking into account storage capacity and production batches to reduce production and inventory costs, production scheduling and sequencing on machines, and to finally make decisions about the delivery flow of finished products to customers or distribution centers [3].

Many real-world combinatorial optimization problems, such as those in transportation and logistics [4-6] and manufacturing [7-9], pose a huge challenge due to the high complexity of most companies' operations given the type of industry to which they belong. They are also subject to not only dynamic conditions, such as customer demands, processing times, returns on investment, but also to uncertainties, such as unavailability of items, changes in market conditions and shortages due to changes in demand [10].

Thus, planning problems seek to maximize profit or gain while minimizing costs and meeting market, environmental and societal constraints. For example, in supply planning problems, there is a direct relation between inventory costs and the costs associated with distribution planning, such as transportation costs and on-time delivery to customers [11]. Therefore, the difficulty of such problems is substantial due to the amount of data they handle [12], nonlinearities and discontinuities, complex constraints, possible conflicting objectives and uncertainty [13]. Hence different types of solvers are used to solve these problems, as are algorithms because of their computational difficulty [14].

Given the large number of algorithms for solving replenishment [15], production [16] and distribution planning problems [17], how to effectively select an algorithm for a given task or a specific problem is an important, but also difficult issue. Peres and Castelli [18] highlight that rules which standardize the formulations of existing combinatorial optimization problems (COP) in planning are lacking, which means that researchers have to start building an algorithm from scratch, which thus limits the interoperability of this field because the algorithms in the literature must be adjusted to solve a specific problem. These authors conclude that the consolidation of combinatorial optimization problems is lacking

and note that this is important for the field of COPs to reach a higher degree of maturity.

The algorithm selection problem (ASP) is an active research area in many fields, such as operations research [19–21] and artificial intelligence (AI) [22, 23]. For many decades, researchers have developed increasingly sophisticated techniques and algorithms to solve difficult optimization problems [18]. These techniques include mathematical programming approaches, heuristics, metaheuristics, nature-inspired metaheuristics, matheuristics and various hybridizations [24]. Literature reviews such as that presented by Jamalnia et al. [25], who reviewed the aggregate production planning problem under uncertainty between 1970 and 2018, detailed the use of approximately 24 different techniques to solve this type of problem out of 92 reviewed papers. Kumar et al. [26] presented a literature review covering the period from 2000 to 2019 of the quantitative approaches used to solve production and distribution planning problems. They found 13 different techniques and types of solvers, including CPLEX and LINGO, to solve this type of problem out of 74 papers. Pereira et al. [27] analyzed the tactical sales and operations planning problem. To do so, they reviewed 103 papers, where the year was not limited. They detailed about 35 different techniques to solve this type of problem. Hussain et al. [28] conducted a literature review of the applications of metaheuristic algorithms and found 140 different metaheuristic algorithms in 1222 publications over a 33-year search period (1983 to 2016).

Different research papers have conducted experimental studies to determine the performance of an algorithm [29–32] or several algorithms according to a problem type with a collection of datasets available in the literature [33–35]. For example, Pan et al. [36] compared three constructive heuristics and four metaheuristics (discrete artificial bee colony, scatter search, iterated local search, iterated greedy algorithm) for the distributed permutation flowshop problem, for which they made extensive comparative evaluations based on 720 instances. However, these comparisons do not provide any enlightening results because they are generally limited to a set of algorithms and to a specific problem set [24].

In practice, algorithm performance vastly varies from one problem state to another. In many cases, heuristic [37], metaheuristic [28] and matheuristic [38] techniques involve randomization, such as genetic algorithm, particle swarm optimization, bee swarm optimization, bat algorithm, artificial tribe algorithm and firefly algorithm [39–43], which results in performance variability, even across repeated trials in a single problem instance [44]. Risk is an important additional feature of algorithms because the planner or the person in charge of selecting an algorithm for planning must be willing to settle for average or lower performance

in exchange for a reasonable answer or may also find a better solution than that expected in the same resolution time. This situation is often encountered in companies that attempt to maximize their profits because these problems are solved by constructing mixed strategies, i.e. strategies that meet the desired risk and return.

Nowadays if a study demonstrates the superiority of one algorithm over other algorithms, that algorithm can be expected to be useful for other problem types for which it has not yet been tested. No-free-lunch (NFL) theorems [45] describe that there is no single algorithm that outperforms all algorithms in all the instances of a problem [24].

Therefore, the selection of the most suitable algorithm to solve an optimization problem for replenishment, production and distribution planning is a very difficult task. Algorithm selection requires advanced knowledge of the efficiency of algorithms, the characteristics of the problem, as well as mathematical and statistical knowledge. However, having the necessary knowledge to find a solution with algorithms does not guarantee success [46].

Algorithm selection depends mainly on the expected results and the data that the company has at the time. Therefore, the properties or characteristics of the business problem must be examined. For this purpose, the linearity of the problem, the number of parameters and the characteristics that the solution supports must be analyzed.

Evaluating algorithms to solve a problem usually involves more than one criterion, such as problem type, problem knowledge, performance, computation time, the quality of the expected solution and programming knowledge. Therefore, algorithm selection can be modeled as a multicriteria decision-making problem [22].

The objective of multicriteria decision making (MCDM) is to identify the most eligible alternatives from a set of alternatives based on qualitative and/or quantitative criteria with different units of measurement to select or rank them [47]. Different techniques such as AHP, ELECTRE, PROMETHEE, SAW, TOPSIS and VIKOR are used to solve MCDM problems [3]. Several studies have been conducted to compare the performance of these techniques; for example, that presented by Zanakis et al. [48], which compared eight MCDM techniques (four variations of AHP, ELECTRE, TOPSIS and SAW). It concluded that different techniques are affected mainly by the number of alternatives because as alternatives increase, methods tend to generate similar final rankings. Opricovic and Tzeng [49] performed a comparative analysis of the VIKOR and TOPSIS methods. Both these

methods are based on an aggregation function that represents the closeness to the ideal. The study revealed that the main differences between the two methods were the employed normalization method types.

Opricovic and Tzeng [50] compared the extended VIKOR method to ELECTRE II, PROMETHEE and TOPSIS. The obtained results showed that ELECTRE II, PROMETHEE and VIKOR gave similar results, while TOPSIS presented different results in some alternatives. Chu et al. [51] made a comparison of the VIKOR, TOPSIS and SAW methods. The study revealed that SAW and TOPSIS presented similar classifications, while VIKOR presented different results. These authors concluded that VIKOR and TOPSIS provided results that were close to reality. Ozcan et al. [52] presented a comparative analysis of the TOPSIS, ELECTRE and Grey Theory techniques for the warehouse location selection problem, where the Grey Theory provided different results to TOPSIS and ELECTRE. Instead, the last two obtained similar results.

In situations in which information is not quantifiable or incomplete, as in realworld problems where data may be incomplete or imprecise, i.e., nondeterministic, data can be represented in a fuzzy way using linguistic variables to represent decision makers' preferences in complex or not well-defined situations. Imprecision in MCDM problems can be modeled using the fuzzy Set Theory, which is used to extend different MCDM techniques. In this background, Ertuğrul and Karakaşoğlu [53] conducted a comparative study of the fuzzy AHP and fuzzy TOPSIS methods for the facility location selection problem. Both methods obtained the same results, i.e., the same rank order of alternatives.

Other studies have used an extension of the classical fuzzy set called the intuitionistic fuzzy set, as proposed by Atanassov [54]. Intuitionistic fuzzy sets have been applied in many fields, such as facility location selection [55], supplier selection [56], evaluation of project and portfolio management information systems [57, 58], and personnel selection [59]. Büyüközkan and Güleryüz [60] compared the performance of ranked fuzzy TOPSIS and intuitionistic fuzzy TOPSIS by detailing how the alternatives ranking barely differed between the two approaches.

From the above comparisons, it is clear that many techniques are available for multi-criteria decision making [61]. These techniques have their advantages and limitations over others depending on the type of problem [62].

Different MCDM techniques have been used for the classification algorithm selection problem, such as the study by Lamba et al. [63] in which TOPSIS and VIKOR were used to evaluate 20 classification algorithms. Both methods obtained

similar results. Peng et al. [22] used four different MCDM techniques (TOPSIS, VIKOR, PROMETHEE II and WSM) to select multiclass classification algorithms. The TOPSIS, VIKOR and PROMETHEE II methods achieved similar classifications, while WSM obtained slightly different ones. Peng et al. [64] evaluated ranking algorithms for financial risk prediction purposes. Using TOPSIS, PROMETHEE and VIKOR, they obtained similar results for the three main ranking algorithms. They concluded that the followed techniques were advantageous for choosing a classification algorithm.

Along these lines, TOPSIS stands out as a widely used technique that is efficient for selecting classification algorithms. It has been successful in different areas such as supply chain and logistics management, environment and energy management, health and safety management, business and marketing management, engineering and manufacturing, human resource management and transportation management [47, 65–67] and, according to Chu et al. [51], is able to represent reality. It is also useful for companies because it can be run with a spreadsheet [68]. For all these reasons and given the fact that the choice of a solution method is subject to vagueness and uncertainty, we use the fuzzy TOPSIS method.

In this context, the present paper aims to answer this question: which solution method is suitable for a replenishment, production and distribution planning problem given a portfolio of algorithms or solvers?

To answer this question, and by taking into account that no research to date has analyzed the selection of algorithms for planning with a multicriteria decision method, a decision-making tool to select algorithms for a planning problem based on fuzzy TOPSIS is presented. To validate the use of the tool herein proposed, an illustrative example is presented, which has been validated by four different manufacturing companies. This paper is organized as follows. Section 4.2 deals with the literature review. The adopted methodology is shown in Section 4.3 and the numerical application of the methodology is presented in Section 4.4. The sensitivity analysis of the results is provided in Section 4.5. Finally, Section 4.6 includes the conclusions and future research lines.

4.2 Algorithm selection problem literature review

Algorithm selection has been widely addressed by the scientific community in both the mathematics [69, 70] and Artificial Intelligence (AI) [71, 72] areas. In the mathematical area, Stützle and Fernandes [73] report how the characteristics of problem instances make the performance of metaheuristics relative to the

properties of instances. Therefore, it is necessary to explore the relation between algorithms and instances. In the AI area, different models have been developed to predict which algorithm is the best one for a problem instance, which is conducted by analyzing the relation between the characteristics of an instance and a set of training data used by an algorithm. In this way, with an algorithms portfolio it is possible to predict which algorithm in a new problem instance is most likely to work [74].

Growing interest has been shown in the ASP to put previously developed algorithms to best use to solve a specific problem instead of developing new ones [75]. According to Leyton-Brown et al. [76], some algorithms are better than others on average, and there is rarely a best algorithm for a given problem. Instead "it is often the case that different algorithms perform well on different problem instances. This phenomenon is most pronounced among algorithms for solving NP-Hard problems, because runtimes for these algorithms are often highly variable from instance to instance". In this context, Rice [77] proposes the first description of methodologies to select algorithms. Kotthoff [75] defines this as the "task of algorithm selection involves choosing an algorithm from a set of algorithms on a per-instance basis in order to exploit the varying performance of algorithms over a set of instances".

In this regard, algorithm selection approaches have been successfully applied in different problem domains [78]. The following table summarizes a literature review of the various papers that have approached ASP from different perspectives (Table 4.1).

Author	Proposal
Lagoudakis and Littman [79]	Algorithm selection using reinforcement learning.
Xu et al. [80]	A scalable and completely automated portfolio construction. The authors improve the ASP methodology by integrating local search solvers as candidate solvers by predicting performance scores instead of runtime, and by using hierarchical hardness models that take into account different types of instances.
Smith-Miles [81]	A unified framework to take the algorithm selection problem as a learning problem and to use this framework to tie together cross- disciplinary developments in tackling the algorithm selection problem. The authors generalize metalearning concepts to algorithms that focus on tasks, including sorting, forecasting, constraint satisfaction and optimization.

Table 4.1. Research studies addressing the algorithm selection problem.

Table 4.1. Cont	inded. Research scudies addressing the algorithm selection problem.
Author	Proposal
Bischl et al. [35]	An algorithm selection problem as a cost-sensitive classification task that is based on an Exploratory Landscape Analysis.
Hoos et al. [82]	A modular open-solver architecture that integrates several different portfolio-based algorithm selection approaches and techniques.
Kotthoff [75]	An algorithm selection for combinatorial search problems.
Tierney and Malitsky [83]	An algorithm selection benchmark based on optimal search algorithms to solve the container premarshalling problem (CPMP), an NP-hard problem from the container terminal optimization field.
Cunha et al. [84]	A metalearning method is used to select the best recommendation algorithms within different scopes to allow to understand the relations between data characteristics and the relative performance of recommendation algorithms, which can be used to select the best algorithm(s) for a new problem. This work analyzes the algorithm selection problem for Recommender Systems by focusing on Collaborative Filtering.
Bożejko et al [85]	A local and optima network analysis and machine learning is used to select appropriate algorithms on an instance-to-instance basis.
Drozdov et al [86]	Graph convolutional network-based generative adversarial networks for the algorithm selection problem in classification terms.
Vilas Boas et al. [87]	Integer programming-based approaches to build decision trees for the algorithm selection problem. These techniques allow the automation of three crucial decisions by discerning the most important problem features to determine problem classes by grouping problems into classes, and then selecting the best algorithm configuration for each class.
Marrero et al. [88]	An efficient parallel genetic algorithm (GA) is proposed as a first step to solve the algorithm selection problem. GA is able to attain competitive results in optimal objective value terms and in a short time. The computational results show that the approach is able to efficiently scale and considerably reduce the average elapsed time to solve Knapsack Problem (KNP) instances.
De Carvalho et al. [21]	A cross-domain evaluation for multi-objective optimization. The authors investigate how four state-of-the-art online hyperheuristics with different characteristics perform to find solutions for 18 real-world multi-objective optimization problems. These hyperheuristics were designed in previous studies and tackle the algorithm selection problem from different perspectives: election-based, based on Reinforcement Learning and based on a mathematical function.

In manufacturing environments, formulations are usually very complex [78] because they present a variety of specific constraints related to the company's scope. Generally, these formulations can serve as blocks or subproblems for other formulations of other specific manufacturing environments. In this way, many formulations or algorithms can obtain similar results to the formulations proposed above. When selecting a formulation or algorithm, tuning the parameters of the different techniques is a very demanding task because each algorithm has different characteristics and the number of times that a parameter tuning has to be performed against different instances of a problem when performing a comparison can exponentially grow [33]. Furthermore, to compare algorithms and select one, the feature set of the instances must be taken into account because the characterization of instances determines a solution approach's performance. In practice, the information needed to establish the characteristics is not always available [89], and experimental results may lead to the fact that there is no single best or worst algorithm for all problem instances [46]. In this context, and as shown in Table 4.1, several approaches have been proposed to address the algorithm selection challenge, including heuristic algorithms, metaheuristics, hybrid metaheuristics, hyperheuristics, and machinelearning techniques. Many of these approaches integrate similarities, such as using a set of instances to learn, measuring or predicting the performance of the best algorithm. The success of algorithm selection approaches for some problem domains has motivated us to develop a decision-making tool to support planners of companies to select a solution method (algorithm or a solver) for replenishment, production and distribution planning problems.

4.3 Solution methodology

For combinatorial optimization problems with realistic discrete decision variables, such as scheduling, sequencing, distribution and transportation planning problems, performing an exhaustive search space for this problem type is not a realistic option despite having a finite search space. The literature includes several heuristic, metaheuristic and matheuristic algorithms, as well as tests with commercial and non-commercial high-performance solvers to solve such problems. So, this question arises: which algorithm is to be chosen for a combinatorial optimization problem?

Generally one way of finding an algorithm to solve a combinatorial optimization problem is to exhaustively run all the available algorithms and choose the best solution [90]. However, this method requires unlimited computational

resources and companies have limited computational, programming and mathematical resources, which makes it impossible to test all the algorithms or to use several solvers to test one instance or several for a specific problem. Weise et al. [12] emphasize that there is a variety of methods to solve different types of problems with acceptable performance, but they can be outperformed by very specialized methods.

Weise et al. [12] consider that there is no optimization method that is better or can outperform others, and the NFL Theorem [45] corroborates this theory. This theorem states that no optimization algorithm is likely to outperform several existing types of methods in different types of problems.

In turn, the same authors mention that the efficiency of an optimization algorithm is based on knowledge of a problem. Radcliffe [91] emphasizes that the algorithm's performance will improve with adequate knowledge of the problem. However, knowledge of one type of problem can be misleading for another type of problem [89] because there is no algorithm that outperforms others in all instances of a problem. Therefore, an algorithm's performance will be based on experience and empirical results.

Algorithm selection schemes are based mainly on approaches that either run a sequence of algorithms in a limited execution time [80, 82] or predict the performance of an algorithm for a given instance and select the algorithm with the best predicted performance [92].

Real-world planning problems are subject to inaccuracies and uncertainties, conflicts between constraints and objectives, discontinuities and nonlinearities [13]. Therefore, determining which algorithm is appropriate poses a challenge that can be analyzed using a multicriteria decision technique for ranking and prioritizing algorithms because algorithm selection involves multiple decisions that require the simultaneous assessment of the various advantages and disadvantages.

In most companies, the complexity of operations has several components that must be addressed at the same time. Evaluating an algorithm to solve a problem often involves more than one criterion, such as problem type, problem knowledge, performance, computation time, the quality of the expected solution and programming knowledge.

MCDM techniques integrate different criteria and an order of preference to evaluate and select the optimal option among multiple alternatives based on the expected outcome. The objective of these techniques is to obtain an ideal solution to a problem in which a decision makers' experience does not allow them to decide

among the various considered parameters. As a result, a ranking is obtained according to the selected criteria, their respective values and the assigned weights [93].

There are many criteria in real-life problems that can directly or indirectly affect the outcome of different decisions. Decision making often involves inaccuracies and vagueness that can be effectively dealt with using fuzzy sets. This method is especially important for clarifying decisions that are difficult to quantify or compare, especially if decision makers have different perspectives, as in this study. Therefore, we herein adopt the fuzzy TOPSIS methodology to model an algorithm or solver selection given a solution methods portfolio to solve replenishment, production and distribution planning problems.

In decision making problems, the Fuzzy Set Theory was introduced by Zadeh [94] to overcome the ambiguity and uncertainty of human thought and reasoning by using linguistic terms to represent decision makers' choices.

The TOPSIS method was originally proposed by Hwang and Yoon [95]. It is based on choosing an alternative that should have the shortest distance between the positive ideal solution (PIS) and the negative ideal solution (NIS), i.e., the selected alternative is obtained with the closest solution to the PIS and is farthest away from the NIS. The main limitation of this technique is that it cannot capture ambiguity in the decision making process [96]. To overcome this limitation, Chen [97] developed the Fuzzy TOPSIS Method to quantitatively evaluate the score of different alternatives by conferring weight to the different criteria described with linguistic variables. This section briefly describes the employed Fuzzy Set Theory and Fuzzy TOPSIS Method.

4.3.1 Fuzzy Set Theory and fuzzy numbers

The Fuzzy Set Theory [94, 98, 99] is associated with the TOPSIS method, and are related to another by the degree of membership of the elements in fuzzy sets. A fuzzy set is characterized by the membership function, which can come in different formats, e.g. triangular, sigmoid or trapezoidal. The membership function assigns a degree of membership to each object according to its relevance $\mu_A(x): x \rightarrow$ [0.0, 1.0]. To represent a fuzzy set, a tilde '~' is placed [68].

For our study, we consider a triangular fuzzy number, \tilde{A} , which is denoted by its vertices (*l*, *m*, *u*), as shown in Figure 4.1. Triangular fuzzy numbers are used to adapt decision makers' preference to capture the vagueness of linguistic evaluations, where *l*, *u* and *m* respectively, denote the lower bound, the upper bound and the crisp central value.



Figure 4.1. Fuzzy triangular number.

Membership function d of triangular fuzzy number \tilde{A} is defined as:

$$\mu_{\bar{A}}(x) = \begin{cases} \frac{x-l}{m-l}, & l \le x \le m, \\ \frac{u-x}{u-m}, & m \le x \le n, \\ 0, & otherwise \end{cases}$$
(1)

where $\tilde{A} = (l_A, m_A, u_A)$ and $\tilde{B} = (l_B, m_B, u_B)$ are two triangular fuzzy numbers with bases l, m, u. Then, the basic operational laws for triangular numbers are defined as:

$$\widetilde{A}(+) \widetilde{B} = (l_A, m_A, u_A)(+)(l_B, m_B, u_B) = (l_A + l_B, m_A + m_B, u_A + u_B)$$
(2)

$$\widetilde{A}(-)\widetilde{B} = (l_A, m_A, u_A)(-)(l_B, m_B, u_B) = (l_A - l_B, m_A - m_B, u_A - u_B)$$
(3)

$$\widetilde{A}(\times) \widetilde{B} = (l_A, m_A, u_A) (\times) (l_B, m_B, u_B) = (l_A \times l_B, m_A \times m_B, u_A \times u_B) \text{ for } l_A, l_B > 0; m_A, m_B > 0; u_A, u_B > 0$$
(4)

$$\widetilde{A}(\div) \widetilde{B} = (l_A, m_A, u_A) (\div) (l_B, m_B, u_B) = \left(\frac{l_A}{u_B}, \frac{m_A}{m_B}, \frac{u_A}{l_B}\right) \text{for } l_A, l_B > 0 ; m_A, m_B >$$

$$0; u_A, u_B > 0$$

$$(5)$$

$$k\widetilde{A} = kl_A, km_A, ku_A \tag{6}$$

$$\widetilde{A}^{-1} = (l_A, m_A, u_A)^{-1} = \left(\frac{1}{u_A}, \frac{1}{m_A}, \frac{1}{l_A}\right) \text{ for } l_A, l_B > 0 ; m_A, m_B > 0; u_A, u_B > 0$$
(7)

By assuming that fuzzy numbers \tilde{A} and \tilde{B} are real numbers, then the distance measure is identical to the Euclidean distance. Therefore, the vertex method is defined to calculate the distance between two fuzzy numbers (see Equation 8). Although there are several ways of measuring distances between fuzzy numbers [100], the vertex method is a simple and efficient method [97, 101].

$$d(\tilde{A},\tilde{B}) = \sqrt{\frac{1}{3} \left[(l_A - l_B)^2 + (m_A - m_B)^2 + (u_A - u_B)^2 \right]}$$
(8)

136 |

4.3.2 The Fuzzy TOPSIS Method

The main fuzzy TOPSIS idea is based on defining the fuzzy positive ideal solution (FPIS) and the fuzzy negative ideal solution (FNIS). The chosen alternative should have the shortest distance to the FPIS and the farthest distance to the FNIS. TOPSIS follows a systematic process and logic that seek to express the logic of human choice [102]. The basic fuzzy TOPSIS method steps are described in the following way (see [97, 103, 104]):

Step 1. Consider a set of *k* decision makers $(D_1, D_2, ..., D_k)$ with m alternatives (A1, A2, ..., An) and n criteria (C1, C2, ..., Cn) for which the decision matrix is established:

$$\widetilde{D} = \begin{array}{cccc} C_{1} & C_{2} & \dots & C_{n} \\ A_{1} & \begin{bmatrix} \widetilde{X}_{11} & \widetilde{X}_{12} & \dots & \widetilde{X}_{1n} \\ \widetilde{X}_{21} & \widetilde{X}_{22} & \dots & \widetilde{X}_{2n} \\ \vdots & \vdots & \vdots & \vdots \\ \widetilde{X}_{n1} & \widetilde{X}_{n2} & \dots & \widetilde{X}_{nm} \end{bmatrix} i = 1, 2, \dots, m; \ j = 1, 2, \dots, n$$

$$\widetilde{W} = [\widetilde{w}_{1}, \widetilde{w}_{2}, \dots, \widetilde{w}_{n}]$$
(9)

Considering that the perception of algorithms and solvers varies according to knowledge and experience with algorithms for planning, the average value method is applied; where \tilde{x}_{ij}^k is the rating or score of the alternative A_i in relation to criterion C_j evaluated by the *K*-th decision maker (Equation 10). The weights of criteria are aggregated using Equation 11, where \tilde{w}_j^k describes the weight of each criterion C_j according to decision makers D_k .

$$\tilde{x}_{ij} = \frac{1}{k} \left(\tilde{x}_{ij}^1 + \tilde{x}_{ij}^2 + \dots + \tilde{x}_{ij}^k \right)$$
(10)

$$\widetilde{w}_j = \frac{1}{k} \left(\widetilde{w}_j^1 + \widetilde{w}_j^2 + \dots + \widetilde{w}_j^k \right) \tag{11}$$

Step 2. Normalize the fuzzy decision matrix. Decision matrix \tilde{D} with m alternatives and n criteria is normalized to eliminate inconsistencies with the different units of measurement or scales to preserve the ranges of the normalized triangular fuzzy numbers. \tilde{R} represents the normalized decision matrix (Equation 12):

$$\tilde{R}_{j} = \left[\tilde{r}_{ij}\right]_{m \times n}, i = 1, 2, \dots, m; j = 1, 2, \dots, n$$
(12)

The normalization process is performed by Equations 13 and 14, where B and C represent the set of benefit and cost criteria, respectively.

$$\tilde{r}_{ij} = \left(\frac{l_{ij}}{u_j^+}, \frac{u_j}{u_j^+}, \frac{u_{ij}}{u_j^+}\right), \text{ and } u_j^+ = \max_i u_{ij} \text{ if } j \in B$$

$$(13)$$

$$\tilde{r}_{ij} = \left(\frac{l_j^-}{u_{ij}}, \frac{l_j^-}{m_{ij}}, \frac{l_j^-}{l_{ij}}\right), \text{ and } l_j^- = min_i l_{ij} \text{ if } j \in C$$

$$(14)$$

Step 3. Construct the weighted normalized fuzzy decision matrix \tilde{V} (Equation 15). \tilde{v}_{ij} is obtained by multiplying the weights of criteria \tilde{w}_j and the normalized fuzzy decision matrix \tilde{r}_{ij} values:

$$\tilde{V} = \left[\tilde{v}_{ij}\right]_{m \times n}, i = 1, 2, ..., m; j = 1, 2, ..., n$$
(15)

$$\tilde{v}_{ij} = \tilde{r}_{ij} \times \tilde{w}_j \tag{16}$$

Step 4. Obtain the FPIS (FPIS, A⁺) and the FNIS (FNIS, A⁻), as shown in Equation 17 and Equation 18, respectively. The ideal solutions can be defined according to Chen [97] as: $\tilde{v}_i^+ = (1, 1, 1)$ and $\tilde{v}_i^- = (0, 0, 0)$

$$A^{+} = \{ \tilde{v}_{1}^{+}, \tilde{v}_{j}^{+}, \dots, \tilde{v}_{m}^{+} \}$$
(17)

$$A^{-} = \left\{ \tilde{v}_{1}^{-}, \tilde{v}_{j}^{-}, \dots, \tilde{v}_{m}^{-} \right\}$$
(18)

Step 5. Calculate the distances for each alternative, where D_i^+ indicates the distance between the scores of alternative A_i to the FPIS (Equation 19), and D_i^- denotes the distances between the values of alternative A_i to the FNIS (Equation 20), where $d(\tilde{v}_a, \tilde{v}_b)$ represents the distance between two fuzzy numbers.

$$D_i^+ = \sum_{j=1}^n d(\tilde{v}_{ij}, \tilde{v}_j^+) , \ i = 1, 2, \dots, m; j = 1, 2, \dots, n$$
(19)

$$D_i^- = \sum_{j=1}^n d(\tilde{v}_{ij}, \tilde{v}_j^-) , \ i = 1, 2, \dots, m; j = 1, 2, \dots, n$$
(20)

Step 6. Determine proximity coefficient CC_i , which evaluates the rank order of all the alternatives A_i according to their overall performance. The proximity coefficient is calculated as shown in Equation 21.

$$CC_{i} = \frac{D_{i}^{-}}{\left(D_{i}^{+} + D_{i}^{-}\right)}$$
(21)

Step 7. Rank alternatives A_i , using a decreasing order of CC_i values, the shortest distances from the FPIS, i.e. close to 1, to indicate that the overall performance of alternative A_i is better because it is farther away from the FNIS. Having obtained the ranking order, decision makers select the most feasible alternative A_i .

4.4 The methodological approach for algorithm selection problem

This paper employs a three-stage methodology to select an algorithm or solver to solve a replenishment, production and distribution planning problem (see Figure

4.2). The objective of this section is to present a numerical analysis to demonstrate the performance of the proposed methodology.

The three stages of the proposed methodology are described in the following subsections.



Figure 4.2. A methodological approach for algorithm selection problem.

4.4.1 Stage 1 - Define criteria and alternatives

We first identify the different criteria that are taken into account when selecting a solution method; these criteria can be identified in the literature and are based

on the opinion of experts in the field [105]. According to each identified criterion, the decision maker evaluates the suitability of a solution method for the type of problem; that is, how algorithms or solvers can be suitable and formulated for a given problem.

In this research, 13 criteria are identified based on an exhaustive review of the literature (see [8, 18, 25, 27, 106, 107]) and the assessments of experts in the field of operations research. These criteria are presented in Table 4.2.

Id	Criteria	Definition
C1	Problem type	The replenishment (source), production (make) and distribution (deliver) planning problem type is determined by the SCOR (Supply Chain Operation Reference) methodology [106, 108] (see Figure 4.3). Each problem type has its own characteristics and computational difficulty.
		According to Weise et al. [12], it is very difficult to make accurate estimates of a problems' computational performance because a solution method's performance will almost always depend on experience, the empirical results based on related research areas and the rules of thumb established for these problems. So a problem's computational performance depends on different factors. Some of the main factors of a problem's complexity are: problem size, linearity, variables and presence of constraints [109]. Based on these considerations, criteria C2- C7 are proposed.
C2	Equation type	It expresses the equations present in the problem. These equations can be linear or non linear.
C3	Variable type	It represents the elements to be modeled. Variables can be integer, binary and continuous. Planning problems generally contain a combination of variables: Continuous + Integer, Integer + Binary, Continuous + Binary, Continuous + Integer + Binary. These combinations normally generate greater computational difficulty. Each combination can generate a different behavior for the solution method because algorithm or solver performance is linked with the amount of resources used. These resources can be: amount of memory, processing time to deal with each type of variable [12].
C4	Number of instantiated variables	It determines the number of variables present in a problem, which is a determining factor when establishing the expected response time to obtain an answer.

Table 4.2. Criteria for algorithm selection.

Models and algorithms for the optimisation of replenishment, production and distribution
plans in industrial enterprises.

	Table 4.2. Continued. Criteria for algorithm selection.				
Id	Criteria	Definition			
C5	Type of constraints and solutions	The constraint type determines the computational difficulty that the problem will have because constraints express limitations of resources. Some constraints can be expressed as follows:			
		decision >= data (e.g. production >= demand)			
		decision <= data (e.g. load <= capacity)			
		decision == data (e.g. production >= demand)			
		decision >= decision (e.g. production of A >= production of B)			
		decision <= decision (e.g. load of M <= load of N)			
		decision == decision (e.g. inventory A == inventory B)			
		continuity equations of some variables (e.g. Inventory == Inventory prior period + Production - Demand)			
		One factor that affects a problem's difficulty is when the expected solutions to the problem contain a route or sequence. These routing planning or sequencing problems are generally NP-hard [110, 111].			
C6	Number of constraints	The number of constraints contained in a problem can be a limiting factor for establishing the problem's difficulty. Therefore, the evaluator analyzes whether the set of constraints can be adapted to an algorithm or to a solver.			
С7	Dataset size	When representing the problem input data size, a problem's computation is directly related to the amount of data.			
C8	Programming knowledge	Programming knowledge is a determining factor when selecting an algorithm because it determines decision makers' ability to program one algorithm or several algorithms when having to test different algorithms in the hope to obtain a solution that meets the company's needs.			
С9	Mathematical knowledge	Mathematical knowledge is important when choosing whether to express the problem as a mathematical model or to directly choose an algorithm. Algorithms generally require certain mathematical knowledge.			
C10	Knowledge of algorithms	One aspect to take into account in companies is knowledge of the different algorithms.			

	Table	4.2. Continued. Criteria for algorithm selection.
Id	Criteria	Definition
C11	Software	This criterion is considered if the company has mathematical modeling software, but is not considered if the company does not.
		If the company has specific optimization software, the decision maker defines the scope of its performance against each alternative to solve a planning problem.
C12	Quality of solutions	This criterion establishes the quality of the expected solutions to the problem. These solutions can be optimal, near-optimal or good.
C13	Calculation time	The computation time sets the amount of expected time to obtain a solution for the problem.

Second, we identify the portfolio of solution methods (alternatives). This portfolio is composed of a set of nine algorithms and four solvers, identified as the most commonly used ones in the planning problems reported in [107, 112]. Alternatives are divided between different algorithm types, which are:

- Heuristic algorithms (HA). They are used when solvers or exact techniques cannot reach solutions in acceptable computation times. These techniques do not provide optimal solutions, but can offer solutions that come very close to the optimum in acceptable computation times [113];
- Metaheuristic algorithms (MA). According to Swan et al. [114], these techniques are: "an iterative master process that guides and modifies the operations of subordinate heuristics to efficiently produce high-quality solutions. At each iteration, it manipulates either a complete (or partial) single solution or a collection of such solutions";
- Matheuristic algorithms (MTA). They combine mathematical programming techniques and heuristic or metaheuristic algorithms [115].

The alternatives in this classification are A1 - HA/Benders decomposition, A2 - HA/LP and Fix, A3 - HA/LP Relaxation, A4 - MA/Tabu Search, A5 - MA/Genetic Algorithm, A6 - MA/Simulated annealing, A7 - MA/Variable Neighborhood Search, A8 - MTA/ Genetic Algorithm + Mathematical Model, A9 - MTA/Simulated annealing + Mathematical Model. Different solver types used to solve planning problems are also considered. For this purpose, commercial and non-commercial solvers are identified to deal with mathematical models with linear and nonlinear equations. These are: A10 - CPLEX (Commercial), A11 - CBC (Non-Commercial), A12 - BONMIN (Non-Commercial - NonLinear), A13 - LINDO (Commercial – Linear /NonLinear).



Models and algorithms for the optimisation of replenishment, production and distribution plans in industrial enterprises.



143 |





Figure 4.4. Hierarchical structure for algorithm selection.
The hierarchical structure that has been defined and constructed to assist in the process of selecting an algorithm or solver is shown in Figure 4.4. This structure is composed of four layers: the first one corresponds to the objective of this study; the second structure corresponds to the characterization of the different dimensions that have been proposed, which is composed of four dimensions (the problem type and its characteristics, programming knowledge, the software and the expected performance of algorithms or solvers); in the third layer comes the categorization of the 13 identified criteria; in the last one, methods or solution alternatives appear. The correlation between layer 3 and 4 is related to the performance of an algorithm or a solution method.

4.4.2 Stage 2 - Problem statement

In this stage, the type of planning problem to be addressed was defined, for which four expert decision makers working in different manufacturing companies in the planning area were invited to propose a planning problem. The decision makers proposed that the problem to be studied should be a production planning problem falling within the make classification, as shown in Figure 4.3.

Once the problem type has been defined, a questionnaire is developed to obtain the weight of preference of criteria and to thus evaluate alternatives according to the criteria. To devise the questionnaire, it is necessary to construct a fuzzy linguistic scale.

Linguistic scales are used to transform linguistic terms into fuzzy numbers [96]. Linguistic terms are subjective categories of the linguistic variable [116]. Zadeh [117] introduced the linguistic variable concept. A linguistic variable is a variable whose values allow computation with words instead of numbers [118]. Linguistic variables are used to represent decision makers' assessments, estimates and subjectivity [119].

To evaluate the criteria, we use a scale between 0 and 1. To rate the alternatives, we employ a scale from 0 to 10 [97]. The linguistic scales that evaluate the weights of the criteria and alternatives are shown in Table 4.3.

Table 4.3. Linguistic scales to assess the criteria and alternatives (Chen [97]).												
Linguistic ex alternativ	apressio ves (alg	n for rat orithms	ting)	Linguistic variable for the relative importance weight of criteria								
Linguistic expression	1	m	u	Linguistic expression	m	u						
Very Low (VL)	0.1	0.1	2.5	Very Low Importance (VLI)	0.01	0.03	0.25					
Low (L)	0.1	2.5	5.0	Low Importance (LI)	0.01	0.25	0.50					
Moderate (M)	2.5	5.0	7.5	Medium Importance (MI)	0.25	0.50	0.75					
High (H)	5.0	7.5	10.0	High Importance (HI)	0.50	0.75	1.00					
Very High (VH) 7.5 10.0 10.0			10.0	Very High Importance (VHI)	0.75	1.00	1.00					

Finally, decision makers were invited to review the questionnaire and to check its content. Based on this review, we were able to adjust the questionnaire.

In this same stage, we invited the four decision makers who worked in the planning area to evaluate the alternatives and to determine the weights of the criteria. For this purpose, we asked the decision makers to use the linguistic scale described in Table 4.3. An extract of the questionnaires used by the decision makers is shown in Table A1 and A2.

Table 4.4 details the fuzzy weights of each criterion based on the linguistic scales selected by the decision makers. The decision makers' ratings of the alternatives against all criteria are shown in Tables A3-A6.

		-		
	D1	D2	D3	D4
C1	MI	MI	MI	MI
C2	LI	LI	LI	LI
С3	VHI	VHI	VHI	VHI
C4	HI	HI	HI	HI
C5	LI	LI	LI	LI
C6	HI	MI	MI	MI
С7	HI	VHI	VHI	MI

Table 4.4. Decision makers' linguistic assessment of the criteria.

Table 4.4. Continued. Decision makers' linguistic assessment of the criteria.												
	D1	D2	D3	D4								
C8	LI	MI	HI	HI								
С9	LI	MI	HI	HI								
C10	MI	MI	LI	HI								
C11	LI	LI	LI	LI								
C12	VHI	VHI	VHI	VHI								
C13	VHI	VHI	VHI	VHI								

Models and algorithms for the optimisation of replenishment, production and distribution plans in industrial enterprises.

4.4.3 Stage 3 - Application of the Fuzzy TOPSIS Method

In this stage, the Fuzzy TOPSIS Method is used to analyze the different alternatives in relation to the identified criteria. The process used to apply the Fuzzy TOPSIS Method consists of five steps, which are detailed below.

Step 1. Based on the linguistic assessments of the alternatives (see Table A3-A6), the linguistic terms are converted into fuzzy numbers according to Table 4.3 and the fuzzy decision matrix is constructed. The aggregation of the ratings is performed using the fuzzy arithmetic mean, and the aggregate ratings for each alternative are obtained using Equation 10 (see Table 4.5).

In order to obtain the aggregate weights of each criterion, the fuzzy weights of each criterion are used, which are extracted by converting the linguistic terms of the four decision makers (see Table 4.4) into fuzzy numbers according to Table 4.3; for example, the fuzzy weights of criterion C7 of the four decision makers are D_1 = (0.50, 0.75, 1.00), D_2 =(0.75, 1.00, 1.00), D_3 =(0.75, 1.00, 1.00), D_4 = (0.25, 0.50, 0.75), and applying Equation 11, the aggregate fuzzy weight of C7= (0.56, 0.81, 0.93) is obtained. The results of the aggregate fuzzy weights of all the criteria are tabulated in Table 4.6.

		C1			C2			C3			C4			C5		[C6			C7			C8			С9			C10			C11			C12			C13	
	I	m	u	I	m	u	I	m	u	I	m	u	I	m	u	I	m	u	I	m	u	I	m	u	I	m	u	I	m	u	I	m	u	I	m	u	I	m	u
A	1 0.10	2.50	5.00	1.33	3.15	5.63	0.70	3.13	3 5.63	0.70	1.93	4.38	1.30	2.55	5.00	2.53	3 5.00	7.50	4.38	86.88	9.38	0.10	2.50	5.00	0.10	2.50	5.00	0.10	1.90	4.38	0.10	0.10	2.50	1.30	2.55	5.00	1.93	4.38	6.88
A	2 3.15	5 5.63	8.13	4.38	6.88	9.38	3.78	6.25	58.75	0.70	2.53	5.00	1.90	3.78	6.25	2.53	3 5.00	7.50	4.38	86.88	9.38	1.33	3.75	6.25	1.33	1.95	4.38	1.93	4.38	10.00	0.10	2.50	5.00	1.30	2.55	5.00	2.53	5.00	7.50
A	3 6.88	3 9.38	10.00	4.38	6.88	9.38	4.38	6.88	3 9.38	2.53	5.00	7.50	2.50	5.00	7.50	3.13	3 5.63	8.13	3 4.38	86.88	9.38	1.95	4.38	6.25	3.13	5.63	8.13	1.93	4.38	7.50	0.10	2.50	5.00	2.50	5.00	7.50	3.13	5.63	8.13
A	4 6.88	3 9.38	10.00	4.38	6.88	9.38	8 0.70	1.33	3.75	3.78	5.65	7.50	4.38	6.88	8.75	3.75	6.25	8.75	5 4.38	86.88	9.38	7.50	010.00	10.00	1.33	1.95	4.38	3.78	5.65	10.00	0.10	0.10	2.50	7.50	10.00	10.00	4.38	6.88	9.38
A	5 6.88	3 9.38	10.00	6.25	8.75	9.38	4.38	6.88	3 9.38	5.00	7.50	9.38	4.38	6.88	8.75	3.75	6.25	8.13	3 4.38	86.88	9.38	5.63	8.13	10.00	1.95	4.38	6.25	5.00	7.50	10.00	7.50	10.00	10.00	5.00	7.50	10.00	6.88	9.38	10.00
A	5 2.50	5.00	7.50	2.50	5.00	7.50	1.33	3.75	56.25	1.90	4.38	6.88	1.90	4.38	6.88	1.30	3.75	6.25	5 4.38	86.88	9.38	0.70	3.13	5.63	2.50	5.00	7.50	1.33	3.75	5.00	1.33	3.75	6.25	2.50	5.00	7.50	2.55	5.00	7.50
A	7 2.50	5.00	7.50	1.90	3.78	6.25	0.10	2.50	5.00	1.90	4.38	6.88	1.90	3.78	6.25	1.90	4.38	8 6.88	3 4.38	86.88	9.38	0.70	3.13	5.63	2.50	5.00	7.50	0.10	0.70	10.00	0.10	2.50	5.00	2.50	5.00	7.50	1.90	4.38	6.88
A	8 0.70	3.13	5.63	0.10	1.90	4.38	8 0.10	2.50	5.00	1.90	4.38	6.88	1.90	3.78	6.25	1.90	4.38	8 6.88	3 4.38	86.88	9.38	0.10	2.50	5.00	0.10	2.50	5.00	0.10	2.50	5.00	0.10	2.50	5.00	2.50	5.00	7.50	1.90	4.38	6.88
A	9 0.70	3.13	5.63	0.10	1.90	4.38	0.10	2.50	5.00	1.90	4.38	6.88	1.90	3.78	6.25	1.90	4.38	6.88	3 4.38	86.88	9.38	0.10	2.50	5.00	1.90	4.38	6.88	0.10	2.50	10.00	0.10	2.50	5.00	2.50	5.00	7.50	1.90	4.38	6.88
A1	0 4.38	3 6.88	9.38	5.63	8.13	9.38	3.75	6.25	58.75	3.13	5.63	8.13	3.75	6.25	8.75	2.50	5.00	7.50	4.38	86.88	9.38	3.13	5.63	8.13	4.38	6.88	9.38	5.00	7.50	10.00	4.38	6.88	9.38	5.63	8.13	9.38	5.63	8.13	9.38
A1	1 2.50	5.00	7.50	2.50	5.00	7.50	1.90	4.38	86.88	2.50	5.00	7.50	2.50	5.00	7.50	2.53	3 5.00	7.50	4.38	86.88	9.38	0.70	3.13	5.63	2.50	5.00	7.50	2.50	5.00	7.50	2.50	5.00	7.50	3.75	6.25	8.13	6.25	8.75	9.38
A1	2 0.10	0 1.90	4.38	0.10	0.70	3.13	0.10	2.50	5.00	0.70	2.53	5.00	0.10	1.90	4.38	0.70	3.13	5.63	3 0.70	3.13	5.63	0.10	2.50	5.00	0.10	1.90	4.38	0.10	2.50	10.00	0.10	2.50	5.00	0.10	2.50	5.00	0.10	2.50	5.00
A1	3 0.10	0 1.90	4.38	0.10	1.90	4.38	80.10	2.50	5.00	0.70	2.53	5.00	0.10	1.90	4.38	0.70	3.13	5.63	3 0.70	3.13	5.63	0.10	2.50	5.00	0.10	2.50	5.00	0.10	2.50	10.00	0.10	2.50	5.00	0.10	2.50	5.00	0.10	2.50	5.00

Table 4.5. Decision matrix with the aggregated scores.

148 |

Criteria	Aggregate fuzzy weights
C1	(0.25, 0.50, 0.75)
C2	(0.01, 0.25, 0.50)
С3	(0.75, 1.00, 1.00)
C4	(0.50, 0.75, 1.00)
C5	(0.01, 0.25, 0.50)
C6	(0.31, 0.56, 0.81)
C7	(0.56, 0.81, 0.93)
C8	(0.32, 0.56, 0.81)
С9	(0.32, 0.56, 0.81)
C10	(0.25, 0.50, 0.75)
C11	(0.01, 0.25, 0.50)
C12	(0.75, 1.00, 1.00)
C13	(0.75, 1.00, 1.00)

Step 2 and Step 3. Using Equations 13 and Equation 14, the normalized fuzzy decision matrix is obtained. For criteria C1-12, Equation 13 is used because the objective of these criteria is to maximize. For criterion C13, Equation 14 is applied because the aim is to minimize the computation time criterion. Table A7 shows the results of the normalized matrix.

After normalization, the weighted normalized decision matrix is calculated using Equation 16. The results are shown in Table A8.

Step 4. It is followed to calculate the FPIS and the FNIS because the positive triangular fuzzy numbers fall within the range [0, 1], and the FPIS and the FNIS are obtained by Equations 17 and 18. Then, the relative distance is calculated between the algorithms (alternatives) and is computed with Equations 19 and 20 (see Table 4.7).

Table 4.7. Distances between alternatives.

	D+	D-
A1	6.051	2.149
A2	5.596	2.724
A3	5.192	3.296

Table 4.7. Co	ontinue	d. Distan	ces betw	een alternatives.
		D+	D-	
	A4	5.136	3.370	
	A5	4.770	3.802	
	A6	5.595	2.748	
	A7	5.677	2.786	
	A8	5.861	2.467	
	A9	5.766	2.654	
	A10	4.918	3.597	
	A11	5.391	3.004	
	A12	6.176	2.279	
	A13	6.144	2.323	

Chapter 4. A decision-making tool for algorithm selection based on a fuzzy TOPSIS approach to solve replenishment, production and distribution planning problems.

	Step 5.	It is follow	ved to de	termine	the	closen	iess co	peffic	ient	using Eq	uatic	n
21.	CC_i The	e obtained	values re	epresent	the	total	score	of e	ach a	algorithm	for	а
pro	duction	planning pi	roblem. T	able 4.8	shov	vs the	obtair	ned r	esult	s.		

Alternative	Algorithm	ССі	Rank
A1	HA/Benders decomposition	0.262	13
A2	HA/LP and Fix	0.327	8
A3	HA/LP Relaxation	0.388	4
A4	MA/Tabu Search	0.396	3
A5	MA/Genetic Algorithm	0.444	1
A6	MA/Simulated annealing	0.329	6
A7	MA/Variable Neighborhood Search	0.329	7
A8	MTA Genetic Algorithm + Mathematical Model	0.296	10
A9	MTA Simulated annealing + Mathematical Model	0.315	9
A10	CPLEX (Commercial)	0.422	2
A11	CBC (Non-Commercial)	0.358	5
A12	BONMIN (Non-Commercial - NonLinear)	0.270	12
A13	LINDO (Commercial – Linear/ NonLinear)	0.274	11

Table 4.8. Closeness quotient and algorithms ranking.

When applying the proposed methodological approach based on the Fuzzy TOPSIS Method for a production planning problem, the GA is the most suitable solution method. This finding is not new because the literature review by Guzman et al. [107] concludes that GAs are the most widely used for this problem. Second in the ranking is CPLEX, which is the most widespread solver [107].

4.5 Sensitivity analysis

This section evaluates the effects of the different weightings of the criteria, i.e. we aim to evaluate the answers given by decision makers and how they influence algorithm selection. The aim of the sensitivity analysis is to make minor variations in the weights and to observe the influence of this variation on algorithm choice. The weights for the rating of algorithms range from very low importance (VLI) to very high importance (VHI). Using this rating, we performed an analysis of 10 combinations, where each combination was expressed as an experiment.

The criteria that were evaluated with the highest weight were the type of variables (C3), the quality of solutions (C12) and computation time (C13) (see Table 4.4). These parameters in a planning problem are dominant when choosing an algorithm. Therefore, we made a minimal variation and looked for criteria with lower scores, such as: problem type (C1), knowledge of algorithms (C10), dataset size (C7), programming knowledge (C8). The details of the experiments are shown in Table 4.9, where the second column details the changes in the weights of the criteria, and the third column shows the results of the proximity coefficient, while the last column expresses the alternatives ranking.

Table 4.9. Quantitative results of the sensitivity analysis.															
Experi- ment No.	Changes in weights of criteria	A1	A2	A3	A4	A5	A6	A7	A8	A9	A10	A11	A12	A13	Alternatives Ranking
E1	C1= (0.50, 0.75, 1.00)	0.267	0.337	0.402	0.409	0.457	0.338	0.337	0.302	0.321	0.433	0.366	0.274	0.279	A5 > A10 > A4 > A3 > A11 > A6 > A7 > A2 > A9 > A8 > A13 > A12 > A1
E2	C1= (0.75, 1.00, 1.00)	0.268	0.341	0.410	0.418	0.465	0.341	0.341	0.304	0.323	0.439	0.370	0.275	0.279	A5 > A10 > A4 > A3 > A11 > A6 > A2 > A7 > A9 > A8 > A13 > A12 > A1
E3	C3= (0.50, 0.75, 1.00)	0.260	0.322	0.383	0.395	0.438	0.327	0.328	0.295	0.314	0.417	0.355	0.268	0.273	A5 > A10 > A4 > A3 > A11 > A7 > A6 > A2 > A9 > A8 > A13 > A12 > A1
E4	C5= (0.25, 0.50, 0.75)	0.268	0.335	0.398	0.408	0.455	0.338	0.337	0.304	0.323	0.434	0.367	0.274	0.279	A5 > A10 > A4 > A3 > A11 > A6 > A7 > A2 > A9 > A8 > A13 > A12 > A1
E5	C10= (0.50, 0.75, 1.00)	0.266	0.336	0.396	0.406	0.455	0.335	0.336	0.301	0.323	0.434	0.366	0.278	0.282	A5 > A10 > A4 > A3 > A11 > A2 > A7 > A6 > A9 > A8 > A13 > A12 > A1
E6	C10= (0.75, 1.00, 1.00)	0.267	0.339	0.399	0.410	0.461	0.338	0.336	0.302	0.324	0.440	0.370	0.278	0.283	A5 > A10 > A4 > A3 > A11 > A2 > A6 > A7 > A9 > A8 > A13 > A12 > A1
E7	C8= (0.01, 0.25, 0.50), C12= (0.50, 0.75, 1.00)	0.254	0.317	0.376	0.370	0.422	0.319	0.319	0.287	0.306	0.404	0.346	0.262	0.267	A5 > A10 > A3 > A4 > A11 > A6 > A7 > A2 > A9 > A8 > A13 > A12 > A1
E8	C10= (0.75, 1.00, 1.00), C11= (0.50, 0.75, 1.00)	0.268	0.342	0.402	0.411	0.473	0.342	0.339	0.305	0.326	0.448	0.375	0.280	0.285	A5 > A10 > A4 > A3 > A11 > A6 > A2 > A7 > A9 > A8 > A13 > A12 > A1
E9	C7= (0.50, 0.75, 1.00)	0.262	0.327	0.388	0.396	0.443	0.329	0.329	0.296	0.315	0.422	0.358	0.270	0.275	A5 > A10 > A4 > A3 > A11 > A6 > A7 > A2 > A9 > A8 > A13 > A12 > A1
E10	C7= (0.75, 1.00, 1.00)	0.268	0.333	0.394	0.402	0.449	0.335	0.335	0.302	0.321	0.428	0.364	0.272	0.277	A5 > A10 > A4 > A3 > A11 > A6 > A7 > A2 > A9 > A8 > A13 > A12 > A1

Chapter 4. A decision-making tool for algorithm selection based on a fuzzy TOPSIS approach to solve replenishment, production and distribution planning problems.

152 |

The sensitivity analysis shows that alternatives A5 (GA), A10 (CPLEX), A4 (Tabu Search) have the best scores and occupy the first three positions. Hence, the variation in the weights in the chosen criteria minimally affects these alternatives; for example, A5 reaches the first position in all the experiments. The main variations occur in the sixth, seventh and eighth positions with alternatives A2, A6 and A7. However, the last ranking positions remain unchanged in the classification. In this context, decision makers can use these variations or make other modifications to weightings to prioritize a criterion and to, thus, facilitate the evaluation process in decision making.

4.6 Conclusions

The complexity of real-world problems should be seen as not only an obstacle, but also as a research challenge for effective solutions for large-scale planning problems. Relatively small companies often face very complex problems.

It is usually very difficult for production planners in companies to determine or choose an algorithm. The algorithm selection process normally involves the experimental evaluation of several algorithms with different dataset sizes. However, these sets of experiments require considerable computational resources and long processing times. This adds to the disadvantage of having fewer resources to invest in commercial solvers. In addition, efforts often have to be duplicated when attempting to replicate the algorithms or models available in the literature.

To overcome these drawbacks, the methodological approach based on the fuzzy TOPSIS proposed herein intends to be a support tool to select a solution method for replenishment, production and distribution planning problems. To this end, 13 different criteria were defined and used to select nine different algorithm types (heuristic, metaheuristic, and matheuristic) and four solvers (commercial and non-commercial) that are often employed in planning problems. All these criteria address several important dimensions when solving a planning problem. These dimensions are related to the computational difficulty of the planning problem, programming skills, mathematical skills, algorithmic skills, mathematical modeling software skills, and also to expected computational performance the solution methods. These criteria were analyzed based on the linguistic values given by four planning experts from different manufacturing companies. The problem selected to apply the proposed approach was that of production planning. For this problem, the results of the methodology showed that the GA was the best alternative, while Benders' decomposition was the worst.

Given our study results, it can be concluded that it is possible to select a set of suitable candidate algorithms for solving optimization problems with the proposed approach. In this way, not only can one algorithm be selected, but so can other algorithms that provide similar solutions at the same time. The results of this methodology can guide companies to choose whether to use a commercial or non-commercial algorithm or solver. This can help companies to determine whether they should invest in a solver or use mathematical modeling or algorithm programming software and, at the same time, to understand planning staff's training needs.

There are different approaches for algorithm selection [44, 70, 75, 79]. These approaches are heuristic, metaheuristic and AI, and they offer benefits and disadvantages. However, these techniques can be restrictive for companies because they involve a large number of computational resources and experiments that can be affected by accuracy, the number of tested instances, instance generation, consistency, and in AI techniques, training time. The proposed approach requires very few resources, is very useful thanks to its simplicity and is easily replicable. The main limitation of this technique is the appropriate selection of criteria and the balance between them, which is a subjective issue that requires experts in the planning problems field, not to mention the personal bias of experts' opinions.

Future research could be conducted to experiment the proposed approach with the portfolio of algorithms and solvers defined in [107], where some 50 algorithms are identified, including optimizing, heuristic, metaheuristic and matheuristic algorithms, as well as different types of commercial solvers. Alternatives and criteria could be evaluated with more decision makers. Other MCDM techniques such as ELECTRE, PROMETHEE, intuitionistic fuzzy TOPSIS, or novel methods such as the performance calculation technique of the integrated multiple multi-attribute decision making (PCIM-MADM) [120], which incorporates four techniques (COPRAS, GRA, SAW and VIKOR) into a single final classification index, could be used.

Appendix A.

Table A1 shows a section of the questionnaire format used by decision makers to evaluate the algorithm selection criteria. Table A2 presents the questionnaire used to score the chosen alternatives, i.e., the selected algorithms and solvers against the 13 identified criteria.

		c	riteria		
	Very Low Importance (VLI)	Low Importance (LI)	Medium Importance (MI)	High Importance (HI)	Very High Importance (VHI)
C1					
C2					
C3					
C4					
C5					
C6					
C7					
C8					
C9					
C10					
C11					
C12					
C13					

Table A1. Questionnaire used to know decision makers' preferences for the identified criteria

Table A2. Questionnaire used to know the decision makers' preferences for the 13
alternatives according to the criteria

			C1		
	Very Low (VL)	Low (L)	Moderate (M)	High (H)	Very High (VH)
A1					
A2					
A3					
A4					
A5					
A6					
A7					
A8					
A9					
A10					
A11					
A12					
A13					

Tables A3-A6 show the decision makers' alternatives ratings against all the criteria

	C1	C2	С3	C4	C5	C6	C7	C8	С9	C10	C11	C12	C13	
A1	L	Н	М	М	М	М	М	L	L	L	VL	М	Н	
A2	М	Н	Н	М	М	М	М	L	VL	L	L	М	Н	
A3	VH	Н	Н	М	М	М	М	L	М	L	L	М	Н	
A4	VH	Н	VL	М	М	М	М	VH	VL	Н	VL	VH	Н	
A5	VH	VH	Н	М	М	М	М	Н	L	Н	VH	Н	VH	
A6	М	М	L	М	М	М	М	L	М	L	L	М	L	
A7	М	М	L	М	М	М	М	L	М	VL	L	М	М	
A8	L	L	L	М	М	М	М	L	L	L	L	М	М	
A9	L	L	L	М	М	М	М	L	М	L	L	М	М	
A10	Н	VH	Н	М	Н	М	М	М	Н	Н	Н	VH	VH	
A11	М	М	М	М	М	М	М	L	М	М	М	М	VH	
A12	L	L	L	М	L	М	М	L	L	L	L	L	L	
A13	L	L	L	М	L	М	М	L	L	L	L	L	L	

Table A3. Decision maker 1's linguistic assessment.

Table A4. Decision maker 2's linguistic assessment.

	C1	C2	С3	C4	C5	C6	C7	C8	С9	C10	C11	C12	C13
A1	L	L	L	L	VL	L	Н	L	L	VL	VL	VL	М
A2	Н	Н	Н	L	М	L	Н	L	VL	М	L	VL	М
A3	VH	Н	Н	Н	М	М	Н	L	М	М	L	М	М
A4	VH	Н	VL	Н	М	М	Н	VH	VL	Н	VL	VH	Н
A5	VH	VH	Н	Н	М	М	Н	Н	L	Н	VH	Н	VH
A6	М	М	L	М	М	L	Н	L	М	L	L	М	L
A7	М	М	L	М	VL	L	Н	L	М	VL	L	М	М
A8	L	L	L	М	VL	L	Н	L	L	L	L	М	М
A9	L	L	L	М	VL	L	Н	L	М	L	L	М	М
A10	Н	VH	Н	М	Н	М	Н	М	Н	VH	Н	VH	VH
A11	М	М	М	М	М	L	Н	L	М	М	М	М	VH
A12	L	VL	L	VL	VL	L	L	L	L	L	L	L	L
A13	L	L	L	VL	VL	L	L	L	L	L	L	L	L

			Table	e A5. D	ecisio)	n mak	er 3's	lingui	stic as	sessme	nt.		
	C1	C2	С3	C4	C5	C6	C7	C8	С9	C10	C11	C12	C13
A1	L	VL	L	VL	VL	М	Н	L	L	L	VL	VL	L
A2	L	М	L	L	VL	М	Н	Н	Н	Н	L	VL	М
A3	Н	М	Μ	L	М	М	Н	VH	Н	Н	L	М	М
A4	Н	М	Μ	VL	VH	Н	Н	VH	Н	VL	VL	VH	Н
A5	Н	М	Μ	VH	Н	VH	Н	VH	VH	Н	VH	Н	VH
A6	М	М	L	L	М	L	Н	М	М	Н	Н	М	Н
A7	М	VL	L	L	М	М	Н	М	М	L	L	М	М
A8	М	VL	L	L	М	М	Н	L	L	L	L	М	М
A9	М	VL	L	L	М	М	Н	L	L	L	L	М	М
A10	М	Н	Μ	Н	М	М	Н	Н	М	М	М	М	М
A11	М	М	L	М	М	М	Н	М	М	М	М	М	М
A12	VL	VL	L	L	L	L	L	L	VL	L	L	L	L
A13	VL	VL	L	L	L	L	L	L	L	L	L	L	L

Models and algorithms for the optimisation of replenishment, production and distribution plans in industrial enterprises.

Table A6. Decision maker 4's linguistic assessment.

	C1	C2	С3	C4	C5	C6	C7	C8	С9	C10	C11	C12	C13
A1	L	L	L	VL	М	Н	Н	L	L	L	VL	М	L
A2	Н	Н	Н	VL	М	Н	Н	L	VL	L	L	М	L
A3	VH	Н	Н	М	М	Н	Н	L	М	L	L	М	М
A4	VH	Н	VL	VH	Н	Н	Н	VH	VL	Н	VL	VH	М
A5	VH	VH	Н	Н	VH	М	Н	Н	L	Н	VH	Н	Н
A6	М	М	Н	М	L	М	Н	L	М	L	L	М	Н
A7	М	М	L	М	М	М	Н	L	М	VL	L	М	L
A8	L	L	L	М	М	М	Н	L	L	L	L	М	L
A9	L	L	L	М	М	М	Н	L	М	L	L	М	L
A10	Н	М	М	М	М	М	Н	М	Н	Н	Н	Н	Н
A11	М	М	М	М	М	Н	Н	L	М	М	М	VH	VH
A12	L	VL	L	L	L	L	L	L	L	L	L	L	L
A13	L	L	L	L	L	L	L	L	L	L	L	L	L

Chapter 4. A decision-making tool for algorithm selection based on a fuzzy TOPSIS approach to solve replenishment, production and distribution planning problems.

_																							_																	
			C1			C2			С3			C4			C5			C6			C7			C8			С9			C10		C11				C12			C13	;
	-	Ι	m	u	I	m	u	I	m	u	I	m	u	I	m	u	I	m	u	I	m	u	I	m	u	T	m	u	I	m	u	I	m	u	I	m	u	I	m	u
	A1 (0.01	0.25	0.50	0.14	0.34	0.60	0.07	0.33	0.60	0.07	0.21	0.47	0.15	0.29	0.57	0.29	0.57	0.86	0.47	0.73	1.00	0.01	0.25	0.50	0.01	0.27	0.53	0.01	0.19	0.44	0.01	0.01	0.25	0.13	0.26	0.50	0.01	0.02	0.05
	A2 ().32	0.56	0.81	0.47	0.73	1.00	0.40	0.67	0.93	0.07	0.27	0.53	0.22	0.43	0.71	0.29	0.57	0.86	0.47	0.73	1.00	0.13	0.38	0.63	0.14	0.21	0.47	0.19	0.44	1.00	0.01	0.25	0.50	0.13	0.26	0.50	0.01	0.02	0.04
	A3 ().69	0.94	1.00	0.47	0.73	1.00	0.47	0.73	1.00	0.27	0.53	0.80	0.29	0.57	0.86	0.36	0.64	0.93	0.47	0.73	1.00	0.20	0.44	0.63	0.33	0.60	0.87	0.19	0.44	0.75	0.01	0.25	0.50	0.25	0.50	0.75	0.01	0.02	0.03
	A4 ().69	0.94	1.00	0.47	0.73	1.00	0.07	0.14	0.40	0.40	0.60	0.80	0.50	0.79	1.00	0.43	0.71	1.00	0.47	0.73	1.00	0.75	1.00	1.00	0.14	0.21	0.47	0.38	0.57	1.00	0.01	0.01	0.25	0.75	1.00	1.00	0.01	0.01	0.02
	A5 ().69	0.94	1.00	0.67	0.93	1.00	0.47	0.73	1.00	0.53	0.80	1.00	0.50	0.79	1.00	0.43	0.71	0.93	0.47	0.73	1.00	0.56	0.81	1.00	0.21	0.47	0.67	0.50	0.75	1.00	0.75	1.00	1.00	0.50	0.75	1.00	0.01	0.01	0.01
	A6 ().25	0.50	0.75	0.27	0.53	0.80	0.14	0.40	0.67	0.20	0.47	0.73	0.22	0.50	0.79	0.15	0.43	0.71	0.47	0.73	1.00	0.07	0.31	0.56	0.27	0.53	0.80	0.13	0.38	0.50	0.13	0.38	0.63	0.25	0.50	0.75	0.01	0.02	0.04
	A7 ().25	0.50	0.75	0.20	0.40	0.67	0.01	0.27	0.53	0.20	0.47	0.73	0.22	0.43	0.71	0.22	0.50	0.79	0.47	0.73	1.00	0.07	0.31	0.56	0.27	0.53	0.80	0.01	0.07	1.00	0.01	0.25	0.50	0.25	0.50	0.75	0.01	0.02	0.05
	A8 ().07	0.31	0.56	0.01	0.20	0.47	0.01	0.27	0.53	0.20	0.47	0.73	0.22	0.43	0.71	0.22	0.50	0.79	0.47	0.73	1.00	0.01	0.25	0.50	0.01	0.27	0.53	0.01	0.25	0.50	0.01	0.25	0.50	0.25	0.50	0.75	0.01	0.02	0.05
	A9 (0.07	0.31	0.56	0.01	0.20	0.47	0.01	0.27	0.53	0.20	0.47	0.73	0.22	0.43	0.71	0.22	0.50	0.79	0.47	0.73	1.00	0.01	0.25	0.50	0.20	0.47	0.73	0.01	0.25	1.00	0.01	0.25	0.50	0.25	0.50	0.75	0.01	0.02	0.05
A	10).44	0.69	0.94	0.60	0.87	1.00	0.40	0.67	0.93	0.33	0.60	0.87	0.43	0.71	1.00	0.29	0.57	0.86	0.47	0.73	1.00	0.31	0.56	0.81	0.47	0.73	1.00	0.50	0.75	1.00	0.44	0.69	0.94	0.56	0.81	0.94	0.01	0.01	0.02
A	11).25	0.50	0.75	0.27	0.53	0.80	0.20	0.47	0.73	0.27	0.53	0.80	0.29	0.57	0.86	0.29	0.57	0.86	0.47	0.73	1.00	0.07	0.31	0.56	0.27	0.53	0.80	0.25	0.50	0.75	0.25	0.50	0.75	0.38	0.63	0.81	0.01	0.01	0.02
A	12	0.01	0.19	0.44	0.01	0.07	0.33	0.01	0.27	0.53	0.07	0.27	0.53	0.01	0.22	0.50	0.08	0.36	0.64	0.07	0.33	0.60	0.01	0.25	0.50	0.01	0.20	0.47	0.01	0.25	1.00	0.01	0.25	0.50	0.01	0.25	0.50	0.02	0.04	1.00
A	13	0.01	0.19	0.44	0.01	0.20	0.47	0.01	0.27	0.53	0.07	0.27	0.53	0.01	0.22	0.50	0.08	0.36	0.64	0.07	0.33	0.60	0.01	0.25	0.50	0.01	0.27	0.53	0.01	0.25	1.00	0.01	0.25	0.50	0.01	0.25	0.50	0.02	0.04	1.00

Table AT. Normalized juzzy decision matrix.

158|

		C1			C2			С3			C4			C5			C6			C7			C8			С9			C10			C11			C12			C13	;
	I	m	u	I	m	u	I	m	u	I	m	u	I	m	u	I	m	u	I	m	u	I	m	u	I	m	u	I	m	u	I	m	u	I	m	u	I	m	u
A	0.00	0.13	0.38	3 0.00	0.08	0.30	0.06	0.33	0.60	0.04	0.15	0.47	0.00	0.07	0.29	0.09	0.32	0.70	0.26	0.60	0.94	0.00	0.14	0.41	0.00	0.15	0.43	0.00	0.10	0.33	0.00	0.00	0.13	0.10	0.26	0.50	0.01	0.02	0.05
A	2 0.08	80.28	0.61	0.00	0.18	0.50	0.30	0.67	0.93	0.04	0.20	0.53	0.00	0.11	0.36	0.09	0.32	0.70	0.26	0.60	0.94	0.04	0.21	0.51	0.04	0.12	0.38	0.05	0.22	0.75	0.00	0.06	0.25	0.10	0.26	0.50	0.01	0.02	0.04
A	3 0.17	0.47	0.75	5 0.00	0.18	0.50	0.35	0.73	1.00	0.13	0.40	0.80	0.00	0.14	0.43	0.11	0.36	0.75	0.26	0.60	0.94	0.06	0.25	0.51	0.11	0.34	0.70	0.05	0.22	0.56	0.00	0.06	0.25	0.19	0.50	0.75	0.01	0.02	0.03
A	4 0.17	0.47	0.75	0.00	0.18	0.50	0.06	0.14	0.40	0.20	0.45	0.80	0.01	0.20	0.50	0.13	0.40	0.81	0.26	0.60	0.94	0.24	0.56	0.81	0.04	0.12	0.38	0.10	0.28	0.75	0.00	0.00	0.13	0.56	1.00	1.00	0.01	0.01	0.02
A	5 0.17	0.47	0.75	0.01	0.23	0.50	0.35	0.73	1.00	0.27	0.60	1.00	0.01	0.20	0.50	0.13	0.40	0.75	0.26	0.60	0.94	0.18	0.46	0.81	0.07	0.26	0.54	0.13	0.38	0.75	0.01	0.25	0.50	0.38	0.75	1.00	0.01	0.01	0.01
A	5 0.06	0.25	0.56	6 0.00	0.13	0.40	0.11	0.40	0.67	0.10	0.35	0.73	0.00	0.13	0.39	0.05	0.24	0.58	0.26	0.60	0.94	0.02	0.18	0.46	0.08	0.30	0.65	0.03	0.19	0.38	0.00	0.09	0.31	0.19	0.50	0.75	0.01	0.02	0.04
A	7 0.06	0.25	0.56	6 0.00	0.10	0.33	0.01	0.27	0.53	0.10	0.35	0.73	0.00	0.11	0.36	0.07	0.28	0.64	0.26	0.60	0.94	0.02	0.18	0.46	0.08	0.30	0.65	0.00	0.04	0.75	0.00	0.06	0.25	0.19	0.50	0.75	0.01	0.02	0.05
A	3 0.02	0.16	0.42	0.00	0.05	0.23	0.01	0.27	0.53	0.10	0.35	0.73	0.00	0.11	0.36	0.07	0.28	0.64	0.26	0.60	0.94	0.00	0.14	0.41	0.00	0.15	0.43	0.00	0.13	0.38	0.00	0.06	0.25	0.19	0.50	0.75	0.01	0.02	0.05
A	9 0.02	0.16	0.42	2 0.00	0.05	0.23	0.01	0.27	0.53	0.10	0.35	0.73	0.00	0.11	0.36	0.07	0.28	0.64	0.26	0.60	0.94	0.00	0.14	0.41	0.06	0.26	0.60	0.00	0.13	0.75	0.00	0.06	0.25	0.19	0.50	0.75	0.01	0.02	0.05
A1	0 0.11	0.34	0.70	0.01	0.22	0.50	0.30	0.67	0.93	0.17	0.45	0.87	0.00	0.18	0.50	0.09	0.32	0.70	0.26	0.60	0.94	0.10	0.32	0.66	0.15	0.41	0.81	0.13	0.38	0.75	0.00	0.17	0.47	0.42	0.81	0.94	0.01	0.01	0.02
A1	1 0.06	0.25	0.56	6 0.00	0.13	0.40	0.15	0.47	0.73	0.13	0.40	0.80	0.00	0.14	0.43	0.09	0.32	0.70	0.26	0.60	0.94	0.02	0.18	0.46	0.08	0.30	0.65	0.06	0.25	0.56	0.00	0.13	0.38	0.28	0.63	0.81	0.01	0.01	0.02
A1	2 0.00	0.10	0.33	8 0.00	0.02	0.17	0.01	0.27	0.53	0.04	0.20	0.53	0.00	0.05	0.25	0.03	0.20	0.52	0.04	0.27	0.56	0.00	0.14	0.41	0.00	0.11	0.38	0.00	0.13	0.75	0.00	0.06	0.25	0.01	0.25	0.50	0.02	0.04	1.00
A1	3 0.00	0.10	0.33	8 0.00	0.05	0.23	0.01	0.27	0.53	0.04	0.20	0.53	0.00	0.05	0.25	0.03	0.20	0.52	0.04	0.27	0.56	0.00	0.14	0.41	0.00	0.15	0.43	0.00	0.13	0.75	0.00	0.06	0.25	0.01	0.25	0.50	0.02	0.04	1.00

Table A8. Weighted normalized fuzzy decision matrix.

159|

4.7 References

- C. Wang and X. B. Liu, "Integrated production planning and control: A multiobjective optimization model," *J. Ind. Eng. Manag.*, vol. 6, no. 4, pp. 815–830, 2013, doi: 10.3926/jiem.771.
- [2] Z. Wang *et al.*, "Multiobjective Optimization-Aided Decision-Making System for Large-Scale Manufacturing Planning," *IEEE Trans. Cybern.*, pp. 1–14, 2021, doi: 10.1109/TCYB.2021.3049712.
- [3] S. Hartmut, C. Kilger, and M. Herbert, Supply Chain Management and Advanced Planning: Concepts, Models, Software and Case Studies. Springer, Berlin, Heidelberg, 2015.
- [4] T. G. Crainic, F. Djeumou Fomeni, and W. Rei, "Multi-period bin packing model and effective constructive heuristics for corridor-based logistics capacity planning," *Comput. Oper. Res.*, vol. 132, no. November 2019, p. 105308, 2021, doi: 10.1016/j.cor.2021.105308.
- [5] S. Pratap, S. K. Jauhar, S. K. Paul, and F. Zhou, "Stochastic optimization approach for green routing and planning in perishable food production," *J. Clean. Prod.*, vol. 333, no. December 2021, p. 130063, 2022, doi: 10.1016/j.jclepro.2021.130063.
- [6] Y. Zarouk, I. Mahdavi, J. Rezaeian, and F. J. Santos-Arteaga, "A novel multiobjective green vehicle routing and scheduling model with stochastic demand, supply, and variable travel times," *Comput. Oper. Res.*, vol. 141, no. August 2020, p. 105698, 2022, doi: 10.1016/j.cor.2022.105698.
- [7] A. Rossi and M. Lanzetta, "Integration of hybrid additive/subtractive manufacturing planning and scheduling by metaheuristics," *Comput. Ind. Eng.*, vol. 144, no. April, p. 106428, 2020, doi: 10.1016/j.cie.2020.106428.
- [8] G. Mirabelli and V. Solina, "Optimization Strategies for the Integrated Management of Perishable Supply Chains: A Literature Review," J. Ind. Eng. Manag., vol. 15, no. 1, pp. 58–91, 2022, doi: 10.3926/jiem.3603.
- [9] M. Vanajakumari, H. Sun, A. Jones, and C. Sriskandarajah, "Supply chain planning: A case for Hybrid Cross-Docks," *Omega (United Kingdom)*, vol. 108, 2022, doi: 10.1016/j.omega.2021.102585.
- [10] A. A. Juan *et al.*, "A review of the role of heuristics in stochastic optimisation: from metaheuristics to learnheuristics," *Ann. Oper. Res.*, 2021, doi: 10.1007/s10479-021-04142-9.
- [11] H. Stadtler, "Supply chain management and advanced planning Basics, overview and challenges," *Eur. J. Oper. Res.*, vol. 163, no. 3, pp. 575–588, 2005, doi: 10.1016/j.ejor.2004.03.001.

- [12] T. Weise, M. Zapf, R. Chiong, and A. J. Nebro, "Why Is Optimization Difficult?," in *Nature-Inspired Algorithms for Optimisation*, R. Chiong, Ed. Berlin, Heidelberg: Springer Berlin Heidelberg, 2009, pp. 1–50.
- [13] Z. Michalewicz and D. B. Foge, *How to Solve It: Modern Heuristics*, 2nd ed. Springer, Berlin, Heidelberg, 2004.
- [14] J. Lohmer and R. Lasch, "Production planning and scheduling in multi-factory production networks: a systematic literature review," *Int. J. Prod. Res.*, vol. 59, no. 7, pp. 2028–2054, 2021, doi: 10.1080/00207543.2020.1797207.
- [15] Y. Adulyasak, J. F. Cordeau, and R. Jans, "The production routing problem: A review of formulations and solution algorithms," *Comput. Oper. Res.*, vol. 55, pp. 141–152, 2015, doi: 10.1016/j.cor.2014.01.011.
- [16] M. Díaz-Madroñero, J. Mula, and D. Peidro, "A review of discrete-time optimization models for tactical production planning," *Int. J. Prod. Res.*, vol. 52, no. 17, pp. 5171–5205, 2014, doi: 10.1080/00207543.2014.899721.
- [17] J. Mula, D. Peidro, M. Díaz-Madroñero, and E. Vicens, "Mathematical programming models for supply chain production and transport planning," *Eur. J. Oper. Res.*, vol. 204, no. 3, pp. 377–390, 2010, doi: 10.1016/j.ejor.2009.09.008.
- [18] F. Peres and M. Castelli, "Combinatorial Optimization Problems and Metaheuristics: Review, Challenges, Design, and Development," *Appl. Sci.*, 2021, doi: https://doi.org/10.3390/app11146449.
- [19] I. I. Huerta, D. A. Neira, D. A. Ortega, V. Varas, J. Godoy, and R. Asín-Achá, "Anytime automatic algorithm selection for knapsack," *Expert Syst. Appl.*, vol. 158, 2020, doi: 10.1016/j.eswa.2020.113613.
- [20] B. T. Tezel and A. Mert, "A cooperative system for metaheuristic algorithms," *Expert Syst. Appl.*, vol. 165, no. September 2020, p. 113976, 2021, doi: 10.1016/j.eswa.2020.113976.
- [21] V. R. de Carvalho, E. Özcan, and J. S. Sichman, "Comparative Analysis of Selection Hyper-Heuristics for Real-World Multi-Objective Optimization Problems," *Appl. Sci.*, vol. 11, no. 19, 2021, doi: 10.3390/app11199153.
- [22] Y. Peng, G. Kou, G. Wang, and Y. Shi, "FAMCDM: A fusion approach of MCDM methods to rank multiclass classification algorithms," *Omega*, vol. 39, no. 6, pp. 677–689, 2011, doi: 10.1016/j.omega.2011.01.009.
- [23] K. Grąbczewski, "Using Result Profiles to Drive Meta-learning," pp. 69–83, 2022, doi: 10.1007/978-3-030-95947-0_6.
- [24] K. Smith-Miles and L. Lopes, "Measuring instance difficulty for combinatorial optimization problems," *Comput. Oper. Res.*, vol. 39, no. 5, pp. 875–889, 2012, doi: 10.1016/j.cor.2011.07.006.

- [25] A. Jamalnia, J.-B. Yang, A. Feili, D.-L. Xu, and G. Jamali, "Aggregate production planning under uncertainty: a comprehensive literature survey and future research directions," *Int. J. Adv. Manuf. Technol.*, vol. 102, no. 1–4, pp. 159–181, 2019, doi: 10.1007/s00170-018-3151-y.
- [26] R. Kumar, L. Ganapathy, R. Gokhale, and M. K. Tiwari, "Quantitative approaches for the integration of production and distribution planning in the supply chain: a systematic literature review," *Int. J. Prod. Res.*, vol. 58, no. 11, pp. 3527–3553, 2020, doi: 10.1080/00207543.2020.1762019.
- [27] D. F. Pereira, J. F. Oliveira, and M. A. Carravilla, "Tactical sales and operations planning: A holistic framework and a literature review of decision-making models," *Int. J. Prod. Econ.*, vol. 228, no. July 2019, p. 107695, 2020, doi: 10.1016/j.ijpe.2020.107695.
- [28] K. Hussain, M. N. Mohd Salleh, S. Cheng, and Y. Shi, "Metaheuristic research: a comprehensive survey," *Artif. Intell. Rev.*, vol. 52, no. 4, pp. 2191–2233, 2019, doi: 10.1007/s10462-017-9605-z.
- [29] J. N. Hooker, "Testing heuristics: We have it all wrong," *J. Heuristics*, vol. 1, no. 1, pp. 33–42, 1995, doi: 10.1007/BF02430364.
- [30] E. Angel and V. Zissimopoulos, "On the hardness of the quadratic assignment problem with metaheuristics," *J. Heuristics*, vol. 8, no. 4, pp. 399–414, 2002, doi: 10.1023/A:1015454612213.
- [31] D. Il Seo and B. R. Moon, "An information-theoretic analysis on the interactions of variables in combinatorial optimization problems," *Evol. Comput.*, vol. 15, no. 2, pp. 169–198, 2007, doi: 10.1162/evco.2007.15.2.169.
- [32] S. Bansal, Performance comparison of five metaheuristic nature-inspired algorithms to find near-OGRs for WDM systems, vol. 53, no. 8. Springer Netherlands, 2020.
- [33] J. Silberholz, B. Golden, S. Gupta, and X. Wang, "Computational Comparison of Metaheuristics," in *Handbook of Metaheuristics*, M. Gendreau and J.-Y. Potvin, Eds. Cham: Springer International Publishing, 2019, pp. 581–604.
- [34] J. Beasley, "OR-Library." http://people.brunel.ac.uk/~mastjjb/jeb/info.html (accessed Mar. 05, 2022).
- [35] B. Bischl *et al.*, "ASlib: A benchmark library for algorithm selection," *Artif. Intell.*, vol. 237, pp. 41–58, 2016, doi: 10.1016/j.artint.2016.04.003.
- [36] Q. K. Pan, L. Gao, L. Wang, J. Liang, and X. Y. Li, "Effective heuristics and metaheuristics to minimize total flowtime for the distributed permutation flowshop problem," *Expert Syst. Appl.*, vol. 124, pp. 309–324, 2019, doi: 10.1016/j.eswa.2019.01.062.
- [37] D. S. Johnson, "Approximation algorithms for combinatorial problems," J.

Comput. Syst. Sci., vol. 9, no. 3, pp. 256–278, 1974, doi: 10.1016/S0022-0000(74)80044-9.

- [38] V. Maniezzo, M. A. Boschetti, and T. Stützle, *Matheuristics*. Springer, Cham, 2014.
- [39] J. de Armas, E. Lalla-Ruiz, S. L. Tilahun, and S. Voß, "Similarity in metaheuristics: a gentle step towards a comparison methodology," *Nat. Comput.*, vol. 9, no. 2013, 2021, doi: 10.1007/s11047-020-09837-9.
- [40] M. A. Rahman, R. Sokkalingam, M. Othman, K. Biswas, L. Abdullah, and E. A. Kadir, "Nature-inspired metaheuristic techniques for combinatorial optimization problems: Overview and recent advances," *Mathematics*, vol. 9, no. 20, pp. 1–32, 2021, doi: 10.3390/math9202633.
- [41] Y. Douek-Pinkovich, I. Ben-Gal, and T. Raviv, "The stochastic test collection problem: Models, exact and heuristic solution approaches," *Eur. J. Oper. Res.*, vol. 299, no. 3, pp. 945–959, 2022, doi: 10.1016/j.ejor.2021.12.043.
- [42] A. F. Silva, J. M. S. Valente, and J. E. Schaller, "Metaheuristics for the permutation flowshop problem with a weighted quadratic tardiness objective," *Comput. Oper. Res.*, vol. 140, no. December 2021, p. 105691, 2022, doi: 10.1016/j.cor.2021.105691.
- [43] İ. Tarhan and C. Oğuz, "A matheuristic for the generalized order acceptance and scheduling problem," *Eur. J. Oper. Res.*, vol. 299, no. 1, pp. 87–103, 2022, doi: 10.1016/j.ejor.2021.08.024.
- [44] P. Kerschke, H. H. Hoos, F. Neumann, and H. Trautmann, "Automated Algorithm Selection: Survey and Perspectives," *Evol. Comput.*, vol. 27, no. 1, pp. 3–45, 2019, doi: 10.1162/evco_a_00242.
- [45] D. H. Wolpert and W. G. Macready, "No free lunch theorems for optimization," *IEEE Trans. Evol. Comput.*, vol. 1, no. 1, pp. 67–82, 1997, doi: 10.1109/4235.585893.
- [46] W. Mostert, K. M. Malan, and A. P. Engelbrecht, "A Feature Selection Algorithm Performance Metric for Comparative Analysis," *Algorithms*, vol. 14, no. 3, 2021, doi: 10.3390/a14030100.
- [47] M. M. Salih, B. B. Zaidan, A. A. Zaidan, and M. A. Ahmed, "Survey on fuzzy TOPSIS state-of-the-art between 2007 and 2017," *Comput. Oper. Res.*, vol. 104, pp. 207– 227, 2019, doi: 10.1016/j.cor.2018.12.019.
- [48] S. H. Zanakis, A. Solomon, N. Wishart, and S. Dublish, "Multi-attribute decision making: A simulation comparison of select methods," *Eur. J. Oper. Res.*, vol. 107, no. 3, pp. 507–529, 1998, doi: 10.1016/S0377-2217(97)00147-1.
- [49] S. Opricovic and G. H. Tzeng, "Compromise solution by MCDM methods: A comparative analysis of VIKOR and TOPSIS," *Eur. J. Oper. Res.*, vol. 156, no. 2, pp.

445-455, 2004, doi: 10.1016/S0377-2217(03)00020-1.

- [50] S. Opricovic and G. H. Tzeng, "Extended VIKOR method in comparison with outranking methods," *Eur. J. Oper. Res.*, vol. 178, no. 2, pp. 514–529, 2007, doi: 10.1016/j.ejor.2006.01.020.
- [51] M. T. Chu, J. Shyu, G. H. Tzeng, and R. Khosla, "Comparison among three analytical methods for knowledge communities group-decision analysis," *Expert Syst. Appl.*, vol. 33, no. 4, pp. 1011–1024, 2007, doi: 10.1016/j.eswa.2006.08.026.
- [52] T. Özcan, N. Elebi, and A. Esnaf, "Comparative analysis of multi-criteria decision making methodologies and implementation of a warehouse location selection problem," *Expert Syst. Appl.*, vol. 38, no. 8, pp. 9773–9779, 2011, doi: 10.1016/j.eswa.2011.02.022.
- [53] I. Ertuğrul and N. Karakaşoğlu, "Comparison of fuzzy AHP and fuzzy TOPSIS methods for facility location selection," *Int. J. Adv. Manuf. Technol.*, vol. 39, no. 7–8, pp. 783–795, 2008, doi: 10.1007/s00170-007-1249-8.
- [54] K. T. Atanassov, "Intuitionistic fuzzy sets," *Fuzzy Sets Syst.*, vol. 20, no. 1, pp. 87–96, 1986, doi: https://doi.org/10.1016/S0165-0114(86)80034-3.
- [55] F. E. Boran, "An Integrated Intuitionistic Fuzzy Multi Criteria Decision Making Method for Facility Location Selection," *Math. Comput. Appl.*, vol. 16, no. 2, pp. 487–496, 2011, doi: 10.3390/mca16020487.
- [56] F. E. Boran, S. Genç, M. Kurt, and D. Akay, "A multi-criteria intuitionistic fuzzy group decision making for supplier selection with TOPSIS method," *Expert Syst. Appl.*, vol. 36, no. 8, pp. 11363–11368, 2009, doi: 10.1016/j.eswa.2009.03.039.
- [57] V. C. Gerogiannis, P. Fitsilis, and A. D. Kameas, "Using a Combined Intuitionistic Fuzzy Set-TOPSIS Method for Evaluating Project and Portfolio Management Information Systems," in *Artificial Intelligence Applications and Innovations*, 2011, pp. 67–81.
- [58] V. C. Gerogiannis, P. Fitsilis, and A. D. Kameas, "Evaluation of project and portfolio Management Information Systems with the use of a hybrid IFS-TOPSIS method," *Intell. Decis. Technol.*, vol. 7, pp. 91–105, 2013, doi: 10.3233/IDT-120153.
- [59] L. Shan, "Research on vendor selection based on intuitionistic fuzzy sets," Adv. Intell. Soft Comput., vol. 110, no. 5, pp. 645–652, 2011, doi: 10.1007/978-3-642-25185-6_82.
- [60] G. Büyüközkan and S. Güleryüz, "Multi Criteria Group Decision Making Approach for Smart Phone Selection Using Intuitionistic Fuzzy TOPSIS," Int. J. Comput. Intell. Syst., vol. 9, no. 4, pp. 709–725, 2016, doi: 10.1080/18756891.2016.1204119.

- [61] D. Jato-Espino, E. Castillo-Lopez, J. Rodriguez-Hernandez, and J. C. Canteras-Jordana, "A review of application of multi-criteria decision making methods in construction," *Autom. Constr.*, vol. 45, pp. 151–162, 2014, doi: 10.1016/j.autcon.2014.05.013.
- [62] M. Velasquez and P. Hester, "An analysis of multi-criteria decision making methods," *Int. J. Oper. Res.*, vol. 10, no. 2, pp. 56–66, 2013.
- [63] M. Lamba, G. Munjal, and Y. Gigras, "ECABC: Evaluation of Classification Algorithms in Breast Cancer for Imbalanced Datasets," in *Data Driven Approach Towards Disruptive Technologies*, 2021, pp. 379–388.
- [64] Y. Peng, G. Wang, G. Kou, and Y. Shi, "An empirical study of classification algorithm evaluation for financial risk prediction," *Appl. Soft Comput. J.*, vol. 11, no. 2, pp. 2906–2915, 2011, doi: 10.1016/j.asoc.2010.11.028.
- [65] M. Behzadian, S. Khanmohammadi Otaghsara, M. Yazdani, and J. Ignatius, "A state-of the-art survey of TOPSIS applications," *Expert Syst. Appl.*, vol. 39, no. 17, pp. 13051–13069, 2012, doi: 10.1016/j.eswa.2012.05.056.
- [66] N. Jigeesh, D. Joseph, and S. K. Yadav, "A review on industrial applications of TOPSIS approach," *Int. J. Serv. Oper. Manag.*, vol. 30, no. 1, p. 23, 2018, doi: 10.1504/ijsom.2018.10012402.
- [67] K. Palczewski and W. Sałabun, "The fuzzy TOPSIS applications in the last decade," *Procedia Comput. Sci.*, vol. 159, pp. 2294–2303, 2019, doi: 10.1016/j.procs.2019.09.404.
- [68] D. Choudhary and R. Shankar, "An STEEP-fuzzy AHP-TOPSIS framework for evaluation and selection of thermal power plant location: A case study from India," *Energy*, vol. 42, no. 1, pp. 510–521, 2012, doi: 10.1016/j.energy.2012.03.010.
- [69] S. Saleem and M. Gallagher, "Using regression models for characterizing and comparing black box optimization problems," *Swarm Evol. Comput.*, vol. 68, no. June 2021, p. 100981, 2022, doi: 10.1016/j.swevo.2021.100981.
- [70] M. A. Muñoz, Y. Sun, M. Kirley, and S. K. Halgamuge, "Algorithm selection for black-box continuous optimization problems: A survey on methods and challenges," *Inf. Sci. (Ny).*, vol. 317, pp. 224–245, 2015, doi: 10.1016/j.ins.2015.05.010.
- [71] S. Raschka, "Model Evaluation, Model Selection, and Algorithm Selection in Machine Learning," *CoRR*, vol. abs/1811.1, 2018, [Online]. Available: http://arxiv.org/abs/1811.12808.
- [72] A. R. S. Parmezan, H. D. Lee, N. Spolaôr, and F. C. Wu, "Automatic recommendation of feature selection algorithms based on dataset characteristics," *Expert Syst. Appl.*, vol. 185, no. June, p. 115589, 2021, doi:

10.1016/j.eswa.2021.115589.

- [73] T. Stützle and S. Fernandes, "New Benchmark Instances for the QAP and the Experimental Analysis of Algorithms," in *Evolutionary Computation in Combinatorial Optimization*, 2004, pp. 199–209.
- [74] C. P. Gomes and B. Selman, "Algorithm Portfolio Design: Theory vs. Practice," CoRR, vol. abs/1302.1, 2013, [Online]. Available: http://arxiv.org/abs/1302.1541.
- [75] L. Kotthoff, "Algorithm Selection for Combinatorial Search Problems: A Survey," in Data Mining and Constraint Programming: Foundations of a Cross-Disciplinary Approach, C. Bessiere, L. De Raedt, L. Kotthoff, S. Nijssen, B. O'Sullivan, and D. Pedreschi, Eds. Cham: Springer International Publishing, 2016, pp. 149–190.
- [76] K. Leyton-Brown, E. Nudelman, G. Andrew, J. McFadden, and Y. Shoham, "A portfolio approach to algorithm select," *IJCAI Int. Jt. Conf. Artif. Intell.*, no. November, pp. 1542–1543, 2003.
- [77] J. R. Rice, "The Algorithm Selection Problem," Comput. Sci. Tech. Reports. Pap., vol. 99, pp. 75–152, 1975, [Online]. Available: http://docs.lib.purdue.edu/cstech/99.
- [78] S. Strassl and N. Musliu, "Instance space analysis and algorithm selection for the job shop scheduling problem," *Comput. Oper. Res.*, vol. 141, no. April 2021, p. 105661, 2022, doi: 10.1016/j.cor.2021.105661.
- [79] M. G. Lagoudakis and M. L. Littman, "Algorithm Selection using Reinforcement Learning," *Proc. Seventeenth Int. Conf. Mach. Learn.*, pp. 511–518, 2000.
- [80] L. Xu, F. Hutter, H. H. Hoos, and K. Leyton-Brown, "SATzilla: Portfolio-based algorithm selection for SAT," J. Artif. Intell. Res., vol. 32, pp. 565–606, 2008, doi: 10.1613/jair.2490.
- [81] K. A. Smith-Miles, "Cross-disciplinary perspectives on meta-learning for algorithm selection," ACM Comput. Surv., vol. 41, no. 1, pp. 1–25, 2008, doi: 10.1145/1456650.1456656.
- [82] H. Hoos, M. Lindauer, and T. Schaub, "Claspfolio 2: Advances in algorithm selection for answer set programming," *Theory Pract. Log. Program.*, vol. 14, no. 4–5, pp. 569–585, 2014, doi: 10.1017/S1471068414000210.
- [83] K. Tierney and Y. Malitsky, "An Algorithm Selection Benchmark of the Container Pre-marshalling Problem," in *Learning and Intelligent Optimization*, 2015, pp. 17–22.
- [84] T. Cunha, C. Soares, and A. C. P. L. F. de Carvalho, "Metalearning and Recommender Systems: A literature review and empirical study on the algorithm selection problem for Collaborative Filtering," *Inf. Sci. (Ny).*, vol. 423, pp. 128–144, 2018, doi: 10.1016/j.ins.2017.09.050.

- [85] W. Bożejko, A. Gnatowski, T. Niżyński, M. Affenzeller, and A. Beham, "Local optima networks in solving algorithm selection problem for TSP," *Adv. Intell. Syst. Comput.*, vol. 761, pp. 83–93, 2019, doi: 10.1007/978-3-319-91446-6_9.
- [86] G. Drozdov, A. Zabashta, and A. Filchenkov, "Graph Convolutional Network Based Generative Adversarial Networks for the Algorithm Selection Problem in Classification," *PervasiveHealth Pervasive Comput. Technol. Healthc.*, pp. 88–92, 2020, doi: 10.1145/3437802.3437818.
- [87] M. G. Vilas Boas, H. G. Santos, L. H. de C. Merschmann, and G. Vanden Berghe, "Optimal decision trees for the algorithm selection problem: integer programming based approaches," *Int. Trans. Oper. Res.*, vol. 28, no. 5, pp. 2759– 2781, 2021, doi: 10.1111/itor.12724.
- [88] A. Marrero, E. Segredo, and C. Leon, "A Parallel Genetic Algorithm to Speed up the Resolution of the Algorithm Selection Problem," in *Proceedings of the Genetic and Evolutionary Computation Conference Companion*, New York, NY, USA: Association for Computing Machinery, 2021, pp. 1978–1981.
- [89] D. Müller, M. G. Müller, D. Kress, and E. Pesch, "An algorithm selection approach for the flexible job shop scheduling problem: Choosing constraint programming solvers through machine learning," *Eur. J. Oper. Res.*, 2022, doi: 10.1016/j.ejor.2022.01.034.
- [90] M. Karimi-Mamaghan, M. Mohammadi, P. Meyer, A. M. Karimi-Mamaghan, and E. G. Talbi, "Machine learning at the service of meta-heuristics for solving combinatorial optimization problems: A state-of-the-art," *Eur. J. Oper. Res.*, vol. 296, no. 2, pp. 393–422, 2022, doi: 10.1016/j.ejor.2021.04.032.
- [91] N. J. Radcliffe, "The algebra of genetic algorithms," *Ann. Math. Artif. Intell.*, vol. 10, no. 4, pp. 339–384, 1994, doi: 10.1007/BF01531276.
- [92] F. Hutter, L. Xu, H. Hoos, and K. Leyton-Brown, "Algorithm runtime prediction: methods and evaluation," *Artif. Intell.*, vol. 206, 2014, doi: 10.1016/j.artint.2013.10.003.
- [93] I. Ozsahin, D. Uzun Ozsahin, B. Uzun, and M. T. Mustapha, "Chapter 1 -Introduction," in Applications of Multi-Criteria Decision-Making Theories in Healthcare and Biomedical Engineering, I. Ozsahin, D. U. Ozsahin, and B. Uzun, Eds. Academic Press, 2021, pp. 1–2.
- [94] L. A. Zadeh, "Fuzzy sets," Inf. Control, vol. 8, no. 3, pp. 338–353, 1965, doi: https://doi.org/10.1016/S0019-9958(65)90241-X.
- [95] C.-L. Hwang and K. Yoon, "Methods for multiple attribute decision making," in *Multiple attribute decision making*, Springer, 1981, pp. 58–191.
- [96] D. Kannan, A. B. L. De Sousa Jabbour, and C. J. C. Jabbour, "Selecting green suppliers based on GSCM practices: Using Fuzzy TOPSIS applied to a Brazilian

electronics company," *Eur. J. Oper. Res.*, vol. 233, no. 2, pp. 432–447, 2014, doi: 10.1016/j.ejor.2013.07.023.

- [97] C.-T. Chen, "Extensions of the TOPSIS for group decision-making under fuzzy environment," *Fuzzy Sets Syst.*, vol. 114, no. 1, pp. 1–9, 2000, doi: https://doi.org/10.1016/S0165-0114(97)00377-1.
- [98] E. Bellman and L. A. Zadeh, "A fuzzy environment," *Manage. Sci.*, vol. 17, no. 4, pp. 141–164, 1970.
- [99] D. Dubois and H. Prade, "Fuzzy sets and systems: theory and applications," 1980.
- [100] C. Chakraborty and D. Chakraborty, "A theoretical development on a fuzzy distance measure for fuzzy numbers," *Math. Comput. Model.*, vol. 43, no. 3–4, pp. 254–261, 2006, doi: 10.1016/j.mcm.2005.09.025.
- [101] N. Ploskas and J. Papathanasiou, "A decision support system for multiple criteria alternative ranking using TOPSIS and VIKOR in fuzzy and nonfuzzy environments," *Fuzzy Sets Syst.*, vol. 377, pp. 1–30, 2019, doi: 10.1016/j.fss.2019.01.012.
- [102] L. Shen, L. Olfat, K. Govindan, R. Khodaverdi, and A. Diabat, "A fuzzy multi criteria approach for evaluating green supplier's performance in green supply chain with linguistic preferences," *Resour. Conserv. Recycl.*, vol. 74, pp. 170–179, 2013, doi: 10.1016/j.resconrec.2012.09.006.
- [103] T. C. Wang and T. H. Chang, "Application of TOPSIS in evaluating initial training aircraft under a fuzzy environment," *Expert Syst. Appl.*, vol. 33, no. 4, pp. 870– 880, 2007, doi: 10.1016/j.eswa.2006.07.003.
- [104] E. Afful-Dadzie, S. Nabareseh, A. Afful-Dadzie, and Z. K. Oplatková, "A fuzzy TOPSIS framework for selecting fragile states for support facility," *Qual. Quant.*, vol. 49, no. 5, pp. 1835–1855, 2015, doi: 10.1007/s11135-014-0062-3.
- [105] S. Piya, A. Shamsuzzoha, and M. Khadem, "An approach for analysing supply chain complexity drivers through interpretive structural modelling," *Int. J. Logist. Res. Appl.*, vol. 23, no. 4, pp. 311–336, 2020, doi: 10.1080/13675567.2019.1691514.
- [106] E. Guzmán, R. Poler, and B. Andres, "Un análisis de revisiones de modelos y algoritmos para la optimización de planes de aprovisionamiento, producción y distribución de la cadena de suministro," *Dir. y Organ.*, vol. 70, pp. 28–52, 2020, doi: https://doi.org/10.37610/dyo.v0i70.567.
- [107] E. Guzman, B. Andres, and R. Poler, "Models and algorithms for production planning, scheduling and sequencing problems: a holistic framework and a systematic review," *J. Ind. Inf. Integr.*, p. 100287, 2021, doi: 10.1016/j.jii.2021.100287.
- [108] G. Stewart, "Supply-chain operations reference model (SCOR): the first cross-

industry framework for integrated supply-chain management," *Logist. Inf. Manag.*, vol. 10, no. 2, pp. 62–67, 1997, doi: 10.1108/09576059710815716.

- [109] Z. Michalewicz and D. B. Fogel, How to solve it: modern heuristics. Springer Science \& Business Media, 2013.
- [110] A. S. Tasan and M. Gen, "A genetic algorithm based approach to vehicle routing problem with simultaneous pick-up and deliveries," *Comput. Ind. Eng.*, vol. 62, no. 3, pp. 755–761, 2012, doi: 10.1016/j.cie.2011.11.025.
- [111] W. Y. Ku and J. C. Beck, "Mixed Integer Programming models for job shop scheduling: A computational analysis," *Comput. Oper. Res.*, vol. 73, pp. 165–173, 2016, doi: 10.1016/j.cor.2016.04.006.
- [112] B. Fahimnia, R. Z. Farahani, R. Marian, and L. Luong, "A review and critique on integrated production-distribution planning models and techniques," *J. Manuf. Syst.*, vol. 32, no. 1, pp. 1–19, 2013, doi: 10.1016/j.jmsy.2012.07.005.
- [113] M. Gavrilas, "Heuristic and metaheuristic optimization techniques with application to power systems," *Int. Conf. Math. Methods Comput. Tech. Electr. Eng. - Proc.*, pp. 95–103, 2010.
- [114] J. Swan et al., "Metaheuristics 'In the Large," Eur. J. Oper. Res., vol. 297, pp. 393– 406, 2021, doi: 10.1016/j.ejor.2021.05.042.
- [115] M. A. Boschetti, V. Maniezzo, M. Roffilli, and A. Bolufé Röhler, "Matheuristics: Optimization, simulation and control," *Lect. Notes Comput. Sci. (including Subser. Lect. Notes Artif. Intell. Lect. Notes Bioinformatics)*, vol. 5818 LNCS, pp. 171–177, 2009, doi: 10.1007/978-3-642-04918-7_13.
- [116] C. C. Sun, "A performance evaluation model by integrating fuzzy AHP and fuzzy TOPSIS methods," *Expert Syst. Appl.*, vol. 37, no. 12, pp. 7745–7754, 2010, doi: 10.1016/j.eswa.2010.04.066.
- [117] L. A. Zadeh, "The concept of a linguistic variable and its application to approximate reasoning—I," *Inf. Sci. (Ny).*, vol. 8, no. 3, pp. 199–249, 1975, doi: https://doi.org/10.1016/0020-0255(75)90036-5.
- [118] S. NÅdÅban, S. Dzitac, and I. Dzitac, "Fuzzy TOPSIS: A General View," *Procedia Comput. Sci.*, vol. 91, no. ltqm, pp. 823–831, 2016, doi: 10.1016/j.procs.2016.07.088.
- [119] G. Torlak, M. Sevkli, M. Sanal, and S. Zaim, "Analyzing business competition by using fuzzy TOPSIS method: An example of Turkish domestic airline industry," *Expert Syst. Appl.*, vol. 38, no. 4, pp. 3396–3406, 2011, doi: 10.1016/j.eswa.2010.08.125.
- [120] H. W. Lo, C. F. Liaw, M. Gul, and K. Y. Lin, "Sustainable supplier evaluation and transportation planning in multi-level supply chain networks using multiattribute- and multi-objective decision making," *Comput. Ind. Eng.*, vol. 162, no.

2020, p. 107756, 2021, doi: 10.1016/j.cie.2021.107756.

Chapter 5

An MILP model for the lotsizing/scheduling of automotive plastic components with raw materials and packaging availability.

Eduardo Guzman, Beatriz Andres, and Raul Poler. 2022. "An MILP model for the lot-sizing/scheduling of automotive plastic components with raw materials and packaging availability." Accepted for publication in Lecture Notes in Management and Industrial Engineering

Abstract:

This paper examines the lot-sizing/scheduling problem for plastic automotive components manufacturing. The scenario in which the problem is tackled refers to a second-tier supplier in the automotive supply chain. Here the studied second-tier supplier is characterized by transforming plastic granules in injection machines using specific moulds that produce components or finished products. Each mould can be set up on distinct machines to inject one same automobile component, or even two different components or more in the same mould. The same mould is assembled on different injection machines and can have distinct production rates subject to the machine on which it is set up. Our research work puts forward a mixed integer linear programming (MILP) model to minimize setup, the inventory of raw materials and plastic components, stockout, backorder costs and machine-mould assignation costs. We demonstrate the usability of this model with randomly generated instances. The results of the experiments show that our MILP converges toward optimal solutions in large instances by reaching efficient solutions in reference to both quality and execution times. The novelty of this model lies in it considering the arrival of materials as raw material for the injection of parts into moulds, the use of raw materials and the availability of containers for packaging finished products. Moulds can also be set up only during specific time periods in accordance with the quantity of available labour during each time period.

Chapter 5. An MILP model for the lot-sizing/scheduling of automotive plastic components with raw materials and packaging availability.

5.1 Introduction

Production and scheduling planning are central functions in manufacturing industries whose relevance is increasingly important due to the complexity of the operations required to manufacture final products from raw materials and the growing attention paid to supply chain management. The production planning and scheduling problem represent an important area of production planning and operations research [1]. Production decisions for a manufacturing environment are concerned about establishing the most efficient utilization of available resources to produce items, while also meeting customer requirements. The lotsizing/scheduling problem frequently appears in manufacturing systems with complex configurations and finite capacities. In both practice and theory, lotsizing/scheduling decisions are often made in parallel at the production planning and scheduling levels. The objective at the planning level is to draw a production plan, i.e., determining the production quantities (corresponding to the batch sizes processed in workshops) for each horizon period to meet demands and to minimize different costs (production, maintenance and setup costs). These batches are sequenced in production assets at the scheduling level [2].

A substantial number of papers have dealt with lot-sizing/scheduling, the majority of which are mathematical models for this problem, where the objective function seeks to minimize production costs. Our study centres on modelling a real industrial case to solve the lot-sizing/scheduling problem that is subject to internal/external materials requirement planning (MRP) restrictions. The problem is linked with an automotive plastic component producer that acts as a second-tier supplier in the automotive supply chain. The herein studied second-tier supplier is characterized for its specific moulds for producing components or finished products. This problem is particularly characteristic of the automotive industry because:

- the aim of having to produce the plastic components of a specific car model is to supply them during most of the model's lifetime, e.g., 5 years;
- the increasing costs of plastic raw materials caused by the pandemic crisis have led second-tier suppliers to purchase larger amounts of plastic pellets (raw materials), which always entails contemplating warehouse space limitations and discount prices;
- specific reusable containers are purchased by the first-tier supplier to receive the second-tier supplier's components. As reusable containers are expensive, there is only a limited number of them. The number of reusable

containers can be slightly adjusted to the agreed demand for the supply period, which normally coincides with a car model's lifetime. When reusable containers are not available for second-tier suppliers, injected parts have to be stored in cardboard containers until reusable containers arrive. Then the components stored in cardboard containers must be moved to the reusable ones, which incurs an extra handling cost [3].

The case study is framed within the European project Zero-Defect Manufacturing Platform (ZDMP) in the Preparation Stage: Start-up optimization, in which tasks like the optimization of equipment, materials, energy and energy efficiency are addressed [4].

This research work proposes a mixed integer linear programming (MILP) model for the lot-sizing/scheduling problem to manufacture plastic automotive components that contemplates the use, availability and arrival of materials, including raw materials, to inject parts into moulds, as well as containers for packing the finished components to be delivered to the first-tier supplier. It aims to minimize setups, the inventory of raw materials and plastic components, stockouts, backorder costs and machine-mould assignation costs.

This work is set out as so. Section 5.2 starts by reviewing the related literature. Section 5.3 describes the studied problem and the mathematical formulation. Section 5.4 discusses the computational experiments and the results. Section 5.5 offers some concluding remarks and future research lines.

5.2 Literature review

Substantial research has been conducted on various aspects of lotsizing/scheduling problems in distinct industries [5] like those presented by Almada-Lobo et al. [6], who studied two linear mixed-integer programming formulations for a multi-item capacitated lot-sizing problem with sequencedependent setup costs and times for the glass container industry. de Armas and Laguna [7] developed an MILP formulation for a capacitated lot-sizing/scheduling problem toward pipe insulation manufacturing, which included multiple- and single-level items processed on parallel machines according to a planning horizon.

The literature also describes several articles that have addressed injection moulding lot-sizing/scheduling problems. They include Nagarur et al. [8], who present a goal programming model for the injection moulding of PVC pipe fittings. This model aimed to minimize total production costs, inventory and shortages. Ghosh Dastidar and Nagi [9] address the production scheduling problem in an injection moulding facility that produces healthcare products. Their work presents

Chapter 5. An MILP model for the lot-sizing/scheduling of automotive plastic components with raw materials and packaging availability.

an MILP model that schedules parallel work centres with changeover costs, sequence-dependent setup times and multiple capacitated resources in a singlestage case. Martínez et al. [10] describe an MILP model that addresses the lotsizing/scheduling problem for a Brazilian moulded pulp packaging plant. With their model, they seek to establish which moulding patterns can be utilized, for how long, and how they can be sequenced. Ríos-Solís et al. [11] present an MILP model and a heuristic method based on a mathematical programming method for a lotsizing/scheduling problem. The aim is to determine the maximum profit made with assembled products during many periods. This model deals with plastic injection moulding as part of manufacturing, pursues precise production assignment from parts to moulds and from moulds to machines, seeks to maximize the total value of manufactured products and deduces maintenance costs. Mula et al. [12] propose an MILP model for solving the capacitated lot-sizing problem with sequence-dependent setups and parallel machines for injection moulding in the automotive industry. Moulding involves injecting two different parts or products into the same mould. Both parts need the same sequence order and available capacity at the same time.

Andres et al. [13] set up an MILP model for the production/lotsizing/scheduling problem on parallel flexible injection moulding machines with common setup operators. To produce automotive plastic components, the model allocates moulds to machines during a given time period and calculates the number of components to be manufactured. A sequence-dependent setup time is followed for this purpose. The model also bears in mind the common setup operators who change moulds on machines.

As far as we know, mathematical models do not contemplate the availability of materials/packaging for the delivery of components from the second-tier to the first-tier supplier. The herein proposed model extends that by Andres et al. [13] because the proposed MILP model takes into account the arrival, use and availability of not only the raw materials for injecting parts in moulds, but also the packaging for the finished components. Moreover, moulds can be changed only during time windows and depend on the amount of labour available during each period. The proposed model contemplates similar assumptions to those reported by Andres et al. [13], which envisages that moulds can be set up on different injection machines, and MILP output supplies mould-machine assignments.

5.3 Problem description and formulation

The proposed MILP the lot-sizing/scheduling of automotive plastic components with common setup labour and limited raw and packaging materials availability to transport components from the second- to the first-tier supplier is incorporated in a source and make scheme which is classified according to SCOR views [14]. The Plan Source (S) deals with the calculation of the raw materials, items or components to be supplied during each time period and on a specific planning horizon so that the Plan Make (M) can be fulfilled with no backorder penalizations. For the Source and Make Plans (SM), the production plan (M) is computed according to the production requirements identified in the procurement plan (S) [15, 16].

The SM planning scheme is followed by the second-tier supplier to deliver automotive plastic components to assemble them at the first-tier supplier and OEMs (original equipment manufacturers). The SM plan is generated to identify the period and quantity of: (i) the materials and components to be purchased from suppliers (plan S); (ii) the components to be manufactured in the company to assemble and produce the final product (plan M); see Figure 5.1.



Figure 5.1. Outline of the Source and Make Plans

Chapter 5. An MILP model for the lot-sizing/scheduling of automotive plastic components with raw materials and packaging availability.

The firm under study has several moulds that are set up on different injection machines to produce the range of plastic components to be delivered to the firsttier supplier, and finally to the various OEMs forming part of the distinct automotive supply chains characterized by selling to several car brands. The MILP model under discussion is based on the following assumptions:

- Plastic components are injected into moulds, which are assembled on parallel flexible injection machines. Injection machines inject plastic granules which are transformed into automotive semifinished products
- The second-tier supplier has specific moulds for producing each automotive plastic component. When two moulds are available to produce the same plastic component, these moulds can come into play at different processing times because of their technical characteristics
- Each mould can produce one part, or two parts or more, in the same mould
- Each mould can be placed on distinct injection machines to manufacture the same automotive component. However, the same mould set up on different machines has several production rates depending on the machine it is assembled on
- The company works three shifts per day 5 days a week and works overtime shifts on day 6 of the week if production does not end during normal working hours. On day 6, no setup operators are available
- One of the company's study requirements is that, after installing the mould on a machine, the mould must remain at least 24 h to not saturate operators' work and to not involve too many setups because the installation time is estimated to go from 1 to 3 h, and it obviously has an associated setup cost. If a longer production time is necessary, the mould is set up for the required time periods without incurring installation costs
- When the production time lasts longer than 24 h, the mould remains assembled for the necessary time periods with no incurred installation costs
- Backorders are highly penalized in the automotive sector because they work with just-in-time (JIT) models
- The mould can be changed only during specific time windows. Mould changes are counted to not exceed setup operators' capacity. Table 5.1 describes the indices, parameters and variables of this problem.

	Table 5.1.Notation
Index:	
i	Index of machines $i \in \{1,, I\}$
j	Index of moulds $j \in \{1,, J\}$
k	Index of parts $k \in \{1,, K\}$
1	Index of setup operators $I \in \{1,, L\}$
r	Index of materials (raw materials / packaging) $r \in \{1,, R\}$
t	Index of time periods $t \in \{1,, T\}$
Model parame	ters:
aj	Total amount of moulds <i>j</i> available for production
Ca _{kr}	Use of material <i>r</i> required to produce each unit of part <i>k</i>
cbk	Backorder cost of part <i>k</i>
cik	Inventory cost of part <i>k</i>
cim _k	Inventory cost of materials <i>r</i>
COVkt	Stock coverage defined as number of time periods for the stock minimum coverage of part k during time period t
CSj	Setup cost of preparing mould <i>j</i>
<i>cst</i> _k	Coverage stockout cost of part <i>k</i>
d _{kt}	Demand of part <i>k</i> during time period <i>t</i>
INV _{k0}	Initial inventory of part <i>k</i>
INVr ₀	Initial inventory of material <i>r</i>
INVMAX _k	Maximum inventory units for part k during time period t
INVMIN _k	Minimum inventory units for part <i>k</i> during time period <i>t</i>
INVMAXmat _r	Maximum inventory units for material <i>r</i> during time period <i>t</i>
nct	Number of mould changes permitted during time period <i>t</i>
<i>p</i> _{jk}	Number of parts <i>k</i> produced when mould <i>j</i> is set up
ro _{ij}	1 if mould <i>j</i> can be set up on machine <i>i</i>
	and 0 otherwise
rc _{ij}	Assignation cost of mould <i>j</i> on machine <i>i</i>
rp _{rt}	Quantity received of material <i>r</i> during each period <i>t</i>
sla _{ijl}	Amount of setup operators / required to setup mould <i>j</i> on machine <i>i</i>
scliji	Cost of setup operator / to setup mould <i>j</i> on machine <i>i</i>
slst	Number of workers <i>l</i> available during each period <i>t</i>

Chapter 5. An MILP model for the lot-sizing/scheduling of automotive plastic components with raw materials and packaging availability.

	Table 5.1. Continued. Notation
Decision vari	ables
B _{kt}	Backorder of part <i>k</i> during time period <i>t</i>
INV _{kt}	Inventory level of part k at the end of time period t
SA _{iljt}	1 if mould <i>j</i> is set up on machine <i>i</i> by setup operator <i>l</i> during time period <i>t</i> , and is not set up on machine <i>i</i> during time period <i>t-1</i> ;
	0 if mould <i>j</i> is set up by setup operator <i>l</i> on machine <i>i</i> during time period <i>t</i> - 1
Siljt	1 if mould <i>j</i> is set up by setup operator <i>l</i> on machine <i>i</i> during time period <i>t</i> ;
	0 otherwise
ST _{kt}	Coverage stockout of part <i>k</i> during time period <i>t</i>
Cam _{rt}	Material (raw material, packaging) r consumed during period t
INVm _{rt}	Inventory of material <i>r</i> during period <i>t</i>
X_{kt}	Amount of part <i>k</i> to be produced during time period <i>t</i>

Table 5.1. Continued. Notation

Next the formulation of the MILP model proposed for the lot-sizing/scheduling of automotive plastic components with available raw materials and packaging takes place. The objective function minimizes the setup and labour costs, machine-mould assignation, raw materials/packaging and plastic components inventory costs, backorder costs and costs for coverage stockouts.

$$Min z = \sum_{i} \sum_{l} \sum_{j} \sum_{t} cs_{j} \cdot SA_{iljt} + \sum_{i} \sum_{j} \sum_{t} \sum_{t} cs_{lijl} \cdot SA_{iljt} + \sum_{i} \sum_{j} \sum_{t} \sum_{t} ci_{ij} \cdot rc_{ij} \cdot SA_{iljt} + \sum_{k} \sum_{t} ci_{k} \cdot INV_{kt}$$
(1)
$$+ \sum_{k} \sum_{t} cim_{r} \cdot INVm_{rt} + \sum_{k} \sum_{t} cst_{k} \cdot ST_{kt} + \sum_{k} \sum_{t} cb_{k} \cdot B_{kt}$$

Subject to:

Sequence constraints

$$\sum_{j} \sum_{l} S_{iljt} \cdot ro_{ij} \le 1 \qquad \forall i, t \qquad (2)$$

178|

$$\sum_{i} \sum_{l} S_{iljt} \cdot ro_{ij} \leq a_j \qquad \forall j, t \qquad (3)$$

Constraint (2) establishes that 1 or 0 moulds j are set up by setup operator l to be produced during each time period t. Constraint (3) guarantees that the total number of available moulds j can only be set up for production as a maximum by setup operator l during each time period t.

Production and capacity constraints

$$X_{kt} = \sum_{i} \sum_{j} \sum_{l} p_{jk} \cdot r o_{ij} \cdot S_{iljk} \qquad \forall k, t$$
(4)

$$Cam_{rt} = \sum_{k} ca_{kr} \cdot X_{kt} \qquad \forall k, r, t$$
(5)

Constraint (4) determines the number of parts k to be manufactured during time period t, and ensures that a specific mould j can be set up on machine i during time period t while producing product k. Constraint (5) establishes the amount of raw materials and packaging r used during time period t.

Setup constraints

$$SA_{iljt} = S_{iljt} \qquad \forall i, l, j, t = 1$$
(6)

$$SA_{iljt} \ge S_{iljt} - S_{iljt-1}$$
 $\forall i, l, j, t > 1$

$$SA_{iljt} \le 1$$
 $\forall i, l, j, t$ (7)

$$\sum_{i} \sum_{j} SA_{iljt} \le nc_t \qquad \forall l, t,$$
(8)

Constraint (6) records the first setup of mould j carried out by operator l on machine i to identify the first time that mould j is set up during time period t on machine i. Constraint (7) ensures that SA_{iljt} takes binary values. Constraint (8) limits the number of mould j changes allowed during time period t, which are set up by operator l on machine i.

Labour constraint

$$\sum_{i} \sum_{j} SA_{iljt} \cdot sla_{ijl} \le sls_t \qquad \forall l, t \qquad (9)$$

Constraint (9) limits the number of mould changes permitted during time period t to the number of available workers l by bearing in mind the number of setup operators l needed to set up mould j on machine i.

Inventory balance equations

Chapter 5. An MILP model for the lot-sizing/scheduling of automotive plastic components with raw materials and packaging availability.

 $INV_{kt} = INV_{k0} + X_{kt} - d_{kt} + B_{kt} \qquad \forall k, t = 1$ (10a)

$$INV_{kt} = INV_{kt-1} + X_{kt} - d_{kt} + B_{kt} - B_{kt-1} \qquad \forall k, t > 1$$
(10b)

$$INVm_{rt} = INV_{r0} + rp_{rt} - \sum_{k} ca_{kr} \cdot X_{kt} \qquad \forall k, r, t = 1$$
(11a)

$$INVm_{rt} = INVm_{kt-1} + rp_{rt} - \sum_{k} ca_{kr} \cdot X_{kt} \qquad \forall k, r, t > 1$$
(11b)

Inventory balance equations (10a) and (10b) limit the appropriate values for inventories, the quantities to produce and the backorders for each time period t=1 and t>1, respectively. Constraints (11a) and (11b) ensure the uninterrupted supply of raw materials and packaging r for time periods t=1 and t>1.

Stock coverage constraint.

$INV_{kt} \ge INVMIN_k$	∀k,t	(12)
$INV_{kt} \leq INVMAX_k$	$\forall k, t$	(13)
$INVm_{rt} \ge \sum_{k} ca_{kr} \cdot X_{kt}$	$\forall k, r, t$	(14)
$INVm_{rt} \le INVMAX_r$	$\forall r, t$	(15)
$ST_{kt} \ge cov_{kt} - INV_{kt}$	$\forall k, t$	(16)

Constraints (12) and (13) restrict the inventory levels for each part k during time period t. Constraint (14) guarantees that the materials inventory corresponds to the quantity of material that need to be produced during the same period by considering a lead time of 0 and the batching technique is lot-for-lot. Constraint (15) limits the inventory levels for raw materials and packaging r during time period t. Constraint (16) is for the stock coverage of parts.

Bound and nature variables.

SA_{iljt} , $S_{iljt} \in \{0,1\}$	∀i, l, j, t	(17)
X_{kt} , INV_{kt} , B_{kt} , ST_{kt} , $\in \mathbb{Z}$	$\forall k, t$	(18)
$Cam_{rt,} INVm_{rt} \in \mathbb{Z}$	∀r,t	(19)

Constraint (17) determines the binary nature of both variables' setup S_{iljt} and setup amount SA_{iljt} Constraints (18) and (19) determine the represented variables' integer nature.
5.4 Computational experiments

An MILP model for the lot-sizing/scheduling of automotive plastic components, along with the availability of raw materials/packaging, was developed in Python 3.9.2 with Pyomo [17], employed as an extensible python-based open-source optimization modelling language for linear programming, and with Gurobi 9.0. All the experiments were run on a PC equipped with an Intel(R) Core (TM) i7-1165G7 CPU @ 2.80 GHz, 16GB of RAM with the Windows 10 Pro operating system.

5.4.1 Generating datasets

This section presents the experimental results. The conducted model's performance is depicted by 13 test problems. Data values are generated to reflect real automotive component industry data (see Table 5.2). The datasets needed for the experiments were built as in Andres et al. [13]. Data values are defined as shown below:

Parameter	Value	Parameter	Value
aj	1	INVMIN _k	Random (10, 100)
Ca _{kr}	1	INVMAXmat _r	99999
cb _k	99999	INVMINmat _r	Random (100, 150)
ci _k , cim _r	<i>U</i> (0.1, 1)	nct	Random (1, 2)
COVkt	Random (10, 100)	p jk	Random (20, 50)
CSj	Random (50, 100)	ro ij	Random (0, 1)
<i>cst</i> _k	1	rc ij	Random (1, 2)
d_{kt}	Random (10, 100) if <i>T</i> = first	rp _{kt}	Random (0, 50)
	of the five periods of the		
	week, otherwise 0 if T =		
	<i>period</i> 6 and <i>T</i> = period 7 of		
	the week		
INV _{k0}	Random (10, 150)	sla _{ilj}	1
INVr ₀	Random (1000, 1500)	scl _{iij}	U (2.5, 8.5)
INVMAX _k	Random (10000, 50000)	slsı	<i>NC</i> t

Table 5.2. Value generation for the data parameters

The algorithm developed to build the synthetic datasets is found at http://hdl.handle.net/10251/172395

5.4.2 Computational results

This section offers details of the case study of a second-tier supplier in an automotive supply chain. The results derived from the run time and the objective function value for solving problems are tabulated in Table 5.3. A simplified view of the solution is seen in Figure 5.2 to provide details of the problem that the second-tier supplier faces.

Chapter 5. An MILP model for the lot-sizing/scheduling of automotive plastic components with raw materials and packaging availability.

The size of datasets, including the number of machines (*I*), moulds (*J*), parts (K), material (*R*), setup labour (*L*) and periods (*T*), appears in the second column of Table 5.3. In most resolved instances (Small - S, Medium - M, Large – L), the model's computational performance (CPU time) is efficient for all instances. The solution for large instances provides optimal solutions in computational times under 20 seconds. The results obtained in the objective function do not include the backorder cost.

Data-	Problem size		Objective	Lower	Upper	GAP	СРИ				
set	1	J	K	R	L	Τ	-	bound	bound	(%)	(sec)
S1	2	4	6	3	1	3	505.84	505.84	505.84	0.00	0.03
S2	6	8	2	3	2	7	6503.13	6503.13	6503.13	0.00	0.29
S3	8	10	30	3	2	7	8673.92	8673.92	8673.92	0.00	0.26
S4	10	12	40	3	2	7	13841.98	13841.98	13841.98	0.00	0.41
M1	12	14	60	3	2	14	40097.58	39696.60	40097.58	0.01	1.64
M2	14	16	80	3	4	14	50359.57	49855.97	50359.57	0.01	1.51
M3	16	18	100	3	4	14	59831.37	59233.05	59831.37	0.01	2.08
M4	18	20	110	3	4	14	68995.23	68305.28	68995.23	0.01	5.67
L1	20	24	140	3	4	14	80587.91	80587.91	80587.91	0.00	2.04
L2	22	26	160	3	4	14	132509.80	131184.70	132509.80	0.01	14.09
L3	25	28	180	3	6	21	165415.31	163761.16	165415.31	0.01	11.94
L4	30	35	200	3	6	21	182957.97	181128.39	182957.97	0.01	18.52
L5	40	45	300	3	6	21	271586.75	268870.88	271586.75	0.01	15.92

Table 5.3. MILP model results.

Small instance S1 comprises two machines and four moulds, six parts, one operator and three periods. Figure 5.2 depicts how moulds can produce one part or more. In this case, the data generated synthetically, mould 1 produces parts 4, 5 and 6, mould 3 produces part 3 and mould 4 generates parts 1 and 2. The obtained results appear in Tables 5.4 - 5.6. With regard to the results of the sequence of the moulds on the machines Figure 5.2 illustrates that the operator puts mould 1 on machine 1 and manufactures for three periods, once mould 1 is placed the operator puts mould 3 on machine 2 and mould 4 is put on the same machine in period 2 (see Table 5.5). Table 5.6 describes the consumption and inventory, where r=1 corresponds to the raw material (plastic granules) and r=2 and r=3 to the packaging of the automotive semifinished products.

Models and algorithms for the optimization of procurement, production and distribution plans in industrial enterprises and supply chains.



Figure 5.2. Representation of the realistic lot-sizing/scheduling model with raw materials and packaging availability

183 |

Chapter 5. An MILP model for the lot-sizing/scheduling of automotive plastic components with raw materials and packaging availability.

			sizing.		
k	t	B_{kt}	INV _{kt}	ST _{kt}	χ_{kt}
1	1	0	20	0	0
1	2	0	30	0	20
1	3	0	40	0	20
2	1	10	10	10	0
2	2	10	10	10	50
2	3	10	10	10	50
3	1	0	10	10	50
3	2	10	10	10	0
3	3	20	10	10	0
4	1	0	10	5	50
4	2	0	10	5	50
4	3	0	10	5	50
5	1	0	10	25	50
5	2	0	10	25	50
5	3	0	10	25	50
6	1	0	10	0	50
6	2	0	10	0	50
6	3	0	10	0	50

 Table 5.4. Numerical results of instance S1: backorders, inventories, stockout, lot

Table 5.5. Numerical results of instance S1: scheduling.

i	1	j	t	Siljt	SA iljt
1	1	1	1	1	1
1	1	1	2	1	0
1	1	1	3	1	0
1	1	2	1	0	0
1	1	2	2	0	0
1	1	2	3	0	0
1	1	3	1	0	0
1	1	3	2	0	0
1	1	3	3	0	0
1	1	4	1	0	0
1	1	4	2	0	0
1	1	4	3	0	0
2	1	1	1	0	0
2	1	1	2	0	0
2	1	1	3	0	0

Models and algorithms for the optimisation of replenishment, produc	ction and distribution
plans in industrial enterprises.	

Table 5.5. Continued. Numerical results of instance S1: scheduling.					
i	1	j	t	Siljt	SAiljt
2	1	2	1	0	0
2	1	2	2	0	0
2	1	2	3	0	0
2	1	3	1	1	1
2	1	3	2	0	0
2	1	3	3	0	0
2	1	4	1	0	0
2	1	4	2	1	1
2	1	4	3	1	0

Table 5.6. Numerical results of instance S1: use and inventory of materials.

r	t	Cam _{rt}	INVm _{rt}
1	1	200	850
1	2	220	680
1	3	220	460
2	1	200	800
2	2	220	630
2	3	220	410
3	1	200	800
3	2	220	580
3	3	220	410

5.5 Conclusion

This research work develops an MILP model to integrate lot-sizing/scheduling decisions about automotive plastic components with raw materials/packaging availability to minimize setup and labour costs, components and raw materials inventory costs, backorder costs, machine-mould assignations and penalization costs for coverage stockouts. Both moulds and parts are employed as central indices for planning/scheduling on parallel machines. This work also contemplates the mould changes time window, the several setup times according to the number of workers assigned to mould change and mould-machine assignments. It also includes the arrival of materials, use of raw materials and availability of packaging containers.

Chapter 5. An MILP model for the lot-sizing/scheduling of automotive plastic components with raw materials and packaging availability.

This paper validates MILP performance and proves computationally efficient for different instance types, including large datasets that replicate the amount of data employed in real automotive industries. In future studies, the model's assumptions can be extended by adopting other practical conditions, such as constraints for transporting finished products, waiting times for containers for packing finished products to be delivered and limited space to store finished products.

5.6 References

- [1] M. Mohammadi, M. Esmaelian, and A. Atighehchian, "Design of mathematical models for the integration of purchase and production lot-sizing and scheduling problems under demand uncertainty," *Appl. Math. Model.*, vol. 84, pp. 1–18, 2020, doi: 10.1016/j.apm.2020.03.021.
- [2] C. Wolosewicz, S. Dauzère-Pérès, and R. Aggoune, "A Lagrangian heuristic for an integrated lot-sizing and fixed scheduling problem," *Eur. J. Oper. Res.*, vol. 244, no. 1, pp. 3–12, 2015, doi: 10.1016/j.ejor.2015.01.034.
- [3] E. Guzman, B. Andres, and R. Poler, "A MILP Model for Reusable Containers Management in Automotive Plastic Components Supply Chain," in 22nd IFIP WG 5.5 Working Conference on VIRTUAL ENTERPRISES, PRO-VE 2021, 2021, p. 8p, [Online]. Available: https://hal-emse.ccsd.cnrs.fr/emse-03338406.
- [4] S. Campbell, "D2.1: Inception and Vision Document," 2019. [Online]. Available: https://portal.effra.eu/result/show/3722.
- K. Copil, M. Wörbelauer, H. Meyr, and H. Tempelmeier, "Simultaneous lotsizing and scheduling problems: a classification and review of models," *OR Spectr.*, vol. 39, no. 1, pp. 1–64, 2017, doi: 10.1007/s00291-015-0429-4.
- [6] B. Almada-Lobo, D. Klabjan, M. A. Carravilla, and J. F. Oliveira, "Single machine multi-product capacitated lot sizing with sequence-dependent setups," *Int. J. Prod. Res.*, vol. 45, no. 20, pp. 4873–4894, 2007, doi: 10.1080/00207540601094465.
- [7] J. de Armas and M. Laguna, "Parallel machine, capacitated lot-sizing and scheduling for the pipe-insulation industry," *Int. J. Prod. Res.*, 2019, doi: 10.1080/00207543.2019.1600763.
- [8] N. Nagarur, P. Vrat, and W. Duongsuwan, "Production planning and scheduling for injection moulding of pipe fittings: A case study," *Int. J. Prod. Econ.*, vol. 53, no. 2, pp. 157–170, 1997, doi: 10.1016/S0925-5273(97)00109-6.
- [9] S. Ghosh Dastidar and R. Nagi, "Scheduling injection molding operations with multiple resource constraints and sequence dependent setup times and costs,"

Comput. Oper. Res., vol. 32, no. 11, pp. 2987–3005, 2005, doi: 10.1016/j.cor.2004.04.012.

- [10] K. Y. P. Martínez, E. A. V. Toso, and R. Morabito, "Production planning in the molded pulp packaging industry," *Comput. Ind. Eng.*, vol. 98, pp. 554–566, 2016, doi: 10.1016/j.cie.2016.05.024.
- [11] Y. Ríos-Solís, O. J. Ibarra-Rojas, M. Cabo, and E. Possani, "A heuristic based on mathematical programming for a lot-sizing and scheduling problem in moldinjection production," *Eur. J. Oper. Res.*, vol. 284, no. 3, pp. 861–873, 2020, doi: 10.1016/j.ejor.2020.01.016.
- [12] J. Mula, M. Díaz-Madroñero, B. Andres, R. Poler, and R. Sanchis, "A capacitated lot-sizing model with sequence-dependent setups, parallel machines and bipart injection moulding," *Appl. Math. Model.*, vol. 100, pp. 805–820, 2021, doi: 10.1016/j.apm.2021.07.028.
- [13] B. Andres, E. Guzman, and R. Poler, "A Novel MILP Model for the Production, Lot Sizing, and Scheduling of Automotive Plastic Components on Parallel Flexible Injection Machines with Setup Common Operators," *Complexity*, vol. 2021, p. 16, 2021, doi: https://doi.org/10.1155/2021/6667516.
- [14] Supply Chain Council SCC, Supply Chain Operations Reference Model SCOR version 11.0. 2012.
- [15] A. Orbegozo, B. Andres, J. Mula, M. Lauras, C. Monteiro, and M. Malheiro, "An overview of optimization models for integrated replenishment and producction planning decisions," in *Building bridges between researchers and practitioners. Book of Abstracts of the International Joint Conference CIO-ICIEOM-IISE-AIM* (IJC2016), 2016, p. 68.
- [16] B. Andres, R. Sanchis, R. Poler, and L. Saari, "A Proposal of Standardised Data Model for Cloud Manufacturing Collaborative Networks," in *Collaboration in a Data-Rich World*, 2017, pp. 77–85.
- [17] W. E. Hart, C. Laird, J.-P. Watson, and D. L. Woodruff, *Pyomo Optimization Modeling in Python*, 1st ed. Springer Publishing Company, Incorporated, 2012.

Chapter 6

A MILP model for reusable containers management in Automotive plastic components supply chain.

Guzman, E., Andres, B., & Poler, R. 2021. *"A MILP Model for Reusable Containers Management in Automotive Plastic Components Supply Chain"*. In: IFIP Advances in Information and Communication Technology, vol 629. Springer, Cham. https://doi.org/10.1007/978-3-030-85969-5_15

Abstract:

The automotive sector operates under the just-in-time (JIT) approach, but variations in demand mean that first-tier suppliers generate an accumulation of stocks at second-tier suppliers. Second-tier suppliers have a limitation of storage space, reason to limit their production to the size of the warehouse, but always attending the firsttier demand plan. A further limitation of the second-tier supplier is the number of empty reusable containers that the first-tier supplier delivers to the second-tier supplier and that are used to package the injected plastic components. The reusable filled containers are returned to the first-tier supplier, according to the plastic components demand plan. Thus, a closed-loop logistic is carried out between first and second-tier suppliers. This study proposes, from the second-tier perspective, a mixed integer linear programming (MILP) model for fleet sizing decisions of the cardboard containers that second-tier supplier has to use when the production is higher than the number of available reusable containers. Chapter 6. A MILP model for reusable containers management in Automotive plastic components supply chain.

6.1 Introduction

An increasing trend for companies is to work towards meeting environmental and economic requirements and reducing the environmental and social impact of their activities. There is also a rapidly growing interest in reusable packaging, such as wooden pallets and plastic crates and others. Several companies sell products in packaging that can be reused. Returnable transport items (RTI), which represent a specific type of reusable packaging material, including pallets, plastic boxes, or containers (air and maritime), are used today in various industries, for example, in the food sector, in the automotive industry or in the consumer goods industry [1].

The use of reusable packaging is justified by the benefits it can generate, such as the amortization of the price of packaging over its useful life [2]. The literature provides several studies showing the environmental benefits associated with reusable containers [3]. Glock and Kim [1] argued that the use of reusable packaging materials rather than single-use packaging materials has the significant contribution of reducing global CO2 emissions from production and transportation, and can significantly minimize the gross energy consumption and the waste generation from transportation.

The difficulty of the reusable container management problem is to have an appropriate supply of empty containers to meet the customer demand. Part of this supply is the result of returns of previously issued containers. A challenging factor is that, during the lead time, the same container may be emitted, returned, reemitted, etc.

The aim of this research is to investigate the production and fleet-sizing of cardboard containers decisions of a production system when reusable containers are utilized. This model has applicability to the automotive industry, which uses reusable containers to protect and transport plastic parts produced by the second-tier supplier and shipped to the first-tier supplier. The focus of our model is to determine the optimal levels of production and storage rate to minimize the setup times and the quantity of cardboard containers to be purchased when reusable containers, which are property of first-tier supplier, are insufficient to store the parts produced by the second-tier supplier.

The structure of the paper is as follows. Section 6.2 provides an overview of related work. Section 6.3 develops a mixed integer linear programming (MILP) model for Reusable Containers Management and contains numerical and negotiation examples. Section 6.4 concludes the article and offers suggestions for future research.

6.2 Literature review

This section provides a literature review of relevant contributions in the related research field. The literature focuses on packaging costs and emissions as a target to optimize the use of packaging. Most relevant studies are presented next. Accorsi et al. [4] propose a mixed-integer linear programming (MILP) model to address the use of reusable packaging in the food industry. The model establishes the number of available packaging and forces to meet the demand for packaging over the planning horizon by encouraging reutilization and recycling. Rajae et al. [2] present a MILP model that addresses the problem of reusable containers in a reverse supply chain, in a multi-tiered network and under a carbon emmision constraint. Goudenege et al. [5] developed a generic reverse logistics management model focused on investing in and managing reusable packaging at the lowest cost in order to reduce the amount of cardboard used by the company under study. Glock et al. [1] examine a supply chain consisting of a single supplier and several retailers that use returnable transportation items, such as containers or boxes, to facilitate the shipment of products from the supplier to the retailers. The paper presents two mathematical models used to determine the cycle time, container size, individual retailer order quantities, and shipping sequence with the intention of minimizing the average total system costs. Park and Kim [6] present an analytical model for fleet-sizing of containers that are used for the protection (storage of finished parts), transportation and storage of parts between a component plant and multiple assembly plants. Atamer et al. [7] analyse-the pricing and production decisions of a manufacturer selling a single product when using reusable containers with stochastic customer demand, and two supply scenarios are analysed: (1) new containers and (2) containers returned by customers. In our paper we consider different characteristics addressed in the literature, with the novelty that integrates the decisions of production scheduling and sequencing to determine the optimal use of reusable containers and cardboard containers required to store and send plastic components. The problem modelled adjusts to a real problem that have transmitted to us the first and second-tier supplier of an automotive supply chain.

6.3 Problem definition

We consider a second-tier supplier that produces plastic components for its assembling in the first-tier supplier. The second-tier supplier produces the parts in moulds that are assembled on injection machines. The machine setup has a high cost associated; therefore, production is constrained by the number of Chapter 6. A MILP model for reusable containers management in Automotive plastic components supply chain.

moulds changed in a specific period and the amount of periods that the mould must be mounted within the machine. The main aim is to minimize the costs of production, storage, and machine setups, without incurring in backorders on the first-tier demand plan. Once the second-tier supplier has produced the plastic components, according to its optimal production plan, the parts are sent to the first-tier supplier in reusable containers, which are of its property. Reusable containers are limited in capacity and number, when the second-tier supplier produces more parts than he can store in the reusable containers, he has to store in temporary cardboard containers, until empty plastic containers arrive. The use of cardboard containers implies that the second-tier supplier must incur handling costs because they must put the parts in the cardboard containers and then switch to the reusable containers. In addition, the manufacturer must purchase the cardboard containers. Figure 6.1 shows the closed loop of reusable containers.

The optimization of the second-tier supplier production and scheduling plan, results in grouping production in batches and thus having to stock products. If there are sufficient reusable containers, the produced parts are stored and wait in the warehouse to be delivered to the first-tier supplier, and after a lead time the reusable containers re-circulate and are returned to the second-tier supplier. When there are not sufficient reusable containers, second-tier supplier stores the parts in the cardboard containers.



Figure 6.1. Closed loop of reusable containers scheme

192 |

Chapter 6. A MILP model for reusable containers management in Automotive plastic components supply chain.

6.3.1 A MILP model for Reusable Containers Management

This study proposes, from the second-tier perspective, a Mixed Integer Linear Programming (MILP) model for the production, lot-sizing, and scheduling of automotive plastic components, which takes into consideration the number of reusable containers in circulation throughout the closed-loop logistic. Moreover, the model also determines the number of cardboard containers that second-tier supplier has to use when the production is higher than the number of available reusable containers. This model allows to deter- mine the optimal number of reusable containers that should be bought by the first-tier supplier in order not to incur in extra costs due to the use of cardboard containers, which will increase the price of plastic components produced in the second-tier supplier, compromising the supply chain sustainability. This information is useful for both first and second-tier suppliers since with this data both suppliers in the supply chain can negotiate the final price of plastic components, which is contractually dependent on the number of returnable containers delivered by the first-tier supplier to the second-tier supplier (Table 6.1).

Index	
t	time period index $t \in \{1,, T\}$
Data	
ch	handling cost of cardboard container
cb	purchase cost of the cardboard container
сс	container capacity
CS	setup cost of preparing tool
сар	warehouse volume storage capacity
d_t	demand of containers at period t
inv0	initial inventory of reusable containers
dl	delay between sending a full reusable container to the first-tier supplier and returning an empty reusable container to the second- tier supplier
invp ₀	initial inventory of parts
invec ₀	initial inventory of empty reusable containers
invfc ₀	initial inventory of filled reusable containers (filed of plastic parts)

Table 6.1. Nomenclature for Reusable Containers Management model

Table	6.1. Continued. Nomenclature for Reusable Containers Management model
Data	
nmaxc	maximum number of reusable containers
nct	number of mould changes permitted during time period t
Xt	number of parts that the machine is able to produce during time period t
V	volume of containers
Decision v	ariables
CBN_t	number of cardboard containers
INVt	inventory of parts at the end of time period t
IECt	inventory of empty reusable containers
ICFt	inventory of filled reusable containers
NCt	number of containers required
SAt	1 when the mould is set up on machine during period <i>t</i> , 0 when mould is set up on machine during period <i>t-1</i>
St	1 when the mould is set up on machine during period <i>t,</i>
	0 otherwise
Xnt	number of parts to produce during period t

The MILP model formulation for managing the availability of reusable containers in automotive plastic components supply chain is represented below. The objective function minimizes total costs, which comprise setup costs, and investment and handling cost of reusable containers.

$$Min z = \sum_{t} cs \cdot SA_{t} + \sum_{t} ch \cdot cb \cdot CBN_{t}$$
⁽¹⁾

Subject to:

Sequence and setup constraints

$S_t \leq 1$	$\forall t$	(2)
$SA_t \geq S_t - S_{t-1}$	$\forall t$	(3)
	_	

$$SA_t \le nc_t \qquad \forall l, t,$$
 (4)

Constraint (2) guarantees that one or neither mould could be set up in production during each period *t*. Constraint (3) guarantees the first tool setup on

Chapter 6. A MILP model for reusable containers management in Automotive plastic components supply chain.

the machine in period t. Constraint (4) guarantees the number of tool changes allowed during period t.

Production constraint

$Xn_t \leq S_t \cdot x_t$	$\forall t$	(5)
Constraint (5) determines the numbe	r of parts produced during time p	eriod <i>t</i>
Inventory constraints		
$INV_t = invp_0 + Xn_t - d_t * cc$	$\forall t = 1$	(6)
$INV_t = INV_{t-1} + Xn_t - d_t * cc$	$\forall t > 1$	
$NC_t = INV_t/cc$	$\forall t$	(7)
$IFC_t = invfc_0 - d_t$	$\forall t = 1 \text{ if } NC_t > invec_0$	(8)
$IFC_t = NC_t - d_t$	$\forall t = 1 \text{ if } NC_t \leq invec_0$	(9)
$IFC_t = IFC_{t-1} + IEC_{t-1} - d_t$	$\forall t > 1 \text{ if } NC_t > IEC_{t-1}$	(10)
$IFC_t = NC_t - d_t$	$\forall t > 1 \text{ if } NC_t \leq IEC_{t-1}$	(11)
$IEC_t = invec_0$	$\forall t = 1 \text{ if } NC_t > invec_0$	(12)
$IEC_t = invec_0 - IFC_t$	$\forall t = 1 \text{ if } NC_t \leq invec_0$	(13)
$IEC_t = d_{t-dl}$	$\forall t > 1 \text{ if } NC_t > IEC_{t-1}$	(14)
$IEC_t = IEC_{t-1} - d_{t-dl}$	$\forall t > 1 \text{ if } NC_t \leq IEC_{t-1}$	(15)
$CBN_t = NC_t - invec_0$	$\forall t = 1 \text{ if } NC_t > invec_0$	(16)
$CBN_t = 0$	$\forall t = 1 \text{ if } NC_t \leq invec_0$	(17)
$CBN_t = NC_t - IEC_{t-1}$	$\forall t > 1 \text{ if } NC_t > IEC_{t-1}$	(18)
$CBN_t = 0$	$\forall t > 1 \text{ if } NC_t \leq IEC_{t-1}$	(19)
$v \cdot (IEC_t + IFC_t + CBN_t) \le cap$	$\forall t$	(20)
$nmaxc \ge IEC_t + IFC_t + d_t$	$\forall t$	(21)

Constraint (6) defines the inventory level of parts that have not been packaged and delivered in a reusable container, i.e., it determines the inventory of overproduction due to the batch sizes. Constraint (7) establishes the required number of reusable and cardboard containers needed for packaging plastic parts. Constraints (8 to 11) manage the inventory of filled reusable containers and control the allocation of plastic parts to reusable and cardboard containers. Constraints (12 to 14) control the inventory of empty reusable containers. Constraints (16 to 19) determine the allocation of parts that have been packaged in cardboard containers, due to the fact that there are missing empty reusable

containers reusable containers on the second-tier supplier side. It is determined that after a delay time a filled reusable container sent to the first-tier supplier is released as empty reusable container to the second-tier supplier. Constraint (20) is referred to as the storage capacity constraint, which guarantees that the reusable and cardboard container inventory in the warehouse in period t is always less than the capacity of the manufacturer's warehouse. Constraint (21) limits the number of filled and empty reusable containers, since there is a limited number of reusable containers delivered from the first-tier supplier to the second-tier supplier.

Bound and nature variables.

$SA_t, S_t \in \{0,1\}$	$\forall t$	(22)
$INV_t, ICF_t, IEF_t, CBN_t, NC_t, Xn_t \in \mathbb{Z}$	$\forall t$	(23)

Constraint (22) denotes the binary character of the variables S_t and SA_t . Constraint (23) specifies the integer character of the variables represented.

6.3.2 Numerical Experiment

The model is formulated in Python and solved with Gurobi. The data in this case study has been generated randomly. Table 6.2 shows the solutions that arrive at one of the generated instances, in this case we have considered 6 periods, in this scenario, as can be seen in Table 6.2, the second level supplier has to use cardboard containers (*CBN*_t) in several periods causing it to incur handling costs. Gurobi takes a few seconds to find the optimum solution on a computer configured with 11th Gen Intel(R) Core (TM) i7-1165G7 @ 2.80GHz processors and 16 GB of RAM.

							3	
t	CBN _t	<i>IEC</i> _t	<i>IFC</i> _t	INVt	NCt	S_t	SAt	Xnt
1	0	1	0	60	15	1	1	32
2	9	15	0	40	10	1	0	20
3	0	10	2	60	15	1	0	72
4	3	13	0	52	13	1	0	44
5	4	13	0	68	17	1	0	84
6	6	17	0	76	19	1	0	84

Table 6.2 Results of MILP model for Reusable Containers Management.

Chapter 6. A MILP model for reusable containers management in Automotive plastic components supply chain.

6.4 Collaboration scheme

Figure 6.2 depicts the process of negotiating the price of plastic components produced by the second-tier supplier. The price of the plastic components is determined by (i) the number of reusable containers that the first-tier supplier delivers to the second-tier supplier; (ii) the number of cardboard containers that the second-tier supplier has to buy (if the number of empty reusable containers is insufficient to store the plastic components produced by the second-tier supplier); (iii) and the costs associated with the plastic components production process, setup costs.



Figure 6.2. Flow chart of negotiation process

6.5 Conclusions

In this paper, we provide an integrated approach to help companies consider options when managing their reusable containers. This paper proposes a MILP model for optimizing the scheduling of automotive plastic components, which takes into account the use of reusable containers that are required for the protection and transportation of finished products from a second-tier manufacturer to a first-tier supplier. It also determines the number of cardboard containers to be purchased by the second-tier supplier when reusable containers are not available, so the second-tier supplier must incur handling costs to store the parts in cardboard containers until reusable containers are available. Future research lines are led to (i) include in the model the carbon emissions derived from the transport of reusable containers; (ii) the consideration of backorders penalization in the objective function; (iii) and the algorithm implementation of the proposed collaboration scheme, in order to determine the optimal number of reusable containers and the competing price of plastic components.

6.6 References

- [1] C. H. Glock and T. Kim, "Container management in a single-vendor-multiplebuyer supply chain," *Logist. Res.*, vol. 7, no. 1, 2014, doi: 10.1007/s12159-014-0112-1.
- [2] E. Rajae, B. Mohamed, and Z. Tarik, "Reverse logistic optimization: Application to the collect and the reuse of reusable containers," ACM Int. Conf. Proceeding Ser., 2018, doi: 10.1145/3230905.3230966.
- [3] C. H. Glock, "Decision support models for managing returnable transport items in supply chains: A systematic literature review," *Int. J. Prod. Econ.*, vol. 183, pp. 561–569, 2017, doi: 10.1016/j.ijpe.2016.02.015.
- [4] R. Accorsi, G. Baruffaldi, and R. Manzini, "A closed-loop packaging network design model to foster infinitely reusable and recyclable containers in food industry," *Sustain. Prod. Consum.*, vol. 24, pp. 48–61, 2020, doi: 10.1016/j.spc.2020.06.014.
- [5] G. Goudenege, C. Chu, and Z. Jemai, "Reusable containers management: From a generic model to an industrial case study," *Supply Chain Forum*, vol. 14, no. 2, pp. 26–38, 2013, doi: 10.1080/16258312.2013.11517313.
- [6] S. J. Park and D. S. Kim, "Container fleet-sizing for part transportation and storage in a two-level supply chain," *J. Oper. Res. Soc.*, vol. 66, no. 9, pp. 1442– 1453, 2015, doi: 10.1057/jors.2014.111.
- [7] B. Atamer, I. S. Bakal, and Z. P. Bayindir, "Optimal pricing and production

Chapter 6. A MILP model for reusable containers management in Automotive plastic components supply chain.

decisions in utilizing reusable containers," *Int. J. Prod. Econ.*, vol. 143, no. 2, pp. 222–232, 2013, doi: 10.1016/j.ijpe.2011.08.007.

Chapter 7

Matheuristic Algorithm for Job-Shop Scheduling Problem Using a Disjunctive Mathematical Model.

Guzman E, Andres B, Poler R. 2022. *"Matheuristic Algorithm for Job-Shop Scheduling Problem Using a Disjunctive Mathematical Model"*. Computers. 11(1):1. https://doi.org/10.3390/computers11010001

Abstract:

This paper focuses on the investigation of a new efficient method for solving machine scheduling and sequencing problems. The complexity of production systems significantly affects companies, especially small- and medium-sized enterprises (SMEs), which need to reduce costs and, at the same time, become more competitive and increase their productivity by optimizing their production processes to make manufacturing processes more efficient. From a mathematical point of view, most real-world machine scheduling and sequencing problems are classified as NP-hard problems. Different algorithms have been developed to solve scheduling and sequencing problems in the last few decades. Thus, heuristic and metaheuristic techniques are widely used, as are commercial solvers. In this paper, we propose a matheuristic algorithm to optimize the job-shop problem which combines a genetic algorithm with a disjunctive mathematical model, and the Coin-OR Branch & Cut opensource solver is employed. The matheuristic algorithm allows efficient solutions to be found, and cuts computational times by using an open-source solver combined with a genetic algorithm. This provides companies with an easy-to-use tool and does not incur costs associated with expensive commercial software licenses.

Chapter 7. Matheuristic Algorithm for Job-Shop Scheduling Problem Using a Disjunctive Mathematical Model.

7.1 Introduction

Nowadays, rapidly growing economic markets, competitive pressures and increasingly challenging business environments are forcing increasingly more companies, especially small- and medium-sized enterprises (SMEs), to innovate their industrial manufacturing systems. SMEs have had to respond and adapt to a constantly changing organizational environment to deliver high-quality customized products. Consequently, SMEs supply chains are not static as they must respond to continuous change by adapting their control techniques, and coordinating and managing change in the way they operate and configure their businesses. Companies also have to manage their evolution toward participation in collaborative networks [1].

The market in which these companies currently operate is intensely volatile, which makes effective supply chain (SC) management critical to improve organizational performance as manufacturing systems become increasingly dynamic [2] due to new challenges in manufacturing industries, such as Industry 4.0 and the Internet of Things (IoT).

Researchers are showing much interest in improving the performance of enterprises and SC to generally cope with these dynamic environments by devising mechanisms and techniques that provide SMEs with affordable tools in cost, easyto-use and computational efficiency terms. The search for solutions for company scheduling problems, such as job-shop scheduling problems (JSP), remains a relevant research topic [3]. This is because most of these real-world scheduling problems are too complex to be optimally solved and are often NP-hard. This means that exact techniques and some algorithms cannot solve them in effective computational times when the problem is too large. At the same time, solving them with commercial solvers is neither economically viable nor computationally efficient.

Mathematical formulations like mixed integer linear programming (MILP) models for JSP, have been around since the 1960s [4]. The leading formulations for this problem type are disjunctive formulation, rank-based formulation and time-indexed formulation [5].

Ku and Beck [5] compared these mathematical formulations with different solvers (CPLEX, GUROBI and SCIP, the first two are commercial and the last one is not), which showed that the disjunctive model outperformed the rank-based and time-indexed models.

In this context, a new matheuristic algorithm combining a genetic algorithm and the disjunctive model (MILP) is proposed in this study. Matheuristic algorithms are constructed by "the interoperation of metaheuristics and mathematical programming techniques" [6]. According to Ball [7] and Talbi [8], combinations or hybridizations of matheuristics can be classified into three approaches: 1) decomposition approaches, where the problem is decomposed into subproblems to be optimally solved; 2) improvement heuristics or metaheuristics, where the mathematical programming model is used to enhance an initial solution obtained by some heuristic or metaheuristic method; and 3) approaches employing the mathematical programming model to provide approximate solutions in which a relaxation of the problem toward optimality is solved.

The method presented in our study consists of a combination of a genetic algorithm and linear programming (LP) model (GA-LP) that is included in approach 2 of this classification. The main objective of this work is to test the non-commercial COIN-OR Branch and Cut (CBC) solver [9] for solving the JSP, combined with a genetic algorithm, in large or real instances. The experimental results confirm the feasibility and effectiveness of the proposed matheuristic compared to the solutions provided by the solver.

Accordingly, the document is structured as follows. Section 7.2 reviews work related to the application of matheuristic algorithms to the JSP. Section 7.3 presents the proposed JSP mathematical model in detail. Section 7.4 describes the matheuristic approach. Section 7.5 presents the computational experiments and discusses the results. Finally, Section 7.6 covers the conclusions of the performed work and future research lines.

7.2 Literature Review: Matheuristic Resolution Approaches

A wide variety of papers describes different models and algorithms to solve scheduling problems [10]. Many of these techniques correspond to mathematical models, heuristic and metaheuristic algorithms [11]. The application of these techniques depends on the application area, i.e., SC planning under uncertainty [12], closed loop SC [13], SC sustainable management [14] or green SC management [15]. These studies reviewed the models and algorithms employed to solve optimization problems in their specific field.

In this paper, we focus on the scheduling problem to be addressed at the operational decision-making level, and we pay particular attention to the JSP which is considered to be NP-hard. To address this problem, heuristic (H) and metaheuristic (MH) approaches have received much attention in the literature.

Chapter 7. Matheuristic Algorithm for Job-Shop Scheduling Problem Using a Disjunctive Mathematical Model.

Indeed, many literature surveys have been carried out over time. Thus, deterministic and stochastic optimization models have been developed to solve the JSP [3, 16, 17]. Moreover, other approaches including decomposition heuristics, dispatching heuristics, disjunctive representations of the problem, discrete simulation or rolling horizon approaches can be found in the literature [18].

To offer readers an overview on the studied topic, we reviewed how the literature has applied matheuristic approaches to solve the JSP. In our research, we applied the keywords "matheuristic" AND "job-shop scheduling problem". Seventeen papers coincide with our research in the Scopus database. Some tackle flow shop scheduling problems, and others refer to reviews. After analyzing the abstracts and the whole contents of the papers, nine papers remained of the initial seventeen (see Table 7.1). Our review research is not without limitations as the search results may not fully cover all the matheuristic proposed in the literature for being named differently from the keywords used in our research "matheuristic", e.g., hybrid algorithms.

As a general overview, the JSP was tackled from different perspectives, namely flexible JSP, dynamic JSP, resource constrained JSP, parallel machine JSP or just-intime JSP. This review revealed that the most widely used metaheuristics are led by genetic algorithms and tabu search algorithms. The matheuristics presented to address the JSP integrates an MILP with a metaheuristic algorithm in most cases. Others consider an MILP combined with a constructive heuristic to increase the intelligence of the MILP and to reduce computational resolution times in large sized experiments.

Table 7.1 highlights some relevant characteristics of the matheuristic proposed in the analyzed works in terms of: (i) the type of addressed JSP; (ii) the proposed matheuristic; (iii) the integrated approaches used to define the matheuristic, including a heuristic algorithm combined with an MILP or a metaheuristics combined with an MILP; (iv) the employed programming language and modeling language, as well as the solver used to compute the exact method; and (v) the experiment size (job x machine). These features are based on the solution approaches defined in the framework proposed by [10].

In order to provide a profounder analysis, Al-Hinai and Elmekkawy [18] propose an approach to obtain a predictive schedule that minimizes machine breakdowns and responds to a flexible JSP. To this extent, a 2-stage hybrid genetic algorithm (HGA) is proposed: (i) the first stage optimizes the primary objective by minimizing the makespan and considering deterministic data without machine

breakdowns; (ii) the second stage optimizes a bi-objective function (by considering robustness and stability) and integrates machine assignments and operations sequencing with the expected machine breakdowns. Continuing with the scope of a flexible JSP, a hybrid tabu search algorithm with a fast public critical block neighborhood structure (TSPCB) is proposed by Li et al. [19].

These authors present a mixture of four machine assignment rules and four operation scheduling rules to improve the quality of the initial solutions and provide the hybrid algorithm with good exploration capability. Then, they put forward an efficient neighborhood structure to perform local searches in the machine allocation module, which integrates three adaptive approaches. Finally, they present a speedup local search method with three types of insertion and swap neighborhood structures based on the public critical block theory. In line with this, Li and Gao [20] report an effective HGA that hybridizes the genetic algorithm (GA) and tabu search (TS) to address the flexible JSP with a view to minimize the makespan. The GA has a powerful global searching ability, and the TS has a valuable local searching ability.

Thiruvady et al. [21] deal with the Resource Constrained Job Scheduling (RCJS) problem by proposing two MIP-based matheuristic approaches that rely on the solution merging concept to learn from a population of solutions and to use an MIP to generate a "merged" solution in the subspace, which is spanned by a pool of heuristic solutions. The first approach is the Merge Search (MS) and the second is Construct, Merge, Solve and Adapt (CMSA).

Rohaninejad et al. [22] address the JSP of parallel machines with incompatible job families and proposes an efficient matheuristic algorithm based on the hybridization of a GA and a local search (LS) method based on mixed integer programming (MIP). The GA is used to optimize the subproblems related to determining the sequence of parts and the allocation of parts to machines. The allocation of parts to batches is performed by an effective heuristic named batching heuristic (GA_BH) by combining a GA with a batching heuristic (BH). Moreover, the authors propose a combination of a GA and a dispatching rule called Apparent Tardiness Cost (GA_ATC). Dang et al. [23] also deal with the JSP of parallel machines with tool replacements to schedule a set of jobs with tool requirements on identical parallel machines in a work center. To do so, the authors propose a mathematical model for the problem and a matheuristic that combines a GA and an integer linear programming (ILP) formulation to solve large datasets. The matheuristic integrates ILP into the GA framework as a local search step to enhance GA performance. Chapter 7. Matheuristic Algorithm for Job-Shop Scheduling Problem Using a Disjunctive Mathematical Model.

Ahmadian and Salehipour [24] deal with the just-in-time job-shop scheduling problem (JIT–JSP) with distinct due dates for operations with earliness and tardiness penalties. For this purpose, the authors propose a matheuristic algorithm that decomposes the problem into smaller subproblems to obtain optimal or near-optimal sequences to perform the operations for the subproblems, which provides a feasible schedule for the complete problem. The algorithm forms the subproblems by applying two neighborhoods. The employed algorithms are the Giffler Thompson (GT) algorithm, the Shifting Bottleneck Heuristic (SBH) algorithm, the variable neighborhood search (VNS), and the relaxation neighborhood.

Son et al. [25] address the problem of scheduling jobs with limited splitting on a single machine in the available time windows. These authors present an MILP formulation for this problem and propose different heuristics related to the assignment strategy, such as: assignment heuristic (AH); heuristic based on the shortest/longest processing time rules (HSLPTR); heuristic based on max flow resolution (HMAXFR); and heuristic based on a matching and assignment approach (MAAS). They also apply a combination between the proposed heuristics and metaheuristics, such as tabu search and the GA.

These authors also introduce another approach called exact for subset-jobs matheuristic, which combines mathematical programming, and a priority heuristic rule called the single-attribute priority rule.

Cota et al. [26] propose a solution to address the JSP with unrelated parallel machines with sequence-dependent setup times, and independent non-preemptible jobs, minimizing the makespan and the total consumption of electricity. The authors define a multi-objective smart pool search matheuristic for finding solutions near the Pareto front, in which different MILP problems are generated with different weights for aggregating both objective functions involved in the proposed formulation.

From the review, we can state that very few papers apply combined or hybrid algorithms, such as matheuristic algorithms. In other production fields, matheuristics have obtained good solutions. The research of Cabrera-Guerrero et al. [27] demonstrates that the combination of techniques, or hybridization, can be advantageous for solving complex problems, which is also demonstrated in [28]. Verbiest et al. [28] used a combination of an iterated local search algorithm (metaheuristics) with an MILP model to optimize production lines, design installed lines and allocate products. Their study compares the matheuristic approach with an exact method (MILP) to verify that the matheuristic offers efficient solutions

and in a shorter calculation time. According to the results of the studied works, we conclude that matheuristic techniques are suitable for solving problems in realistic instances and allow good results to be obtained in acceptable computing times. Nevertheless, experiments are carried out on commercial solvers, which can be a drawback for those enterprises that cannot afford these tools. Moreover, the maximum data size used for experiments are 120 jobs on six machines, and 20 jobs on 10 machines, which cannot be completely representative of enterprises' realistic data.

Although matheuristics is becoming increasingly well-known for its effectiveness and computational efficiency when dealing with large and NP-hard problems, there is still a long way to go in this field. The contribution of this research aims to provide a solution to the NP-hard JSP with the proposed matheuristic approach by combining a GA and an LP model using a non-commercial solver (CBC) and an open-source operating system (Linux) for a large set of instances.

Table 7.1 Literature review of matheuristic to solve the job-shop problem.							
Reference	Job-Shop Problem	Matheuristic	Integrated Ap	proaches	Programming	Experiment Size	
	Туре		H+MILP	MH+MILP	Languages /Modeling Language/ Solver	(Job × Machine)	
Al-Hinai and Elmekkawy [18]	Flexible JSP	HGA		MILP + GA	C++ / - / - /	20 × 15	
Li et al. [19]	Flexible JSP	TSPCB		MILP + TSPCB	C++ / - / - /	20 × 15	
Li and Gao [20]	Flexible JSP	HGA TS		GA + TS	C++ / GAMS / CPLEX	15 × 10	
Thiruvady et al. [21]	Resource-constrained JSP	MS /CMSA / ACO	Constructive heuristic	MILP + ACO / MS / CMSA	C++ / OpenMP / Gurobi	6 × 20	
Rohaninejad et al. [22]	JSP Parallel Machines	GA_BH / GA_ATC / GA_MLS		MIP+GA	CPLEX	6 × 4	
Dang et al. [23]	JSP Parallel machine	GA		ILP + GA	/ - /IBM ILOG / CPLEX	120 × 6	
Cota et al. [26]	JSP with unrelated parallel machines	-	Multi-objective smart pool search matheuristic + MILP	-	/- / IBM ILOG / CPLEX	15 × 5	
Ahmadian and Salehipour [24]	Just-in-time JSP	-	GT algorithm / SBH / VNS / Relaxation neighborhoods	_	C++ /- / CPLEX	20 × 10	
Son et al. [25]	Bounded-splitting jobs scheduling problem on a single machine in available time windows	GA / TS	AH / HSLPTR / HMAXFR MAAS	TS + MAAS / GA + MAAS	/- /-/ CPLEX	200 × 1	

Chapter 7. Matheuristic Algorithm for Job-Shop Scheduling Problem Using a Disjunctive Mathematical Model.

207 |

Models and algorithms for the optimization of procurement, production and distribution plans in industrial enterprises and supply chains

7.3 Job-Shop Scheduling Problem: Disjunctive Mathematical Formulation

The JSP is an optimization problem in which a set of jobs to manufacture products is assigned to machines at particular times, while attempting to minimize the makespan [17]. Job-shop scheduling is still a problem that has been analyzed since 1954 [29], and is currently tackled given its impact on production costs and efficiency. Nowadays, the JSP remains in essence, but researchers focus on proposing resolution methods that enable enterprises to obtain optimal or near-optimal solutions in shorter computational times to boost the principles of agility, responsiveness and flexibility, all of which are framed within achieving resilience.

The literature proposes different mathematical formulations to model the JSP. Pan [30] presents a comparative analysis of these formulations, namely timeindexed formulation, rank-based formulation and disjunctive formulation. He concluded that the disjunctive model is more efficient because it has the fewest binary variables. More recent studies, such as that presented by Ku and Beck [5], confirm the functionality and effectiveness of the disjunctive model. Although other mathematical formulations exist, they are often combinations or variations of the formulations that we reviewed.

With this background, disjunctive formulation was chosen to model the JSP. To solve the JSP disjunctive problem, we propose combining disjunctive formulation (MILP model) and the GA to generate a matheuristic solution method, whose main aim is to find efficient solutions for large-sized problems and achieve shorter computational times.

The work is then validated by comparing the obtained solutions among the acquired results to solve the JSP disjunctive MILP with a coin-OR Branch & Cut open-source solver.

In this section, we formally define the JSP disjunctive MILP (see Table 7.2). The disjunctive model is presented in Ku and Beck [5]. The JSP is given by a / finite set of *n* jobs or parts, and a finite set *M* of m machines or work centers. For each job *j* \in *J*, the list $(\sigma_1^j, \ldots, \sigma_h^j, \ldots, \sigma_m^j)$ of machines with the processing order of job *j* is provided. Only one job can be processed by each machine at a time. Once started, it must finish processing on that machine without any interruptions.

Chapter 7. Matheuristic Algorithm for Job-Shop Scheduling Problem Using a Disjunctive Mathematical Model.

Sets	
J	set of jobs, $J \in \{1, \dots, n\}$.
М	set of machines $M \in \{1,, m\}$
Parameters	
pij	represents the processing time of job <i>j</i> on machine <i>i</i> .
σ_h^j	denotes the <i>h-th</i> operation of job <i>j</i>
σ_m^j	means the final operation of job <i>j</i>
V	sum of the processing times of all the operations V
	$\sum_{j \in I} \sum_{i \in M} p_{ij}$
Variables	
Xij	start time of job <i>j</i> on machine <i>i</i> .
Zijk	1 if job <i>j</i> is before job <i>k</i> on machine <i>i</i> ; 0 otherwise

Table 7.2. The mathematical notations used in the JSP formulation.

•	~
min	Cmax

Inni ontax		(.)
s.t. $x_{ij} \ge 0$,	$\forall j \in J, i \in M$	(2)
$x_{\sigma_{h}^{j},j} \ge x_{\sigma_{h-1}^{j},j} + p_{\sigma_{h-1}^{j},j},$	$\forall j \in J, h = 2, \dots, m$	(3)
$x_{ij} \ge x_{ik} + p_{ik} - V \cdot z_{ijk},$	$\forall j,k \in J, j < k, i \in M$	(4)
$x_{ik} \ge x_{ij} + p_{ij} - V \cdot (1 - z_{ijk}),$	$\forall j,k \in J, j < k, i \in M$	(5)
$Cmax \geq x_{\sigma_m^j,j} + p_{\sigma_m^j,j},$	$\forall j \in J$	(6)
$z_{ijk} \in \{0,1\}$	$\forall j,k \in J, i \in M$	(7)
$x_{ij} \in \mathbb{Z} \ \forall j, k \in J$		(8)

(1)

The purpose is to obtain a scheduling of jobs on machines to minimize the makespan (*Cmax*). Constraint (2) guarantees that each job's start time equals or exceeds 0. Constraint (3) assures that each operation of a job is carried out in the required order. Disjunctive Constraints (4) and (5) establish that there cannot be two jobs scheduled on one machine at the same time. It is necessary to assign V a large enough value to guarantee the correctness of (4) and (5). The completion time of any operation must not exceed the sum of the processing times of all the operations. Constraint (6) guarantees that the makespan is the longest completion time of the last operation of all the jobs as a minimum [5].

7.4 Materials and Methods

7.4.1 Proposed Matheuristic Approach

The aim of this section is to provide a matheuristic approach to solve the JSP quickly and efficiently, particularly for large-sized problems. To do so, we design a matheuristic approach by applying the metaheuristic procedure (GA) in an LP model (GA-LP). The flowchart of the matheuristic approach is shown in Figure 7.1. All the elements of the proposed matheuristic are separately detailed in the following subsections. The general procedure of the proposed approach is as follows:

Step 1: set the input parameters of the matheuristic (GA-LP);

Step 2: generate the initial population: generate individuals using dispatching heuristic rules and generate individuals randomly;

Step 3: evaluate whether the individuals forming the initial population are feasible;

Step 4: eliminate nonviable individuals and insert the feasible ones into the population;

Step 5: convert the integer chromosome generated by the GA into a binary chromosome;

Step 6: evaluate binary individuals using the LP model;

Step 7: normalize individuals' fitness;

Step 8: select two individuals from the population (parents) and use genetic operators (crossover and mutation);

Step 9 evaluate the chromosomes of the offspring and check if chromosomes are feasible. Then, go to step 3.

Step 10: Are the termination criteria met?

If the termination criteria are met, the solution is obtained; otherwise, go to step 8.

To properly define the proposed matheuristic, we pose a simple JSP example shown in Table 7.3. The data presented in Table 7.3 indicate that there are n = 4 jobs (J_1 , J_2 , J_3 , J_4). The processing order of the jobs on the machines (σ_{mo}) are seen in the second column; for example, J_1 has σ_{11} , in which the first index represents the machine and the second denotes the processing order; i.e., job 1 has to be processed first on machine 1, followed by machines m_2 , m_3 , and m_4 , respectively.

Chapter 7. Matheuristic Algorithm for Job-Shop Scheduling Problem Using a Disjunctive Mathematical Model.

The processing time of job j on machine $i(p_{ij})$ is shown in the third column of this table.

Table 7.3. The job-shop scheduling problem data.						
Job	Processing Order of Jobs	Processing Times				
1	σ11, σ22, σ33, σ44	p11 = 1; p21 = 4; p31 = 2; p41 = 1				
2	σ14, σ23, σ32, σ41	p12 = 2; p22 = 3; p32 = 6; p42 = 2				
3	σ11, σ23, σ32, σ44	p13 = 3; p23 = 7; p33 = 2; p43 = 3				
4	σ14, σ22, σ33, σ41	p14 = 4; p24 = 1; p34 = 5; p44 = 8				



Figure 7.1. Flow chart of the matheuristic algorithm.

7.4.2 Initial Population

Genetic algorithms consist of a set of individuals. Each individual has a chromosome structure composed of genes where the value of each gene represents the jobs performed by each machine. The whole chromosome represents the solution to the problem (see Figure 7.2).



Figure 7.2. Individual's structure.

The GA starts by randomly generating a set of individuals, which is called the initial population. The chromosome of the randomly created individuals can cause the fitness function value to be deficient and can also generate infeasible solutions. Therefore, in the proposed methodology, we use heuristic priority rules to obtain better fitness values by, thus, employing genetic operators so that better solutions can be obtained. In our approach, 80% of the initial population is randomly generated and the rest is generated with the following heuristic priority rules:

- First In First Out—FIFO: the first job to arrive is the first to be served;
- Last In First Out—LIFO: the last job to arrive is the first to be served;
- Shortest Operation Time—SOT: the job that has the shortest processing time is selected. It achieves high flow rate and utilization rates;
- Longest Operation Time—LOT: the job with the longest processing time is selected. The longest operations are considered to be the most important and should be processed first;
- Shortest Remaining Operation Time—SROT: the priority job is the job with the lowest sum of the processing times for all the remaining operations to be performed;
- Longest Remaining Operation Time—LRPT: the priority job is the job with the largest sum of the processing times for all the remaining operations to be performed;
- Less Remaining Operations—LRO: the priority job is that with the fewest remaining operations to be performed;
- Most Remaining Operations—MRO: the priority job is that with the most remaining operations to be performed;
- Work In Next Queue—WINQ: the highest priority is given to the job that would be moved to the machine with the least work to do;

Chapter 7. Matheuristic Algorithm for Job-Shop Scheduling Problem Using a Disjunctive Mathematical Model.

- **Due Date**—**DD**: the job with the closest delivery date is selected;
- **Static Slack -SS:** the job with the shortest time remaining until the delivery date is selected;
- Dynamic Slack—DS: time remaining until the delivery date minus the sum of all the remaining operation times. That with the shortest DS is selected;
- SS/Remaining Operation Time—SS/TPR: Static Slack divided by the sum of the remaining operation times of the remaining operations. The smallest one is selected;
- DS/Remaining Operation Time—DS/TPR: Dynamic Slack divided by the sum of the remaining operation times of the remaining operations. The smallest one is selected;
- SS/Remaining Operations—SS/RO: Static Slack divided by the number of remaining operations. The smallest one is selected;
- DS/Remaining Operations—DS/RO: Dynamic Slack divided by the number of remaining operations. The smallest one is selected.

7.4.3 Feasibility Tester

Randomly generated individuals in the initial population or individuals generated by the crossover and mutation operators may generate infeasible solutions. To avoid the LP having to evaluate infeasible solutions, which makes the matheuristic processing time longer, we present an approach to check the feasibility of individuals. To exemplify the feasibility checker, Table 7.4 shows the one feasible sequence and one infeasible sequence that should be corrected.

Machines	Fea	sible	Sequ	ence	Infe	asible	Sequ	ence	Corr	ected	Sequ	ence
1	J 3	J 1	J 2	J 4	J 3	J 1	J 2	J 4	J 3	J1	J 2	J 4
2	J1	J 2	J 4	J 3	J1	J 2	J 4	J 3	J1	J 2	J 4	J 3
3	J 2	J 3	J 1	J 4	J 2	J 1	J 3	J 4	J 3	J 2	J1	J 4
4	J 2	J 4	J 1	J 3	J1	J 2	J 4	J 3	J 2	J 1	J 4	J 3

Table 7.4. Sequence of jobs on machines.

Infeasibility occurs when jobs do not satisfy the processing order on machines. Table 7.3 shows the processing order of the jobs on machines where, for example, J_1 has the processing order: σ_{11} , σ_{22} , σ_{33} , σ_{44} , i.e., J_1 should be processed first on M_1 and then on machines M_2 , M_3 , M_4 respectively. Figure 7.3 illustrates the feasible solution for all the jobs to fulfill the processing constraints.





Figure 7.3. Feasible solution representation and its Gantt chart.

To exemplify the feasibility tester, we present an unviable solution (see Table 7.4). In Figure 7.4, we represent the solution, but, as observed, the sequence of J_1 does not comply with the processing order. In the same way, J_2 cannot be located as the processing order of J_2 is σ_{14} , σ_{23} , σ_{32} , σ_{41} . This means that it must first be processed on M_4 and then on M_3 , M_2 , M_1 . However, J_1 on machine 4 leads to the processing order not being fulfilled because predecessor J_1 on machine 3 is processed after the job of its successor J_1 on machine 4. This is what causes the infeasibility in the chromosome of the individuals. Therefore, the chromosome must be repaired.



Figure 7.4. Infeasible solution representation and its Gantt chart.

After verifying infeasibility, the feasibility tester changes the location of J_1 and J_2 on machine 4 and, in the same way, the positions of J_1 , J_2 , J_3 on machine 3 (see Table 7.4). After using the feasibility checker in Figure 7.5, the representation of the solution that meets all the precedence constraints is shown.



Figure 7.5. Repaired solution representation and its Gantt chart.

Chapter 7. Matheuristic Algorithm for Job-Shop Scheduling Problem Using a Disjunctive Mathematical Model.

7.4.4 Fitness Function

The individuals in the population are evaluated with the fitness function, which measures the quality of solutions. The evaluation of individuals is performed using the disjunctive relaxed MILP model, i.e., an LP model. Thus, binary variable z_{ijk} represents whether job *j* is prior to job k on machine *i*, and is calculated by the GA. Hence, this variable is fixed to the LP. The binary variable is calculated sequentially with the GA, i.e., while the GA generates individuals, the LP evaluates that the chromosome meets the constraints of the disjunctive model described in Section 7.3.

As individuals have an integer chromosome and variable z_{ijk} is binary in nature, we convert the chromosome. For this purpose, we use the position of each gene as shown in Figure 7.6. Machine 1 has sequences J_3 , J_1 , J_2 , J_4 . We start by looking for the location of the first predecessor job, that is job 1, and this gene is in position 2. Then, we look for the successor job, which is job 2 that meets the condition of the position of the predecessor job being inferior to the successor job. Thus, we assign 1. This same condition is met by predecessor job J_1 and successor job J_4 , but this condition is not met by J_3 , which is in position 1. Table 7.5 shows the result obtained with this process.



Figure 7.6. Position of genes on integer chromosomes.

		,	· · · · ·
i	j	k	Z
1	1	2	1
1	1	4	1
1	2	4	1
1	3	1	1
1	3	2	1
1	3	4	1
2	1	2	1
2	1	3	1
2	1	4	1
2	2	3	1
2	2	4	1

Table 7.5. Precedence of jobs with binary array *z*_{ij}.

Table 7.5. Continued. Precedence of jobs with binary array <i>zij</i> .						
i	j	k	Z			
2	4	3	1			
3	1	4	1			
3	2	1	1			
3	2	4	1			
3	3	1	1			
3	3	2	1			
3	3	4	1			
4	1	3	1			
4	1	4	1			
4	2	1	1			
4	2	3	1			
4	2	4	1			
4	4	3	1			

7.4.5 Selection

Before applying the selection operator, the normalized fitness of the individuals in the population is calculated with the difference between the highest fitness value and the fitness value of each individual.

The selection operator is in charge of deciding which individuals in the population will have the opportunity to reproduce. As a selection operator, we employ a roulette wheel approach [31]. This approach consists of the best individuals, according to their fitness, having the best opportunity to be selected with a uniform selection probability within the range [0...1].

7.4.6 Crossover Operator

The crossover operator used by the GA is the Partially Mapped Crossover Operator. Given the fact that the chromosome of the individuals has an ordered set of permutations, this operator allows for the creation of non-repeated permutation, which it does by choosing two crossover points at random that delimit the area to be inherited. The offspring takes any value of this area from one parent and the rest from the other, which can produce duplicates. To remove duplicates, this method uses a map, on which it checks the relation between the copied sections, and verifies if there is a duplicate gene or a missing gene in the chromosome.
Chapter 7. Matheuristic Algorithm for Job-Shop Scheduling Problem Using a Disjunctive Mathematical Model.

7.4.7 Mutation Operator

In this paper, we use swap mutation. This procedure is as shown in Figure 7.7, where we randomly select two positions from each machine, and then swap the genes at the selected positions to generate a mutated offspring. Our GA employs a mutation probability (pm = 1). As the mutated offspring can give a worse fitness value than the normal offspring [32], we insert the normal and the mutated one into the population if they do not exist after passing the feasibility tester.



Figure 7.7. An example of a swap mutation operator.

7.5 Computational Experiments

The purpose of this study is to evaluate the performance of the non-commercial CBC solver with both the mathematical model and the matheuristic one on large or similar instances to those used by SMEs. To evaluate the performance of the proposed matheuristic, we test the performance of the disjunctive MILP model by using a CBC solver. For this purpose, we generate experiments that consist of a set of different sized problems (20×15 , 20×20 , 30×20). To do so, we use the large-scale instances of Taillard [33], specifically the instances labeled Ta11-Ta13, Ta26-Ta28 and Ta41-Ta43. The dataset can be found in [34]. The JSP is NP-hard for $n \ge 3$ and $m \ge 2$ [5].

The software followed in this research is a non-commercial optimization solver from the Computational Infrastructure for Operation Research (COIN-OR) community called the COIN-OR Branch and Cut Solver [9]. This open-source solver is generally employed for MILP problems. The MILP model and the matheuristic were implemented in Python with the Pyomo package [35]. Experiments were run by an Intel Core i7 2.80 GHz processor (8 GB RAM) in the Ubuntu 20.04.1 LTS operating system.

The GA-LP was run 10 times with the same problem instances. The stopping criterion of the mathematical model and matheuristic is 3600 s. The parameters used in the GA-LP are shown in Table 7.6. The average solutions (*Cmax*) of the GA-

LP and the time in which the methods reached the best solutions are shown in Table 7.7.

The results show that the CBC solver cannot obtain good results for the Ta12, Ta27, Ta42 and Ta43 instances because of its computational difficulty. It is noteworthy that the matheuristic algorithm obtained good solutions in relatively shorter computational times than the CBC solver.

	•
Population size	100
Crossover operator	Partially Mapped Crossover
Selection operator	Roulette wheel
Mutation operator	Swap
Mutation ratio	1

Table 7.6. GA-LP parameters.

Figure 7.8 offers the results obtained with instances Ta11 (20×15), Ta25 (20×20) and Ta41 (30×20). The computational results of CBC for instances Ta12, Ta27, Ta42 and Ta43, are relatively bad, and do not converge to good solutions. For these instances, we changed the stopping criterion to check if the CBC solver can obtain better results, with a computing time of 4 h. We confirm that the result is still the same. The deviation value of these instances is not shown in Figure 7.8 as it cannot be compared with the matheuristic.

		Methods					
Problem	<i>n</i> × m	CBC			Matheuristic (GA-LP)		
		Cmax	CPU (sec)	D1 (%)	Cmax	CPU (sec)	D (%)
Ta11	20 × 15	2219	3567.57	35.55%	1637	198.67	0%
Ta12	20 × 15	-		-	1627	196.09	0%
Ta13	20 × 15	1902	3565.46	15.06%	1653	64.16	0%
Ta26	20 × 20	2483	2673.88	29.32%	1920	152.79	0%
Ta27	20 × 20	-		-	1982	234.88	0%
Ta28	20 × 20	1978	2948.2	3.55%	1910.2	267.64	0%
Ta41	30 × 20	3282	3227.24	32.82%	2471	366.47	0%
Ta42	30 × 20	-			2415	361.36	0%
Ta43	30 × 20	-			2350	373.08	0%

Table 7.7. Comparison of how the proposed approaches perform.

¹ Deviation = [(Obtained Value—Best Value)/Best Value]



Chapter 7. Matheuristic Algorithm for Job-Shop Scheduling Problem Using a Disjunctive Mathematical Model.

Figure 7.8. Experimental results.

From Figure 7.8, it is deduced that the matheuristic produces better results and allows good solutions in short computational times. Table 7.7 shows the results of deviations, where the GA-LP approach provides the best solution for each problem size. GA-LP provides better solutions than CBC, especially with rising computational difficulty. All these results indicate that, by using 20% of the individuals in the initial population with heuristic priority rules, we can improve the efficiency of the proposed method for large instances.

The Figure 7.9 shows the box plots of the two proposed methods for instances Ta11, Ta26 and Ta41. The distribution of the results can be observed in the box plot. The stability of the matheuristic algorithm results is more stable than the CBC. According to the results. we conclude that GA-LP provides better solutions for all the instances in quality and solution time terms.



Figure 7.9. Box plot of all the methods.

7.6 Conclusions and Further Work

The new production paradigms offer plenty of opportunities and challenges as they support the transformation of technology and market conditions for Chapter 7. Matheuristic Algorithm for Job-Shop Scheduling Problem Using a Disjunctive Mathematical Model.

companies. The adaptation of companies to Industry 4.0 means that companies must look for technological tools that help to optimize their manufacturing processes. The adaptation to this technology is determined by adapting different technological tools to the companies. In many cases, SMEs cannot cope with all the technological changes given their cost. Thus, the use of open-source software can act as a valuable tool for companies.

In order to contribute to the literature, this paper presents a matheuristic that combines a GA with a relaxed MILP, solved using a non-commercial solver. We apply different priority heuristic rules to provide faster and more efficient solutions for large problems. The proposed matheuristic achieves good results for large instances. In short computational times, the CBC solver does not offer good results for large instances, but the CBC solver-GA combination provides better solutions in shorter computational times. In the literature, no experiments appear with a non-commercial solver for this instance size. This means that matheuristic can be a useful tool for those SMEs that do not wish to pay for commercial solvers as matheuristic is a useful tool that is easily implemented.

The comparison of the mathematical model, and the matheuristic approach shows that the GA-LP with heuristic priority rules provides good results compared to the CBC results. CBC for the instances of 30 jobs and 20 machines provides the best results in almost 1 h, while the matheuristic approach achieves the best results for these instances in under 400 s. After analyzing the two approaches presented to solve the JSP, we see that the GA-LP is a robust method, is able to achieve good results on instances with different complexities and has a faster convergence rate compared to CBC.

Therefore, future research lines include: improving the GA as the applied genetic operators are standard ones and the operators designed for the concrete problem would perform better; attempting other hybridizations can be performed using: other metaheuristics, such as GRASP, Memetic Algorithm, Particle Swarm Optimization, Tabu Search, Variable Neighborhood Search, and others identified in [10]; testing instances with different job and machine sizes and varying processing times. Other non-commercial solvers can be tested, such as SCIP (Solving Constraint Integer Programs) with commercial solvers like Gurobi and CPLEX.

7.7 References

[1] B. L. MacCarthy, C. Blome, J. Olhager, J. S. Srai, and X. Zhao, "Supply chain evolution – theory, concepts and science," *Int. J. Oper. Prod. Manag.*, vol. 36, no.

12, pp. 1696–1718, 2016, doi: 10.1108/IJOPM-02-2016-0080.

- [2] A. Dolgui, D. Ivanov, S. P. Sethi, and B. Sokolov, "Scheduling in production, supply chain and Industry 4.0 systems by optimal control: fundamentals, stateof-the-art and applications," *Int. J. Prod. Res.*, vol. 57, no. 2, pp. 411–432, 2019, doi: 10.1080/00207543.2018.1442948.
- [3] M. M. Ahmadian, M. Khatami, A. Salehipour, and T. C. E. Cheng, "Four decades of research on the open-shop scheduling problem to minimize the makespan," *Eur. J. Oper. Res.*, vol. 295, no. 2, pp. 399–426, 2021, doi: 10.1016/j.ejor.2021.03.026.
- [4] J. Stastny, V. Skorpil, Z. Balogh, and R. Klein, "Job shop scheduling problem optimization by means of graph-based algorithm," *Appl. Sci.*, vol. 11, no. 4, pp. 1–16, 2021, doi: 10.3390/app11041921.
- [5] W. Y. Ku and J. C. Beck, "Mixed Integer Programming models for job shop scheduling: A computational analysis," *Comput. Oper. Res.*, vol. 73, pp. 165–173, 2016, doi: 10.1016/j.cor.2016.04.006.
- [6] M. A. Boschetti, V. Maniezzo, M. Roffilli, and A. Bolufé Röhler, "Matheuristics: Optimization, simulation and control," *Lect. Notes Comput. Sci. (including Subser. Lect. Notes Artif. Intell. Lect. Notes Bioinformatics)*, vol. 5818 LNCS, pp. 171–177, 2009, doi: 10.1007/978-3-642-04918-7_13.
- [7] M. O. Ball, "Heuristics based on mathematical programming," *Surv. Oper. Res. Manag. Sci.*, vol. 16, no. 1, pp. 21–38, 2011, doi: 10.1016/j.sorms.2010.07.001.
- [8] E.-G. Talbi, "A Unified Taxonomy of Hybrid Metaheuristics with Mathematical Programming, Constraint Programming and Machine Learning," in *Hybrid Metaheuristics*, E.-G. Talbi, Ed. Berlin, Heidelberg: Springer Berlin Heidelberg, 2013, pp. 3–76.
- [9] J. Forrest *et al.*, "coin-or/Cbc: Version 2.9.9," Jul. 19, 2018. https://zenodo.org/record/1317566 (accessed May 27, 2021).
- [10] E. Guzman, B. Andres, and R. Poler, "Models and algorithms for production planning, scheduling and sequencing problems: a holistic framework and a systematic review," *J. Ind. Inf. Integr.*, p. 100287, 2021, doi: 10.1016/j.jii.2021.100287.
- [11] J. Mula, D. Peidro, M. Díaz-Madroñero, and E. Vicens, "Mathematical programming models for supply chain production and transport planning," *Eur. J. Oper. Res.*, vol. 204, no. 3, pp. 377–390, 2010, doi: 10.1016/j.ejor.2009.09.008.
- [12] D. Peidro, J. Mula, R. Poler, and F. C. Lario, "Quantitative models for supply chain planning under uncertainty," *Int. J. Adv. Manuf. Technol.*, vol. 43, no. 3–4, pp. 400–420, 2009, doi: 10.1007/s00170-008-1715-y.
- [13] D. Stindt and R. Sahamie, "Review of research on closed loop supply chain

Chapter 7. Matheuristic Algorithm for Job-Shop Scheduling Problem Using a Disjunctive Mathematical Model.

management in the process industry," *Flex. Serv. Manuf. J.*, vol. 26, no. 1–2, pp. 268–293, 2014, doi: 10.1007/s10696-012-9137-4.

- [14] M. Brandenburg, K. Govindan, J. Sarkis, and S. Seuring, "Quantitative models for sustainable supply chain management: Developments and directions," *Eur. J. Oper. Res.*, vol. 233, no. 2, pp. 299–312, 2014, doi: 10.1016/j.ejor.2013.09.032.
- [15] R. K. Malviya and R. Kant, "Green supply chain management (GSCM): a structured literature review and research implications," *Benchmarking An Int. J.*, vol. 22, no. 7, pp. 1360–1394, Oct. 2015, doi: 10.1108/BIJ-01-2014-0001.
- [16] S. Abdullah and M. Abdolrazzagh-Nezhad, "Fuzzy job-shop scheduling problems: A review," Inf. Sci. (Ny)., vol. 278, pp. 380–407, 2014, doi: 10.1016/j.ins.2014.03.060.
- [17] J. Zhang, G. Ding, Y. Zou, S. Qin, and J. Fu, "Review of job shop scheduling research and its new perspectives under Industry 4.0," *J. Intell. Manuf.*, vol. 30, no. 4, pp. 1809–1830, 2019, doi: 10.1007/s10845-017-1350-2.
- [18] N. Al-Hinai and T. Y. Elmekkawy, "Robust and stable flexible job shop scheduling with random machine breakdowns using a hybrid genetic algorithm," *Int. J. Prod. Econ.*, vol. 132, no. 2, pp. 279–291, 2011, doi: 10.1016/j.ijpe.2011.04.020.
- [19] J. Q. Li, Q. K. Pan, P. N. Suganthan, and T. J. Chua, "A hybrid tabu search algorithm with an efficient neighborhood structure for the flexible job shop scheduling problem," *Int. J. Adv. Manuf. Technol.*, vol. 52, no. 5–8, pp. 683–697, 2011, doi: 10.1007/s00170-010-2743-y.
- [20] X. Li and L. Gao, "An effective hybrid genetic algorithm and tabu search for flexible job shop scheduling problem," *Int. J. Prod. Econ.*, vol. 174, pp. 93–110, 2016, doi: 10.1016/j.ijpe.2016.01.016.
- [21] D. Thiruvady, C. Blum, and A. T. Ernst, "Solution merging in matheuristics for resource constrained job scheduling," *Algorithms*, vol. 13, no. 10, pp. 1–31, 2020, doi: 10.3390/A13100256.
- [22] M. Rohaninejad, Z. Hanzálek, and R. Tavakkoli-Moghaddam, "Scheduling of Parallel 3D-Printing Machines with Incompatible Job Families: A Matheuristic Algorithm," *IFIP Adv. Inf. Commun. Technol.*, vol. 630 IFIP, pp. 51–61, 2021, doi: 10.1007/978-3-030-85874-2_6.
- [23] Q. V. Dang, T. van Diessen, T. Martagan, and I. Adan, "A matheuristic for parallel machine scheduling with tool replacements," *Eur. J. Oper. Res.*, vol. 291, no. 2, pp. 640–660, 2021, doi: 10.1016/j.ejor.2020.09.050.
- [24] M. M. Ahmadian and A. Salehipour, "The just-in-time job-shop scheduling problem with distinct due-dates for operations," *J. Heuristics*, vol. 27, no. 1–2, pp. 175–204, 2021, doi: 10.1007/s10732-020-09458-6.
- [25] T. H. Son, T. Van Lang, N. Huynh-Tuong, and A. Soukhal, "Resolution for

bounded-splitting jobs scheduling problem on a single machine in available time-windows," *J. Ambient Intell. Humaniz. Comput.*, vol. 12, no. 1, pp. 1179–1196, 2021, doi: 10.1007/s12652-020-02162-0.

- [26] L. P. Cota, V. N. Coelho, F. G. Guimarães, and M. J. F. Souza, "Bi-criteria formulation for green scheduling with unrelated parallel machines with sequence-dependent setup times," *Int. Trans. Oper. Res.*, vol. 28, no. 2, pp. 996– 1017, 2021, doi: 10.1111/itor.12566.
- [27] G. Cabrera-Guerrero, C. Lagos, C. Castañeda, F. Johnson, F. Paredes, and E. Cabrera, "Parameter tuning for local-search-based matheuristic methods," *Complexity*, vol. 2017, 2017, doi: 10.1155/2017/1702506.
- [28] F. Verbiest, T. Cornelissens, and J. Springael, "A matheuristic approach for the design of multiproduct batch plants with parallel production lines," *Eur. J. Oper. Res.*, vol. 273, pp. 933–947, 2018, doi: 10.1016/j.ejor.2018.09.012.
- [29] S. M. Johnson, "Optimal two- and three-stage production schedules with setup times included," *Nav. Res. Logist. Q.*, vol. 1, no. 1, pp. 61–68, 1954, doi: https://doi.org/10.1002/nav.3800010110.
- [30] C.-H. PAN, "A study of integer programming formulations for scheduling problems," *Int. J. Syst. Sci.*, vol. 28, no. 1, pp. 33–41, 1997, doi: 10.1080/00207729708929360.
- [31] Z. Jinghui, H. Xiaomin, G. Min, and Z. Jun, "Comparison of performance between different selection strategies on simple genetic algorithms," *Proc. - Int. Conf. Comput. Intell. Model. Control Autom. CIMCA 2005 Int. Conf. Intell. Agents, Web Technol. Internet*, vol. 2, pp. 1115–1120, 2005, doi: 10.1109/cimca.2005.1631619.
- [32] A. Valero-Gomez, J. Valero-Gomez, A. Castro-Gonzalez, and L. Moreno, "Use of genetic algorithms for target distribution and sequencing in multiple robot operations," 2011 IEEE Int. Conf. Robot. Biomimetics, ROBIO 2011, pp. 2718– 2724, 2011, doi: 10.1109/ROBIO.2011.6181716.
- [33] E. Taillard, "Benchmarks for basic scheduling problems," *Eur. J. Oper. Res.*, vol. 64, no. 2, pp. 278–285, 1993, doi: https://doi.org/10.1016/0377-2217(93)90182-M.
- [34] "Job Shop Instances and Solutions." http://jobshop.jjvh.nl/index.php (accessed Dec. 01, 2021).
- [35] W. E. Hart, C. Laird, J.-P. Watson, and D. L. Woodruff, *Pyomo Optimization Modeling in Python*, 1st ed. Springer Publishing Company, Incorporated, 2012.

Chapter 8

A matheuristic approach to production and distribution planning.

Guzman E, Poler R, Andres B. "*A matheuristic approach to production and distribution planning*". Submitted to Advances in Production Engineering & Management. 2022

Abstract:

A number of research studies has addressed supply chain planning from various perspectives (strategical, tactical, operational) and demonstrated the advantages of integrating both production and distribution planning (PDP). The globalisation of supply chains and the fourth industrial revolution (Industry 4.0) mean that companies must be more agile and resilient to adapt to volatile demand, and to improve their relation with customers and suppliers. Hence the growing interest in coordinating production-distribution processes in supply chains. To deal with the new market's requirements and to adapt business processes to industry's regulations and changing conditions, more efforts should be made towards new methods that optimise PDP processes. This paper proposes a matheuristic approach for solving the PDP problem. Given the complexity of this problem, combining a genetic algorithm was tested using the Coin-OR Branch & Cut open-source solver. The computational outcomes revealed that the presented matheuristic algorithm may be used to solve real sized problems.

8.1 Introduction

The globalisation of markets has lee to companies to optimise their processes and resources to remain competitive. Nowadays, optimisation is a relevant factor for improving firms' performance, and for turning the challenges that they face into competitive advantages [1]. One optimal strategy for profits that can minimise a company's total costs is to integrate different business functions, such as purchasing, inventory management, production and distribution [2]. Therefore, it is an important factor for optimizing supply chain enterprises to establish greater integration production and distribution planning (PDP) [3]. The PDP problem is defined by Safaei et al. [4] as a firm's scheduling process to manufacture the right products and to ship the quantities of the right products to the right place at the right time.

Increasing pressure to minimise total production and logistic costs means that supply chain agents are having to re-examine production-distribution policies, and to maximise the use of physico-technological assets [5]. Properly coordinating PDP in supply chains is a challenging problem because companies expand internationally and move to a competitive environment that requires greater collaboration. Efficient supply chain cooperation involves many coordinated decisions being made at several decision levels (e.g., strategical, tactical, operational) about products, financial resources and information.

In this context, the present research work develops an efficient matheuristic approach to solve the integrated PDP problem. A matheuristic algorithm is defined by Boschetti et al. [6] as "the interoperation of metaheuristics and mathematical programming techniques". There are different approaches for combining metaheuristics with exact methods, and each technique has its individual advantages and disadvantages. However, the aim is to benefit from synergy. Several researchers present a taxonomy that classifies this type of cooperation. Table 8.1 shows several approaches by different authors, although some have similar characteristics.

Table 8.1. Classification of matheuristic approaches.						
Approach	Classes					
Cooperation between exact and local search methods. [7] based on Dumitrescu & Stützle [8]	Exact algorithms to browsing through neighbourhoods in local search algorithms. Exact algorithms intended for specific hybrid metaheuristics procedures. Explore boundaries in constructive heuristics. Local searches or constructive algorithms guided by data from integer programming model relaxations. Exact algorithms for smaller problems using solutions from local searches					
Combination between exact	Collaborative combinations. Algorithms exchange information. None is contained in any other. Both procedures can be executed sequentially, interlaced or in parallel	Subclasses Sequential execution Parallel and intertwined execution				
techniques and metaheuristic algorithms [9], [10]	Integrative combinations. One technique is component- integrated into another technique with a master-slave structure. An exact or metaheuristic algorithm can be presented with a master- type structure and at least one integrated slave	Incorporating exact algorithms into metaheuristics Incorporating metaheuristics into exact algorithms				
MASTER-SLAVE" structure with a guiding process and application process [11]	- Metaheuristics operate at the control and guide actions to the - The exact method operates as a the metaheuristic approach	master level and, thereby, e exact technique a master to call/control by				

When designing a matheuristic, the question is, which components can work together to generate an efficient algorithm? Although supplying a collaboration rule does not seem a feasible approach, the matheuristic design involves functionality and architecture. Thus the cooperation level can be ranked according to its hierarchy, as in Table 8.2 [12].

Chapter 8. A matheuristic approach to production and distribution planning.

Table 8.2. Hierarchical matheuristic classification [12].					
Hierarchy	Description				
LRH (low-level relay hybrid)	It depicts hybrid schemes in which a metaheuristic approach is included in an exact approach to improve the search strategy				
LTH (low-level teamwork hybrid)	It describes one search element of a metaheuristic to be replaced with another exact algorithm				
HRH (high-level relay hybrid)	Autonomous algorithms are executed in a sequence. The stage can be either pre-processing or post-processing, i.e., two groups of algorithms (metaheuristics + exact algorithms) are provided with some data in sequence				
HTH (High-level Teamwork Hybrid)	A combination of metaheuristics and exact algorithms that performs a parallel search and cooperate to find relaxed optimal solutions, better lower or upper bounds, optimum subproblem solutions, partial solutions, etc. Metaheuristics and exact algorithms solve partial, specialised or global optimisation problems and exchange helpful information.				

Recent studies, such as that presented by Kumar et al. [13], provide a literature review of the quantitative approaches applied to combined PDP. These authors concluded that the main modelling approaches for this problem type are MILP (mixed integer linear programming), while the main applied solution approaches are those that resort to optimisation software, followed by genetic algorithms (GAs). Computational experiments in small instances use mainly LINGO and CPLEX to solve MILP and Matlab and C++ in large instances to solve heuristic and metaheuristic methods. In this review, we observe that matheuristic methods have not yet been discussed in-depth. As far as the authors know, to date no research has addressed the PDP problem using this type of matheuristic.

In this context, we propose developing a solution strategy for the PDP problem with a mathematical algorithm that is positioned in the hierarchical classification described in Table 8.2 as a High-level Teamwork Hybrid. This strategy is useful because the search space of the MILP model is considered to be too big for a solver to solve it. Therefore, we employ a GA that exchanges information in parallel to the MILP model to diminish the search space.

Given the complexity of PDP problems, they prove difficult when implementing large datasets or solving real SME problems with a MILP model. For

this reason, some companies choose to use commercial solvers for this type of problem. However, some SMEs cannot afford to buy a commercial solver because of its high cost but, as digitisation needs are accelerating, many companies are considering how to be equipped with a digital infrastructure insofar as it does not constrain them and does not cost too much. So those SMEs that have implemented open-source software have made significant savings in technology spending because they do not have to pay annual software licences and have not run the risk of software becoming obsolete when licences expire [14].

Accordingly, this article contributes: (i) a new matheuristic approach to solve the PDP problem; (ii) the matheuristic algorithm was tested and compared to a non-commercial Coin-Branch & Cut (CBC) solver and employs a free open-source operating system (Linux). The proposed approach's effectiveness is proven by solving randomly generated test datasets with real data sizes.

The rest of this article is arranged as so: Section 8.2 briefly presents a literature review about the integrated approach to supply chain PDP; Section 3 offers a mathematical model; Section 8.4 details the matheuristic algorithm for solving the planning-distribution problem; Section 8.5 presents the evaluation of the matheuristic algorithm using large instances to simulate real-life companies. Finally, Section 8.6 defines some conclusions and future research directions.

8.2 Related works

This section reviews the literature about integrating decisions from PDP functions, along with the solution approaches suggested for these problems. This problem has been paid plenty of attention in recent years. Literature reviews like that by Chen [15] indicate several future research lines. One of them states that more effort should be made to create heuristic or metaheuristic methods for this type of problems, which are NP-hard, as there are very few solution algorithms for this type of problems. Years later Fahimnia et al. [16] describe that the use of heuristic, metaheuristic and simulation techniques predominate in the literature, but propose employing new techniques, and suggest having to extend the effectiveness of solution techniques to deal with realistic PDP problems as most techniques have been applied to deal with small- and medium-sized problems. Lastly, the work by Kumar et al. [13] indicates the extensive use of metaheuristic algorithms, GAs and exact methods, but does not reveal the use of matheuristic algorithms.

Accordingly, related work like that of Raa et al. [5] proposes an aggregate PDP model for injection moulding production in the many facilities of a plastics

manufacturer. This MILP is solved by the Gurobi solver for small instances. For large instances, these authors employ an iterative matheuristic that utilises a decomposition heuristic. Bilgen and Çelebi [1] offer a combined simulation and MILP approach for integrated production and distribution problems in the dairy industry. The MILP model is solved with CPLEX and the hybrid approach employs ARENA.

Su et al. [17] propose combining distinct algorithms like the GA and particle swarm optimisation (PSO). The GA comes with a learning scheme, and a hybrid algorithm that combines PSO techniques with the GA and a learning scheme to solve both partner selection and the PDP problem in a manufacturing chain design. Moattar Husseini et al. [18] put forward bi-objective MILP for integrated PDP with manufacturing partners. One of the objectives of this model is to minimise the total cost by covering production, inventory holding purchases from partners and transport-distribution costs. Another objective aims to maximise the quality level of the products that partners supply on the planning horizon. For this problem, they employ LINGO to solve the model in small instances. However, as the problem in large instances is classified as NP-hard, the authors solve it by a Non-Dominant GA II (NSGA-II) and a Multi-Objective PSO (MOPSO) algorithm. The computational results confirm the suitability and practicality of these two algorithms, but the MOPSO algorithm obtains better results in most instances. Devapriya et al. [19] report a PDP problem with a perishable product. The problem is modelled by MILP, is solved with CPLEX, employs a memetic algorithm to solve the problem in large instances, and obtains good solutions in a relatively shorter computational time.

Kazemi et al. [3] put forward a hybrid algorithm that combines a multi-agent system and three metaheuristic algorithms, including a GA, a tabu search and simulated annealing. They propose a MILP model that is solved with LINGO. They employ Matlab to evaluate the hybrid approach. Their results reveal that LINGO better works in small instances, while the hybrid approach delivers better solutions in large instances. In a multifactory supply chain, Gharaei and Jolai [20] study a multi-agent scheduling problem with distribution decisions. To do so, they propose using a MILP formulation to solve the problem with CPLEX by employing small and medium instances. They also develop a multi-objective evolutionary algorithm based on decomposition by combining the Bees algorithm and using Matlab, which well performs in long instances. Marandi and Fatemi Ghomi [21] put forward an integrated production-distribution scheduling problem. They aim to simultaneously find a production schedule and a vehicle routing solution to minimise the sum of delay and transportation costs. They apply CPLEX for small problems and propose a new algorithm for medium and large problems, namely

the Improved Imperialist Competitive Algorithm, which applies a local search algorithm based on a simulated annealing algorithm.

The literature review highlights a growing research tendency to integrate PDP functions. It reveals that companies tend to collaborate with manufacturing partners to better respond to demanding market conditions, and they focus more on their core activities. Interest is shown in heuristic and metaheuristic methods, which are frequently employed to solve these problems with large instances. These instances normally represent the size of the data actually employed by real companies, although different variants of the PDP problem exist. Models tend to be solved mostly with a commercial solver, of which CPLEX is the most widespread. Despite previous works having discussed some combinations of the above algorithms, other combinations have not yet been addressed by the literature, such as those using matheuristic algorithms in practice.

Based on these results, this study considers an integrated PDP problem formulated as a MILP model. As the literature reports the potential effectiveness of GA-based algorithms [18], a combined solution approach with a GA and a mathematical model is herein considered. A non-commercial solver and an opensource operating system are also implemented. The next sections discuss the particulars of the posed problem, its formulation and the solution approach.

8.3 Problem definition

This section offers details of the studied problem and formulates the proposed model. The PDP problem herein contemplated is based on Park [22].

The MILP model takes these assumptions:

- For the production stage: many production plants produce multiple items with a limited capacity per time period. Each product type has a setup cost, while production plants have a limited storage capacity, and produced items are shipped directly to points of sale.
- For the distribution stage: distribution is performed with a fleet of homogeneous vehicles, which are parked in production plants.
- The vehicle movement incurs on: (i) a fixed cost in relation to the depreciation of vehicles, insurance, etc.; (ii) a variable cost according to the transported item, quantity and route.

For points of sale: an item's demand during a period at a point of sale consists of two components: (i) "core demand": the amount of main demand that the point of sale must meet by loyal customers in the long term; (ii) "forecasted demand": the total amount, including core demand. Unmet demand at a point of

sale is considered a stockout (rejected demand) and does not allow deferred demand. Each point of sale can maintain a limited amount of inventory at a very high cost. An overview of the considered problem is demonstrated in Figure 8.1.



Figure 8.1. Illustration of the integrated production and distribution planning problem

8.3.1 Notation

The PDP problem nomenclature is shown below.

Notation	Description
Sets	
i	Index of plants $i \in \{1,, I\}$
j	Index of points of sale, $j \in \{1,, J\}$
k	Index of products (parts) $k \in \{1,, K\}$
t	Index of time periods $t \in \{1,, T\}$
Parameters	
C _{ik}	cost of producing 1 unit of product <i>k</i> in plant <i>i</i>
Sik	cost of setup for product <i>k</i> in plant <i>i</i>
Oik	time of producing 1 unit of product <i>k</i> in plant <i>i</i>
Uik	time of setup for product <i>k</i> in plant <i>i</i>
hpik	unit holding cost per period for product k in plant <i>i</i>
Li	production capacity per period of plant <i>i</i>
dijk	cost of transporting 1 unit of product <i>k</i> from plant <i>i</i> to point of sale <i>j</i>

ire

Table 8.3. Continued. The MILP model nomenclature.					
Notation	Description				
Parameters					
g	fixed cost per vehicle				
В	capacity per vehicle				
Ejkt	core demand of product <i>k</i> at point of sale <i>j</i> during period <i>t</i>				
F _{jkt}	forecasted demand of product <i>k</i> at point of sale <i>j</i> during period <i>t</i>				
P jk	price of sale of 1 unit of product <i>k</i> at point of sale <i>j</i>				
hrjk	unit holding cost per period for product <i>k</i> at point of sale <i>j</i>				
Wrj	inventory capacity of point of sale <i>j</i>				
Vjk	unit stockout cost of item <i>k</i> at point of sale <i>j</i>				
Variables					
Xikt	amount of product <i>k</i> produced at plant <i>i</i> during period <i>t</i>				
q ijkt	amount of product <i>k</i> transported from plant <i>i</i> to point of sale <i>j</i> during period <i>t</i>				
Y _{ikt}	1 if product <i>k</i> is produced a plant <i>i</i> during period <i>t</i> ; 0 otherwise				
ap_{ikt}	inventory level of product k at plant i during period t				
ar _{jkt}	inventory level of product <i>k</i> at point of sale <i>j</i> during period <i>t</i>				
Zijt	number of vehicles needed for distribution from plant i to point of sale j during period t				

The objective function maximises sales revenues at points of sale, minus the costs of setup, production and inventory at plants, the costs of inventory and stockout at points of sale, and the costs of vehicles and transport.

$$MaxZ = \sum_{j} \sum_{k} p_{jk} \cdot \sum_{t} \left(ar_{jkt-1} + \sum_{i} q_{ijkt} - ar_{jkt} \right) - \left(\sum_{i} \sum_{k} c_{ik} \cdot \sum_{t} x_{ikj} + \sum_{i} \sum_{k} S_{ik} \cdot \sum_{t} Y_{ikt} + \sum_{i} \sum_{k} hp_{ik} \right) \cdot \sum_{t} ap_{ikt} \right) - \left(\sum_{j} \sum_{k} hr_{jk} \cdot \sum_{t} ar_{jkt} + \sum_{j} \sum_{k} v_{jk} \cdot \sum_{t} \left(F_{jkt} - ar_{jkt-1} \right) + \sum_{t} q_{ijkt} + ar_{ijkt} \right) \right) - \left(g \cdot \sum_{i} \sum_{j} \sum_{t} z_{ijt} + \sum_{i} \sum_{j} \sum_{k} d_{ijk} \cdot \sum_{t} q_{ijkt} \right)$$

$$(1)$$

Subject to:

Material flow constraints

233 |

$$ap_{ikt} = ap_{ikt-1} + x_{ikt} - \sum_{j} q_{ikt} \qquad \forall i, k, t$$
(2)

$$ar_{jkt-1} + \sum_{i} q_{ijkt} - ar_{jkt} \ge E_{jkt} \qquad \forall j, k, t$$
(3)

$$ar_{jkt-1} + \sum_{i} q_{ijkt} - ar_{jkt} \le F_{jkt} \qquad \forall j, k, t$$
(4)

Constraint (2) guarantees the inventory of all products in each plant at the end of every period. Constraint (3) ensures meeting "core demand" at each point of sale per product during each time period. Constraint (4) ensures that the demand served for any product at any point in time at any point of sale never exceeds the expected demand ("forecast demand").

Physical resource limitations

$\sum_{k} o_{ik} \cdot x_{ikt} + \sum_{k} u_{ik} \cdot Y_{ikt} \leq L_i$	$\forall i, t$	(5)
$x_{ikt} \leq = M \cdot Y_{ikt}$	$\forall i, k, t$	(6)

$$\sum_{k} ar_{jkt} \le Wr_j \qquad \qquad \forall j, t \tag{7}$$

$$z_{ijt} \sum_{k} \frac{q_{ijkt}}{B} \qquad \qquad \forall i, j, t \tag{8}$$

$$ap_{ik0} = 0, ar_{ik0} = 0 \qquad \qquad \forall i, j, k \tag{9}$$

Constraint (5) guarantees that, per plant during each period, the capacity consumption due to the processing and preparation times of processed items never exceeds the plant's available production capacity. Constraint (6) ensures that if a quantity of a certain product is produced in a plant during a period, a setup of this product is necessary. Constraint (7) ensures that the amount of products stored at a point of sale during every period must never exceed the point of sale's storage capacity. Constraint (8) computes the number of vehicles required to transport products from every plant to each point of sale during all periods. Constraint (9) represents the initial inventory levels in plants and at points of sale.

$$\begin{aligned} & x_{ikt}, q_{iikt}, a p_{ikt}, a r_{ikt}, z_{iit} \in \mathbb{Z} \\ \end{aligned} \qquad \forall i, j, k, t \end{aligned} \tag{10}$$

Constraints (10) and (11) indicate the binary nature of Y_{ikt} and the integer nature of some variables.

8.4 Matheuristic solution method

The PDP problem is a complex one to solve given the number of integer variables that corresponds to produced and transported products, the inventory level in the plant and at points of sale, and the vehicles needed for distribution, plus the binary

variable that indicates in which plants products are produced. Given the difficulty of this problem, a solution methodology is offered and describes how the GA is combined with the MILP model to evaluate the solutions for the PDP problem. Figure 8.2 illustrates the flow chart of the matheuristic approach. The particulars of elements are described below.



Figure 8.2. The flow chart of the proposed matheuristic approach

8.4.1 Initial population

Each individual in the population corresponds to binary decision variable Y_{ikt} , which takes 1 if product k is produced in plant i during period t, and 0 otherwise. An individual's length is a one-dimensional matrix of size $I \times K \times T$ (multiplication of the quantities of every index). The population takes a binary structure and is generated randomly from a uniform distribution with a 50% probability of 1 appearing on an individual's chromosome. The computational results indicate that fewer infeasible individuals are generated if this probability is applied. Population size N_{pop} equals 10, so we employ this small size to increase the GA's speed. Koljonen and Alander [23] confirm that a small population size increases the optimisation speed to a certain extent. We prove that using this population size suffices to obtain good solutions.

8.4.2 Evaluation function

The fitness function measures the quality of an individual in the population. The problem looks for solutions that maximise the benefits that the objective function represents. The PDP problem's computational difficulty focuses mostly on binary variable Y_{ikt} , which refers to the decisions made about which product to produce in which plant. This means that the GA is in charge of producing a suitable binary chromosome with equal dimensions to the binary variable.

As this binary chromosome corresponds to each individual in the population, the evaluation of each individual is made by formulating the mathematical model. The computational and execution times of a MILP *versus* a linear programming (LP) model are longer given the SIMPLEX algorithm's computational efficiency *versus* the algorithms dedicated to solve problems with integer or binary variables, along with the problem's difficulty, which is considered NP-hard. The proposed MILP model comprises one binary variable and five integer variables. Thus, to improve matheuristic performance, we apply MILP model relaxation. The MILP relaxation to obtain LP is given by transforming integer variables into continuous variables, and by transforming the binary variable into data.

At this time the solver is in charge of solving LP and the GA is responsible for supplying the binary variable. The binary variable of the GA is fixed to LP. Thus, when executing the matheuristic, it can be quickly solved even for very large problems. In our experiments, on average LP obtains better results than MILP by 3.84%. Thus, to obtain a final result, we employ the best binary chromosome obtained during the evaluation process and launch MILP to gain a final result. This is explained in Section 8.4.6.

8.4.3 Selection

In the selection stage, a set of individuals from the current population is chosen to be used as the parents for the crossing stage. The roulette wheel approach [24] is taken to select the individuals with the best fitness values in accordance with the uniform probability of selection distributed over the range [0...1], and the worst individuals are eliminated from one generation to another so that the best individuals are more probably selected.

8.4.4 Crossover

The single point crossover technique [25] is applied. Two parents (P1 and P2) are selected by the fortune roulette wheel selection technique. Then the P1 and P2 chromosomes are cut at a point that is randomly generated and a new offspring (OF) is generated with the genetic information of its parents, as illustrated in Figure 8.3.



Figure 8.3. Single-Point Crossover

8.4.5 Mutation

Mutation allows the GA to explore a bigger region of the ranges of potential solutions by including random genetic changes, which are produced by introducing variations into individuals, and thus allowing the GA to not fall into local optima [26]. The swap mutation operator is implemented here. This mutation method randomly selects two genes from offspring and then exchanges the gene content in its offspring, as shown in Figure 8.4.



Figure 8.4. The Swap mutation operator

The offspring that undergo mutation are selected with uniform probability Pm=1. This means that all offspring are mutated but, in order not to lose the normal offspring, both the normal and mutated offspring remain in the resulting population. This avoids losing a good solution obtained by the crossing process as a mutation can provide a worse fitness value [27]. Then the two offspring (normal and mutated) are included in the population by replacing the two worst individuals in that population to leave a constant population size.

8.4.6 The matheuristic approach procedure

The best individual with the best fitness is selected at the end of the calculation time (stopping criterion) of the matheuristic GA. With this binary chromosome, MILP is launched.

In the evaluation function, MILP is relaxed to LP to reduce computational effort and to obtain a sufficiently good solution. To obtain a definitive solution, MILP is used, i.e., by removing relaxation and employing the binary chromosome of the individual with the best fitness provided by the GA.

The binary chromosome is set at the Yikt variable of MILP. This means that the binary chromosome becomes a parameter for the model. Then the MILP model is launched, the solver solves the integer variables and the GA provides the binary variable.

The advantage of using matheuristic, and not directly using the solver, lies in the search space of the MILP model being significantly reduced when a matheuristic is employed to deal with binary variables.

8.5 Numerical experiments

In the present section, a set of synthetic data is used to evaluate our approach. In this type of problem, real data sets are generally large, which renders it unsolvable with many plants, products, outlets and periods. To assess how the matheuristic and the non-commercial solver perform, in computational tests we apply large instances, which are randomly generated according to the outlined parameters and formulations in Park (2005). To create these data, we created a synthetic data generator, which appears at: https://bit.ly/3qBn363.

Park [22] analysed big datasets with similar characteristics to those in Table 8.4. Park used CPLEX to solve MILP but did not present any results for large instances because of the problems' computational difficulty, which is why he applied this solver only for small instances. We use the same data for plant size

(5), points of sales (from 40 to 65), products (5) and periods (from 10 to 12), as presented in the above-mentioned study.

The software followed in this research is a non-commercial optimisation solver from the Computational Infrastructure for Operation Research (COIN-OR) community called COIN-OR Branch and Cut Solver (CBC) [28]. This open-source solver is generally employed for MILP problems. The MILP model and matheuristic were implemented in Python with the Pyomo package [29]. Experiments were run by an Intel Core i7 2.80 GHz processor (6 GB RAM) in an Ubuntu 20.04.1 LTS operating system.

The performances of the matheuristic approach and the proposed MILP through computational experiments were compared to one another to identify the best performing method. The resulting GAP of the MILP solved by CBC and the matheuristic is calculated as indicated in equation (12).

$$GAP(\%) = \frac{|UB - Best_{sol}|}{|Best_{sol}|}$$
(12)

Where UB indicates the upper solver bound, and *Best_{sol}* refers to the best solution generated by either the mathematical model or the matheuristic approach.

8.5.1 Experimental results

In order to demonstrate the proposed approach's efficiency and performance, the computational experiments with different large instances are provided. Table 8.4 compares the solution's efficiency among the solutions obtained by solving MILP with CBC and the matheuristic one with CBC. The first column in this table denotes the name of the instance, followed by the number of plants (*I*), points of sale (*J*), products (*K*) and periods (*T*). For all the instances, the applied criterion is the same calculation time that corresponds to 14,400 seconds. We executed the matheuristic algorithm 20 times for each instance in order to evaluate and avoid atypical performance.

Table 8.4. Performance comparison between CBC and Matheuristic										
In- stance		Problem				CBC			Matheuristic	
	1	J	K	Т	Total profit	Upper bound	GAP	Total profit	GAP	
11	5	40	5	12	Unfeasi- ble solu- tion found	5746740.5	-	5117847	12.28 %	
12	5	50	5	10	5873321	6876383.6	17.08 %	6347862	8.33 %	
13	5	45	5	12	Unfeasi- ble solu- tion found	6785206.6	-	6157783	10.18 %	
14	5	60	5	10	6643133	8037979.9	21.00 %	7487111	7.36 %	
15	5	50	5	12	Unfeasi- ble solu- tion found	6932259.4	-	6294622	10.13 %	
16	5	65	5	10	Unfeasi- ble solu- tion found	8272162.9	-	7640947	8.26 %	

Chapter 8. A matheuristic approach to production and distribution planning.

The MILP solved by CBC obtained solutions for two (I2, I4) of the six instances, but it was unable to find optimal or good solutions. The matheuristic gave good solutions for all the instances. Figure 8.5 illustrates the total profit obtained by matheuristic and CBC. I2a and I4a show how the matheuristic approach evolves and converges towards good solutions, along with how CBC performs at around 7,200 computation seconds, while I4b and I4b show the behaviour of both the matheuristic and CBC at 14,400 processing seconds. For the I2 instance, CBC gave a feasible solution at 5,152.76 seconds (see Figure 8.5) with GAP = 17.10%. GAP improved up to 14,400 seconds by 0.02%. For the matheuristic for the same instance, it obtained feasible solutions from 71.05 seconds, with GAP less than 10% at 1,880 seconds (see Figure 8.6).



Figure 8.5. Time spent by matheuristic to find a feasible solution.

With the I4 instance, CBC performance visibly improves. It obtains solutions in 4,097 seconds and becomes the best solution in 5,538 seconds (see Figure 8.5). Matheuristic better performs than CBC by reaching feasible solutions in shorter computational times (see Figure 8.5) and reaches a GAP below 10% after 2,735 seconds (see Figure 8.6). This means that matheuristic outperforms CBC by 13.64%.



Chapter 8. A matheuristic approach to production and distribution planning.

Figure 8.6. Comparison of matheuristic and MILP-CBC performance.

The complexity of the instances and the size of the problem mean that CBC is unable to find feasible solutions. Nevertheless, the combination of a GA with CBC gives better results with feasible solutions in shorter computational times. A matheuristic's efficiency is linked with the solver's speed because the solver is in charge of evaluating solutions by the GA's evaluation function. Moreover, as the evaluation function is the principal component of GAs [30], employing a noncommercial solver combined with a GA offers good results, as herein shown, and the matheuristic is more efficient in solving problems with many variables and parameters, and can be a useful alternative for large instances. When utilising a non-commercial solver like CBC, a matheuristic can support the solver to find better solutions.

In order to further demonstrate the efficiency of the proposed matheuristic, we compare it to Gurobi 9.1.1, i.e., the MILP and LP of the matheuristic are solved with Gurobi. We employ the same aforementioned computational conditions and apply a processing time of 14,400 seconds. The computational results given by Gurobi are better than those of CBC. Thus Gurobi obtains feasible solutions in all the instances in much shorter solution times. However, the matheuristic is better for achieving a lower GAP than Gurobi in all instances (see Figure 8.7).







Chapter 8. A matheuristic approach to production and distribution planning.

Figure 8.7. Comparison of matheuristic and Gurobi performance.

8.6 Conclusion

The PDP problem has long since been studied for the practical applications that it can offer industry. One such case is enterprises with different manufacturing plants in several locations, perhaps in the same city or country, or in others, which must decide the amount of products to be produced in plants, the quantity of products to be stored in plants during each period, the number of products to deliver to various points of sale according to product demand and the inventory of finished products at points of sales. Although several resolution techniques have been used for this problem and its variations, heuristic and metaheuristic algorithms can provide excellent results as combined or hybrid approaches. Likewise, combinations between metaheuristic techniques and exact approaches can offer better results for real-life problems because these combinations make the most of the benefits of both techniques [7]. In this context, the present paper intends to solve the PDP problem in real-world large data sizes. The problem is modelled as MILP, and a matheuristic solution approach is presented that combines a GA and an LP model.

Computational tests were performed on a large dataset capable of simulating real-world problems. The development of this approach stems from SMEs having to use open-source tools and the need to digitise companies because they must compete in today's market. Many SMEs cannot have access to software

with paid licenses, due to the high-costs they may have to adapt the software to the needs of the enterprises. The main research contribution is about applying a matheuristic approach by employing a non-commercial solver (CBC). We also tested the performance of the non-commercial solver with an NP-hard MILP model. The computational tests run on different instances showed that our approach offers markedly improved results than the exact method. Matheuristic obtained competitive results in a short time. When solving MILP, CBC is unable to acquire feasible solutions for four of the six computed instances. However with our proposed matheuristic, and by also using CBC for solving relaxed LP, our results were good for all instances. Matheuristic can perform better even when using a commercial solver like Gurobi. Therefore, matheuristic can offer a real technical and economical application and is affordable mainly for SMEs that cannot pay a commercial solver or do not recurrently resort to one. This approach is feasible thanks to the proposed model's simplicity. The matheuristic also offers the benefit of making the most of the solver's features, regardless of them being commercial or not, because the matheuristic improves the solver's performance.

Other metaheuristics can be used for future work, such as memetic algorithms, ant colony optimisation or tabu search, and other highly complex problems can also be tested. Other genetic operators can be evaluated, or specific heuristics can be used to improve the GA's performance.

8.7 References

- B. Bilgen and Y. Çelebi, "Integrated production scheduling and distribution planning in dairy supply chain by hybrid modelling," *Ann. Oper. Res.*, vol. 211, no. 1, pp. 55–82, 2013, doi: 10.1007/s10479-013-1415-3.
- [2] V. A. Armentano, A. L. Shiguemoto, and A. Løkketangen, "Tabu search with path relinking for an integrated production distribution problem," *Comput. Oper. Res.*, vol. 38, no. 8, pp. 1199–1209, 2011, doi: 10.1016/j.cor.2010.10.026.
- [3] A. Kazemi, M. H. F. Zarandi, and M. Azizmohammadi, "A hybrid search approach in production-distribution planning problem in supply chain using multi-agent systems," *Int. J. Oper. Res.*, vol. 28, no. 4, pp. 506–527, 2017, doi: 10.1504/IJOR.2017.082611.
- [4] A. S. Safaei, S. M. Moattar Husseini, R. Z.-Farahani, F. Jolai, and S. H. Ghodsypour, "Integrated multi-site production-distribution planning in supply chain by hybrid modelling," *Int. J. Prod. Res.*, vol. 48, no. 14, pp. 4043–4069, 2010, doi: 10.1080/00207540902791777.

- [5] B. Raa, W. Dullaert, and E. H. Aghezzaf, "A matheuristic for aggregate production-distribution planning with mould sharing," *Int. J. Prod. Econ.*, vol. 145, no. 1, pp. 29–37, 2013, doi: 10.1016/j.ijpe.2013.01.006.
- [6] M. A. Boschetti, V. Maniezzo, M. Roffilli, and A. Bolufé Röhler, "Matheuristics: Optimization, simulation and control," *Lect. Notes Comput. Sci. (including Subser. Lect. Notes Artif. Intell. Lect. Notes Bioinformatics)*, vol. 5818 LNCS, pp. 171–177, 2009, doi: 10.1007/978-3-642-04918-7_13.
- [7] L. Jourdan, M. Basseur, and E. G. Talbi, "Hybridizing exact methods and metaheuristics: A taxonomy," *Eur. J. Oper. Res.*, vol. 199, no. 3, pp. 620–629, 2009, doi: 10.1016/j.ejor.2007.07.035.
- [8] I. Dumitrescu and T. Stützle, "Combinations of Local Search and Exact Algorithms," in *Applications of Evolutionary Computing*, 2003, pp. 211–223.
- [9] G. R. Raidl and J. Puchinger, "Combining (Integer) Linear Programming Techniques and Metaheuristics for Combinatorial Optimization," in *Hybrid Metaheuristics: An Emerging Approach to Optimization*, C. Blum, M. J. B. Aguilera, A. Roli, and M. Sampels, Eds. Berlin, Heidelberg: Springer Berlin Heidelberg, 2008, pp. 31–62.
- [10] J. Puchinger and G. R. Raidl, "Combining Metaheuristics and Exact Algorithms in Combinatorial Optimization: A Survey and Classification," pp. 41–53, 2005, doi: 10.1007/11499305_5.
- M. Caserta and S. Voß, "Metaheuristics: Intelligent Problem Solving," in Matheuristics: Hybridizing Metaheuristics and Mathematical Programming, V. Maniezzo, T. Stützle, and S. Voß, Eds. Boston, MA: Springer US, 2010, pp. 1–38.
- [12] E. G. Talbi, "Combining metaheuristics with mathematical programming, constraint programming and machine learning," *Ann. Oper. Res.*, vol. 240, no. 1, pp. 171–215, 2016, doi: 10.1007/s10479-015-2034-y.
- [13] R. Kumar, L. Ganapathy, R. Gokhale, and M. K. Tiwari, "Quantitative approaches for the integration of production and distribution planning in the supply chain: a systematic literature review," *Int. J. Prod. Res.*, vol. 58, no. 11, pp. 3527–3553, 2020, doi: 10.1080/00207543.2020.1762019.
- [14] A. Reshad and S. Sinha, "Open Source Software Solution for Small and Medium Enterprises," Int. J. Comput. Sci. Eng., vol. 8, no. 6, pp. 86–90, 2020, doi: https://doi.org/10.26438/ijcse/v8i6.8690.
- [15] Z. L. Chen, "Integrated production and outbound distribution scheduling: Review and extensions," Oper. Res., vol. 58, no. 1, pp. 130–148, 2010, doi:

10.1287/opre.1080.0688.

- B. Fahimnia, R. Z. Farahani, R. Marian, and L. Luong, "A review and critique on integrated production-distribution planning models and techniques," *J. Manuf. Syst.*, vol. 32, no. 1, pp. 1–19, 2013, doi: 10.1016/j.jmsy.2012.07.005.
- [17] W. Su, S. X. Huang, Y. S. Fan, and K. L. Mak, "Integrated partner selection and production-distribution planning for manufacturing chains," *Comput. Ind. Eng.*, vol. 84, pp. 32–42, 2015, doi: 10.1016/j.cie.2015.01.015.
- [18] Z. Moattar Husseini, B. Karimi, S. M. Moattar Husseini, and S. H. Ghodsipour, "Multi-objective integrated production distribution planning concerning manufacturing partners," *Int. J. Comput. Integr. Manuf.*, vol. 28, no. 12, pp. 1313–1330, 2015, doi: 10.1080/0951192X.2014.972460.
- [19] P. Devapriya, W. Ferrell, and N. Geismar, "Integrated production and distribution scheduling with a perishable product," *Eur. J. Oper. Res.*, vol. 259, no. 3, pp. 906–916, 2017, doi: 10.1016/j.ejor.2016.09.019.
- [20] A. Gharaei and F. Jolai, "A multi-agent approach to the integrated production scheduling and distribution problem in multi-factory supply chain," *Appl. Soft Comput. J.*, vol. 65, pp. 577–589, 2018, doi: 10.1016/j.asoc.2018.02.002.
- [21] F. Marandi and S. M. T. Fatemi Ghomi, "Integrated multi-factory production and distribution scheduling applying vehicle routing approach," *Int. J. Prod. Res.*, vol. 57, no. 3, pp. 722–748, 2019, doi: 10.1080/00207543.2018.1481301.
- [22] Y. B. Park, "An integrated approach for production and distribution planning in supply chain management," *Int. J. Prod. Res.*, vol. 43, no. 6, pp. 1205–1224, 2005, doi: 10.1080/00207540412331327718.
- [23] J. Koljonen and J. T. Alander, "Effects of population size and relative elitism on optimization speed and reliability of genetic algorithms," *Publ. Finnish Artif. Intell. Soc.*, pp. 54–60, 2006.
- [24] Z. Jinghui, H. Xiaomin, G. Min, and Z. Jun, "Comparison of performance between different selection strategies on simple genetic algorithms," *Proc. Int. Conf. Comput. Intell. Model. Control Autom. CIMCA 2005 Int. Conf. Intell. Agents, Web Technol. Internet*, vol. 2, pp. 1115–1120, 2005, doi: 10.1109/cimca.2005.1631619.
- [25] Z. C. Dagdia and M. Mirchev, *When Evolutionary Computing Meets Astro- and Geoinformatics*. Elsevier Inc., 2020.

- [26] S. Wang and M. Liu, "A genetic algorithm for two-stage no-wait hybrid flow shop scheduling problem," *Comput. Oper. Res.*, vol. 40, no. 4, pp. 1064– 1075, 2013, doi: 10.1016/j.cor.2012.10.015.
- [27] A. Valero-Gomez, J. Valero-Gomez, A. Castro-Gonzalez, and L. Moreno, "Use of genetic algorithms for target distribution and sequencing in multiple robot operations," 2011 IEEE Int. Conf. Robot. Biomimetics, ROBIO 2011, pp. 2718–2724, 2011, doi: 10.1109/ROBIO.2011.6181716.
- [28] J. Forrest *et al.*, "coin-or/Cbc: Version 2.9.9," Jul. 19, 2018. https://zenodo.org/record/1317566 (accessed May 27, 2021).
- [29] W. E. Hart, C. Laird, J.-P. Watson, and D. L. Woodruff, *Pyomo Optimization Modeling in Python*, 1st ed. Springer Publishing Company, Incorporated, 2012.
- [30] Z. Michalewicz, "The Significance of the Evaluation Function in Evolutionary Algorithms BT - Evolutionary Algorithms," L. D. Davis, K. De Jong, M. D. Vose, and L. D. Whitley, Eds. New York, NY: Springer New York, 1999, pp. 151–166.

Chapter 9 Conclusions.

Abstract:

This chapter presents the conclusions that derive from this doctoral thesis, as well as its contributions, to meet the overall objective set out in Chapter 1: design and implement new models and algorithms to calculate replenishment, production and distribution plans in industrial companies. Finally, future research lines are presented.

Chapter 9. Conclusions.

9.1 Main contributions

This doctoral thesis focuses on the main optimisation challenges in replenishment, production and distribution planning. To tackle the vast field of this topic, we set five specific objectives, which are listed in the introductory chapter, all of which have been successfully met. The different chapters of the thesis derive from these objectives, and the main contributions and results are presented below:

9.1.1 The replenishment, production and distribution planning holistic conceptual framework

The first objectives (*O1, O2*) of this doctoral thesis are to categorise the types of plans and to identify the different types of models and algorithms used for replenishment, production and distribution planning problems, and to know the current research state. To do so these aspects are discussed in Chapters 2 and 3.

Replenishment, production and distribution planning problems are conceptualised to identify the main dimensions, characteristics, formulations and resolution methods used in the different types of supply chains to evaluate the current state of these problems based on a tertiary study (analysis of literature reviews).

Subsequently, a holistic framework was generated, which includes the different works found in the literature on production planning, scheduling and sequencing. The proposed framework considers several categories to provide a taxonomy to classify the different planning problems studied in this thesis, including: (i) the planning aggregation level; (ii) the decision-making level associated with the plan; (iii) the type of models used to represent the plan; (iv) the objectives that characterise each modelling approach; (v) the resolution techniques followed; (vi) the software used for the computational calculation of plans; (vii) the areas and sectors of the application of production plans; (viii) the level of plans integration along the supply chain with other partners; (ix) the experiments performed to test real cases; (x) the size of the data with which planning problems are solved; (xi) the quality of the obtained solutions.

The state of the art developed in this thesis addresses an extremely wide variety of considerations related to the planning problem that has not been addressed to date, which is intended to answer the following research question:

RQ1. How can replenishment, production and distribution planning problems and solution methods be categorised?

In response to RQ1, Chapters 2 and 3 are presented from which we can discern those numerous conceptual frameworks are found in the literature, many of which generate a very broad theoretical basis, but have yet to be validated. It is also evident that more and more approaches take into account aspects like sustainability and the environment, while topics such as the rolling horizon are less recurrent. Lots of papers present mathematical models, and mixed integer linear programming and analytical models are the most frequently used ones, while simulation models and their combination with other methods are less frequent. Similarly, heuristic or metaheuristic methods are widely used, and CPLEX is the most frequently used solver. The systematic literature review generally highlights that there are still a few works that address planning process problems based on real cases. Nevertheless, the review allowed many of the presented model solving methods that can be used in different industrial sectors to be identified.

9.1.2 Methodology for selecting algorithms to solve planning problems

To fulfil O3, and in response to the following research question, Chapter 4 is presented.

RQ2. How can the most suitable algorithm or solution method be selected to solve replenishment, production and distribution planning according to its complexity?

Nowadays, algorithm selection is a very difficult task for companies because it involves a lot of mathematical, statistical and programming knowledge. The literature offers several algorithms and different techniques for solving planning problems, where the performance of algorithms has been tested in a defined area. However, companies that wish to use the available algorithms do not know which one is suitable and for which type of problem because the performance of an algorithm varies depending on the type of problem and the expected solution. Studies show that there is no single algorithm that outperforms all algorithms for all problem cases.

The algorithm selection process generally involves an experimental phase which, for small- and medium-sized enterprises (SMEs), can prove to be an obstacle because they have fewer resources to invest in commercial solvers or specific software for planning. The experimental phase implies programming and computing a set of algorithms and making comparisons of their performance,

Chapter 9. Conclusions.

which is time-consuming for programmers, as well as the additional time needed for evaluating different datasets, which also requires significant computing resources. Then there are the difficulties of replicating an algorithm or mathematical model from the literature.

Given these problems, we generated a decision-making support tool based on the fuzzy TOPSIS method. This approach provides companies with a method and allows them to select an algorithm or solution method given a portfolio of them. The decision-making tool is based on four dimensions and 13 different decision criteria are defined according to these dimensions. The four dimensions are related to: (i) the type of problem and its characteristics; (ii) the planner's degree of knowledge related to scheduling and mathematical modelling; (iii) the company's endowment in mathematical modelling software and knowledge of its operation; (iv) the expected gap performance that the planner is willing to achieve by solving the planning problem.

With this background, the models and algorithms proposed in the literature, and the new ones developed within the scope of the present doctoral thesis with our novel approach based on Fuzzy TOPSIS for algorithm selection, a solution method can be selected from a set of methods or algorithms suitable to solve planning problems. Thus, it is possible to choose not only one algorithm, but others that offer similar solutions at the same time. The results of this decisionmaking support tool guide companies to decide whether to use a commercial or non-commercial algorithm or solver. In this way, companies can determine whether to invest in a commercial solver or use an open-source mathematical modelling or algorithm programming software and, at the same time, can understand planning staff's training needs with respect to the different types of software.

9.1.3 Models and algorithms for replenishment, production and distribution planning

In order to meet objectives O4 - O5 and in response to:

RQ3: What new algorithms should be developed to solve real replenishment, production and distribution planning problems?

A number of mathematical models and matheuristic algorithms were formulated to bridge the gaps encountered in replenishment, production and distribution planning problems from both individual enterprise perspective and a collaborative perspective, i.e., in which replenishment plans from the supplier affect the manufacturer's production plans.
Models and algorithms for the optimisation of replenishment, production and distribution plans in industrial enterprises.

We firstly formulate a mixed integer linear programming (MILP) model for a combined replenishment and production planning problem (see Chapter 5), in which we analyse the sizing and batch scheduling to manufacture plastic components. Our model is based on the case of a second-tier supplier in the automotive supply chain. This supplier produces plastic automotive components. To do so, it uses plastic granules that are placed inside injection moulding machines by means of moulds. The main characteristics of the addressed problem are that moulds can be assembled on different injection machines but, depending on the machine having a different production rate, two components or more can also be injected into a mould, and moulds can be assembled during specific time periods according to the manpower availability during that period. The problem considers the arrival of material and its consumption also takes into account the availability of containers for plastic components packaging.

The output of the model provides the allocation of moulds to machines for a specific time period, calculates the quantity of the parts to be produced, and designates the manpower needed for mould changes, the consumption and inventory of raw material, and the containers needed for the packaging of parts. The proposed formulation fulfils the objective of representing a real problem of a plastic component supplier. The problem is validated using different dataset sizes that were randomly generated. In general, all the data sizes provided optimal results in computation times of less than 1 minute.

Secondly, a mixed integer linear programming model is formulated (see Chapter 6) for a combined production and delivery planning problem in the aforementioned second-tier supplier. In this case, a model is formulated to analyse the number of containers needed to store the produced components.

The second-tier supplier employs two types of containers: the first type is plastic containers that are reusable and belong to the first-tier supplier; the second type is cardboard containers, which are temporary containers to store parts until the first-tier supplier sends plastic containers. The difficulty of the problem lies in managing the reusable containers and the adequate supply of empty available containers to meet the second-tier supplier's needs and operating only according to the just-in-time philosophy.

The main problems of the combined production and delivery plan are related to: (i) variation in the demand from the first-tier supplier, which involves using large safety stocks in the second-tier supplier's warehouse; (ii) the second-tier supplier produces many automotive component parts to minimise start-up costs, which involves having to produce more parts than demanded by the first-tier

Chapter 9. Conclusions.

supplier; (iii) overproduction results in excess stock, which has to be stored in cardboard containers, with subsequent processing when plastic containers become available again at the second-tier supplier's production plant. As plastic containers become scarce, the second-tier supplier has to incur the costs associated with purchasing the cardboard containers, as well as the storage and handling costs of having to switch automotive components from cardboard containers to plastic containers when they arrive. The limitation in this combined production and delivery plan arises in the amount of empty reusable containers that the first-tier supplier sends to the second-tier supplier.

In this context, a model was generated that allows to determine the number of reusable containers that should remain in circulation and to determine the number of plastic containers that the first-tier supplier should have to cover the demand of parts and, alternatively, the number of cardboard containers that the second-tier supplier should buy to meet demand. The model is useful for both the first-tier and second-tier suppliers because they can know the number of containers needed and can, therefore, negotiate the price of parts because buying containers increases the price of parts. Accordingly, the model is useful for setting the selling price of plastic parts on a planning horizon.

Thirdly, by considering the computational difficulty of scheduling problems and that (commercial and free) solvers cannot find good solutions in reasonable computational times, an efficient method (see Chapter 7) is developed to tackle the complex job shop problem. This is one of the most studied problems in the optimisation field and is presented as NP-hard.

A matheuristic method is proposed. It consists of the hybridisation of a genetic algorithm (GA) with a mixed integer linear programming solver for a disjunctive MILP and a priority heuristic.

The priority heuristic is generated to help to generate individuals from the initial population, which allows the GA to arrive at an initial solution more quickly. These heuristic rules are designed specifically for the sequencing problem. The exact solver plays a very relevant role for efficiently dealing with the continuous variable (simplex method) of the disjunctive mixed integer linear programming model. In this context the model retains its optimality and generality in the continuous variable. So, the combinatorial part is handled by the GA. While developing the problem, an open-source solver was used, in this case Coin-OR Branch & Cut (CBC), which gives good results. When comparing the results of CBC and the proposed matheuristic, the proposed algorithm is superior in all instances, and in both computational times and gaps. In this problem, we test our approach

Models and algorithms for the optimisation of replenishment, production and distribution plans in industrial enterprises.

with large instances. Therefore, the proposed method is a significant contribution to the literature because the combination between a non-commercial solver and a metaheuristic algorithm is an easy-to-use method that does not generate any costs associated with licences or commercial software. Furthermore, the matheuristic method herein proposed is also a method that is easily scalable to other problems.

Finally, a matheuristic algorithm is designed for the combined production and distribution planning problem (see Chapter 8) using the open-source solver Coin-OR Branch & Cut (CBC) and the GA. In this case the GA evaluates the population by a relaxed mixed integer linear programming (MILP) model; that is, the integer variables of the model are transformed into continuous ones, and the binary variable is calculated with the GA. The result of the combined use of the GA and the relaxed MILP provides us with the best chromosome. This chromosome is then fixed to the MILP to obtain a final solution. The results obtained from this hybridisation compared to CBC show the superiority of the matheuristic. To reinforce our results, we use Gurobi (commercial solver) in which the matheuristic can obtain better results compared to this solver. Consequently, our proposal provides an efficient method that converges to good results in an acceptable computational time with real size instances. It is also a feasible model that can be replicated for other industrial sectors.

In synthesis, the general research question (GRQ) has been answered.

GRQ: What suitable approach could effectively solve replenishment, production and distribution planning, which are computationally difficult to solve by exact solvers?

From the analysis of chapters 2 and 3, the importance of the study of matheuristic algorithms was identified by determining the need to develop new tools for replenishment, production and distribution planning. Considering that many studies detail that commercial or non-commercial solvers cannot reach optimal solutions in reasonable computation times, due to the difficulty of the problems that are generally presented as NP-Hard and the amount of data that real problems have, the need arises to develop new tools that are efficient and can meet the needs of companies.

In order to establish which method could be efficient to solve replenishment, production and distribution planning problems (GRQ), and in order for companies to take advantage of the mathematical models created and not have to create a new heuristic or metaheuristic algorithm and leave aside the mathematical model, since creating an algorithm from scratch can cause an unaffordable expense for Chapter 9. Conclusions.

companies, a new matheuristic algorithm that combines a genetic algorithm with a MILP was built, thus achieving the general objective of the research (GRO). In this way, we generated a matheuristic algorithm that reuses the mathematical model and together with a genetic algorithm that is easy to replicate, since it complies with the basic fundamentals of the genetic algorithm, reaches good solutions in reasonable computational times. So, it can be inferred that it is an efficient method and that combined with a non-commercial solver such as the CBC can achieve good results.

9.2 Future research lines

Based on the work carried out in this thesis, several research lines have been opened up:

- In the mathematical modelling field, robust optimisation and fuzzy programming methods are an important area to explore, as is developing new modelling approaches that address and associate the parameters related to production and sustainability, which can also address uncertain parameters.
- It is also necessary to generate mathematical models that take into account collaboration. This means, models that represent the integration of the information transmitted between partners in a supply chain.
- Another research and development area is evaluating the mathematical models that are called transversal in this classification (presented in Chapter 3) and testing them in other areas of industry because these models have features that make them easily reproducible.
- Future research should be related to generate a portfolio of models and algorithms. In this portfolio, the algorithms that are present in the literature should be programmed so they can be used by companies, especially SMEs, as a source in their decision-making process. Likewise, the use of the Fuzzy TOPSIS methodological approach can be adapted to company-specific criteria. To do so, more dimensions and criteria can be considered.
- The formulation of matheuristic algorithms remains a large field to be explored. Matheuristic algorithms depend directly on the employed solver, although commercial solvers are widely used. It is important to verify not only the performance of non-commercial solvers, but also the combination of non-linear models and solvers for this type of models.
- Finally, many algorithms and models appear in the literature. However,

Models and algorithms for the optimisation of replenishment, production and distribution plans in industrial enterprises.

the combination of simulation models with different algorithms [1] is a broad area to explore, as is using combinatorial optimisation and neural networks, where optimisation methods can be employed to train the neural network.

9.3 Reference

[1] A. A. Juan, J. Faulin, S. E. Grasman, M. Rabe, and G. Figueira, "A review of simheuristics: Extending metaheuristics to deal with stochastic combinatorial optimization problems," Oper. Res. Perspect., vol. 2, pp. 62– 72, 2015, doi: 10.1016/j.orp.2015.03.001.

Funding Acknowledgement

Finally, this work would not have been possible without the following funding sources:

- Conselleria de Educación, Investigación, Cultura y Deporte Generalitat Valenciana for hiring predoctoral research staff with Grant (ACIF/2018/170) and the European Social Fund with the Grant Operational Programme of FSE 2014-2020.
- Conselleria de Educación, Investigación, Cultura y Deporte Generalitat Valenciana for predoctoral contract students to stay in research centers outside the research centers outside the Valencian Community (BEFPI/2021/040) and the European Social Fund.